Imagine

2020 ANNUAL REPORT



About Us

Founded in 1988, Tricon Residential is a rental housing company focused on serving the middle-market demographic. Tricon owns and operates approximately 31,000 single-family rental homes and multi-family rental apartments in 21 markets across the United States and Canada, managed with an integrated technologyenabled operating platform.

More information about Tricon is available at www.triconresidential.com.

This document contains forward-looking statements and information relating to expected future events and the Company's financial and operating results and projections. This document also presents key performance indicators used by the Company to measure its performance that are not recognized under International Financial Reporting Standards ("IFRS"). Please refer to the Management's Discussion and Analysis section of this report for further information and disclaimers concerning these forward-looking statements and non-IFRS measures.

Imagine a company that recognizes economic uncertainty for what it is – a time to inspire hope and go the extra mile.

Dear Shareholders,

While 2020 was a challenging and heart-wrenching year for so many, Tricon's 10th year as a public company will be remembered more constructively as the year an unforeseen global health crisis accelerated demographic trends in our favour. With the COVID-19 pandemic spreading to every corner of the planet, Americans left dense urban centres in droves in search of the perceived safety of suburbia and single-family housing. Suddenly, powerful de-urbanization and de-densification trends took hold, driving future demand and uprooting long-held attitudes about cities and housing preferences. Although the longer-term ramifications of this pandemic will only be known in time, it seems clear in the near term that the real estate landscape has been irreversibly altered and that telecommuting will challenge conventional wisdom associated with density and location.

Resilience Where it Matters Most

The single-family rental industry, which Tricon helped establish in the aftermath of the Great Financial Crisis, is a clear beneficiary of this real estate reversal. In a recessionary period where many traditional real estate businesses have struggled with deteriorating fundamentals, demand for single-family rentals soared as families prioritized health during the pandemic, sought more space to socially distance, and moved to more affordable locales. Moreover, with government-mandated work-from-home orders and the advancement of videoconferencing technology, working from home suddenly became socially acceptable and the single-family home became the preferred place to live, work and play. In essence, the great experiment of work-fromhome exceeded all expectations and ushered in a new era where "workers are untethered from the plant," freeing both employers and employees

to benefit from more flexible approaches to work. While offices will continue to be viable and necessary for most enterprises, the expanding gig economy and the merits of work-from-home are likely to strengthen these powerful nesting trends and hasten growth in the less dense Sun Belt markets where Tricon operates.

We expected our single-family rental business (75% of Tricon's real estate assets) to exhibit defensive characteristics in a downturn but the response to the pandemic and related economic recession exceeded our wildest imagination or rosiest forecasts. Even after a very strong 2019, our same home portfolio experienced a 110-basis point increase in occupancy to 97.2%, a 460-basis point decrease in turnover to 22.8%, and a record net operating income ("NOI") margin of 66.3% compared to 59.8% in 2017. Our demand metrics also accelerated, with applications up 8% and gross leases per available home up 18% year-over-year. But perhaps most telling is that our average rent growth on new leases reached 10.7% in the fourth guarter, notwithstanding that the U.S. unemployment rate had roughly doubled from pre-COVID levels. Simply put, we have moved into an environment where demand for single-family rentals far outstrips supply and we expect this favourable outlook to continue for many years given the reinforcing nature of migration trends and the difficulty of bringing on new, affordable housing supply.

66

Demand for singlefamily rentals soared as families prioritized health during the pandemic, sought more space to socially distance, and moved to more affordable locales.

99

There's Always Room to Improve

Although the pandemic has been kind to our single-family rental business, fundamentals in our U.S. multi-family business (20% of Tricon's real estate assets, prior to syndicating an 80% interest) have deteriorated, as the periodic closure of amenities made communal living less attractive and de-densification trends put pressure on leasing. Further, our portfolio has a relatively high concentration of properties in Orlando and northwest Houston (~32% of units), which have

66

We remain confident that meaningful synergies can be harnessed by combining single-family and multi-family rental operations in our Sun Belt markets.

99

been disproportionately impacted by the pandemic given their high exposure to the tourism and energy sectors. Our U.S. multi-family portfolio experienced a 6% reduction in NOI year-overyear as a result of lower occupancy, negative lease trade-outs, higher use of concessions to stabilize occupancy in the summer months, and elevated bad debt. The good news is that we believe our operational metrics have troughed, and barring

a further deterioration in the economy, we expect to see stronger results over the course of 2021 as concessions burn off and we leverage our operating capabilities to drive the performance of the portfolio. Our single-family rental operations team led by Chief Operating Officer Kevin Baldrige has extensive experience in multi-family property management and is looking forward to implementing many of the innovations that have driven our success in single-family rental. It will take time, but we remain confident that meaningful synergies can be harnessed by combining single-family and multi-family rental operations in our Sun Belt markets.

Building for Future Value

The Selby, the first stabilized multi-family asset in our burgeoning build-to-core Canadian apartment business (2% of Tricon's real estate assets), showed relative strength amid a very weak downtown Toronto apartment and condo rental market. The Selby's best-in-class status enabled it to achieve superlative operating results, with occupancy and lease trade-outs largely flat year-over-year, notwithstanding the periodic closure of its amenities and major increases in market rental supply as de-urbanization, de-densification and COVID-19 restrictions worked against the downtown multi-family sector. This is a very difficult environment in which to lease up a building and we are fortunate that The Selby was largely leased before the onset of the pandemic and, perhaps more importantly, that our other ~3,720 units in development/construction are not scheduled for completion until 2022 to 2026. Unlike the U.S., where we believe de-urbanization trends will have a longer-term impact on migration patterns and real estate fundamentals, it is our view that Toronto's apartment market will recover relatively quickly and ultimately return to the lowvacancy environment of February 2020 once a vaccine is rolled out and COVID-19-related visa restrictions are lifted for new permanent residents and international students.

Similar to single-family rental circa 2015, we believe we are incubating a major business in purpose-built Canadian multi-family rental that will be differentiated by its superior asset quality and will be highly sought-after once it is largely stabilized and generating income. Moreover, using conservative estimates for development yields (4.75%) and cap rates (3.5–4.0%), it is our belief that our Canadian multi-family portfolio will be worth approximately "C\$3 per share upon completion compared to its book value of C\$0.84 per share today, which provides our shareholders with significant upside and option value.

From Strength to Strength

With our single-family rental business exhibiting resilience and strengthening fundamentals, we decided to aggressively capitalize on our good fortune and ensure that we emerge from the pandemic stronger than ever. Our approach centred around two key objectives - shoring up our balance sheet and positioning Tricon for significant growth ahead. When we were approached over the summer of COVID-19 with the opportunity to raise additional public equity, we elected to run a formal process to secure capital that could i) improve our liquidity position and leverage metrics, ii) be structured at a "forward share price" to minimize dilution, and iii) validate our business in a time of uncertainty. The process was far more competitive than we initially anticipated; it was clear that large institutional investors were favourably disposed to single-family rental and were interested in allocating significant capital to the sector. In the end, Blackstone's core real estate vehicle, BREIT, emerged as the natural winner and led a group of investors in a \$300 million exchangeable preferred stock transaction that was applauded by the markets and enabled us to largely pay off our revolving corporate credit facility. No other investor has more experience or conviction in our rental housing strategy than Blackstone and we are excited to partner with them, and work with our newest board member, Frank Cohen, to help grow our business and take it to the next level.

In my 2019 annual letter to shareholders, I discussed our gradual transformation from a residential real estate asset manager to a rental housing company. As part of that transformation, we adopted consolidated accounting and revised our financial disclosure – major undertakings designed to simplify our business and surface value for shareholders. With this completed, we turned our attention in 2020 to transforming our balance sheet. By way of background, we never set out to run our company with relatively higher leverage but took on additional debt through two transformative acquisitions where we saw significant operational upside and sought to minimize share dilution -Silver Bay Realty Trust in 2017 and the Starlight U.S. Multi-Family (No. 5) Core Fund in 2019. With the Blackstone-led exchangeable preferred stock transaction, we successfully reduced net debt to assets from 61% in Q2 of 2020 to 55% by Q4, and expect to bring this leverage metric down to approximately 50% with the announced syndication of an 80% interest in our U.S. multi-family portfolio. In addition, we seized the opportunity to refinance and extend existing debt in an ultra-low interest rate environment by completing two single-family rental securitizations. Over the course

of the year, we reduced our weighted average interest rate by 83 basis points to 3.12% and laddered out our weighted-average debt maturity from 3.6 years to 3.7 years – a significant accomplishment in the midst of a pandemic and worldwide recession. Lastly and most importantly, our liquidity (including unrestricted cash on hand) increased by \$295 million during the year, to \$529 million by year-end. 66

No other investor has more experience or conviction in our rental housing strategy than Blackstone and we are excited to partner with them.

99

Accelerating our Growth

With our business on solid footing and ample dry powder to fund future growth, we spent the second half of 2020 marketing new private investment vehicles to add scale across our rental housing platform. In addition to the announced syndication of the U.S. multi-family portfolio to two institutional investors, we intend to establish an accompanying growth vehicle to acquire garden-style properties and enlarge our U.S. core plus/value add multifamily portfolio. North of the border, we are

66

We believe our ability to secure meaningful amounts of third-party capital for rental housing is an important differentiator and competitive advantage.

99

actively working with another major institutional investor to continue our build-to-core multi-family development strategy in Toronto and to capitalize on potential dislocation in a softer market. Even more significantly, we are in advanced stages with existing and new institutional investors to create two new investment vehicles for single-family rental, including a joint venture to purchase new homes directly from public and private homebuilders ("homebuilder direct") and a follow-on

vehicle to our \$750 million SFR JV-1, which will likely contain an enhanced "buy box" and enable us to ramp up our acquisition volume.

We intend to allocate the vast majority of new capital to single-family rental and to operate three distinct investment strategies concurrently, including the organic acquisition of older single-family homes (average vintage of ~20 years),

the acquisition of new homes from builders, and the development and/or acquisition of dedicated single-family rental communities (also known as build-to-rent communities) through our existing joint venture with Arizona State Retirement System. Our goal of raising \$1 billion of third-party equity is on track to be achieved in 2021, 12–18 months ahead of schedule, largely because of the favourable investment conditions for our rental housing strategies and the significant preparation undertaken by our team in 2020. It should be noted that Tricon's co-investment in these investment vehicles can be financed with internally driven cash flow, including proceeds from our legacy for-sale housing assets and our corporate credit facility, allowing us the opportunity to generate meaningful FFO growth in future years without being dependent on tapping the public markets.

While we believe that our ability to secure meaningful amounts of third-party capital for rental housing is an important differentiator and competitive advantage, we are strategic in our approach to raising investment vehicles and are mindful of the impact on our shareholders. For instance, we believe it is sensible to raise thirdparty capital for growth when our own stock is trading below our net asset value. If our stock trades at or above net asset value or we enjoy a more favourable cost of capital, we are more likely to increase our balance sheet investment or make larger co-investments alongside third-party investors so that our shareholders participate more directly in our investment activities. Regardless of our stock performance, we believe it is prudent to use third-party capital for development, including build-to-rent development ventures, to minimize the FFO drag during the development period.

In structuring our partnerships with third-party investors, we are insistent on retaining an option to acquire their limited partner interests and increase our own holdings should they want to exit.

We have been inundated by inquiries from third-party investors to form new investment vehicles for rental housing. With many sectors of commercial real estate facing uncertainty, private investors have continued to favour residential and industrial real estate strategies, or what they call "beds and sheds." Tricon is uniquely positioned to capitalize on these private capital inflows as one of the very few institutional operators with meaningful experience in single-family rental and build-to-rent investments. We are also proud to report that many of these investors consider us to be a best-in-class operator, which illustrates how significant our transformation to a rental housing company has been.

Best-in-Class Operations

We believe that our people-first culture and our tech-enabled operating platform are key drivers of our industry-leading performance metrics in single-family rental. A culture of innovation runs deep through Tricon and has fostered the development of a "suite of apps" for single-family rental which enables us to buy consistent product, manage resident relationships and communication, and seamlessly organize workflows, standardize scope, and manage vendor relationships for renovations and repairs and maintenance. Our desire to deliver superior customer service and continuously refine and improve our property management processes is at the heart of our success because it facilitates higher revenues and better expense containment. Interestingly, we have learned through our experience operating in a pandemic that we may be better off to run

our portfolio closer to full occupancy by sacrificing some renewal rent growth so that less stress is placed on the operational apparatus. When the portfolio is run at higher occupancy, we have less turnover and associated friction costs, which in turn reduces the workload for what is still a peopleintensive business. We naturally build "loss to lease" in the portfolio over time, which is subsequently captured as rent growth on new leases.

Doing Good

Perhaps the most important corollary of the pandemic is that it has given us the freedom to create a more significant platform to do good. This has been a difficult and stressful time for many of us. We have been on an emotional rollercoaster and living through a surreal time beset by social division, misinformation and unanswered questions, but more positively, one in which there is a loud call for

change. The senseless murder of George Floyd over the summer led to a global movement to seriously confront anti-black systemic racism and create a better society that is more diverse, more inclusive and more just. And in the arena of housing and associated ESG policy, investors have begun to prioritize social factors rather than focusing narrowly on the environment. Our industry responded in kind by championing social factors during the pandemic and sought to balance the needs of both residents and investors. For instance, we elected to forego or limit rent



We are proud to support Black Girls CODE and the BlackNorth Initiative. increases on renewals and halt evictions to alleviate economic hardship for our residents, but at the same time capture higher rents on new leases. Inspired by the beliefs of our co-founder and Executive Chairman David Berman, I'm proud to report that we also introduced a "living wage" for our employees, establishing a minimum annual base salary threshold of \$36,400 in the U.S. and C\$46,000 in Canada, so that our front-line workers can live with dignity and set aside funds for unforeseen expenses or retirement.





ORLANDO, FL After losing everything in a fire, Chelsie and her sons were supported by Tricon Residential.

Moreover, our team expanded the Tricon Resident Emergency Assistance Fund to help residents who have fallen upon hard times, including those facing medical emergencies or dealing with unexpected death or trauma.

We face a further significant challenge, as the vast majority of leaders in our industry are white men. Tricon as an organization is diverse, with 47% of our employees identifying as women, 28% as Hispanic/ Latino and 14% being African-American/Black, but like many other companies in real estate or finance, we fall short in more senior positions. We believe wholeheartedly in the merits of diversity at all levels of an organization and this is why it was easy for us to sign the BlackNorth Initiative CEO Pledge, which commits us to longer-term diversity targets and other measures to combat anti-black systemic racism. We believe further that companies with a diverse workforce and inclusive policies will perform better over time because they will be more reflective and understanding of the communities they serve and better positioned to create a culture of transparency, learning and authenticity so that employees can be the best version of themselves. While our approach to diversity and inclusion is a long-term journey, we have accomplished much in a short time and I'm proud of the many initiatives launched this year, including i) establishing a diversity council, ii) curating an electronic library so our employees can learn more about our shared legacy of slavery and indigenous racism, iii) declaring Juneteenth a company-wide holiday, and iv) supporting organizations such as Black Girls CODE, which is helping young black girls learn about computer science so they can pursue careers in Science, Technology, Engineering and Mathematics fields.

United in our Purpose

In a strange time that has been marred by uncertainty, disruption and anxiety our team has remained focused, productive, and connected in order to ensure that our business culture not only endures the pandemic but also strengthens as we emerge from the downturn. Nothing exemplifies this more than our massive internal project called "Tricon One," which is being led by CFO Wissam Francis and encapsulates all processes required to transform Tricon from its old investment entity structure into a resident-centric unified business under our new banner, Tricon Residential. As part of Tricon One, we have already i) adopted consolidated accounting and REIT-like disclosure, a Herculean effort at any time, but especially during an environment when our team has been working

from home, ii) undertaken significant tax and legal reorganization, iii) established a new purpose statement - imagine a world where housing unlocks life's potential - new guiding principles, and other policies to unite our workforce and facilitate hiring, retention and performance reviews and, last but not least, iv) launched a new brand and website, www.triconresidential.com, that enables prospective residents to find a home with just "two clicks" and creates a leasing storefront coupled with back-end integration for both single-family and multi-family rental operations. As we look forward to 2021 and beyond, we are focused on upgrading our IT backbone infrastructure so we can become even more efficient and harness data, artificial intelligence and machine learning to provide better service to our residents. For instance, we will be rolling out "intelligent voice automation" in 2021 so that the intake process for leasing and maintenance inquiries is automated, enabling our call centre employees to focus on more personalized and higher-value tasks.

The Future is Bright

In completing our transformation to a rental housing company, it is our strong belief that we will appeal to a broader group of investors and, over time, surface more value in our stock. Our performance in 2020 was an early indication of our success, with Core FFO per share increasing 69% year-over-year and our stock returning 11% (including dividends), significantly outpacing the Canadian real estate index's total return by 25%. Our value proposition to shareholders is increasingly clear and centres around three key pillars: i) operational excellence, underpinned by our technology-enabled operating platform that enhances the service experience for our residents and drives better operating metrics for our investors; ii) our people and culture as a competitive advantage; and iii) our growth strategy, which targets a vast and highly fragmented market

in rental housing and is enhanced by the use of third-party capital to grow faster and scale our operations more efficiently. We believe this is a winning and differentiated formula that will lead to significantly more growth and superior risk-adjusted returns over time.

I would like to thank our team of 788 employees for the passion they bring to work every day, their commitment to excellence, and their willingness to go above and beyond to leave our residents and investors better off. We strive to create a work environment where our employees can excel, and we are honoured that they in turn have recognized Tricon as a "Great Place to Work" in the U.S. and Canada and bestowed on us the highest Glassdoor rating in the industry (4.8 out of 5.0) for employee satisfaction. As always, we remain incredibly grateful to our Board of Directors, shareholders and private investors for their support and confidence in us. We will get through this pandemic and look forward to meeting with you in person on the other side. Until then, please keep well and stay positive!

Gary Berman

Gary Berman PRESIDENT AND CHIEF EXECUTIVE OFFICER, DIRECTOR Toronto, March 2021

66

We are honoured by our employees to have the highest Glassdoor rating in the industry (4.8 out of 5) for employee satisfaction.

99

Imagine leading with purpose.

OUR **LEADERSHIP** GARY BERMAN President and Chief Executive Officer, Director



WISSAM FRANCIS Executive Vice President and Chief Financial Officer



JONATHAN ELLENZWEIG Chief Investment Officer



10 2020 ANNUAL REPORT TRICON RESIDENTIAL

KEVIN BALDRIDGE Chief Operating Officer

TEAM



SHERRIE SUSKI Chief People Officer



DAVID VENEZIANO Chief Legal Officer







GARY BERMAN President and Chief Executive Officer, Director



OUR BOARD OF DIRECTORS

FRANK COHEN Independent Director



CAMILLE DOUGLAS Independent Director



IRA GLUSKIN Independent Director



MICHAEL KNOWLTON Chair of the Audit Committee



SIÂN M. MATTHEWS Chair of the Compensation, Nominating and Corporate Governance Committee



GEOFF MATUS Director and Co-Founder



PETER D. SACKS Independent Lead Director



TRACY SHERREN Independent Director



TRICON RESIDENTIAL 2020 ANNUAL REPORT 11

Imagine investing in a business that has never been more essential.



12 2020 ANNUAL REPORT TRICON RESIDENTIAL

*Tricon has \$8.5 billion of Assets Under Management, of which \$2.6 billion is managed on behalf of third-party investors. **Represents single-family homes and stabilized multi-family rental apartments.



Imagine continuously innovating to drive operations.

Consistently Strong Operating Metrics



Same Home Average Occupancy

Sustaining near-record occupancy throughout the pandemic.

Same Home NOI Margin

Margin improvement driven by strong top-line growth and stringent expense control.

Same Home Annualized Turnover

Lower turnover driven by a focus on customer service and resident retention.

Technology-Driven Operating Platform

Technology and innovation are at the core of our success. Our technology-enabled operating platform enables us to scale our business, drive operating efficiencies and continuously improve our resident experience.



Imagine a company that delivers results today and invests in growth for tomorrow.

Growth in Funds From Operations per Share

69% increase in 2020



Growth in Book Value per Share

18% annualized growth since entering single-family rental in 2012



2020 Performance Highlights

Growth in Rental Units

Ongoing acquisition program adding 800+ homes per quarter



Imagine a company where doing what's right, not what's easy, is rewarded.

Our Environmental, Social and Governance (ESG) priorities are reflected in all aspects of our business and how we conduct ourselves. Our inaugural ESG roadmap was published in 2020, and we will be releasing our first ESG Annual Report in 2021.

Our 5 ESG Priorities

1

Our People

Be good stewards in the communities in which we operate and set our people up for success so that they are empowered to enrich the lives of our residents.

2

Our Residents

Build meaningful communities where people can connect, grow and prosper. Our goal is to take care of life's little tasks so that our residents have more time to focus on what's important to them.

3

Our Innovation

Leverage innovative technologies and housing solutions to meet the current and future needs of our residents and the broader housing market in North America.

4

Our Impact

Embrace smarter ways to reduce the environmental impact of our buildings by leveraging technology to minimize our resource consumption and our carbon footprint.

5

Our Governance

Proactively understand and manage the risks to our business, while acting in a manner that exemplifies our commitment to ethics, integrity, trust and transparency.



Our Innovation

Our ESG Priorities

Imagine spending more time on what matters most.

















Management's Discussion and Analysis

for the year ended December 31, 2020

| | Non-IFRS measures and forward-looking statements | 24 |
|-----------|---|---|
| <u>1.</u> | INTRODUCTION11Vision and guiding principles1.2Business objectives and strategy1.3Environmental, Social and Governance1.4COVID-19 developments | 27 27 29 35 38 |
| <u>2.</u> | HIGHLIGHTS | 40 |
| <u>3.</u> | CONSOLIDATED FINANCIAL RESULTS 3.1 Review of income statements 3.2 Review of selected balance sheet items 3.3 Subsequent events | 43 43 53 59 |
| <u>4.</u> | OPERATING RESULTS OF BUSINESSES 4.1 Single-Family Rental 4.2 Multi-Family Rental 4.3 Residential Development 4.4 Private Funds and Advisory | 61 63 72 81 87 |
| <u>5.</u> | SUMMARY OF NON-IFRS SEGMENT INFORMATION | 92 |
| <u>6.</u> | LIQUIDITY AND CAPITAL RESOURCES 6.1 Financing strategy 6.2 Liquidity 6.3 Capital resources | 103 103 103 103 |
| <u>7.</u> | OPERATIONAL KEY PERFORMANCE INDICATORS 71 Key performance indicators 7.2 Assets under management | 104 104 105 |
| | ACCOUNTING ESTIMATES AND POLICIES, CONTROLS AND PROCEDURES, AND RISK ANALYSIS 8.1 Revenue and income recognition 8.2 Accounting estimates and policies 8.3 Controls and procedures 8.4 Transactions with related parties 8.5 Dividends 8.6 Compensation incentive plans 8.7 Risk definition and management | 106 107 110 110 110 111 111 |
| <u>9.</u> | HISTORICAL FINANCIAL INFORMATION | 121 |

NON-IFRS MEASURES AND FORWARD-LOOKING STATEMENTS

The Company has included herein certain supplemental measures of key performance, including, but not limited to, net operating income ("NOI"), funds from operations ("FFO"), core funds from operations ("Core FFO"), adjusted funds from operations ("AFFO"), Core FFO per share, AFFO per share, Core FFO payout ratio and AFFO payout ratio, as well as certain key indicators of the performance of our businesses. We utilize these measures in managing our business, including performance measurement and capital allocation. In addition, certain of these measures are used in measuring compliance with our debt covenants. We believe that providing these performance measures on a supplemental basis is helpful to investors and shareholders in assessing the overall performance of the Company's business. However, these measures are not recognized under International Financial Reporting Standards ("IFRS"). Because non-IFRS measures do not have standardized meanings prescribed under IFRS, securities regulations require that such measures be clearly defined, identified, and reconciled to their nearest IFRS measure. The definition, calculation and reconciliation of the non-IFRS measures used in this MD&A are provided in <u>Sections 4</u> and <u>5</u> and the key performance indicators presented are discussed in detail in Section 7.

The supplemental measures presented herein should not be construed as alternatives to net income (loss) or cash flow from the Company's activities, determined in accordance with IFRS, as indicators of Tricon's financial performance. Tricon's method of calculating these measures may differ from other issuers' methods and, accordingly, these measures may not be comparable to similar measures presented by other publicly-traded entities.

Certain statements in this MD&A may be considered "forward-looking information" as defined under applicable securities laws ("forward-looking statements"). Statements other than statements of historical fact contained in this document may be forward-looking statements. Wherever possible, words such as "may", "would", "could", "will", "anticipate", "believe", "plan", "expect", "intend", "estimate", "aim", "endeavour", "project", "continue" and similar expressions have been used to identify these forward-looking statements. These statements reflect management's expectations, intentions and beliefs concerning anticipated future events, results, circumstances, economic performance or expectations with respect to Tricon and its investments and are based on information currently available to management and on assumptions that management believes to be reasonable.

This MD&A includes forward-looking statements pertaining to: anticipated operational and financial performance; the Company's strategic and operating plans and growth prospects; expected demographic and economic trends impacting the Company's key markets; project plans, timelines and sales/rental expectations; projected development costs, timelines, plans and development yields; estimated stabilized NOI from development and rental properties; expected performance fees; future cash flows; transaction timelines; anticipated demand for homebuilding and lots; the anticipated growth of the Company's rental businesses; the acquisition of build-to-rent projects; the intentions to attract third-party capital to the Company's businesses, including the syndication of its current investments; the Company's key priorities over the next three years and the manner in which they might be achieved; the intended internalization of property management of the Company's U.S. multi-family rental portfolio and any resulting synergies; expected future acquisitions, occupancy and turnover rates, and capital expenditure programs for single-family rental homes and U.S. multi-family rental apartments; and the ongoing impact of the current COVID-19 pandemic. The assumptions underlying these forward-looking statements and a list of factors that may cause actual business performance to differ from current projections are discussed in this MD&A and in the Company's Annual Information Form dated March 2, 2021 (the "AIF") which is available on SEDAR at www.sedar.com. The continuing impact of COVID-19 on the operations, business and financial results of the Company may cause actual results to differ, possibly materially, from the results discussed in the forward-looking statements.

Forward-looking statements involve significant known and unknown risks, uncertainties and assumptions. Should one or more of these risks or uncertainties materialize, or should assumptions underlying the forward-looking statements prove incorrect, actual results, performance or achievements could vary materially from those expressed or implied by the forward-looking statements contained in this MD&A. See the AIF and the continuous disclosure documents referenced in <u>Section 8.7</u> for a more complete list of risks relating to an investment in the Company and an indication of the impact the materialization of such risks could have on the Company, and therefore cause actual results to deviate from the forward-looking statements.

Certain statements included in this MD&A may be considered a "financial outlook" for purposes of applicable Canadian securities laws, and as such, the financial outlook may not be appropriate for purposes other than this document. Although the forward-looking statements contained in this MD&A are based upon what management currently believes to be reasonable assumptions, there can be no assurance that actual results, performance or achievements will be consistent with these forward-looking statements. The forward-looking statements contained in this document in this document are expressly qualified in their entirety by this cautionary statement.

When relying on our forward-looking statements to make decisions with respect to the Company, investors and others should carefully consider the foregoing factors and other uncertainties and potential events. The forward-looking statements in this MD&A are made as of the date of this document and the Company does not intend to, or assume any obligation to, update or revise these forward-looking statements or information to reflect new information, events, results or circumstances or otherwise after the date on which such statements are made to reflect the occurrence of unanticipated events, except as required by law, including securities laws.

Market and industry data

This MD&A includes certain market and industry data and forecasts obtained from third-party sources, industry publications and publicly available information as well as industry data prepared by management on the basis of its knowledge of the industry in which the Company operates (including management's estimates and assumptions relating to the industry based on that knowledge). Management's knowledge of the North American residential real estate industry has been developed through its experience and participation in the industry. Management believes that its industry data is accurate and that its estimates and assumptions are reasonable, but there can be no assurance as to the accuracy or completeness of this data. Third-party sources generally state that the information contained therein has been obtained from sources believed to be reliable, but there can be no assurance as to the accuracy or completeness of included information. Although management believes it to be reliable, the Company has not independently verified any of the data from management or third-party sources referred to in this MD&A, or analyzed or verified the underlying studies or surveys relied upon or referred to by such sources, or ascertained the underlying economic assumptions relied upon by such sources.

Other

Select photos in this document are presented for illustrative purposes only, may be artists' renditions, and may not be representative of all properties in the Company's portfolio.

INTRODUCTION

1. INTRODUCTION

This Management's Discussion and Analysis ("MD&A") is dated as of March 2, 2021, the date it was approved by the Board of Directors of Tricon Residential Inc. ("Tricon", "us", "we" or the "Company"), formerly Tricon Capital Group Inc., and reflects all material events up to that date.

In January 2020, the Company completed its transition to an owner and operator of diversified rental housing, resulting in the Company determining that it no longer meets the criteria for being an investment entity under International Financial Reporting Standards 10, *Consolidated Financial Statements* ("IFRS 10"). As a result, the Company began consolidating the financial results of controlled subsidiaries including those holding its investments in single-family rental homes and U.S. multi-family rental properties, resulting in the inclusion of these subsidiaries' assets, liabilities and non-controlling interests in the balance sheet of the Company on a prospective basis in accordance with the relevant guidance of IFRS 10.

The requirement to consolidate the financial results of the Company's single-family rental and U.S. multi-family rental businesses is not applied on a retrospective basis in accordance with IFRS 10. Therefore, comparative balances on the balance sheet and income statement continue to present the financial results of these businesses as investments in, and investment income from, Tricon American Homes ("TAH") and Tricon Lifestyle Rentals ("TLR"). For the purpose of comparability, where applicable in this MD&A, the comparative balances have been recast to show the financial results as if the consolidation of Tricon's single-family rental and multi-family rental businesses had been in effect in prior periods.

This MD&A should be read in conjunction with the Company's audited consolidated financial statements and related notes for the year ended December 31, 2020, which were prepared using International Financial Reporting Standards ("IFRS") accounting policies.

Additional information about the Company, including its Annual Information Form, is available on the Company's website at www.triconresidential.com, and on the Canadian Securities Administrators' website at www.sedar.com.

All dollar amounts in this MD&A are expressed in U.S. dollars unless otherwise indicated.

1.1 Vision and guiding principles

Tricon was founded in 1988 as a fund manager for private clients and institutional investors focused on for-sale residential real estate development. The pursuit of continuous improvement as well as a desire to diversify and facilitate succession planning drove the Company's decision to become publicly traded in 2010. While the U.S. for-sale housing industry was decimated in the Great Recession of 2007–2009, Tricon's strong foundation and its leaders' resilience helped it endure the downturn and learn valuable lessons that informed the Company's decision to ultimately focus on rental housing.

In the decade that followed, Tricon embarked on a deliberate transformation away from for-sale housing, which is inherently cyclical, to a rental housing company that addresses the needs of a new generation facing reduced home affordability and a desire for meaningful human connections, convenience and a sense of community. Today, Tricon provides high-quality, essential shelter to residents. It's a defensive business that is designed to outperform in good times and perform relatively well in more challenging times like today.

Tricon was among the first to enter into and institutionalize the U.S. single-family rental industry. Our success has been built on a culture of innovation and our willingness to adopt new technologies to drive efficiencies and improve our residents' lives. We believe that our ability to bring together capital, ideas, people and technology under one roof is unique in real estate and allows us to improve the resident experience, safeguard our stakeholders' investments, and drive superior returns.

We were also first to recognize the benefits of combining single-family and multi-family rental operations to create a pure play on "middle-market" rental housing. By focusing on the similarities of collecting monthly rent from residents and the complementary nature of property management, we believe that Tricon can deliver a superior experience at all stages of the resident lifecycle. Our properties and residents may be diverse but our commitment to enrich the lives of our residents through caring service and a simplified, connected lifestyle is consistent.

Tricon strives to be North America's pre-eminent rental housing company focused on the middle-market demographic by owning quality properties in attractive markets, focusing on operational excellence, and delivering exceptional customer service. Tricon is driven by its purpose statement – **Imagine a world where housing unlocks life's potential** – and expects its employees to conduct themselves according to the following guiding principles:

- Go above and beyond to enrich the lives of our residents
- Commit to and inspire excellence in everything we do
- · Ask questions, embrace problems, thrive on the process of innovation
- Do what is right, not what is easy
- · Elevate each other so together we leave an enduring legacy

Tricon's guiding principles underpin our business strategy and culture of taking care of our employees first so they in turn are empowered and inspired to provide residents with superior service and to positively impact local communities. When our residents are satisfied, they rent with us longer, they are more likely to treat our properties as their own, and they are more willing to refer new customers. We have realized that the best way to drive returns for our investors and shareholders is to ensure our team and residents are fulfilled.

At Tricon, we have always sought to improve the lives of our employees, our residents, and those in our broader communities. We strive to make the world a better place through our guiding principles, which inspire us to go above and beyond and commit to excellence in everything we do.

Living our corporate purpose every day starts with our own employees. And at a time when the world seems so full of uncertainty, it is more important than ever that our employees feel comfortable that they can pay their bills, and save for retirement and unforeseen expenses. We recognize that our employees and their families can live with dignity when their lives are financially secure. Tricon's newly introduced "Living wage" program embodies these guiding principles (see <u>Section 1.3</u>).

We also know that true diversity improves corporate performance, drives growth, and enhances employee engagement. Accordingly, Tricon is also dedicated to the continuation of learning about our society's historic and current systemic prejudices. Recognizing important dates such as June 19th, which is Freedom Day in the U.S., and June 21st, which is National Indigenous People's Day in Canada, was the impetus for declaring June 19th as a paid holiday for all our employees to take the opportunity to learn more about these important issues (see <u>Section 1.3</u>). Only through education can we achieve a greater understanding of, and appreciation for, races, genders, and all groups that find themselves disadvantaged.

As another important, concrete step toward building more truly diverse and inclusive workplaces, Tricon has committed to promoting the BlackNorth Initiative in order to help eliminate persistent inequities across Canada that have resulted from anti-Black systemic racism (see Section 1.3).

Tricon's activities are also guided by Environmental, Social and Governance ("ESG") factors, as outlined in our inaugural ESG roadmap, published in January 2020. This roadmap will guide the Company's ESG initiatives over the next three years and will provide a framework for robust data collection and reporting on Tricon's ongoing progress and performance (see Section 1.3).

In addition, to guide its efforts of building shareholder value over the near term, Tricon has defined the following key priorities for the next three years. Progress toward these goals remains subject to potential ongoing economic instability and uncertainty related to the novel coronavirus global pandemic ("COVID-19") and other risks and uncertainties (see "<u>Non-IFRS measures and forward-looking statements</u>" on page 24 and <u>Section 8.7</u>).

- Growing core funds from operations ("Core FFO", a key performance indicator ("KPI"); refer to <u>Section 7.1</u>) Tricon is focused on growing Core FFO per share by increasing the net operating income of its rental properties, increasing its Private Funds and Advisory fee streams, and acquiring additional rental properties;
- Increasing third-party assets under management ("AUM") Tricon aims to raise third-party capital in all of its businesses to enhance scale and improve operational efficiency, reduce its balance sheet exposure to development activities, and drive its return on equity with incremental fee income;
- Growing book value per share Over time, Tricon plans to redeploy the majority of its free cash flow into accretive growth opportunities focused primarily on rental housing; and
- **Reducing leverage** Tricon plans to reduce corporate-level debt by maintaining prudent and largely non-recourse leverage at the subsidiary level, with a mid-term leverage target of 50–55% net debt to assets.

1.2 Business objectives and strategy

Tricon is a residential real estate company primarily focused on owning and operating rental housing in the United States and Canada. Since the Company's initial public offering in 2010, Tricon has evolved from an asset manager focused on investing in "for-sale" housing development to a growth-oriented rental housing company with a comprehensive technology-enabled operating platform. Tricon currently owns and operates approximately 31,000 single-family rental homes and multi-family rental units in 21 markets across the United States and Canada. As at December 31, 2020, about 95% of the Company's real estate assets are stabilized rental housing assets, and the remaining 5% or less are invested in residential development projects.



(Based on the fair value of single-family homes, multi-family properties, investments in Canadian multi-family developments, Canadian development properties (net of debt) and investments in for-sale housing.)

Through its fully integrated operating platform, the Company earns rental income and ancillary revenue from single-family and multi-family rental properties as well as fees from managing third-party capital co-invested in its real estate assets.

RENTAL PORTFOLIO



*Excludes 28 single-family rental homes held for sale.

**Includes estimated Canadian multi-family rental units under development based on development plans

as of December 31, 2020. See Section 4.3 for details.

Rental housing strategy

Tricon's U.S. rental strategy, in both single-family and multi-family rental, is focused on select geographic markets in the U.S. Sun Belt and targets the "middle-market" resident demographic. The U.S. Sun Belt has experienced significant population and job growth over time, driven by a friendly business environment, lower tax rates, enhanced affordability and a warm climate. It is home to about 40% of all U.S. households, and is expected to see 60% of the growth in U.S. households over the next decade (source: John Burns Real Estate Consulting, 2019). In many ways, the COVID-19 pandemic has accelerated these demographic trends and is expected to help drive even stronger relative population growth over the next five years in Tricon's core markets as Americans de-urbanize and seek out the safety of suburban living in less dense markets. Furthermore, the Company believes that growing work-from-home trends will likely strengthen in-migration to the Sun Belt states as employers permit more flexible work arrangements and employees gravitate towards more affordable housing markets.

Within its targeted geographic markets, Tricon is focused on serving the middle-market resident demographic which consists of nearly nine million working-class U.S. renter households (source: U.S. Census Bureau). The Company defines the middle-market cohort as those households earning between \$60,000 and \$100,000 per year and with monthly rental payments of \$1,200 to \$1,800. These rent levels typically represent approximately 20–25% of household income, which provides each household with meaningful cushion to continue paying rent in times of economic hardship and when experiencing a decline in income. Conversely, Tricon has the flexibility to increase rents and defray higher operating costs in a stronger economic environment without significantly impacting its residents' financial well-being. Focusing on qualified middle-market families who are likely to be long-term residents is expected to result in lower turnover rates, thereby reducing turn costs and providing stable cash flows for the Company.

Tricon's Canadian "build-to-core" rental strategy targets markets that are underpinned by strong economic fundamentals, including robust job and population growth over an extended period, and attractive supply and demand fundamentals. The Company is currently developing all of its Canadian multi-family properties in downtown Toronto, and believes that the confluence of Canadian urbanization trends, strong population growth, a robust and diversified economy, and major for-sale housing affordability issues will support attractive, long-term rental fundamentals. In addition, Tricon's high-quality, service-oriented rental offerings are well-positioned to cater towards an urban workforce seeking condo-quality, highly amenitized apartments but with professional property management.



A description of each of the Company's businesses is provided below.

Single-Family Rental

Tricon owns and operates one of the largest portfolios of single-family rental homes in the U.S. Sun Belt, with 22,766 homes (excluding 28 homes held for sale) in 18 markets across ten states as of December 31, 2020. Tricon offers middle-market families the convenience of renting a high-quality, renovated home without costly overhead expenses such as maintenance and property taxes, and with a focus on superior customer service.

Since entering the single-family rental business in 2012, Tricon has built a technology-enabled platform to support its growth and manage its properties efficiently. The Company's proprietary technology automates home acquisitions, leasing activities (such as virtual tours and/or self-showings), resident underwriting, revenue management, call centre services, repairs and maintenance and workflow management, among other activities. Management believes the Company has a significant competitive advantage arising from its technology-enabled property management platform that is difficult to replicate yet highly scalable, and it intends to apply these capabilities across both its single-family and multi-family rental portfolios.



Multi-Family Rental

In the U.S., Tricon owns a portfolio of high-quality, affordably priced garden-style apartments primarily in the U.S. Sun Belt, comprised of 23 properties totalling 7,289 suites in 13 major markets. The current portfolio consists of new vintage garden-style complexes featuring resort-style amenities, including swimming pools and well-appointed fitness and common areas, located in desirable suburban sub-markets that have experienced strong employment and population growth over an extended period of time. These assets are currently property managed by leading third-party firms, overseen by Tricon's internal asset management team. However, the Company intends to internalize property management to produce additional synergies by leveraging existing technology, infrastructure and centralized management functions. Tricon's long-term strategy is to continue to grow this business and drive operating synergies through incremental scale.

In Canada, Tricon operates one 500-unit Class A rental property, The Selby, located in Toronto. The Selby is currently managed through Tricon's vertically integrated platform, including local property management employees.



Residential Development

In its Residential Development business, Tricon develops new residential real estate properties, predominantly rental housing intended for long-term ownership. Such developments include (i) Class A multi-family rental apartments in Canada, (ii) its recently launched strategy to develop single-family rental communities in the U.S., and (iii) (legacy) land development and homebuilding projects predominantly in the U.S.

(i) Canadian Class A multi-family rental apartments:

Tricon is one of the most active multi-family rental developers in downtown Toronto with eight projects under development, totalling approximately 3,720 units (including select condominium units). Tricon is focused on developing, owning and operating the leading portfolio of Class A rental apartments in the Greater Toronto Area, Canada's economic engine and one of its fastest-growing metropolitan areas. The Company's "build-to-core" strategy targets institutional-quality development of well-located rental properties near major employment nodes and/or public transit that will ultimately be held over the long term as part of an income-producing portfolio. Through its vertically integrated operations, including land acquisition/entitlement, development, oversight of vertical construction, and property management, we believe that Tricon has a major competitive advantage and is able to develop properties designed specifically to serve rental residents in a Toronto market saturated with investor-driven condominium projects. Tricon holds these assets in partnerships with pension plans and strategic partners who have an investment bias towards long-term ownership and stable recurring cash flows. These institutional investors or strategic partners pay Tricon development management fees, asset management fees and possibly performance fees, enabling the Company to enhance its return on investment.

(ii) U.S. single-family rental communities:

The Company's innovative build-to-rent strategy, which is focused on developing a portfolio of well-designed, dedicated single-family home rental communities, commenced in the third quarter of 2019, following the establishment of a joint venture arrangement with an institutional investor. Such developments, which typically include a cluster of rental homes with shared amenities, combine the privacy and convenience of single-family rental living with the community experience of the multi-family rental model. This strategy leverages the Company's complementary expertise in land development, homebuilding, and single-family rental and multi-family rental property management. The Company closed on its first investment under this strategy in 2020 and expects to commit to approximately ten development communities per year in 2021 and 2022.

(iii) U.S. land development and homebuilding:

The Company's legacy business provides equity or equity-type financing to experienced local or regional developers and builders of for-sale housing primarily in the U.S. These investments are typically made through Investment Vehicles that hold an interest in land development and homebuilding projects, including master-planned communities ("MPCs"). Tricon also serves as the developer of certain of its MPCs through its Houston-based subsidiary, The Johnson Companies LP ("Johnson"), an integrated development platform with expertise in land entitlement, infrastructure, municipal bond finance and placemaking, and deep relationships with public and regional homebuilders and commercial developers.

Johnson's reputation for developing high-quality MPCs is further evidenced by Johnson having four MPCs ranked in the top 50 based on homebuilder sales in 2020 according to RCLCO Real Estate Advisors and John Burns Real Estate Consulting.



(Residential development investments of \$293.0 million represent 5% of Tricon's real estate asset value. The investment balance includes Tricon's investments in Canadian multi-family developments, investments in Canadian development properties (net of debt) and investments in for-sale housing as at December 31, 2020. Refer to Section 4.3.)

Private Funds and Advisory

Through its Private Funds and Advisory ("PF&A") business, Tricon earns fees from managing third-party capital co-invested in its real estate assets through commingled funds, separate accounts and joint ventures ("Investment Vehicles"). Activities of this business include:

(i) Asset management of third-party capital: Tricon manages capital on behalf of American, Canadian and international institutional investors, including pension funds, sovereign wealth funds, insurance companies, endowments and foundations, as well as family offices and high net-worth accredited investors who seek exposure to the residential real estate industry. Tricon currently manages \$2.6 billion of third-party capital (of total AUM of \$8.5 billion) across its single-family rental, multi-family rental and residential development business segments.

Tricon manages third-party capital for ten of the top 100 largest institutional real estate investors in the world (source: PERE 2020 Top 100 Global Investor report, October 2020). Tricon ranked 65th globally and second in Canada (compared to 68th and third, respectively, in 2019) among global real estate investment managers based on the amount of private real estate direct investment capital raised since 2015 (source: PERE 100 report, June 2020). In aggregate, the Company has approximately 17 institutional investors in its active Investment Vehicles.

For its services, Tricon earns asset management fees and performance fees, provided targeted investment returns are achieved.

Tricon believes it is prudent to use a combination of balance sheet and third-party capital across its businesses. In particular, third-party capital allows the Company to generate scale and drive operational synergies, diversify its investor base, capitalize on opportunities that would otherwise be too large for the Company, reduce its balance sheet exposure to development activities, and enhance Tricon's return on equity by earning asset management and other fees.

When co-investing with institutional partners, Tricon prefers to invest a higher relative portion of its commitment in incomeproducing rental strategies and a lower portion in development. This approach allows Tricon's balance sheet investments to immediately generate regular income streams and help grow FFO, while minimizing exposure to longer-term development assets, which do not generate immediate cash flow.

- (ii) Development management and related advisory services: Tricon earns development management fees from its rental development projects in Toronto, which leverage its fully integrated development team. In addition, Tricon earns contractual development fees and sales commissions from the development and sale of single-family lots, residential land parcels, and commercial land within the MPCs managed by its Johnson subsidiary.
- (iii) Property management of rental properties: Tricon provides integrated property management services to its entire single-family rental portfolio (including homes owned through joint ventures with third-party capital partners) and Canadian multi-family assets and is planning the internalization of property management for its U.S. multi-family rental portfolio. The property management business is headquartered in Orange County, California, and provides resident-facing services including marketing, leasing, and repairs and maintenance delivered through a dedicated call centre and local field offices. For its services, Tricon earns property management fees, typically calculated as a set percentage of the gross revenues of each property, as well as leasing, construction and acquisition fees.



FEE REVENUE BY SOURCE FOR THE YEAR ENDED DECEMBER 31, 2020*

*Property management fees paid by the single-family rental business segment and certain development management fees paid by Canadian development properties are eliminated upon consolidation and are excluded from revenue from private funds and advisory services. Refer to <u>Section 4.4</u> for a summary of revenue from private funds and advisory services for the three and twelve months ended December 31, 2020.

1.3 Environmental, Social and Governance

Environmental, Social and Governance (**"ESG**") principles have guided Tricon's decision-making and strategy for the past 32 years. In January 2020, the Company published its ESG roadmap, formalizing our approach to ESG and highlighting our commitment to five strategic priorities: Our People, Our Residents, Our Innovation, Our Impact and Our Governance. The ESG roadmap will guide the Company's ESG initiatives through 2022 and will provide a framework for data collection and reporting on the Company's ongoing progress and performance.

Over the course of this year, we have established the ESG programs and related performance measures intended to fulfill our commitments. The design of these programs is substantially complete and will form our three-year ESG implementation plan. We aim to share our key initiatives and discuss our ESG performance in our inaugural ESG annual report in the coming months.

Our People

The Company is committed to engaging, supporting and enriching the lives of its employees so they can thrive. Tricon recognizes that creating a strong and healthy culture is an ongoing journey that must be firmly rooted in the concept of continuous improvement. Examples of accomplishments to date include:

- A continued focus on talent management and a succession planning framework to build leadership capacity and strengthen retention.
- The implementation of a number of recognition programs to promote workplace culture and values. These programs include the "Good Gotcha" program, which celebrates individual examples of day-to-day employee excellence, and the "Pay It Forward" program, whereby every employee receives \$100 annually to give to the charity of their choice or a person in need. We are proud to have donated over \$100,000 as a group to a broad range of organizations and individuals in need during the past holiday season. Just over half of our donations went to organizations that have missions ranging from poverty reduction, animal welfare and health causes, benefiting our local communities in diverse ways. We may be "social distancing" today but the human connections we value and the communities we live in remain firmly intact.
- Health, safety and well-being initiatives including programs such as web-based medical services, fitness benefits, employee assistance programs, Best Doctors medical counselling and life balance naturopathic services.
- A corporate office designed with employee health and well-being as a primary consideration, including a spacious open-concept floor plan that increases employees' access to natural sunlight, ergonomic solutions for all employees (including sit-stand desks), and the promotion of face-to-face interactions and walking meetings when possible.
- We continuously monitor employee engagement and satisfaction, using the results to drive our actions. Our annual employee
 engagement survey was completed through Great Places to Work in the past quarter. The survey focuses on instilling a culture
 where employees both trust and feel trusted by their managers and co-workers. Our Tricon Residential teams in the U.S. and
 now in Canada have been Great Place to Work-certified.

At Tricon, living our corporate purpose every day starts with our own employees. In 2020, the Company embarked on several initiatives focused on equality, diversity and inclusion:

- Living wage we established a minimum base salary threshold of \$36,400 in the U.S. and C\$46,000 in Canada per year, providing financial security for our employees and their families.
- BlackNorth Initiative's CEO Pledge we participated in the BlackNorth Initiative and have joined several of Canada's largest
 businesses in signing a "CEO pledge" committing Tricon to take demonstrable and positive action to acknowledge and counter
 systemic anti-Black racism.
- Juneteenth holiday on June 19, we observed the Juneteenth holiday which marks the day in 1865 when anti-slavery laws were enforced in Texas. We invited our employees company-wide to learn about Black history and the challenges that racialized communities face.
- Black Girls Code we donated to Black Girls Code, a charity focused on helping young Black girls gain exposure to computer science and technology and encouraging careers in Science, Technology, Engineering and Mathematics.

Founders' Day

Each year, we celebrate Tricon's founders by devoting one day towards making a positive difference in our communities.

This year, Tricon celebrated Founders' Day on September 23 with nearly 700 employees across North America participating virtually. The main theme of 2020 was anti-Black systemic racism, featuring discussions with the heads of the Canadian Council of Business Leaders Against Anti-Black Systemic Racism as well as Black Girls Code.
In addition, during Founders' Day, we featured Red Door Family Shelter as Tricon's charity of choice in Canada. Red Door Family Shelter is one of the largest family shelters in Toronto, providing emergency shelter and support for women and children affected by domestic abuse, families experiencing a housing crisis, and refugee claimants. The COVID-19 pandemic has put even more pressure on families at risk and so Tricon partnered with Red Door Family Shelter in 2020 to respond to the growing need for homeless shelters in the City of Toronto.

This year, we also focused on Tricon's guiding principles and, as part of the Founders' Day celebration, employees across the U.S. and Canada submitted short videos of themselves in which they explained what Tricon's new purpose statement and guiding principles mean to them and why they are important in their work at Tricon. Select content from these inspirational videos is posted on Tricon's website at www.triconresidential.com.

Our Residents

Tricon's goal is to build meaningful communities where people can connect, grow and prosper. In light of the widespread economic uncertainty related to COVID-19, we have focused our efforts this year on assisting residents in need through several initiatives:

- Comprehensive suite of resident surveys we implemented a comprehensive suite of resident surveys in our single-family rental business that are used throughout the resident lifecycle, including after a property tour, move-in, maintenance technician visit, seven-month checkpoint, renewal and post move-out communications. We believe this program helps drive significantly higher resident retention, higher revenues and a lower turnover rate.
- Resident Emergency Assistance Fund in response to the COVID-19 pandemic, we expanded our Resident Emergency Assistance Fund to \$200,000 per year which provides emergency assistance and financial relief to residents experiencing unexpected hardship. The fund helps residents and their families meet their rent obligations and stay in their homes.
- Tricon Residential Giving Back Fund in the third quarter of 2020, we established the Tricon Residential Giving Back Fund. The Fund gives Tricon employees the opportunity to automatically deduct a portion of their pay to make a tax-deductible donation to a selection of non-profit organizations, including our partner charities Black Girls Code and Red Door Family Shelter.
- Self-governing rent growth on renewals recognizing that many of our residents may be facing financial pressures during the COVID-19 pandemic, in the second and third quarters of 2020 we offered to renew many expiring leases at nominal increases, or forego rent increases altogether, and plan to continue our practice of "self-governing" on rent increases related to renewals.
- Late fees and deferral plans among our various initiatives intended to alleviate financial pressure for our residents, we waived late fees and offered flexible rent deferral plans for those in need. We also temporarily halted all evictions and currently observe eviction moratoriums according to federal and municipal mandates. Moreover, we have waived early termination fees throughout the pandemic to select residents who encountered COVID-19 hardships.
- Affordable Housing Lands Program we partnered with investors and the Ontario government under the Affordable Housing Lands Program to deliver an innovative solution to housing affordability in Toronto. Our West Don Lands project is one of the largest affordable housing projects in Canada and will include 30% affordable units delivered at the same quality and standard as the market rate units. Block 10 of the West Don Lands will feature Toronto's first purpose-built Indigenous Hub which will include an Indigenous Health Centre and community gardens as well as an Indigenous Employment, Education and Training Centre.

Our Innovation

Tricon is strongly committed to leveraging innovative technologies and housing solutions in order to drive convenience, connectivity and affordability. Core service offerings are guided by two key desired outcomes: (i) delivering superior service that creates exceptional resident experiences and (ii) developing offerings that enhance the lives of residents while addressing their housing needs. Examples include:

- Intelligent Virtual Agent technology deployed at our call centre to automate the leasing process. This technology improves the customer communication experience, enabling residents to contact us 24 hours a day, seven days a week, with inquiries related to home statistics, tour scheduling and account information.
- Proprietary self-showing and virtual move-in process that allows potential residents to: (i) find a Tricon home online and perform a 360-degree walkthrough from the comfort of their smart phone or computer; (ii) schedule and conduct a self-showing of a Tricon home at their preferred time and without a leasing agent; (iii) complete the leasing documentation process seamlessly and 100% electronically; and (iv) move in to a Tricon home using a virtual concierge who can conduct a home walkthrough via videoconference.
- Customized Smart Home system which provides: convenient and controlled access to our homes through smart locks and door sensors, remote thermostat access which enhances comfort and generates energy savings, and moisture sensors that identify and allow us to fix hard-to-detect water leaks before they cause damage.

- Investment in new fleet tracking technology that enhances capacity and demand planning to ensure much higher on-time service delivery to our residents. This technology also helps decrease emissions through reduced vehicle idling times.
- Augmented reality is used to provide field training to our maintenance technicians virtually. This program helps us standardize training across all 21 markets and identify suitable candidates through a virtual interview process.
- Partnerships with organizations such as Toronto Life magazine, Eye Buy Art, Roy Thomson Hall, Massey Hall and the Evergreen Brick Works to provide our Toronto multi-family residents access to cultural activities and events. We also have partnerships with companies such as Last Box Moving, Casper Mattress and Wayfair, aimed at providing discounted access to services needed for elevated apartment living. Our full-sized commercial-grade gyms, designed in partnership with Biosteel Fitness, offer a combination of weight training, cardiovascular and group fitness options through our Fitness on Demand app.
- Predictive Index, an analytical measurement tool, is used to assist with recruiting and retaining high-performing employees.
 A behavioural assessment is completed for every position we fill, and by every candidate who applies for that specific position.
 The Predictive Index assesses a candidate on four dimensions dominance, extroversion, patience and formality and compares the results to the job profile to identify areas of mismatch during the interview process so that we can identify the candidate best suited for the position and address misalignment that may lead to turnover.

Our Impact

The Company is committed to making a material sustainability impact across all of its business activities over the long term. This effort will involve embracing smarter ways to reduce the environmental impact of our buildings by minimizing both our resource consumption and carbon footprint. Tricon is dedicated to ensuring its developments are built to LEED standards and that wildlife and biodiversity are protected by creating parks, green spaces and natural ecosystems. Examples of accomplishments to date include:

- The Viridian master-planned community is a Certified Gold Signature Sanctuary. This certification is only awarded to new developments that are designed, constructed and maintained according to Audubon International's standards for planning and environmental stewardship.
- In the fourth quarter of 2020, Tricon's first purpose-built residential development, The Selby, received LEED gold certification. Several sustainable design strategies were deployed to improve the building's performance, taking into consideration its energy consumption, water efficiency, carbon emissions and indoor environmental quality. The building also features a green roof with drought-tolerant plants as well as bike storage and electric vehicle charging stations, providing additional opportunities for residents to reduce their carbon footprint.
- The West Don Lands mixed-use development is being built to achieve LEED Gold status, with a strong emphasis on sustainability, energy efficiency and walkability. Key sustainability and energy efficiency features have been incorporated into the design and development, including efficient chillers, temperature-moderating façade systems, in-suite heat recovery, low-flow hot water fixtures, LED fixtures in communal areas, locally sourced materials, bike parking, storm water retention, solar wall technology, a self-shading façade, green roofs, native plant species, urban farming and a city tram connection.
- We are using smart home technology in our single-family rental homes across the United States to reduce our carbon footprint. Sensors under sinks and near hot water heaters reduce needless water consumption by providing early leak detection, while smart thermostats allow temperature management when homes are vacant.

Our Governance

Tricon is firmly committed to acting in an ethical manner across all of its business dealings, and to working transparently with stakeholders and investors to enhance trust and reduce risks. We have established a governance framework to hold our organization, leadership and staff accountable. The governance framework includes four key elements:

- Code of Business Conduct and Ethics and Compliance Manual outline the Company's business practices and procedures to ensure compliance with all securities laws, legal requirements and our own standards.
- Whistleblower policy sets out expectations for the reporting of any illegal or unethical behaviour, in addition to a confidential complaint procedure through which people can report concerns about accounting or auditing matters or potential violations of the Company's policies without the threat of retaliation.
- Diversity of leadership exemplifying the Company's commitment to diversity throughout its business across a range of factors, including expertise and experience, gender, geography, age, race and ethnicity. It also confirms our commitment to meeting or exceeding the expectations of the 30% Club Canada, a campaign to increase gender diversity at board and senior management levels, and our pledge made as part of the BlackNorth Initiative.
- Risk management including the use of prudent and disciplined investment practices, diversifying capital across product types and market locations, diligently structuring transactions, conducting comprehensive due diligence and market research, and taking an active role in the ongoing management of our investments.

For further details, please refer to the Company's ESG roadmap, which was published on January 28, 2020. The ESG roadmap is available on our website at www.triconresidential.com/investors and on SEDAR at www.sedar.com.

1.4 COVID-19 developments

During 2020, the outbreak of COVID-19 and its rapid spread around the globe caused unprecedented disruption to the world's economies and capital markets. The ultimate consequences of the COVID-19 pandemic are still unknown; however, management believes that the Company's strong leadership team, its diverse sources of recurring cash flow and its flexible liquidity profile will help mitigate the impact that COVID-19 may have on Tricon's near-term business performance (see also <u>Section 8.7</u>). Tricon's response to COVID-19 has been as follows:

Supporting Tricon's employees

Tricon is committed to the health and safety of over 750 employees across our U.S. and Canadian operations (including approximately 100 employees at Johnson). The Company's employees began working from home as early as March 16, 2020, leveraging Tricon's investments in technology to conduct operations without interruption. Tricon's call centre staff are fully equipped to work from home, and leasing activities are largely conducted using virtual tours and self-showings. In-person contact is being minimized for local market staff, and personal protective equipment is being used where necessary to continue providing essential maintenance activities.

Supporting Tricon's residents

Tricon is focused on providing its residents with a safe living environment and helping to mitigate the financial impact of COVID-19. The Company has moved to a strong occupancy bias in its rental businesses, and temporarily halted evictions, waived late fees, and offered flexible payment plans for residents whose financial well-being has been directly impacted by the pandemic.

The pandemic is a highly dynamic and evolving situation. Tricon will continue to monitor and act according to the direction of relevant federal, state, provincial and municipal governments. The Company remains steadfast and is committed to implementing the necessary actions to protect its employees and residents during this unprecedented time (see Section 3.3).





2. HIGHLIGHTS

The following section presents highlights for the quarter on a consolidated and proportionate basis. Throughout this section, comparative balances have been recast to conform with the current period presentation.

In response to the COVID-19 pandemic, a business update for the period subsequent to year-end has been presented in Section 3.3.

Core funds from operations ("Core FFO"), Core FFO per share, Adjusted funds from operations ("AFFO") and AFFO per share are KPIs as defined in <u>Section 7.1</u>. The Company uses guidance specified by the National Association of Real Estate Investment Trusts ("NAREIT").

| For the periods ended December 31 | Thre | e months | Twelve | months |
|--|-------------|---------------------|-------------------|----------------------------------|
| (in thousands of U.S. dollars, except per share amounts which are in U.S. dollars, unless otherwise indicated) | 2020 | 2019 ⁽¹⁾ | 2020 | 2019 ⁽¹⁾ |
| Financial highlights on a consolidated basis | | | | |
| Net income, including: | \$ 81,478 | \$ 43,557 | \$ 116,413 | \$ 110,335 |
| Fair value gain on rental properties | 106,995 | 32,025 | 198,314 | 116,548 |
| Income (loss) from investments in for-sale housing | 10,191 | 2,964 | (61,776) | 9,646 |
| Basic earnings per share | 0.41 | 0.22 | 0.58 | 0.62 |
| Diluted earnings per share | 0.39 | 0.21 | 0.58 | 0.61 |
| Dividends per share | \$ 0.07 | \$ 0.07 | \$ 0.28 | \$ 0.28 |
| Weighted average shares outstanding – basic | 194,679,682 | 195,269,680 | 194,627,127 | 172,735,776 |
| Weighted average shares outstanding – diluted | 212,445,547 | 213,682,237 | 195,795,473 | 191,081,128 |
| Non-IFRS ⁽²⁾ measures on a proportionate basis | | | | |
| Core funds from operations ("Core FFO") | \$ 39,910 | \$ 21,748 | \$ 109,584 | \$ 55,011 |
| Adjusted funds from operations ("AFFO") | 32,465 | 15,923 | 81,709 | 28,388 |
| Core FFO per share ⁽³⁾ | 0.16 | 0.10 | 0.49 | 0.29 |
| Core FFO per share (CAD) ^{(3),(4)} | 0.21 | 0.13 | 0.66 | 0.38 |
| AFFO per share ⁽³⁾ | 0.13 | 0.07 | 0.36 | 0.15 |
| AFFO per share (CAD) ^{(3),(4)} | 0.17 | 0.09 | 0.48 | 0.20 |
| Select balance sheet items reported on a consolidated basis | | | December 31, 2020 | December 31, 2019 ⁽¹⁾ |
| Total assets | | | \$ 7,174,834 | \$ 6,486,396 |
| Total liabilities | | | 5,431,596 | 4,825,214 |
| Net assets attributable to shareholders of Tricon | | | 1,735,096 | 1,653,138 |
| Rental properties | | | 6,321,918 | 5,682,525 |
| Debt | | | 4,137,506 | 3,955,261 |

(1) The comparative period results have been recast to present the consolidated results in conformity with the current period presentation. The reconciliation of the prior period figures under investment entity accounting to consolidated accounting can be found in <u>Sections 31</u> and <u>3.2</u>.

Non-IFRS measures are presented to illustrate alternative relevant measures to assess the Company's performance and ability to generate cash. Refer to Section 5.
 Core FFO per share and AFFO per share are calculated using the total number of weighted average potential dilutive shares outstanding, including the assumed conversion of convertible debentures and exchange of preferred units issued by Tricon PIPE LLC, which were 248,247,018 and 224,015,498 for the three and twelve months ended December 31, 2020, respectively, and 213,682,237 and 191,081,128 for the three and twelve months ended December 31, 2019, respectively.

(4) USD/CAD exchange rates used are 1.3030 and 1.3415 for the three and twelve months ended December 31, 2020, respectively, and 1.3200 and 1.3269 for the three and twelve months ended December 31, 2019, respectively.

IFRS measures on a consolidated basis

Net income for the fourth quarter of 2020 was \$81.5 million compared to \$43.6 million in the fourth quarter of 2019, and included:

- Revenue from rental properties of \$122.0 million compared to \$109.9 million in the fourth quarter of 2019 driven primarily by the single-family rental business, reflecting 8.3% growth in the number of rental homes to 22,766, combined with a 4.2% increase in average effective monthly rent and a 1.9% increase in occupancy.
- Direct operating expenses of \$42.7 million compared to \$40.1 million in the fourth quarter of 2019, mainly representing
 incremental costs to manage the larger single-family rental home portfolio and a 5.0% increase in property taxes, partially
 offset by a decrease in repairs, maintenance and turnover expenses driven by improved cost containment discipline as
 well as lower resident turnover.
- Income from investments in for-sale housing of \$10.2 million compared to \$3.0 million in the fourth quarter of 2019 driven by higher valuations at certain projects, as historically low mortgage rates and de-urbanization trends are increasing demand for new single-family housing.
- Fair value gain on rental properties of \$107.0 million compared to \$32.0 million in the fourth quarter of 2019, owing to significant home price appreciation in Tricon's core markets for its single-family rental homes. The increase in home prices is underpinned by population growth in Tricon's Sun Belt markets driven by in-migration, de-densification and de-urbanization trends, all of which have fuelled demand for spacious suburban homes.

Net income for the twelve months ended December 31, 2020 was \$116.4 million compared to net income of \$110.3 million for the twelve months ended December 31, 2019, and included:

- Revenue from rental properties of \$478.2 million and direct operating expenses of \$169.5 million compared to \$361.8 million and \$130.5 million in the prior year, respectively, as a result of the continued growth in the single-family rental business as discussed above and the addition of the U.S. multi-family rental portfolio in the second quarter of 2019.
- Income from investments in Canadian multi-family developments of \$14.1 million compared to \$7.7 million in the prior year, attributable to fair value gains recognized across multiple projects upon achieving key development milestones.
- Fair value gain on rental properties of \$198.3 million compared to \$116.5 million in the prior year driven by home price appreciation in the single-family rental portfolio, partially offset by a fair value loss of \$22.5 million recognized on the U.S. multi-family rental portfolio in the second quarter of 2020, as COVID-19-related impacts to the Company's U.S. multi-family business contributed to a downward adjustment in NOI assumptions.
- Loss from investments in for-sale housing of \$61.8 million compared to income of \$9.6 million in 2019, as a significant write-down was recognized in the first quarter of 2020 in the context of a precipitous drop in sales and uncertainty over the timing of future cash flows brought on by the pandemic.

Non-IFRS measures on a proportionate basis

Core funds from operations ("Core FFO") for the fourth quarter of 2020 was \$39.9 million, an increase of \$18.2 million or 84% compared to \$21.7 million in the fourth quarter of 2019. The increase was attributable to solid operating results from Tricon's growing single-family rental business, reflecting strong rent growth and higher occupancy, coupled with the improved performance of the Company's investments in for-sale housing and a decrease in corporate interest expense. For the twelve months ended December 31, 2020, Core FFO was \$109.6 million, an increase of \$54.6 million or 99% compared to \$55.0 million in the prior year. This increase was mainly attributable to the items noted above, along with the inclusion of a full year of results from the U.S. multi-family rental portfolio compared to a seven-month inclusion in 2019.

Adjusted funds from operations ("AFFO") for the three and twelve months ended December 31, 2020 was \$32.5 million and \$81.7 million, respectively, an increase of \$16.5 million and \$53.3 million from the same periods in the prior year. These variances reflect the increase in Core FFO discussed above, along with higher recurring capital expenditures attributable to the full-year inclusion of the U.S. multi-family rental portfolio results. While Tricon's single-family rental portfolio has expanded in 2020, the Company was able to lower recurring capital expenditures as a result of a targeted reduction in elective capital projects during the COVID-19 pandemic.

Change in net assets

As at December 31, 2020, Tricon's net assets totalled \$1,735 million compared to \$1,653 million as at December 31, 2019. The \$82.0 million increase includes \$113.3 million of net income attributable to Tricon's shareholders reported for the twelve months ended December 31, 2020 (including a \$198.3 million fair value gain on rental properties and a \$61.8 million loss from investments in for-sale housing), offset by dividends of \$35.8 million, among other items.

Tricon's net asset value for its for-sale housing investments decreased by \$135.8 million from \$300.7 million as at December 31, 2019 to \$164.8 million as at December 31, 2020, attributable to (i) distributions of \$77.4 million largely from the proceeds of syndicating a balance sheet investment in the first quarter, and (ii) the above-noted fair value loss of \$61.8 million driven by the pandemic-related write-down in the first quarter. As a result, Tricon's for-sale housing assets now represent approximately 2.3% of the total assets of the Company.



CONSOLIDATED FINANCIAL RESULTS



3. CONSOLIDATED FINANCIAL RESULTS

The following section should be read in conjunction with the Company's consolidated financial statements and related notes for the year ended December 31, 2020.

3.1 Review of income statements

Consolidated statements of income

The comparative figures in the Company's consolidated statements of comprehensive income in the table below have been recast as if the current reporting framework under IFRS 10, *Consolidated Financial Statements* ("IFRS 10"), first applied by the Company effective January 1, 2020 on a prospective basis, had been in effect for the three and twelve months ended December 31, 2019.

| For the periods ended December 31 | Three months | | | | | | | | Twelve | months | | |
|---|--------------|------------------|------|---------------------|----|----------|-----|-----------|--------------|--------------------|------|----------------|
| (in thousands of U.S. dollars, except per share amounts which are in U.S. dollars) | | 2020 | | 2019 | ١ | /ariance | | 2020 | | 2019 | | Variance |
| | | | (Scl | Recast hedule A) | | | | | | Recast dule A) | | |
| Revenue from rental properties | \$ 1 | 121,983 | \$ | 109,915 | \$ | 12,068 | \$ | 478,187 | \$ 36 | 51,766 | \$ | 116,421 |
| Direct operating expenses | | (42,660) | | (40,093) | | (2,567) | | (169,538) | (1 | 30,468) | | (39,070) |
| Net operating income from rental properties | | 79,323 | | 69,822 | | 9,501 | | 308,649 | 2 | 31,298 | | 77,351 |
| Revenue from private funds and advisory services | | 10,339 | | 12.138 | | (1,799) | | 34,090 | | 41,060 | | (6,970) |
| Income from investments in Canadian multi-family developments ⁽¹⁾ | | 8,720 | | 7.416 | | 1,304 | | 14,124 | | 7.714 | | 6,410 |
| Other income from Canadian development properties ⁽²⁾ | | 309 | | 91 | | 218 | | 791 | | 725 | | 66 |
| Income (loss) from investments in for-sale housing ⁽³⁾ | | 10,191 | | 2,964 | | 7,227 | | (61,776) | | 9,646 | | (71,422) |
| Other income ⁽⁴⁾ | | 1.774 | | 2,304 | | 1,774 | | 1,774 | | 9,040 | | 1.774 |
| Property management overhead | | (5,872) | | (5,675) | | (197) | | (22,654) | | | | 3,221 |
| Compensation expense | | (14,940) | | (9,744) | | (5,196) | | (40,100) | , | 23,873) 37,681) | | (2,419) |
| General and administration expense | | | | | | 177 | | | , | | | |
| | | (5,748) (791) | | (5,925) (1,004) | | 213 | | (23,569) | (| 20,846) | | (2,723) 818 |
| Other expense | | | | | | (770) | | (3,173) | (1 | (3,991) | | |
| Interest expense | | (44,421) | | (43,651) | | () | | (170,610) | | 52,309) | | (18,301) |
| Fair value gain on rental properties | | 106,995 | | 32,025 | | 74,970 | | 198,314 | 1 | 16,548 | | 81,766 |
| Gain on sale of U.S. multi-family developments | | - | | 1,113 | | (1,113) | | - | | 9,718 | | (9,718) |
| Fair value (loss) gain on derivative financial instruments | | (10 410) | | (1.400) | | (14050) | | (7.404) | | 2 257 | | (0.010) |
| and other liabilities | | (16,418) | | (1,462) | | (14,956) | | (7,461) | , | 2,357 | | (9,818) |
| Transaction costs | | (2,491) | | (6,532) | | 4,041 | | (14,016) | , | 36,415) | | 22,399 |
| Amortization and depreciation expense | | (2,614) | | (2,733) | | 119 | | (10,848) | (| 10,543) | | (305) |
| Realized and unrealized foreign exchange gain (loss) | | 948 | | 178 | | 770 | | (166) | | 42 | | (208) |
| Net change in fair value of limited partners' interests | | (47 700) | | (1.0.10) | | (10 570) | | (50 504) | | (0.70.0) | | (40.707) |
| in rental business | | (17,780) | | (4,210) | | (13,570) | _ | (50,581) | | (3,784) | | (46,797) |
| | | 17,862 | | (37,149) | | 55,011 | | (189,951) | | 44,694) | | (45,257) |
| Income before income taxes | \$ 1 | 107,524 | \$ | 44,811 | \$ | 62,713 | \$ | 152,788 | \$ 12 | 27,664 | \$ | 25,124 |
| Income tax recovery (expense) – current | | 7,087 | | 1,974 | | 5,113 | | 4,050 | | (5,395) | | 9,445 |
| Income tax expense – deferred | | (33,133) | | (3,228) | | (29,905) | | (40,425) | (| 11,934) | | (28,491) |
| Net income | \$ | 81,478 | \$ | 43,557 | \$ | 37,921 | \$ | 116,413 | \$ 11 | 10,335 | \$ | 6,078 |
| Attributable to: | | | | | | | | | | | | |
| Shareholders of Tricon | \$ | 79,678 | \$ | 42,354 | \$ | 37,324 | \$ | 113,322 | \$ 1 | 07,762 | \$ | 5,560 |
| Non-controlling interest | | 1,800 | | 1,203 | | 597 | | 3,091 | | 2,573 | | 518 |
| Net income | \$ | 81,478 | \$ | 43,557 | \$ | 37,921 | \$ | 116,413 | \$ 1′ | 10,335 | \$ | 6,078 |
| Other comprehensive income | | | | | | | | | | | | |
| Items that will be reclassified subsequently to net income | | | | | | | | | | | | |
| Cumulative translation reserve | | 5,256 | | 1,669 | | 3,587 | | 3,999 | | 3,671 | | 328 |
| Comprehensive income for the period | \$ | 86,734 | \$ | 45,226 | \$ | 41,508 | \$ | 120,412 | \$ 1′ | 14,006 | \$ | 6,406 |
| Attributable to: | | | | | | | | | | | | |
| Shareholders of Tricon | \$ | 84,934 | \$ | 44,023 | \$ | 40,911 | \$ | 117,321 | \$ 1 | 11,433 | \$ | 5,888 |
| Non-controlling interest | | 1,800 | | 1,203 | | 597 | | 3,091 | | 2,573 | | 518 |
| Comprehensive income for the period | \$ | 86,734 | \$ | 45,226 | \$ | 41,508 | \$ | 120,412 | \$ 11 | 14,006 | \$ | 6,406 |
| Basic EPS attributable to shareholders of Tricon | \$ | 0.41 | \$ | 0.22 | \$ | 0.19 | \$ | 0.58 | \$ | 0.62 | \$ | (0.04) |
| Diluted EPS attributable to shareholders of Tricon | \$ | 0.39 | \$ | 0.21 | \$ | 0.18 | \$ | 0.58 | \$ | 0.61 | \$ | (0.03) |
| Weighted average shares outstanding – basic | 194 6 | 579,682 | 195 | 269,680 | /5 | 89,998) | 19/ | ,627,127 | 172 73 | 35,776 | 21 9 | 891,351 |
| Weighted average shares outstanding – blast Weighted average shares outstanding – diluted ⁽⁵⁾ | | 145,547 | | 682,237 | • | 36,690) | | ,795,473 | 191,08 | | | 714,345 |

(1) Includes income from The Selby (Section 4.2.2) and income from The Taylor, West Don Lands, The Ivy and 7 Labatt (Section 4.3.1).

(2) Includes other income from Canadian development properties, The James (Scrivener Square) and The Shops of Summerhill (Section 4.3.1).

(3) Reflects the net change in the fair values of the underlying investments in the legacy THP business (Section 4.3.2).

(4) Includes government assistance received by Johnson.

(5) For the three months ended December 31, 2020, the Company's 2022 convertible debentures were dilutive and the exchangeable preferred units of Tricon PIPE LLC were anti-dilutive, whereas for the twelve months ended December 31, 2020, both were anti-dilutive. For the three and twelve months ended December 31, 2019, the 2022 convertible debentures were dilutive. Refer to Note 29 to the consolidated financial statements.

Schedule A

The table below provides a reconciliation of the consolidated statements of comprehensive income for the three and twelve months ended December 31, 2019 from figures previously reported under investment entity accounting in accordance with IFRS 10 to the recast figures shown in the table above.

| | | Three | e months | | | Twelve months | | | | | | |
|---|-------------|-------|----------|------|----------|---------------|------------|----|------------|-------------|--|--|
| For the periods ended December 31, 2019 | Previously | | | | | | Previously | | | . . | | |
| (in thousands of U.S. dollars) | reported | | ustments | - | Recast | | reported | | ljustments | Recast | | |
| Revenue from rental properties | \$ - | \$ ' | 109,915 | \$ | 109,915 | \$ | - | - | 361,766 | \$ 361,766 | | |
| Direct operating expenses | - | | (40,093) | | (40,093) | | - | | (130,468) | (130,468) | | |
| Net operating income from rental properties | - | | 69,822 | | 69,822 | | - | | 231,298 | 231,298 | | |
| Revenue from private funds and advisory services | 11,716 | | 422 | | 12,138 | | 39,895 | | 1,165 | 41,060 | | |
| Income from investments in Canadian multi-family developments | - | | 7,416 | | 7,416 | | - | | 7,714 | 7,714 | | |
| Investment income – Tricon American Homes | 42,451 | | (42,451) | | - | | 162,193 | | (162,193) | - | | |
| Investment income – Tricon Lifestyle Rentals | 16,812 | | (16,812) | | - | | 34,980 | | (34,980) | - | | |
| Other income from Canadian development properties | - | | 91 | | 91 | | - | | 725 | 725 | | |
| Income from investments in for-sale housing | 2,964 | | - | | 2,964 | | 9,646 | | - | 9,646 | | |
| Property management overhead | - | | (5,675) | | (5,675) | | - | | (25,875) | (25,875) | | |
| Compensation expense | (9,744) | | - | | (9,744) | | (37,681) | | - | (37,681) | | |
| General and administration expense | (2,876) | | (3,049) | | (5,925) | | (11,683) | | (9,163) | (20,846) | | |
| Other expense | - | | (1,004) | | (1,004) | | - | | (3,991) | (3,991) | | |
| Interest expense | (8,908) | | (34,743) | | (43,651) | | (32,439) | | (119,870) | (152,309) | | |
| Fair value gain on rental properties | - | | 32,025 | | 32,025 | | - | | 116,548 | 116,548 | | |
| Gain on sale of U.S. multi-family developments | - | | 1,113 | | 1,113 | | - | | 9,718 | 9,718 | | |
| Fair value (loss) gain on derivative financial instruments | | | | | | | | | | | | |
| and other liabilities | (1,348) | | (114) | | (1,462) | | 2,961 | | (604) | 2,357 | | |
| Transaction costs | (3,713) | | (2,819) | | (6,532) | | (32,626) | | (3,789) | (36,415) | | |
| Amortization and depreciation expense | (1,589) | | (1,144) | | (2,733) | | (6,274) | | (4,269) | (10,543) | | |
| Realized and unrealized foreign exchange gain | 178 | | - | | 178 | | 42 | | - | 42 | | |
| Net change in fair value of limited partners' interests | | | | | | | | | | | | |
| in rental business | | | (4,210) | | (4,210) | | - | | (3,784) | (3,784) | | |
| | 34,227 | | (71,376) | | (37,149) | | 89,119 | | (233,813) | (144,694) | | |
| Income before income taxes | \$ 45,943 | \$ | (1,132) | \$ | 44,811 | \$ | 129,014 | \$ | (1,350) | \$ 127,664 | | |
| Income tax recovery (expense) – current | 1,974 | | - | | 1,974 | | (5,410) | | 15 | (5,395) | | |
| Income tax expense – deferred | (2,658) | | (570) | | (3,228) | | (9,469) | | (2,465) | (11,934) | | |
| Net income ⁽¹⁾ | \$ 45,259 | \$ | (1,702) | \$ | 43,557 | \$ | 114,135 | \$ | (3,800) | \$ 110,335 | | |
| Attributable to: | | | | | | | | | | | | |
| Shareholders of Tricon | \$ 44,056 | \$ | (1,702) | \$ | 42,354 | \$ | 111,562 | \$ | (3,800) | \$ 107,762 | | |
| Non-controlling interest | 1,203 | | - | | 1,203 | | 2,573 | | - | 2,573 | | |
| Net income ⁽¹⁾ | \$ 45,259 | \$ | (1,702) | \$ | 43,557 | \$ | 114,135 | \$ | (3,800) | \$ 110,335 | | |
| Other comprehensive income | | | | | | | | | | | | |
| Items that will be reclassified subsequently to net income | | | | | | | | | | | | |
| Cumulative translation reserve ⁽¹⁾ | (33) | | 1,702 | | 1,669 | | (129) | | 3,800 | 3,671 | | |
| Comprehensive income for the period | \$ 45,226 | \$ | - | \$ | 45,226 | \$ | 114,006 | \$ | - | \$ 114,006 | | |
| Attributable to: | | | | | | | | | | | | |
| Shareholders of Tricon | \$ 44,023 | \$ | - | \$ | 44,023 | \$ | 111,433 | \$ | - | \$ 111,433 | | |
| Non-controlling interest | 1,203 | | - | | 1,203 | | 2,573 | | - | 2,573 | | |
| Comprehensive income for the period | \$ 45,226 | \$ | - | \$ | 45,226 | \$ | 114,006 | \$ | - | \$ 114,006 | | |
| Basic EPS attributable to shareholders of Tricon ⁽¹⁾ | \$ 0.23 | \$ | (0.01) | \$ | 0.22 | \$ | 0.65 | \$ | (0.03) | \$ 0.62 | | |
| Diluted EPS attributable to shareholders of Tricon ⁽¹⁾ | \$ 0.22 | \$ | (0.01) | \$ | 0.21 | \$ | 0.63 | \$ | (0.02) | \$ 0.61 | | |
| Weighted average shares outstanding – basic | 195,269,680 | | - | 195, | ,269,680 | 172 | ,735,776 | | - | 172,735,776 | | |
| Weighted average shares outstanding – diluted | 213,682,237 | | - | 213, | ,682,237 | 191 | ,081,128 | | - | 191,081,128 | | |

(1) The effects of changes in foreign exchange rates for Canadian multi-family developments were accounted for as investment income under investment entity accounting. Such exchange differences are recognized in other comprehensive income for the Company upon adoption of consolidation accounting framework. As a result, basic and diluted EPS as recast have decreased compared to the amounts under investment entity accounting, as other comprehensive income is not included in net income.

Revenue from rental properties

The following table provides further details regarding revenue from rental properties for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | | Three months | | Twelve months | | | | | |
|-----------------------------------|------------|--------------|-----------|---------------|------------|------------|--|--|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | | | |
| Single-family rental | \$ 94,400 | \$ 81,348 | \$ 13,052 | \$ 366,982 | \$ 297,956 | \$ 69,026 | | | |
| Multi-family rental – U.S. | 27,583 | 28,567 | (984) | 111,205 | 63,810 | 47,395 | | | |
| Revenue from rental properties | \$ 121,983 | \$ 109,915 | \$ 12,068 | \$ 478,187 | \$ 361,766 | \$ 116,421 | | | |

Revenue from rental properties for the three months ended December 31, 2020 totalled \$122.0 million, an increase of \$12.1 million compared to \$109.9 million for the same period in the prior year. The increase is attributable to:

- An increase of \$13.1 million in rental revenue from single-family rental properties reflecting (i) an 8.3% portfolio expansion (22,766 rental homes compared to 21,014), (ii) 4.2% growth in average effective rent per home (\$1,464 compared to \$1,405), and (iii) a 1.9% increase in occupancy (96.4% compared to 94.5%).
- A partially offsetting decrease of \$1.0 million in rental revenue from the U.S. multi-family rental portfolio, driven by (i) a 1.3% decline in occupancy (93.6% compared to 94.9%), (ii) a \$0.5 million (or 130%) increase in bad debt expense (\$0.8 million compared to \$0.3 million) as a result of higher resident delinquency, and (iii) a \$0.4 million (or 169%) increase in leasing concessions (\$0.6 million compared to \$0.2 million) from an effort to drive occupancy.

Revenue from rental properties for the twelve months ended December 31, 2020 totalled \$478.2 million, an increase of \$116.4 million from the prior year as a result of (i) the expansion of the single-family rental portfolio as well as improvements in occupancy and average monthly rent as discussed above, and (ii) the inclusion of twelve months of revenue from the U.S. multi-family portfolio in 2020 compared to a seven-month inclusion in the comparative period, as the portfolio was acquired in the second quarter of 2019.

Direct operating expenses

The following table provides further details regarding direct operating expenses for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | | Three months | | | Twelve months | | | | |
|-----------------------------------|-----------|--------------|------------|------------|---------------|-------------|--|--|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | | | |
| Single-family rental | \$ 30,681 | \$ 28,490 | \$ (2,191) | \$ 121,242 | \$ 104,605 | \$ (16,637) | | | |
| Multi-family rental – U.S. | 11,979 | 11,603 | (376) | 48,296 | 25,863 | (22,433) | | | |
| Direct operating expenses | \$ 42,660 | \$ 40,093 | \$ (2,567) | \$ 169,538 | \$ 130,468 | \$ (39,070) | | | |

Direct operating expenses for the three months ended December 31, 2020 were \$42.7 million, an increase of \$2.6 million compared to the same period in the prior year. The variance is attributable to:

- An increase of \$2.2 million from the single-family rental portfolio, driven by (i) an 8.3% growth in the size of the portfolio (1,752 more rental homes in service in Q4 2020 compared to Q4 2019), (ii) a 5.0% increase in property taxes, and (iii) a partially offsetting decrease in repairs, maintenance and turnover expenses owing to a 3.1% decrease in turnover as well as improved cost discipline and controlled scoping of maintenance work.
- An increase of \$0.4 million from the U.S. multi-family rental portfolio, reflecting a 27.9% increase in property insurance premiums due to industry-wide price escalation and normal-course salary increases for on-site property management personnel.

Direct operating expenses for the twelve months ended December 31, 2020 were \$169.5 million, an increase of \$39.1 million compared to the prior year. This variance is primarily attributable to (i) the acquisition of the U.S. multi-family rental portfolio in the second quarter of 2019, resulting in the inclusion of twelve months of operating expenses in 2020 compared to seven months in 2019, and (ii) the expansion of the single-family rental portfolio year-over-year along with the property tax increases discussed above.

Revenue from private funds and advisory services

The following table provides further details regarding revenue from private funds and advisory services for the three and twelve months ended December 31, 2020, net of inter-segment revenues eliminated upon consolidation. Under certain arrangements, asset-based fees that are earned from third-party investors in Tricon's subsidiary entities are billed directly to those investors and are therefore not recognized in the accounts of the applicable subsidiary. These amounts are included in the asset management fees revenue recognized in the statements of comprehensive income.

| For the periods ended December 31 | | Three months | | Twelve months | | | | | |
|-----------------------------------|-----------|--------------|------------|---------------|-----------|------------|--|--|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | | | |
| Asset management fees | \$ 2,815 | \$ 3,355 | \$ (540) | \$ 12,061 | \$ 15,099 | \$ (3,038) | | | |
| Performance fees | 1,691 | 2,565 | (874) | 2,836 | 7,448 | (4,612) | | | |
| Development fees ⁽¹⁾ | 5,653 | 5,876 | (223) | 18,298 | 17,736 | 562 | | | |
| Property management fees | 180 | 342 | (162) | 895 | 777 | 118 | | | |
| Revenue from private funds | | | | | | | | | |
| and advisory services | \$ 10,339 | \$ 12,138 | \$ (1,799) | \$ 34,090 | \$ 41,060 | \$ (6,970) | | | |

(1) Development fees are comprised of fees earned by:

| For the periods ended December 31 | | Three months | | Twelve months | | | | | |
|-----------------------------------|----------|--------------|----------|---------------|-----------|------------|--|--|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | | | |
| The Johnson Companies ("Johnson") | \$ 4,833 | \$ 5,509 | \$ (676) | \$ 14,586 | \$ 15,726 | \$ (1,140) | | | |
| Tricon Development Group ("TDG") | 820 | 367 | 453 | 3,712 | 2,010 | 1,702 | | | |
| Development fees | \$ 5,653 | \$ 5,876 | \$ (223) | \$ 18,298 | \$ 17,736 | \$ 562 | | | |

Revenue from private funds and advisory services for the three months ended December 31, 2020 totalled \$10.3 million, a decrease of \$1.8 million from the same period in the prior year. The variance is primarily attributable to:

- A decrease of \$0.9 million in performance fees generated from the Company's for-sale housing investments portfolio, owing to higher performance fees paid in the fourth quarter of 2019 from commingled funds and separate accounts compared to the current period. Performance fees are paid when realized returns from an Investment Vehicle exceed third-party investor return thresholds, and are therefore episodic in nature and can vary substantially from period to period.
- A decrease of \$0.5 million in asset management fees, as significant distributions were made from investments in for-sale housing to third-party investors over the past twelve months, thereby reducing the outstanding invested capital upon which asset management fees are based.
- A decrease of \$0.2 million in development fees driven by (i) a \$0.7 million reduction from Johnson attributable to fewer lot sales in the fourth quarter of 2020 compared to the comparative period (see below), and (ii) a partially offsetting increase of \$0.5 million in development fees earned from Canadian multi-family developments at Blocks 3/4/7 of the West Don Lands project.

Johnson's development fees are generated based on the number of lots sold to homebuilders. While Johnson does not generate revenues from third-party homes sales, the number of homes sold is indicative of Johnson's expected future performance as homebuilders must replenish inventories to accommodate future demand. In spite of the COVID-19 pandemic, the for-sale housing market has fared well, underpinned by ultra-low mortgage rates, de-densification and de-urbanization trends and extended work-from-home orders, which have all led to higher demand for detached houses. Third-party home sales at Johnson communities increased by 24% year-over-year (2020 – 4,876 vs. 2019 – 3,920), which is expected to drive homebuilders' demand for lot inventory in the coming periods.

Revenue from private funds and advisory services for the twelve months ended December 31, 2020 totalled \$34.1 million, a decrease of \$7.0 million from the prior year largely for the reasons discussed above.

Income from investments in Canadian multi-family developments

Investments in Canadian multi-family developments include joint ventures and equity holdings in development projects, which are equity-accounted for in accordance with IAS 28 (as defined in <u>Section 8</u>), namely The Taylor, West Don Lands, The Ivy and 7 Labatt. The Selby, a Canadian multi-family rental property, is also accounted for under the equity method while its operational results are discussed within the Canadian multi-family rental segment in <u>Section 4.2</u> as the property is now substantially stabilized. The James (Scrivener Square) and The Shops of Summerhill are accounted for as Canadian development properties. The income earned from The Shops of Summerhill is grouped with Other income (expenses) given its immaterial nature.

The following table provides further details regarding income from investments in Canadian multi-family developments for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | | Three months | | Twelve months | | | | | | |
|---|----------|--------------|----------|---------------|----------|----------|--|--|--|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | | | | |
| Multi-family rental – Canada (The Selby) ⁽¹⁾ | \$ 427 | \$ 535 | \$ (108) | \$ 746 | \$ (564) | \$ 1,310 | | | | |
| Multi-family rental – Canada | | | | | | | | | | |
| (under development) | 8,293 | 6,881 | 1,412 | 13,378 | 8,278 | 5,100 | | | | |
| Income from investments in | | | | | | | | | | |
| Canadian multi-family developments | \$ 8,720 | \$ 7,416 | \$ 1,304 | \$ 14,124 | \$ 7,714 | \$ 6,410 | | | | |

(1) See Section 4.2.2, "Canadian multi-family rental - The Selby", for details of the operational performance of The Selby.

Income from investments in Canadian multi-family developments for the three months ended December 31, 2020 was \$8.7 million, an increase of \$1.3 million from the same period in the prior year. This variance was driven by:

- An increase of \$1.4 million in income from the Company's development projects, primarily attributable to fair value gains recognized on Blocks 3/4/7 of the West Don Lands project, which achieved development milestones in the current period. In comparison, the fair value gains in the comparative period were driven by land value increases across multiple projects, which moderated in 2020.
- A partially offsetting decrease of \$0.1 million in income from The Selby, as higher fair value gains were recognized in the fourth quarter of 2019 while no fair value gains were recorded in the same period in 2020.

Income from investments in Canadian multi-family developments for the twelve months ended December 31, 2020 was \$14.1 million, an increase of \$6.4 million from the prior year. The variance is attributable to:

- An increase of \$5.1 million in income from the Company's development projects, reflecting fair value gains on multiple blocks of the West Don Lands project, including Block 8 which commenced construction in 2020.
- An additional increase of \$1.3 million in income from The Selby. The Selby generated positive net operating income ("NOI") in 2020 following its substantial stabilization, compared to a loss during the lease-up period in 2019. Tricon's share of NOI was \$0.9 million, reflecting occupancy of 87.0% and average monthly rent of \$2,663 (in Canadian dollars) for the year (see Section 4.2.2).

Income (loss) from investments in for-sale housing

The following table presents the income (loss) from investments in for-sale housing for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | | Three months | | Twelve months | | | | | |
|-----------------------------------|-----------|--------------|----------|---------------|----------|-------------|--|--|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | | | |
| Income (loss) from investments | | | | | | | | | |
| in for-sale housing | \$ 10,191 | \$ 2,964 | \$ 7,227 | \$ (61,776) | \$ 9,646 | \$ (71,422) | | | |

Income from investments in for-sale housing for the three months ended December 31, 2020 was \$10.2 million, an increase of \$7.2 million from the same period in the prior year. This increase was driven largely by higher valuations at certain projects which are experiencing improved performance, reflecting stronger housing demand buoyed by historically low mortgage rates, de-urbanization trends and extended work-from-home mandates.

Loss from investments in for-sale housing for the twelve months ended December 31, 2020 was \$61.8 million, a decrease of \$71.4 million compared to income of \$9.6 million for 2019. The variance was driven by a fair value loss of \$79.6 million recorded in the first quarter of 2020, which was partially recovered in the latter part of the year through improvements in project performance as discussed above. While the for-sale housing market outlook for 2021 appears favourable, management is also mindful of rising labour and material costs which could partially offset rising home prices and high absorption rates.

Property management overhead

Property management overhead costs are corporate-level costs (including salaries of employees engaged in leasing, acquisition, disposition and other direct property management-related activities) and are not direct property-level costs included in NOI.

The following table provides further details regarding property management overhead for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | | Three months | | Twelve months | | | | | |
|---|----------|--------------|----------|---------------|-----------|----------|--|--|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | | | |
| Property management salaries | | | | | | | | | |
| and benefits | \$ 3,384 | \$ 3,432 | \$ 48 | \$ 12,903 | \$ 15,873 | \$ 2,970 | | | |
| Other property management overhead ⁽¹⁾ | 2,488 | 2,243 | (245) | 9,751 | 10,002 | 251 | | | |
| Property management overhead | \$ 5,872 | \$ 5,675 | \$ (197) | \$ 22,654 | \$ 25,875 | \$ 3,221 | | | |

(1) Includes general and administration expenses, marketing and other expenses attributable to the property management platform.

Property management overhead for the three months ended December 31, 2020 was \$5.9 million, an increase of \$0.2 million compared to the same period in the prior year. The increase was primarily attributable to incremental IT subscription costs in scaling the operating platform and online marketing initiatives to drive virtual leasing activities.

Property management overhead for the twelve months ended December 31, 2020 was \$22.7 million, a decrease of \$3.2 million compared to the prior year. The favourable variance is primarily driven by increased allocation of property management salaries to direct operating costs, as the rental portfolio under management continued to expand. The amount of property management overhead allocated to direct operating costs correlates with the amount of revenue earned from rental properties during the period. In addition, lower travel costs and contained overhead expenses due to constrained property management activities during the COVID-19 pandemic drove savings in overhead expenses.

Compensation expense

The following table provides further details regarding compensation expense, excluding the compensation expense for direct property management employees noted above, for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | | | Three months | | | Twelve months | |
|------------------------------------|-----|-----------|--------------|------------|-----------|---------------|------------|
| (in thousands of U.S. dollars) | | 2020 | 2019 | Variance | 2020 | 2019 | Variance |
| Salaries and benefits | A | \$ 6,319 | \$ 5,139 | \$ (1,180) | \$ 21,451 | \$ 19,198 | \$ (2,253) |
| Cash-settled ⁽¹⁾ | | 6,370 | 2,584 | (3,786) | 15,721 | 11,661 | (4,060) |
| Equity-settled ⁽²⁾ | | 198 | 290 | 92 | 2,066 | 2,194 | 128 |
| Annual incentive plan ("AIP") | B | 6,568 | 2,874 | (3,694) | 17,787 | 13,855 | (3,932) |
| Cash-settled ⁽¹⁾ | | 1,549 | 1,317 | (232) | (2,051) | 2,843 | 4,894 |
| Equity-settled ⁽²⁾ | | 504 | 414 | (90) | 2,913 | 1,785 | (1,128) |
| Long-term incentive plan | | | | | | | |
| ("LTIP") | C | 2,053 | 1,731 | (322) | 862 | 4,628 | 3,766 |
| Total compensation expense 🛛 🗛 + 🖪 | + C | \$ 14,940 | \$ 9,744 | \$ (5,196) | \$ 40,100 | \$ 37,681 | \$ (2,419) |

(1) Includes cash component and performance share units.

(2) Includes deferred share units, stock options and restricted shares.

Compensation expense for the three months ended December 31, 2020 was \$14.9 million, an increase of \$5.2 million compared to the same period in the prior year. The variance is attributable to:

- An increase of \$3.7 million in AIP expense, which represents: (i) an expanded base salary pool on which the AIP is measured;
 (ii) an increase to previously accrued expenses throughout the year, as the year-end AIP payable was finalized based on actual performance; and (iii) an increase in cash-settled PSUs, which incorporates existing and new entitlements as well as an increase from revaluation owing to the Company's share price increase. As at December 31, 2020, the Company's share price was \$8.98 (C\$11.43), a 26% increase compared to \$7.10 (C\$9.69) as at the end of 2018, the year in which the PSU plan was first implemented.
- An increase of \$1.2 million in payroll costs related to additional staffing to support Tricon's continued growth as well as normal course salary adjustments.

Compensation expense for the twelve months ended December 31, 2020 was \$40.1 million, an increase of \$2.4 million compared to the prior year. The variance is driven by:

- An increase of \$3.9 million and \$2.3 million in AIP expense and salaries and benefits, respectively, for the reasons discussed above.
- A partially offsetting decrease of \$3.8 million in LTIP expense, which corresponds to the significant write-down of Tricon's investments in for-sale housing, resulting in lower estimated future performance fees to be paid to management under the LTIP.

General and administration expense

The following table presents general and administration expense for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | Three months | | | | | | | Twel | ve months | |
|------------------------------------|--------------|-------|----|-------|----|----------|--------------|------|-----------|---------------|
| (in thousands of U.S. dollars) | | 2020 | | 2019 | ١ | /ariance | 2020 | | 2019 | Variance |
| General and administration expense | \$ | 5,748 | \$ | 5,925 | \$ | 177 | \$ 23,569 | \$ | 20,846 | \$ (2,723) |

General and administration expense for the three months ended December 31, 2020 was \$5.7 million, a decrease of \$0.2 million compared to the same period in the prior year. This favourable variance is driven by lower professional fees incurred in the fourth quarter of 2020 compared to the comparative period, which included additional expenses related to the implementation of the Company's ESG roadmap, among other items.

General and administration expense for the twelve months ended December 31, 2020 increased by \$2.7 million compared to the prior year, driven by higher consulting costs for various improvement initiatives and increased franchise tax in certain states, both of which corresponded with the Company's growing business activities. Notably, the comparative period includes only seven months of additional overhead costs from Tricon's expansion into the U.S. multi-family rental business in June 2019, compared to a full year of cost inclusion in 2020.

Interest expense

The following table provides details regarding interest expense for the three and twelve months ended December 31, 2020 by borrowing type.

| For the periods ended December 31 | ded December 31 Three months | | | | Twelve months | | | | |
|-----------------------------------|------------------------------|-----------|----------|------------|---------------|-------------|--|--|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | | | |
| Corporate borrowings | \$ 1,018 | \$ 5,297 | \$ 4,279 | \$ 13,032 | \$ 18,173 | \$ 5,141 | | | |
| Property-level borrowings | 34,209 | 34,677 | 468 | 135,562 | 119,661 | (15,901) | | | |
| Convertible debentures | 3,687 | 3,595 | (92) | 14,572 | 14,201 | (371) | | | |
| Due to Affiliate | 5,429 | _ | (5,429) | 7,116 | - | (7,116) | | | |
| Lease obligations | 78 | 82 | 4 | 328 | 274 | (54) | | | |
| Total interest expense | \$ 44,421 | \$ 43,651 | \$ (770) | \$ 170,610 | \$ 152,309 | \$ (18,301) | | | |
| Weighted average interest rate | | | | 3.12% | 3.95% | 0.83% | | | |

The following table provides further details regarding interest expense by its nature (cash interest and non-cash interest expense, such as amortization).

| For the periods ended December 31 | | Three months | | Twelve months | | | |
|---|-----------|--------------|----------|---------------|------------|-------------|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | |
| Corporate and property-level borrowings | \$ 33,873 | \$ 39,104 | \$ 5,231 | \$ 144,107 | \$ 135,326 | \$ (8,781) | |
| Convertible debentures | 2,506 | 2,492 | (14) | 9,927 | 9,902 | (25) | |
| Due to Affiliate | 4,312 | _ | (4,312) | 5,654 | _ | (5,654) | |
| Amortization of deferred financing costs, | | | | | | | |
| discounts and lease obligations | 3,730 | 2,055 | (1,675) | 10,922 | 7,081 | (3,841) | |
| Total interest expense | \$ 44,421 | \$ 43,651 | \$ (770) | \$ 170,610 | \$ 152,309 | \$ (18,301) | |

Interest expense was \$44.4 million for the three months ended December 31, 2020, an increase of \$0.8 million compared to \$43.7 million for the same period last year. The variance is primarily attributable to:

- A \$5.2 million decrease in interest expense on corporate and property-level borrowings, despite higher borrowings compared to the same period in the prior year. The \$5.2 million decrease is mainly driven by the following:
 - (i) Corporate credit facility interest expense decreased by \$4.3 million as a result of a \$271.0 million reduction in the carrying balance of this facility in the past year, from \$297.0 million to \$26.0 million as at December 31, 2020. In addition, the average effective interest rate on this facility decreased by 0.8% (from 3.82% in Q4 2019 to 3.06% in Q4 2020).
 - (ii) Interest expense on property-level debt decreased by \$0.5 million as a result of a 1.0% decrease in the average effective interest rates, which outweighed a \$0.2 billion increase in the average debt balance. The reduction in the effective interest rate was driven by advantageous refinancing of properties at lower fixed-rate terms (see Section 3.2) along with a 1.6% decrease in LIBOR (from 1.79% in Q4 2019 to 0.15% in Q4 2020). These savings underscore management's efforts to refinance existing debt at lower prevailing interest rates as well to stagger Tricon's debt maturities.
- An offsetting \$4.3 million increase in interest expense on the balance Due to Affiliate in connection with the preferred share issuance in September 2020 (see Section 3.2). These interest payments are to fund dividend payments by PIPE LLC.

Interest expense was \$170.6 million for the twelve months ended December 31, 2020, an increase of \$18.3 million compared to \$152.3 million for the prior year. The variance is attributable to:

- An \$8.8 million increase in interest expense on corporate and property-level borrowings, driven primarily by (i) additional debt assumed in relation to the Company's acquisition of the U.S multi-family rental portfolio in the second quarter of 2019, and (ii) additional debt incurred to finance the Company's growing portfolio of single-family rental homes.
- A \$5.6 million increase in interest expense on the balance Due to Affiliate as discussed above.
- A \$3.8 million increase in the amortization of deferred financing costs and discounts, attributable to costs incurred for the aforementioned Due to Affiliate and incremental debt for the acquisition of single-family rental homes.

Fair value gain on rental properties

The following table presents the fair value gain on rental properties held by the Company for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | Three months | | | | Twelve months | |
|--------------------------------------|--------------|-----------|-----------|------------|---------------|-----------|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance |
| Fair value gain on rental properties | \$ 106,995 | \$ 32,025 | \$ 74,970 | \$ 198,314 | \$ 116,548 | \$ 81,766 |

Fair value gain on rental properties was \$107.0 million for the three months ended December 31, 2020, an increase of \$75.0 million compared to \$32.0 million for the same period last year, which is attributable to the single-family rental portfolio. The fair value of single-family rental homes is typically determined by using a combination of Broker Price Opinion ("BPO") and the Home Price Index ("HPI") methodologies.

Higher home pricing is attributable to population growth in Sun Belt markets owing to in-migration, de-densification and de-urbanization trends, all of which have strengthened demand for suburban homes. This increased demand coupled with limited supply drove HPI growth during the three months ended December 31, 2020, which was 1.5% (6.0% annualized), net of capital expenditures, compared to 0.7% (2.8% annualized) in the same period in the prior year. The HPI and BPO methodologies were also applied to a larger portfolio of homes in the three months ended December 31, 2020 than in the comparative period, driving even higher fair value gains.

Fair value gain on rental properties for the twelve months ended December 31, 2020 totalled \$198.3 million, an increase of \$81.8 million from the prior year. The increase was primarily attributable to a \$220.8 million fair value gain on the single-family rental portfolio driven by home price appreciation influenced by the factors discussed above. The variance was partially offset by a fair value loss of \$22.5 million on the U.S. multi-family rental portfolio as reduced demand for multi-family living contributed to a downward adjustment in stabilized NOI assumptions in the second quarter of 2020.

Fair value (loss) gain on derivative financial instruments and other liabilities

The following table presents the fair value (loss) gain on derivative financial instruments and other liabilities for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | Three months | | | Twelve months | | | | |
|--|--------------|------------|-------------|---------------|----------|------------|--|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | | |
| Fair value (loss) gain on derivative financial | | | | | | | | |
| instruments and other liabilities | \$ (16,418) | \$ (1,462) | \$ (14,956) | \$ (7,461) | \$ 2,357 | \$ (9,818) | | |

For the three months ended December 31, 2020, the fair value loss on derivative financial instruments and other liabilities increased by \$15.0 million to \$16.4 million compared to \$1.5 million in the same period in the prior year. This unfavourable variance is attributable to a new derivative liability added in the third quarter in connection with the exchangeable preferred units issued by Tricon PIPE LLC (see Section 3.2).

The fair value loss on the derivative financial instruments was driven by an increase in Tricon's share price, on a USD converted basis, which served to increase the probability of exchange of the exchangeable preferred units into Tricon's common shares. This increased conversion probability drove the increase in the derivative liability of the Company.

For the twelve months ended December 31, 2020, the fair value loss on derivative financial instruments and other liabilities increased by \$9.8 million to \$7.5 million compared to a \$2.4 million gain in the prior year. The variance is primarily driven by:

- A \$7.9 million increase in the derivative liability in connection with the aforementioned exchangeable preferred units issued by Tricon PIPE LLC in the third quarter of 2020.
- A \$1.0 million loss on a previously-outstanding put liability, compared to a \$0.3 million loss in 2019, which was redeemed on March 4, 2020 in connection with the Company's acquisition and cancellation of 1,867,675 outstanding common shares (see Section 3.2).
- A partially offsetting \$1.5 million gain on the embedded derivative on the 2022 convertible debentures, reflecting an increase in the value of Tricon's redemption option relative to the holders' conversion option. In the comparative period, the fair value gain of \$3.3 million related primarily to the conversion option, driven by a reduction in the time remaining until option expiration, among other factors.

Transaction costs

The following table presents the transaction costs for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | Three months | | | Twelve months | | | |
|-----------------------------------|--------------|----------|----------|---------------|-----------|-----------|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | |
| Transaction costs | \$ 2,491 | \$ 6,532 | \$ 4,041 | \$ 14,016 | \$ 36,415 | \$ 22,399 | |

For the three months ended December 31, 2020, transaction costs were \$2.5 million, a decrease of \$4.0 million compared to the same period in the prior year. The decrease was primarily driven by costs incurred for the acquisition of the U.S. multi-family rental portfolio in the comparative period.

For the twelve months ended December 31, 2020, transaction costs were \$14.0 million, a decrease of \$22.4 million compared to the prior year, driven primarily by \$28.0 million of transaction costs incurred for the acquisition of the U.S. multi-family rental portfolio in 2019. This decrease was partially offset by transaction costs incurred in connection with the issuance of the exchangeable preferred units by Tricon PIPE LLC. The Company incurred \$15.2 million of transaction costs in connection with the Transaction costs in connection with the State of the transaction, of which \$12.2 million was capitalized and \$3.0 million was expensed as transaction costs (see Section 3.2).

Net change in fair value of limited partners' interests in rental business

Ownership interests in the single-family rental joint venture ("SFR JV-1") are in the form of non-controlling limited partnership interests which are classified as liabilities under the provisions of IFRS. The following table presents the net change in fair value of limited partners' interests in rental business for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | Three months | | | Twelve months | | | |
|--|--------------|------------|-------------|---------------|------------|-------------|--|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | |
| Net change in fair value of limited | | | | | | | |
| partners' interests in rental business | \$ (17,780) | \$ (4,210) | \$ (13,570) | \$ (50,581) | \$ (3,784) | \$ (46,797) | |

For the three months ended December 31, 2020, the change in fair value of limited partners' interests in rental business was \$17.8 million compared to \$4.2 million for the same period in the prior year, representing an increase in non-controlling limited partners interests of \$13.6 million. This increase in non-controlling limited partners' interests mainly reflects additional income earned from SFR JV-1 during the year that is attributable to the Company's joint venture partners. The increase in income was driven largely by NOI growth associated with a larger portfolio and a higher fair value gain on rental properties in which SFR JV-1 invests.

For the twelve months ended December 31, 2020, the fair value of limited partners' interests in rental business increased by \$50.6 million compared to an increase of \$3.8 million for the prior year, for the same reasons noted above.

Income tax expense

| For the periods ended December 31 | | Three months | | | | Twelve months | | | | | |
|---|--------|--------------|----|---------|----|---------------|----|---------|----|--------|----------------|
| (in thousands of U.S. dollars) | 2 | 2020 | | 2019 | | Variance | | 2020 | | 2019 | Variance |
| Income tax (recovery) expense – current | \$ (7, | ,087) | \$ | (1,974) | \$ | 5,113 | \$ | (4,050) | \$ | 5,395 | \$ 9,445 |
| Income tax expense – deferred | 33, | ,133 | | 3,228 | | (29,905) | | 40,425 | | 11,934 | (28,491) |
| Income tax expense – total | \$26, | 046 | \$ | 1,254 | \$ | (24,792) | \$ | 36,375 | \$ | 17,329 | \$ (19,046) |

For the three months ended December 31, 2020, income tax expense was \$26.0 million, an increase of \$24.8 million compared to the same period in the prior year driven primarily by a higher deferred tax expense of \$29.9 million related to the fair value gain on rental properties. This variance was partially offset by an increase in the current tax recovery of \$5.1 million that arose from the utilization of a loss carryback provision which enabled the Company to apply current year tax losses from certain corporate entities against taxes paid in previous periods.

For the twelve months ended December 31, 2020, income tax expense was \$36.4 million, an increase of \$19.0 million compared to the prior year for the reasons discussed above. In addition, the comparative period included the current tax impact of the gain on the sale of U.S. multi-family developments.

3.2 Review of selected balance sheet items

The comparative figures in the Company's consolidated balance sheets in the table below have been recast as if the current reporting framework under IFRS 10, which was first applied by the Company effective January 1, 2020 on a prospective basis, had been in effect as at December 31, 2019.

| As at | | |
|---|-------------------|--------------------|
| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
| | | Recast |
| ASSETS | | (Schedule B) |
| Non-current assets | | |
| Rental properties | \$ 6,321,918 | \$ 5,682,525 |
| Investments in Canadian multi-family developments | 94,868 | 75,141 |
| Canadian development properties | 110,018 | 35,625 |
| Investments in for-sale housing | 164,842 | 300,653 |
| Restricted cash | 116,302 | 84,082 |
| Goodwill | 108,838 | 108,838 |
| Intangible assets | 12,363 | 16,396 |
| Other assets | 47,990 | 42,071 |
| Deferred income tax assets | 102,444 | 84,749 |
| Derivative financial instruments | 841 | - |
| Total non-current assets | 7,080,424 | 6,430,080 |
| Current assets | | |
| Cash | 55,158 | 31,107 |
| Amounts receivable | 25,593 | 13,851 |
| Prepaid expenses and deposits | 13,659 | 11,358 |
| Total current assets | 94,410 | 56,316 |
| Total assets | \$ 7,174,834 | \$ 6,486,396 |
| LIABILITIES | | |
| Non-current liabilities | | |
| Long-term debt | \$ 3,863,316 | \$ 3,954,977 |
| Convertible debentures | 165,956 | 161,311 |
| Due to Affiliate | 251,647 | - |
| Derivative financial instruments | 45,494 | 629 |
| Limited partners' interests in rental business | 356,305 | 285,774 |
| Long-term incentive plan | 17,930 | 21,409 |
| Other liabilities | 4,599 | 19,764 |
| Deferred income tax liabilities | 298,071 | 240,723 |
| Total non-current liabilities | 5,003,318 | 4,684,587 |
| Current liabilities | | |
| Amounts payable and accrued liabilities | 98,290 | 97,744 |
| Resident security deposits | 45,157 | 32,125 |
| Dividends payable | 10,641 | 10,474 |
| Current portion of long-term debt | 274,190 | 284 |
| Total current liabilities | 428,278 | 140,627 |
| Total liabilities | 5,431,596 | 4,825,214 |
| Equity | | |
| Share capital | 1,192,963 | 1,201,061 |
| Share capital reserve | _ | (13,057) |
| Contributed surplus | 19,738 | 20,223 |
| Cumulative translation adjustment | 23,395 | 19,396 |
| Retained earnings | 499,000 | 425,515 |
| Total shareholders' equity | 1,735,096 | 1,653,138 |
| Non-controlling interest | 8,142 | 8,044 |
| Total equity | 1,743,238 | 1,661,182 |
| Total liabilities and equity | \$ 7,174,834 | \$ 6,486,396 |
| | φ /,1/4,034 | <i>↓</i> 0,480,390 |

Schedule B

The table below provides a reconciliation of balance sheet results as at December 31, 2019 from figures previously disclosed under investment entity accounting in accordance with IFRS 10 to the recast figures shown above.

| (in thousands of U.S. dollars) | Previously reported | Adjustments | Recast |
|---|------------------------|--------------|-----------------|
| ASSETS | reported | Aujustinents | |
| Non-current assets | | | |
| Rental properties | \$ – | \$ 5,682,525 | \$ 5,682,525 |
| Investments – Tricon American Homes | ↓ | (1,365,007) | φ 3,002,323 |
| Investments – Tricon Lifestyle Rentals | 525,932 | (525,932) | _ |
| Investments – fricon Ellestyle kentals | 525,552 | 75,141 | |
| Investments in canadian multi-lamity developments | 300,653 | 75,141 | 300,653 |
| Canadian development properties | 500,055 | 35,625 | 35.625 |
| Restricted cash | - | 84,082 | 84,082 |
| Goodwill | 219 | 108,619 | 108,838 |
| Intangible assets | 16,396 | 100,015 | 16,396 |
| Other assets | 30,677 | _ 11,394 | 42,071 |
| Deferred income tax assets | 44,749 | 40,000 | 84,749 |
| Total non-current assets | 2,283,633 | 4,146,447 | 6,430,080 |
| | 2,203,033 | 4,140,447 | 0,430,080 |
| Current assets | | | |
| Cash | 8,908 | 22,199 | 31,107 |
| Amounts receivable | 8,952 | 4,899 | 13,851 |
| Prepaid expenses and deposits | 796 | 10,562 | 11,358 |
| Total current assets | 18,656 | 37,660 | 56,316 |
| Total assets | \$ 2,302,289 | \$ 4,184,107 | \$ 6,486,396 |
| LIABILITIES | | | |
| Non-current liabilities | | | |
| Long-term debt | \$ 307,869 | \$ 3,647,108 | \$ 3,954,977 |
| Convertible debentures | 161,311 | - | 161,311 |
| Derivative financial instruments | 657 | (28) | 629 |
| Limited partners' interests in rental business | _ | 285,774 | 285,774 |
| Long-term incentive plan | 21,409 | - | 21,409 |
| Other liabilities | 14,329 | 5,435 | 19,764 |
| Deferred income tax liabilities | 98,584 | 142,139 | 240,723 |
| Total non-current liabilities | 604,159 | 4,080,428 | 4,684,587 |
| Current liabilities | | | |
| Amounts payable and accrued liabilities | 26,190 | 71,554 | 97,744 |
| Resident security deposits | | 32,125 | 32,125 |
| Dividends payable | 10,474 | - | 10,474 |
| Current portion of long-term debt | 284 | _ | 284 |
| Total current liabilities | 36,948 | 103,679 | 140,627 |
| Total liabilities | 641,107 | 4,184,107 | 4,825,214 |
| | ••••,••• | .,, | ., |
| Equity | 1 201 004 | | 1 204 004 |
| Share capital | 1,201,061 | - | 1,201,061 |
| Share capital reserve | (13,057) | - | (13,057) |
| Contributed surplus | 20,223 | - | 20,223 |
| Cumulative translation adjustment | 19,396 | _ | 19,396 |
| Retained earnings | 425,515 | _ | 425,515 |
| Total shareholders' equity | 1,653,138 | - | 1,653,138 |
| Non-controlling interest | 8,044 | _ | 8,044 |
| Total equity | 1,661,182 | - | 1,661,182 |
| Total liabilities and equity | \$ 2,302,289 | \$ 4,184,107 | \$ 6,486,396 |

Rental properties

The table below presents the changes in the fair value of rental properties by business segment for the years ended December 31, 2020 and December 31, 2019. The comparative figures in the table below have been recast as if the current reporting framework under IFRS 10, which was first applied by the Company effective January 1, 2020 on a prospective basis, had been in effect for the year ended December 31, 2019.

| | C | ecember 31, 202 | 0 | I | December 31, 2019 | | | |
|---|-------------------------|------------------------|--------------|-------------------------|------------------------|--------------|--|--|
| (in thousands of U.S. dollars) | Single-Family Rental | Multi-Family Rental | Total | Single-Family Rental | Multi-Family Rental | Total | | |
| Balance, beginning of year | \$ 4,337,681 | \$ 1,344,844 | \$ 5,682,525 | \$ 3,357,967 | \$ - | \$ 3,357,967 | | |
| Initial recognition for business combinations | - | - | - | - | 1,338,683 | 1,338,683 | | |
| Acquisitions | 356,514 | - | 356,514 | 733,370 | - | 733,370 | | |
| Capital expenditures | 93,568 | 9,067 | 102,635 | 115,238 | 6,161 | 121,399 | | |
| Dispositions | (18,070) | - | (18,070) | (18,809) | - | (18,809) | | |
| Fair value adjustments | 220,849 | (22,535) | 198,314 | 149,915 | - | 149,915 | | |
| Balance, end of year | \$ 4,990,542 | \$ 1,331,376 | \$ 6,321,918 | \$ 4,337,681 | \$ 1,344,844 | \$ 5,682,525 | | |

Rental properties increased by \$639.4 million to \$6.3 billion as at December 31, 2020, from \$5.7 billion as at December 31, 2019. The increase was driven by:

- Acquisitions of 1,836 single-family rental homes for \$356.5 million, partially offset by the disposition of 119 properties with an aggregate carrying value of \$18.1 million.
- Capital expenditures of \$102.6 million of which \$93.6 million was attributable to the renovation of newly-acquired single-family homes as well as the maintenance and improvement of homes across the single-family rental portfolio. In addition, \$9.1 million was invested in U.S. multi-family properties to enhance common area amenities and restore units.
- Fair value gain of \$220.8 million on the single-family rental portfolio driven by higher demand for single-family homes, as previously discussed, combined with relatively limited supply in the Company's Sun Belt markets that contributed to significant home price appreciation. This significant fair value gain was partially offset by a \$22.5 million fair value loss on the multi-family portfolio recognized in the second quarter of 2020, reflecting the negative impact of COVID-19 on NOI assumptions.

Investments in Canadian multi-family developments

The table below presents the change in investments in Canadian multi-family developments for the twelve months ended December 31, 2020.

| (in thousands of U.S. dollars) | As at December 31, 2019 | Advances | Distributions | Income | Translation adjustment | As at December 31, 2020 |
|--|----------------------------|----------|---------------|-----------|---------------------------|----------------------------|
| Multi-family rental – Canada (The Selby) ⁽¹⁾ | \$ 19,733 | \$ – | \$ (935) | \$ 746 | \$ 369 | \$ 19,913 |
| Multi-family rental – Canada (under development) ⁽²⁾ | 55,408 | 4,294 | _ | 13,378 | 1,875 | 74,955 |
| Investments in Canadian multi-family developments | \$ 75,141 | \$ 4,294 | \$ (935) | \$ 14,124 | \$ 2,244 | \$ 94,868 |

(1) See Section 4.2.2, "Canadian multi-family rental – The Selby", for details of the operational performance of The Selby.

(2) See Section 4.3.1, "Canadian Class A multi-family developments", for details of Canadian multi-family projects under development.

Investments in Canadian multi-family developments increased by \$19.7 million to \$94.9 million as at December 31, 2020 compared to \$75.1 million as at December 31, 2019. The increase was primarily attributable to (i) income of \$14.1 million, mainly related to fair value gains from the West Don Lands projects, which achieved significant development and construction milestones during the year, (ii) advances to development projects of \$4.3 million, and (iii) a favourable foreign exchange translation adjustment of \$2.2 million driven by a stronger Canadian dollar, partially offset by (iv) distributions of \$0.9 million from The Selby.

Canadian development properties

The table below presents the change in investments in Canadian development properties, which are comprised of The James (Scrivener Square) and The Shops of Summerhill, for the twelve months ended December 31, 2020.

| | As at | | Development | Translation | As at |
|---------------------------------|-------------------|--------------|--------------|-------------|-------------------|
| (in thousands of U.S. dollars) | December 31, 2019 | Acquisitions | expenditures | adjustment | December 31, 2020 |
| Canadian development properties | \$ 35,625 | \$ 65,861 | \$ 2,998 | \$ 5,534 | \$ 110,018 |

The Company's Canadian development properties increased by \$74.4 million to \$110.0 million as at December 31, 2020 compared to \$35.6 million as at December 31, 2019. The increase was attributable to (i) the acquisition of third-party ownership interests in The James and The Shops of Summerhill for \$65.9 million during the second quarter of 2020, (ii) a favourable foreign exchange translation adjustment of \$5.5 million driven by a stronger Canadian dollar, and (iii) development expenditures of \$3.0 million at The James.

Investments in for-sale housing

The table below presents the change in investments in for-sale housing for the twelve months ended December 31, 2020.

| | | | Loss from | | |
|---------------------------------|-------------------|----------|------------------|---------------|-------------------|
| | As at | | investments in | | As at |
| (in thousands of U.S. dollars) | December 31, 2019 | Advances | for-sale housing | Distributions | December 31, 2020 |
| Investments in for-sale housing | \$ 300,653 | \$ 3,408 | \$ (61,776) | \$ (77,443) | \$ 164,842 |

Investments in for-sale housing decreased by \$135.8 million to \$164.8 million as at December 31, 2020 compared to \$300.7 million as at December 31, 2019. The decrease was attributable to (i) distributions of \$77.4 million primarily from the syndication of a balance sheet investment in the first quarter, (ii) a cumulative fair value loss of \$61.8 million driven by the write-down recognized in the first quarter of 2020 as a result of negative revisions in expected project performance and uncertainty of cash flows caused by the onset of COVID-19, partially offset by (iii) advances to projects of \$3.4 million.

Debt

The following table summarizes the consolidated net debt position of the Company.

| As | at |
|----|----|
| | |

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 | Variance |
|---|-------------------|-------------------|--------------|
| Single-family rental properties borrowings | \$ 3,156,601 | \$ 2,728,717 | \$ (427,884) |
| Multi-family rental properties borrowings | 910,340 | 916,340 | 6,000 |
| Canadian development properties borrowings | 60,037 | 13,935 | (46,102) |
| Corporate borrowings | 37,089 | 308,153 | 271,064 |
| | 4,164,067 | 3,967,145 | (196,922) |
| Transaction costs (net of amortization) | (25,019) | (9,896) | 15,123 |
| Debt discount (net of amortization) | (1,542) | (1,988) | (446) |
| Total debt per balance sheet ⁽¹⁾ | \$ 4,137,506 | \$ 3,955,261 | \$ (182,245) |
| Cash and restricted cash | (171,460) | (115,189) | 56,271 |
| Net debt | \$ 3,966,046 | \$ 3,840,072 | \$ (125,974) |
| Total assets | \$ 7,174,834 | \$ 6,486,396 | \$ 688,438 |
| Net debt to assets ⁽²⁾ | 56.6 % | 60.3% | |

(1) Excludes the 2022 convertible debentures and Due to Affiliate.

(2) Calculated by dividing net debt by total assets (net of cash and restricted cash).

Net debt increased by \$126.0 million to \$4.0 billion as at December 31, 2020, from \$3.8 billion as at December 31, 2019. The variance was primarily attributable to:

- An increase of \$427.9 million in single-family rental properties borrowings driven by two new securitization transactions completed during the year with a total face value of \$993.9 million. Of the net proceeds, \$352.5 million was used to pay down a higher-coupon securitization facility and a portion was used to reduce short-term debt.
- An increase of \$46.1 million in Canadian development properties borrowings, attributable to assumed debt and vendor take-back loans as part of the Company's purchase of its partners' 50% and 75% respective interests in The James and The Shops of Summerhill during the second quarter of 2020 (see Section 4.3.1).

- A reduction in corporate borrowings of \$271.1 million. The Company used the \$287.8 million net proceeds of Tricon PIPE LLC's issuance of exchangeable preferred units to repay the majority of the balance outstanding on the Company's revolving credit facility.
- An increase of \$56.3 million in cash and restricted cash mainly attributable to cash reserved to pay for upcoming property tax bills and capital renovation projects on a larger portfolio of single-family rental homes, as well as a higher cash balance at SFR JV-1 to finance the acquisition of single-family rental homes after year-end.

The weighted average interest rate applicable to debt owed by the Company as at December 31, 2020 was 3.12%. The following table summarizes the debt structure and leverage position as at December 31, 2020:

| (in thousands of U.S. dollars) | | | Weighted average | Weighted average time to maturity |
|---|--------------|------------|------------------|--------------------------------------|
| Debt structure | Balance | % of total | interest rate | (years) |
| Fixed (including floating swapped to fixed) | \$ 3,152,455 | 75.7% | 3.24% | 4.4 |
| Floating | 1,011,612 | 24.3% | 2.76% | 1.5 |
| Total/Weighted average | \$ 4,164,067 | 100.0% | 3.12% | 3.7 |

During the quarter, Tricon negotiated additional extension options of its two single-family warehouse credit facilities, extending their maturities into the fourth quarter of 2022. As at December 31, 2020, Tricon's near-term debt maturities include a land loan and a vendor take-back loan totalling \$47.6 million in connection with Tricon's Canadian multi-family developments, the U.S. multi-family credit facility of \$109.9 million and the SFR JV-1 subscription facility of \$116.0 million. The SFR JV-1 subscription facility will be repaid jointly with the limited partners as per the joint venture agreement and the U.S. multi-family credit facility will be repaid upon syndication of the Company's interest in the portfolio (see Section 4.4).

As a result of the transactions during the fourth quarter described above, Tricon extended the weighted average time to maturity of its debt to 3.7 years as at December 31, 2020, representing an increase of 0.3 years from the previous quarter. In addition, Tricon reduced its weighted average interest rate by 0.25% to 3.12% compared to 3.37% in the previous quarter, due in large part to favourable rate financing transactions entered into during the quarter, along with a decrease in LIBOR.

Tricon's debt maturities as at December 31, 2020 are presented below, assuming the exercise of all extension options.



DEBT MATURITY ANALYSIS*

*Assumes the exercise of all extension options.

**Single-family rental borrowings maturing in 2026 include securitized debt totalling \$887 million.

Goodwill

Goodwill was \$108.8 million as at December 31, 2020, comprised primarily of the goodwill recognized upon the deemed acquisitions of the single-family rental and multi-family rental businesses on January 1, 2020 as a result of converting to consolidated accounting. The Company tested its goodwill and concluded that there was no impairment of goodwill as at December 31, 2020. The goodwill testing model is sensitive to underlying assumptions, such as changes in the discount rate and long-term growth rate. Refer to Note 12 to the consolidated financial statements.

Other liabilities

Other liabilities decreased by \$15.2 million to \$4.6 million as at December 31, 2020, from \$19.8 million as at December 31, 2019, primarily attributable to the settlement of the \$13.4 million put liability in relation to common shares issued by Tricon in connection with its acquisition of the Starlight U.S. Multi-Family (No. 5) Core Fund on June 11, 2019. These put rights were exercised by their holders during the first quarter of 2020 (see Section 6.3).

Limited partners' interests in rental business

The following table provides details regarding the change in limited partners' interests in rental business for the twelve months ended December 31, 2020.

| (in thousands of U.S. dollars) | As at December 31, 2019 | Contributions | Distributions | Net change in fair value | As at December 31, 2020 |
|---|----------------------------|---------------|---------------|-----------------------------|----------------------------|
| Limited partners' interests in rental business | \$ 285.774 | \$ 66.112 | \$ (46,162) | \$ 50.581 | \$ 356.305 |
| in rental business | \$ 285,774 | \$ 66,112 | \$ (46,162) | \$ 50,581 | \$ 356,305 |

Limited partners' interests in rental business were \$356.3 million as at December 31, 2020, an increase of \$70.5 million from December 31, 2019. Limited partners' interests in the SFR JV-1 single-family rental joint venture are classified as liabilities under the provisions of IAS 32.

Due to Affiliate

On August 26, 2020, Tricon and its affiliate, Tricon PIPE LLC (the "Affiliate" or "LLC") entered into subscription agreements with each investor in a syndicate of investors (the "Investors"), pursuant to which the Investors subscribed for exchangeable preferred units of the Affiliate (the "Preferred Units") for an aggregate subscription price of \$300 million (the "Transaction"). The Transaction was completed on September 3, 2020, on which date the Company and the Affiliate entered into various agreements with the Investors in connection with the Transaction (together with the subscription agreements, the "Transaction Documents"). The material terms of the Transaction Documents are summarized in the Company's material change report which, together with the material Transaction Documents, is available on SEDAR at www.sedar.com.

In connection with the Transaction, the Company borrowed the subscription proceeds of \$300 million from the Affiliate. This indebtedness, which is evidenced by a promissory note (the "Promissory Note" or "Due to Affiliate"), has a maturity of September 3, 2032 (permitting prepayment at any time pursuant to its terms) and bears interest at a rate of 5.75% per annum, calculated and payable quarterly for the first seven years following Closing with increases thereafter matching the applicable increases of the dividend rate applicable to the Preferred Units, described below. The Company incurred \$15.2 million of transaction costs in connection with the Transaction, of which \$12.2 million was capitalized, which reduced the initial fair value of the Promissory Note, and the remaining portion was expensed as it was attributed to the derivative component of the Promissory Note.

On September 3, 2020, the fair value of the Promissory Note was \$262.4 million, and \$37.6 million of the subscription price was allocated to the fair value of the derivative (see <u>Derivative financial instruments in connection with the Transaction</u>). As at December 31, 2020, the carrying value of the Due to Affiliate was \$251.6 million, net of unamortized discount and transaction costs.

Derivative financial instruments in connection with the Transaction

Pursuant to the Transaction Documents, the holders of the Preferred Units have the right to exchange the Preferred Units into common shares of the Company at any time at the option of the holder (the "Exchange Right") at an initial exchange price of \$8.50 per common share, as may be adjusted from time to time in accordance with the terms of the Transaction Documents (the "Exchange Price"), subject to shareholder approval, where applicable. Holders of Preferred Units are also entitled to receive a cash dividend equal to 5.75% of the Liquidation Preference of the Preferred Units (as defined in the Transaction Documents), per annum, calculated and payable quarterly for the first seven years following closing of the Transaction ("Closing"), with a prescribed annual increase to the dividend rate of 1% per year thereafter, up to a maximum rate of 9.75% per year.

The Affiliate has the right to force the exchange (the "Forced Exchange Right") of the outstanding Preferred Units beginning after the fourth anniversary of Closing, provided the 20-day volume-weighted average price of Tricon's shares exceeds 135% of the Exchange Price (reducing to 115% following the fifth anniversary of Closing). The Affiliate also has the right to redeem the Preferred Units ("Redemption Right") at any time following the fifth anniversary of Closing for cash equal to 105% of the Liquidation Preference of the Preferred Units (as defined in the Transaction Documents).

The Promissory Note contains mandatory prepayment provisions ("Mandatory Prepayment") applicable in connection with certain provisions of the Transaction Documents requiring the redemption of all or a portion of the outstanding Preferred Units. This Mandatory Prepayment is a derivative, which incorporates assumptions in respect of the Exchange Right, Forced Exchange Right and Redemption Right, and is measured separately from the Promissory Note, with a fair value at inception of \$37.6 million. From September 3, 2020 to December 31, 2020, there was a fair value loss on the derivative on the Due to Affiliate of \$7.9 million, increasing the balance to \$45.5 million, primarily driven by an increase in Tricon's common share price, on a USD converted basis, from September 3rd to December 31st, which served to increase the probability of exchange of the preferred units into Tricon's common shares (see Section 3.1).

3.3 Subsequent events

COVID-19 related business update

In light of the ongoing COVID-19 pandemic, the Company is providing a more current update on its rental operations.

Single-family rental

In the single-family rental business, same home occupancy for January remained stable at 97.3%. As of February 28, 2021, the Company had collected 97% of January rents and fewer than 1% of Tricon's single-family rental residents had requested a rent deferral plan because of economic hardship in 2021. Average blended rent growth for the same home portfolio in January increased to 6.0%, driven by 10.6% growth on new move-ins and 4.0% growth on renewals.

U.S. multi-family rental

In the U.S. multi-family rental business, same property occupancy for January improved to 94.6%. As of February 28, 2021, the Company had collected 96% of January rents and none of Tricon's multi-family rental residents had requested a rent deferral plan because of economic hardship in 2021. Average blended rent growth for the same property portfolio has also increased in January to 1.1%, registering the first month of positive blended rent growth since February of 2020, driven by 2.8% growth on new move-ins and (0.4%) growth on renewals.

Texas storm update

In February of 2021, a severe winter storm hit Texas that devastated the state's power grid and natural gas production, leaving thousands of people without power across the state and millions experiencing water disruptions. Based on assessments completed to date, approximately 570 of Tricon's single-family rental homes and 200 multi-family units in Texas were affected. The Company is expecting no material financial impact as a result of this storm as Tricon's rental properties are insured under property and casualty insurance policies, subject to certain deductibles and limits. The Company is managing the restoration processes, while remaining focused on our employees' and residents' well-being.

U.S. multi-family rental portfolio syndication

On February 25, 2021, the Company announced that it had reached an agreement in principle to enter into a joint venture arrangement with two institutional investors. Under the joint venture, the investors will acquire a combined 80% ownership interest in Tricon's existing portfolio of 23 U.S. multi-family apartments and Tricon will retain a 20% ownership interest. The transaction reflects a total portfolio value of \$1.331 billion including in-place debt, and is expected to generate gross proceeds of approximately \$425 million to Tricon, which will be used to repay outstanding debt and for general corporate purposes. The transaction is expected to close in March of 2021, subject to finalizing definitive documentation and customary closing conditions including obtaining the necessary lender consents.

Quarterly dividend

On March 2, 2021, the Board of Directors of the Company declared a dividend of seven cents per common share in Canadian dollars payable on or after April 15, 2021 to shareholders of record on March 31, 2021.



OPERATING RESULTS OF BUSINESSES

4. OPERATING RESULTS OF BUSINESSES

Management believes that information concerning the underlying activities within each of the Company's operating businesses is useful for investors in understanding the Company's overall performance, and this section presents key operating highlights for the quarter on a segment-by-segment basis. Although the Company's performance is primarily measured by Core FFO per share, as set out in <u>Section 1.1</u>, management also monitors the underlying activities within those businesses using KPIs to provide a better understanding of the performance of the Company. A list of these KPIs, together with a description of the information each measure reflects and the reasons why management believes the measure to be useful or relevant in evaluating the underlying performance of the Company's businesses, is set out in <u>Section 7.1</u>. The supplemental measures presented herein are not recognized under IFRS and should not be construed as alternatives to net income determined in accordance with IFRS as indicators of Tricon's financial performance. Tricon's method of calculating these measures may differ from other issuers' methods and, accordingly, these measures may not be comparable to similar measures presented by other publicly-traded entities.

The financial results and performance metrics in Section 4 and throughout this document reflect Tricon's proportionate share of results, unless otherwise stated.

Operational highlights by segment

The following table summarizes Tricon's proportionate share of operating results and key performance metrics for each business segment. In previous years, operating highlights by segment were disclosed on a consolidated or portfolio-wide basis, and have been recast to conform with the current period presentation. Refer to <u>Section 5</u> for a reconciliation of Tricon's proportionate financial results from each segment to consolidated figures under IFRS.

| For the periods ended December 31 | Three | months | Twelve months | | | |
|--|------------|------------|---------------|------------|--|--|
| (in thousands of U.S. dollars, except percentages and units) | 2020 | 2019 | 2020 | 2019 | | |
| SINGLE-FAMILY RENTAL (Refer to Section 4.1) | | | | | | |
| | | | | | | |
| Total rental homes managed | 22,766 | 21,014 | | | | |
| Net operating income (NOI) ⁽¹⁾ | \$ 50,476 | \$ 45,493 | \$ 197,528 | \$ 173,865 | | |
| Same home net operating income (NOI) margin ⁽¹⁾ | 66.8% | 65.3% | 66.3% | 65.5% | | |
| Same home net operating income (NOI) growth ⁽¹⁾ | 5.1% | N/A | 5.6% | N/A | | |
| Same home occupancy ⁽¹⁾ | 97.3% | 95.9% | | | | |
| Same home annualized turnover ⁽¹⁾ | 22.2% | 25.7% | | | | |
| Same home average quarterly rent growth – blended ⁽¹⁾ | 5.6% | 5.3% | | | | |
| MULTI-FAMILY RENTAL (Refer to Section 4.2) | | | | | | |
| U.S. multi-family rental ^{(2),(3)} – See <u>Section 4.2.1</u> | | | | | | |
| Total suites managed | 7,289 | 7,289 | | | | |
| Net operating income (NOI) | \$ 15,604 | \$ 16,964 | \$ 62,909 | \$ 67,170 | | |
| Net operating income (NOI) margin | 56.6% | 59.4% | 56.6% | 59.0% | | |
| Occupancy | 93.6% | 94.9% | | | | |
| Annualized turnover | 46.5% | 51.3% | | | | |
| Average quarterly rent growth – blended | (1.8%) | 1.1% | | | | |
| Canadian multi-family rental ⁽⁴⁾ – See Section 4.2.2 | | | | | | |
| Total suites managed | 500 | - | | | | |
| Net operating income (NOI) ⁽⁵⁾ | \$ 220 | \$ – | \$ 927 | \$ – | | |
| Net operating income (NOI) margin ⁽⁵⁾ | 55.6% | - | 58.6% | - | | |
| Occupancy ⁽⁵⁾ | 87.0% | - | | | | |
| Annualized turnover ⁽⁵⁾ | 41.6% | - | | | | |
| Average quarterly rent growth – blended ⁽⁵⁾ | (5.1%) | - | | | | |
| RESIDENTIAL DEVELOPMENT (Refer to Section 4.3) | | | | | | |
| Investments in residential developments ⁽⁶⁾ | \$ 292,958 | \$ 397,815 | | | | |
| Core funds from operations (Core FFO) | 11,532 | 3,076 | \$ 18,913 | \$ 8,240 | | |
| Cash distributions from investments to Tricon | 12,720 | 24,284 | 77,443 | 51,946 | | |
| PRIVATE FUNDS AND ADVISORY (Refer to Section 4.4) | | | | | | |
| Revenue from private funds and advisory services | \$ 10,339 | \$ 12,138 | \$ 34,090 | \$ 41,060 | | |
| Third-party AUM ⁽⁷⁾ | 2,553,358 | 2,434,610 | | | | |
| | | | | | | |

(1) Operating metrics are stated at Tricon's proportionate share of the managed portfolio and exclude limited partners' interests in the SFR JV-1 portfolio.

(2) The financial information presented in the table includes prior-year results for comparability although Tricon's U.S. multi-family rental portfolio was acquired on June 11, 2019.

(3) For the three and twelve months ended December 31, 2020, the total property results equate to same property results for the U.S. multi-family rental portfolio.

(4) Presented within investments in Canadian multi-family developments and income from Canadian multi-family developments, respectively, on the Company's balance sheet and income statement. Tricon's proportionate share of the operating results and key performance metrics is presented to provide more insight into underlying property operations.

(5) Operating metrics are stated at Tricon's proportionate share of the managed portfolio and exclude limited partner's interest in The Selby.

(6) Represents Tricon's investments in Canadian multi-family developments, investments in Canadian development properties (net of debt) and investments in for-sale housing.

(7) KPI measure; see Section 7.2.



4.1 Single-Family Rental

The discussion and presentation of the single-family rental operating metrics and results throughout this section reflect Tricon's proportionate share of the business, including its proportionate share of the Company's single-family rental joint venture ("SFR JV-1"), unless otherwise stated. Prior period metrics have also been recast to reflect Tricon's proportionate share.

Operating results

The Company's single-family rental business continued to benefit from secular tailwinds which drove higher occupancy, rent growth and retention in the quarter. De-urbanization, de-densification and work-from-home trends have accelerated over the course of the COVID-19 pandemic, making Tricon's Sun Belt markets and suburban rental homes an increasingly attractive option for many families. Tighter supply, stringent mortgage underwriting and affordability pressures in the for-sale market have also contributed to growing demand for rental housing.

Above all, Tricon's Sun Belt middle-market strategy has proven to be resilient throughout this economic down cycle. Tricon's relatively low average rent-to-income ratio of 23% and stable resident base with average household incomes of \$85,000 (based on the prior six months of new move-ins) have allowed the single-family rental business to weather a large portion of the negative economic effects caused by the COVID-19 pandemic thus far.

Tricon's single-family rental business finished the year on a strong note, reporting a record-high NOI margin of 67.1% in the fourth quarter. Both occupancy and blended rent growth remained healthy at 96.4% and 5.4%, respectively. Blended rent growth was comprised of 10.7% growth on new leases, attributable to strong demand and a scarcity of homes available for rent, as well as 2.9% growth on renewals, reflecting the Company's policy of moderating rent increases for current residents. Management expects that a favourable supply-demand imbalance coupled with inherent portfolio loss-to-lease, estimated conservatively to be 7% to 9% of current rents, will continue to drive rent growth in 2021 and beyond (see "Non-IFRS measures and forward-looking statements" on page 24).

The annualized turnover rate was 22.3% during the quarter, a 3.1% decrease from 25.4% recorded during the same period in 2019. On a full-year basis, resident turnover was 23.4% compared to 26.9% in the prior year, reflecting Tricon's focus on superior customer service and effective renewal management, as well as a strong occupancy bias throughout the pandemic, which has provided residents with housing security during these challenging times.

The Company acquired 842 homes (at an average cost per home of \$226,000 including upfront renovations) during the fourth quarter as acquisition volumes returned to pre-COVID-19 levels. Management continues to see strong opportunities for home-buying and expects to acquire approximately 800 homes in the first quarter of 2021.

The table below presents key operational metrics that drive revenue and NOI for the single-family rental segment (KPI measure; refer to <u>Section 7.1</u>). *The operating metrics below reflect Tricon's proportionate share of the single-family rental portfolio, with the exception of the total number of rental homes comprising the portfolio.*

| Proportionate operating metrics | Q4 2020 | Q3 2020 | Q2 2020 | Q1 2020 | Q4 2019 ⁽¹⁾ | Q3 2019 ⁽¹⁾ | Q2 2019 ⁽¹⁾ | Q1 2019 ⁽¹⁾ |
|--|----------|----------|----------|----------|------------------------|------------------------|------------------------|------------------------|
| Rental homes | 22,766 | 21,948 | 21,582 | 21,535 | 21,014 | 19,886 | 19,016 | 18,094 |
| Occupancy ⁽²⁾ | 96.4% | 97.3% | 97.1% | 95.5% | 94.5% | 94.4% | 95.1% | 94.5% |
| Annualized turnover rate | 22.3% | 26.3% | 23.5% | 21.4% | 25.4% | 30.0% | 29.7% | 22.3% |
| Average monthly rent ⁽³⁾ | \$ 1,464 | \$ 1,450 | \$ 1,432 | \$ 1,420 | \$ 1,405 | \$ 1,389 | \$ 1,371 | \$ 1,348 |
| Average quarterly rent growth – renewal ⁽⁴⁾ | 2.9% | 2.4% | 3.2% | 5.3% | 5.3% | 5.1% | 5.2% | 5.3% |
| Average quarterly rent growth – new move-in ⁽⁴⁾ | 10.7% | 11.6% | 7.5% | 7.5% | 5.5% | 8.4% | 9.1% | 8.8% |
| Average quarterly rent growth – blended ⁽⁴⁾ | 5.4% | 5.1% | 4.5% | 5.9% | 5.3% | 6.1% | 6.5% | 6.3% |

(1) Prior period metrics have been recast to reflect Tricon's proportionate share of the single-family rental segment.

(2) The decrease in occupancy from Q3 2020 to Q4 2020 was primarily driven by the resumption of acquisition activity to pre-COVID-19 levels.

(3) Average monthly rent represents average monthly rental income per unit for occupied units and reflects the impact of rent concessions amortized over the life of the related leases.

(4) Average rent growth during the period represents the percentage difference between the monthly rent from an expiring lease and the monthly rent from the next lease and reflects the impact of rent concessions amortized over the life of the related lease.



The table below presents a breakdown of Tricon's NOI (KPI measure; refer to Section 7.1) for the single-family rental business.

| | Three months | | | | | Twelve months | | | | |
|--|--------------|---------|---------------------|---------|----------|---------------|---------|---------------------|---------|-----------|
| For the periods ended December 31 | | % of | | % of | | | % of | | % of | |
| (in thousands of U.S. dollars) | 2020 | revenue | 2019 ⁽¹⁾ | revenue | Variance | 2020 | revenue | 2019 ⁽¹⁾ | revenue | Variance |
| Rental revenue | \$ 74,906 | | \$ 67,982 | | \$ 6,924 | \$ 292,112 | | \$ 260,719 | | \$ 31,393 |
| Concessions and abatements | (326) | | (473) | | 147 | (1,231) | | (1,621) | | 390 |
| Fees and other revenue | 2,754 | | 2,907 | | (153) | 10,804 | | 10,028 | | 776 |
| Bad debt expense | (2,080) | | (560) | | (1,520) | (4,745) | | (2,145) | | (2,600) |
| Total revenue from rental properties | 75,254 | 100% | 69,856 | 100% | 5,398 | 296,940 | 100% | 266,981 | 100% | 29,959 |
| Property taxes | 11,415 | 15% | 10,873 | 16% | (542) | 45,768 | 15% | 42,307 | 16% | (3,461) |
| Repairs, maintenance and turnover | 5,044 | 7% | 5,547 | 8% | 503 | 20,555 | 7% | 20,408 | 8% | (147) |
| Property management expenses | 5,249 | 7% | 4,935 | 7% | (314) | 20,646 | 7% | 18,870 | 7% | (1,776) |
| Property insurance | 1,168 | 2% | 1,095 | 2% | (73) | 4,606 | 2% | 4,230 | 2% | (376) |
| Homeowners' association (HOA) costs | 976 | 1% | 894 | 1% | (82) | 3,858 | 1% | 3,259 | 1% | (599) |
| Other direct expenses | 926 | 1% | 1,019 | 1% | 93 | 3,979 | 1% | 4,042 | 2% | 63 |
| Total direct operating expenses | 24,778 | | 24,363 | | (415) | 99,412 | | 93,116 | | (6,296) |
| Net operating income (NOI) ⁽²⁾ | \$ 50,476 | | \$ 45,493 | | \$ 4,983 | \$ 197,528 | | \$ 173,865 | | \$ 23,663 |
| Net operating income (NOI) margin ⁽²⁾ | 67.1% | | 65.1% | | | 66.5% | | 65.1% | | |

(1) The comparative period has been reclassified to conform with the current period presentation.

(2) KPI measures; see Section 7.1.

NOI was \$50.5 million for the three months ended December 31, 2020, an increase of \$5.0 million or 11.0% compared to the same period in 2019. The variance in NOI is attributable to an increase of \$6.9 million or 10.2% in rental revenue, mainly explained by (i) a larger rental portfolio (Tricon's proportionate share of rental homes was 17,698 in Q4 2020 compared to 17,054 in Q4 2019), (ii) higher average monthly rent (\$1,464 in Q4 2020 compared to \$1,405 in Q4 2019), and (iii) a 1.9% increase in occupancy. The higher rental revenue was partially offset by a \$1.5 million increase in bad debt expense, as a result of higher resident delinquency from ongoing unemployment related to COVID-19 and the wind-down of various government stimulus programs. The Company has taken a conservative approach and has reserved 100% of residents' accounts receivable balances aged more than 30 days. The bad debt expense in the fourth quarter represented 2.7% of revenue, compared to historical bad debt levels (pre-COVID-19) of approximately 0.8%.

Direct operating expenses in the quarter increased by \$0.4 million or 1.7% driven by higher costs incurred on a larger portfolio of homes, which were partially offset by savings on repairs, maintenance and turnover as well as other direct expenses. The savings were attributable to a lower turnover rate as well as improved cost discipline and controlled scoping of maintenance work.

Managed portfolio

The following tables provide a summary of the single-family rental home portfolio, reflecting information for all homes managed by Tricon, including all homes owned by SFR JV-1 and homes wholly-owned by Tricon.

| | Q4 2020 | Q3 2020 | Q2 2020 | Q1 2020 | Q4 2019 | Q3 2019 | Q2 2019 | Q1 2019 |
|-----------------------------------|---------|---------|---------|---------|---------|---------|---------|---------|
| Tricon wholly-owned homes | 15,355 | 15,384 | 15,410 | 15,429 | 15,453 | 15,500 | 15,535 | 15,563 |
| SFR JV-1 homes | | | | | | | | |
| (34% TCN/66% JV Partners) | 7,439 | 6,597 | 6,212 | 6,154 | 5,624 | 4,462 | 3,545 | 2,568 |
| Total homes managed | 22,794 | 21,981 | 21,622 | 21,583 | 21,077 | 19,962 | 19,080 | 18,131 |
| Less homes held for sale | 28 | 33 | 40 | 48 | 63 | 76 | 64 | 37 |
| Rental homes | 22,766 | 21,948 | 21,582 | 21,535 | 21,014 | 19,886 | 19,016 | 18,094 |
| Homes acquired | 842 | 388 | 68 | 538 | 1,162 | 918 | 977 | 730 |
| Less homes disposed | (29) | (29) | (29) | (32) | (47) | (36) | (28) | (41) |
| Net homes acquired | | | | | | | | |
| during the quarter ⁽¹⁾ | 813 | 359 | 39 | 506 | 1,115 | 882 | 949 | 689 |

(1) Of the net homes acquired during the quarter, 842 were acquired by SFR JV-1 and 29 wholly-owned homes were disposed.

Tricon acquired 842 homes (or 813 net of dispositions) during the quarter. As of December 31, 2020, Tricon managed 22,794 homes (22,766 rental homes and 28 homes held for sale) of which 15,355 were wholly-owned by Tricon and 7,439 were owned by SFR JV-1, where Tricon has a one-third equity interest.



As at December 31, 2020, Tricon's single-family rental portfolio is diversified across 18 markets. Market-level details for all homes managed by Tricon are presented below.

| Geography | Rental homes | Average vintage | Average total cost per home (in U.S. dollars) | | Average size (sq. feet) | Tricon % ownership |
|-------------------------|--------------|-----------------|---|---------|----------------------------|-----------------------|
| Atlanta | 5,253 | 1997 | \$ | 158,000 | 1,762 | 79.3% |
| Charlotte | 2,630 | 1999 | | 171,000 | 1,590 | 69.0% |
| Nashville | 1,031 | 2009 | | 286,000 | 1,945 | 33.7% |
| Columbia | 922 | 1996 | | 137,000 | 1,505 | 65.2% |
| Raleigh | 280 | 2006 | | 212,000 | 1,507 | 33.7% |
| Southeast United States | 10,116 | 1999 | \$ | 174,000 | 1,705 | 69.4% |
| Phoenix | 2,040 | 1995 | \$ | 184,000 | 1,689 | 100.0% |
| Northern California | 999 | 1970 | | 224,000 | 1,304 | 100.0% |
| Las Vegas | 601 | 1996 | | 182,000 | 1,649 | 100.0% |
| Southern California | 269 | 1962 | | 191,000 | 1,307 | 100.0% |
| Reno | 248 | 1981 | | 181,000 | 1,549 | 100.0% |
| Western United States | 4,157 | 1986 | \$ | 194,000 | 1,557 | 100.0% |
| Dallas | 1,760 | 1991 | \$ | 169,000 | 1,580 | 76.7% |
| Houston | 1,510 | 1993 | | 161,000 | 1,610 | 72.1% |
| San Antonio | 549 | 1998 | | 165,000 | 1,616 | 64.1% |
| Texas | 3,819 | 1993 | \$ | 165,000 | 1,597 | 73.1% |
| Tampa | 1,797 | 1987 | \$ | 180,000 | 1,555 | 87.0% |
| Jacksonville | 731 | 1995 | | 166,000 | 1,515 | 73.9% |
| Southeast Florida | 695 | 1969 | | 173,000 | 1,417 | 100.0% |
| Orlando | 477 | 1988 | | 185,000 | 1,484 | 93.0% |
| Florida | 3,700 | 1985 | \$ | 177,000 | 1,512 | 87.6% |
| Indianapolis | 974 | 2002 | \$ | 156,000 | 1,638 | 63.3% |
| Midwest United States | 974 | 2002 | \$ | 156,000 | 1,638 | 63.3% |
| Total/Weighted average | 22,766 | 1993 | \$ | 176,000 | 1,626 | 78.3% |

Operating results – Same home portfolio

"Same home" or "same home portfolio" includes homes that were stabilized 90 days prior to the first day of the prior-year comparative period as per the guidelines of the National Rental Home Council. It excludes homes that have been sold and homes that have been designated for sale. This same home portfolio is defined on January 1 of each reporting year. Based on this definition, any home included in the same home portfolio will have satisfied the conditions described above prior to September 30, 2018, and those homes are held in operations throughout the full periods presented in both 2019 and 2020.

The operating metrics below reflect Tricon's proportionate share of the same home portfolio, with the exception of the total number of homes comprising the same home portfolio.

| For the periods ended December 31 | | Three months | | Twelve months | | | |
|--|----------|--------------|----------|---------------|----------|----------|--|
| (in U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance | |
| Operating metrics – same home | | | | | | | |
| Tricon wholly-owned rental homes | 14,804 | 14,804 | - | 14,804 | 14,804 | - | |
| SFR JV-1 homes (34% TCN/66% JV Partners) | 530 | 530 | _ | 530 | 530 | - | |
| Rental homes | 15,334 | 15,334 | - | 15,334 | 15,334 | - | |
| Occupancy | 97.3% | 95.9% | 1.4% | 97.2% | 96.1% | 1.1% | |
| Annualized turnover rate | 22.2% | 25.7% | 3.5% | 22.8% | 27.4% | 4.6% | |
| Average monthly rent ⁽¹⁾ | \$ 1,464 | \$ 1,407 | \$ 57 | \$ 1,440 | \$ 1,383 | \$ 57 | |
| Average rent growth – renewal ⁽²⁾ | 3.0% | 5.1% | (2.1%) | 3.4% | 5.1% | (1.7%) | |
| Average rent growth – new move-in ⁽²⁾ | 11.3% | 5.5% | 5.8% | 9.9% | 8.1% | 1.8% | |
| Average rent growth – blended ⁽²⁾ | 5.6% | 5.3% | 0.3% | 5.3% | 6.1% | (0.8%) | |

(1) Average monthly rent represents average monthly rental income per unit for occupied units and reflects the impact of rent concessions amortized over the life of the related leases.

(2) Average rent growth during the period represents the percentage difference between the monthly rent from an expiring lease and the monthly rent from the next lease and reflects the impact of rent concessions amortized over the life of the related lease.

For the same home portfolio, blended rent growth for the quarter was 5.6% (including 11.3% on new leases and 3.0% on renewals), accompanied by a 1.4% increase in occupancy to 97.3% from 95.9% recorded in the same period in 2019. Robust demand for single-family rental homes in suburban markets, combined with limited supply of rental homes and embedded loss-to-lease in the portfolio, drove stronger new-lease rent growth. In addition, the Company's continued focus on resident retention and its inherent occupancy bias helped it achieve an annualized turnover rate of 22.2%, a 3.5% decrease compared to the same period in the prior year.

The following table provides details of the same home portfolio results for the three and twelve months ended December 31, 2020 and December 31, 2019.

| | Three months | | | | | Twelve months | | | | |
|--------------------------------------|--------------|---------|-----------|---------|----------|---------------|---------|------------|---------|-----------|
| For the periods ended December 31 | | % of | | % of | | | % of | | % of | |
| (in thousands of U.S. dollars) | 2020 | revenue | 2019 | revenue | Variance | 2020 | revenue | 2019 | revenue | Variance |
| Rental revenue | \$ 63,959 | | \$ 60,593 | | \$ 3,366 | \$ 251,405 | | \$ 238,989 | | \$ 12,416 |
| Concessions and abatements | (263) | | (250) | | (13) | (819) | | (1,088) | | 269 |
| Fees and other revenue | 2,261 | | 2,520 | | (259) | 8,863 | | 8,852 | | 11 |
| Bad debt expense | (1,846) | | (502) | | (1,344) | (4,144) | | (1,971) | | (2,173) |
| Total revenue from rental properties | 64,111 | 100% | 62,361 | 100% | 1,750 | 255,305 | 100% | 244,782 | 100% | 10,523 |
| Property taxes | 9,972 | 16% | 9,769 | 16% | (203) | 40,119 | 16% | 38,451 | 16% | (1,668) |
| Repairs, maintenance and turnover | 4,355 | 7% | 4,881 | 8% | 526 | 18,200 | 7% | 18,898 | 8% | 698 |
| Property management expenses | 4,455 | 7% | 4,396 | 7% | (59) | 17,660 | 7% | 17,230 | 7% | (430) |
| Property insurance | 1,053 | 2% | 1,003 | 2% | (50) | 4,167 | 2% | 3,929 | 2% | (238) |
| Homeowners' association (HOA) costs | 737 | 1% | 725 | 1% | (12) | 2,944 | 1% | 2,759 | 1% | (185) |
| Other direct expenses | 710 | 1% | 846 | 1% | 136 | 3,036 | 1% | 3,301 | 1% | 265 |
| Total direct operating expenses | 21,282 | | 21,620 | | 338 | 86,126 | | 84,568 | | (1,558) |
| Net operating income (NOI) | \$ 42,829 | | \$ 40,741 | | \$ 2,088 | \$ 169,179 | | \$ 160,214 | | \$ 8,965 |
| Net operating income (NOI) growth | | | | | 5.1% | | | | | 5.6% |
| Net operating income (NOI) margin | 66.8% | | 65.3% | | | 66.3% | | 65.5% | | |

4.1 Single-Family Rental

Total revenue for the same home portfolio increased by \$1.8 million or 2.8% to \$64.1 million in the fourth quarter of 2020 compared to \$62.4 million for the same period in the prior year. This favourable change was primarily attributable to a \$3.4 million or 5.6% increase in rental revenue as a result of higher average monthly rent (\$1,464 in Q4 2020 compared to \$1,407 in Q4 2019) and a 1.4% increase in occupancy as discussed above.

These positive variances were partially offset by a \$1.3 million increase in bad debt expense as the Company has reserved higher bad debt amounts for the reasons discussed previously. The bad debt expense represented 2.8% of revenue in the fourth quarter of 2020 compared to 0.8% in the same period in the prior year. In addition, fees and other revenue declined by \$0.3 million as the Company has decided to forego early termination and late fees in the quarter. Late fees and early termination fees are typically enforced in accordance with lease agreements but were not charged to many residents in light of the pandemic.

Same home operating expenses decreased by \$0.3 million or 1.6% to \$21.3 million in the fourth quarter of 2020 from \$21.6 million during the same period in 2019. This variance is largely attributable to the following:

- Property taxes Property tax expense increased nominally by \$0.2 million or 2.1% to \$10.0 million as the 2019 comparative period expense included an unfavourable one-time local market reassessment. On a full-year basis, property taxes increased by \$1.7 million or 4.4%, reflecting higher assessments primarily as a result of home price appreciation.
- Repairs, maintenance and turnover These costs decreased by \$0.5 million or 10.8% to \$4.4 million during the quarter, attributable to a \$1.0 million reduction in turnover costs driven by a lower turnover rate and a heightened attention to scope and expense management. The Company has focused on expense management by refining and managing work scopes on turns and increasing purchasing efficiencies through its centralized procurement group. The turnover savings were partially offset by a \$0.5 million increase in repairs and maintenance costs owing to damage from severe weather in the Atlanta and Southeast Florida markets.
- Property insurance The increase in property insurance expense of \$0.1 million or 5.0% to \$1.1 million was attributable to higher insurance premium rates across the industry. Based on the Company's recently negotiated 2021 renewals, 2021 insurance pricing is expected to increase by approximately 10% compared to 2020, which is in line with the broader single-family rental market.

With strong revenue growth and controlled expenses, same home NOI increased by 5.1% year-over-year to \$42.8 million in the fourth quarter of 2020 compared to \$40.7 million in the fourth quarter of 2019. Same home NOI margin increased to a record-high 66.8% in the fourth quarter of 2020 from 65.3% in the same period in the prior year.

As at December 31, 2020, the same home portfolio is diversified across 16 target markets. Same home market-level details are presented below.

| Year-over-year comparison | | | NOI ⁽¹⁾ | | NOI margin ⁽¹⁾ | | | | | |
|---------------------------|--------|-----------|--------------------|--------|---------------------------|---------|--------|--|--|--|
| Geography | Homes | Q4 2020 | Q4 2019 | Change | Q4 2020 | Q4 2019 | Change | | | |
| Atlanta | 3,695 | \$ 9,782 | \$ 9,123 | 7.2% | 68.4% | 66.9% | 1.5% | | | |
| Charlotte | 1,514 | 3,904 | 3,737 | 4.5% | 70.9% | 70.4% | 0.5% | | | |
| Columbia | 462 | 869 | 791 | 9.9% | 57.8% | 53.4% | 4.4% | | | |
| Southeast United States | 5,671 | \$ 14,555 | \$ 13,651 | 6.6% | 68.3% | 66.8% | 1.5% | | | |
| Phoenix | 1,766 | \$ 5,439 | \$ 5,255 | 3.5% | 74.4% | 75.3% | (0.9%) | | | |
| Northern California | 991 | 4,262 | 4,033 | 5.7% | 77.1% | 74.0% | 3.1% | | | |
| Las Vegas | 585 | 1,865 | 1,823 | 2.3% | 74.7% | 74.9% | (0.2%) | | | |
| Reno | 247 | 987 | 958 | 3.0% | 78.4% | 78.7% | (0.3%) | | | |
| Southern California | 237 | 935 | 820 | 14.0% | 71.2% | 64.6% | 6.6% | | | |
| Western United States | 3,826 | \$ 13,488 | \$ 12,889 | 4.6% | 75.3% | 74.3% | 1.0% | | | |
| Tampa | 1,479 | \$ 4,152 | \$ 4,023 | 3.2% | 61.6% | 61.4% | 0.2% | | | |
| Southeast Florida | 671 | 1,908 | 1,730 | 10.3% | 55.2% | 49.6% | 5.6% | | | |
| Jacksonville | 465 | 1,153 | 1,132 | 1.9% | 64.6% | 63.3% | 1.3% | | | |
| Orlando | 432 | 1,207 | 1,217 | (0.8%) | 65.0% | 65.1% | (0.1%) | | | |
| Florida | 3,047 | \$ 8,420 | 8,102 | 3.9% | 60.8% | 59.2% | 1.6% | | | |
| Dallas | 1,198 | \$ 2,855 | \$ 2,798 | 2.0% | 56.7% | 56.9% | (0.2%) | | | |
| Houston | 888 | 1,960 | 1,829 | 7.2% | 56.8% | 53.0% | 3.8% | | | |
| San Antonio | 236 | 550 | 502 | 9.6% | 61.5% | 56.3% | 5.2% | | | |
| Texas | 2,322 | \$ 5,365 | \$ 5,129 | 4.6% | 57.2% | 55.4% | 1.8% | | | |
| Indianapolis | 468 | \$ 1,001 | \$ 970 | 3.2% | 60.1% | 59.6% | 0.5% | | | |
| Midwest United States | 468 | \$ 1,001 | \$ 970 | 3.2% | 60.1% | 59.6% | 0.5% | | | |
| Total/Weighted average | 15,334 | \$ 42,829 | \$ 40,741 | 5.1% | 66.8% | 65.3% | 1.5% | | | |

Year-over-vear comparison

(1) Represents Tricon's proportionate share of NOI and NOI margin of the same home portfolio.

MANAGEMENT'S DISCUSSION AND ANALYSIS

for the year ended December 31, 2020



| | Rental | | Avera | ige mo | onthly rent | 1),(2) | Occupancy ⁽¹⁾ | | | | |
|-------------------------|--------|----|---------|--------|-------------|------------|--------------------------|---------|------------|--|--|
| Geography | homes | (| Q4 2020 | (| 24 2019 | Change (%) | Q4 2020 | Q4 2019 | Change (%) | | |
| Atlanta | 3,695 | \$ | 1,355 | \$ | 1,297 | 4.5% | 97.3% | 95.3% | 2.0% | | |
| Charlotte | 1,514 | | 1,315 | | 1,270 | 3.5% | 97.6% | 95.1% | 2.5% | | |
| Columbia | 462 | | 1,204 | | 1,161 | 3.7% | 96.8% | 95.0% | 1.8% | | |
| Southeast United States | 5,671 | \$ | 1,332 | \$ | 1,278 | 4.2% | 97.4% | 95.2% | 2.2% | | |
| Phoenix | 1,766 | \$ | 1,402 | \$ | 1,311 | 6.9% | 98.2% | 97.4% | 0.8% | | |
| Northern California | 991 | | 1,912 | | 1,833 | 4.3% | 98.9% | 98.5% | 0.4% | | |
| Las Vegas | 585 | | 1,459 | | 1,388 | 5.1% | 98.0% | 97.6% | 0.4% | | |
| Reno | 247 | | 1,731 | | 1,655 | 4.6% | 98.3% | 97.0% | 1.3% | | |
| Southern California | 237 | | 1,894 | | 1,801 | 5.2% | 99.4% | 96.5% | 2.9% | | |
| Western United States | 3,826 | \$ | 1,595 | \$ | 1,511 | 5.6% | 98.4% | 97.6% | 0.8% | | |
| Татра | 1,479 | \$ | 1,577 | \$ | 1,534 | 2.8% | 97.4% | 95.4% | 2.0% | | |
| Southeast Florida | 671 | | 1,799 | | 1,755 | 2.5% | 96.1% | 95.2% | 0.9% | | |
| Jacksonville | 465 | | 1,366 | | 1,303 | 4.8% | 96.8% | 97.0% | (0.2%) | | |
| Orlando | 432 | | 1,501 | | 1,428 | 5.1% | 96.4% | 97.7% | (1.3%) | | |
| Florida | 3,047 | \$ | 1,583 | | 1,533 | 3.3% | 96.9% | 95.9% | 1.0% | | |
| Dallas | 1,198 | \$ | 1,498 | \$ | 1,446 | 3.6% | 96.0% | 95.6% | 0.4% | | |
| Houston | 888 | | 1,371 | | 1,359 | 0.9% | 96.6% | 93.0% | 3.6% | | |
| San Antonio | 236 | | 1,342 | | 1,321 | 1.6% | 95.5% | 94.9% | 0.6% | | |
| Texas | 2,322 | \$ | 1,434 | \$ | 1,400 | 2.4% | 96.2% | 94.5% | 1.7% | | |
| Indianapolis | 468 | \$ | 1,264 | \$ | 1,223 | 3.4% | 97.1% | 96.4% | 0.7% | | |
| Midwest United States | 468 | \$ | 1,264 | \$ | 1,223 | 3.4% | 97.1% | 96.4% | 0.7% | | |
| Total/Weighted average | 15,334 | \$ | 1,464 | \$ | 1,407 | 4.1% | 97.3% | 95.9% | 1.4% | | |

(1) Represents Tricon's proportionate share of average monthly rent and occupancy of the same home portfolio.

(2) Average monthly rent represents average monthly rental income per unit for occupied units and reflects the impact of rent concessions amortized over the life of the related leases. The year-over-year change in average monthly rent does not equal the average quarterly rent growth, which is calculated as the percentage difference between the monthly rent from an expiring lease and the monthly rent from the next lease.

MANAGEMENT'S DISCUSSION AND ANALYSIS

for the year ended December 31, 2020

4.1 Single-Family Rental

N

Quarter-over-quarter comparison

| | Rental | Avera | ge monthly rent | (1),(2) | Occupancy ⁽¹⁾ | | | | |
|-------------------------|--------|----------|-----------------|------------|--------------------------|---------|------------|--|--|
| Geography | homes | Q4 2020 | Q3 2020 | Change (%) | Q4 2020 | Q3 2020 | Change (%) | | |
| Atlanta | 3,695 | \$ 1,355 | \$ 1,334 | 1.6% | 97.3% | 97.7% | (0.4%) | | |
| Charlotte | 1,514 | 1,315 | 1,297 | 1.4% | 97.6% | 97.6% | - | | |
| Columbia | 462 | 1,204 | 1,189 | 1.3% | 96.8% | 97.5% | (0.7%) | | |
| Southeast United States | 5,671 | \$ 1,332 | \$ 1,312 | 1.5% | 97.4% | 97.6% | (0.2%) | | |
| Phoenix | 1,766 | \$ 1,402 | \$ 1,375 | 2.0% | 98.2% | 98.5% | (0.3%) | | |
| Northern California | 991 | 1,912 | 1,885 | 1.4% | 98.9% | 98.5% | 0.4% | | |
| Las Vegas | 585 | 1,459 | 1,439 | 1.4% | 98.0% | 98.0% | - | | |
| Reno | 247 | 1,731 | 1,718 | 0.8% | 98.3% | 98.5% | (0.2%) | | |
| Southern California | 237 | 1,894 | 1,873 | 1.1% | 99.4% | 99.9% | (0.5%) | | |
| Western United States | 3,826 | \$ 1,595 | \$ 1,570 | 1.6% | 98.4% | 98.5% | (0.1%) | | |
| Tampa | 1,479 | \$ 1,577 | \$ 1,562 | 1.0% | 97.4% | 97.3% | 0.1% | | |
| Southeast Florida | 671 | 1,799 | 1,787 | 0.7% | 96.1% | 96.6% | (0.5%) | | |
| Jacksonville | 465 | 1,366 | 1,350 | 1.2% | 96.8% | 98.0% | (1.2%) | | |
| Orlando | 432 | 1,501 | 1,481 | 1.4% | 96.4% | 98.1% | (1.7%) | | |
| Florida | 3,047 | \$ 1,583 | 1,568 | 1.0% | 96.9% | 97.4% | (0.5%) | | |
| Dallas | 1,198 | \$ 1,498 | \$ 1,482 | 1.1% | 96.0% | 96.0% | - | | |
| Houston | 888 | 1,371 | 1,366 | 0.4% | 96.6% | 95.0% | 1.6% | | |
| San Antonio | 236 | 1,342 | 1,343 | (0.1%) | 95.5% | 94.5% | 1.0% | | |
| Texas | 2,322 | \$ 1,434 | \$ 1,423 | 0.8% | 96.2% | 95.5% | 0.7% | | |
| Indianapolis | 468 | \$ 1,264 | \$ 1,248 | 1.3% | 97.1% | 97.6% | (0.5%) | | |
| Midwest United States | 468 | \$ 1,264 | \$ 1,248 | 1.3% | 97.1% | 97.6% | (0.5%) | | |
| Total/Weighted average | 15,334 | \$ 1,464 | \$ 1,445 | 1.3% | 97.3% | 97.5% | (0.2%) | | |

(1) Represents Tricon's proportionate share of average monthly rent and occupancy of the same home portfolio.

(2) Average monthly rent represents average monthly rental income per unit for occupied units and reflects the impact of rent concessions amortized over the life of the related leases. The year-over-year change in average monthly rent does not equal the average quarterly rent growth, which is calculated as the percentage difference between the monthly rent from an expiring lease and the monthly rent from the next lease.

MANAGEMENT'S DISCUSSION AND ANALYSIS

for the year ended December 31, 2020



| Geography | Renewal | New move-in | Blended | |
|-------------------------|---------|-------------|---------|--|
| Atlanta | 2.9% | 17.1% | 6.6% | |
| Charlotte | 3.3% | 14.6% | 7.0% | |
| Columbia | 2.7% | 8.4% | 5.4% | |
| Southeast United States | 3.0% | 15.4% | 6.6% | |
| Phoenix | 4.5% | 21.4% | 9.5% | |
| Northern California | 4.7% | 8.4% | 5.4% | |
| Las Vegas | 3.7% | 11.7% | 5.9% | |
| Reno | 3.2% | 17.0% | 5.4% | |
| Southern California | 4.5% | 9.0% | 5.0% | |
| Western United States | 4.4% | 16.5% | 7.3% | |
| Tampa | 2.1% | 9.6% | 4.8% | |
| Southeast Florida | 1.3% | 3.3% | 1.9% | |
| Jacksonville | 2.3% | 7.6% | 4.2% | |
| Orlando | 2.4% | 9.3% | 4.6% | |
| Florida | 2.0% | 7.8% | 4.0% | |
| Dallas | 2.2% | 6.3% | 3.9% | |
| Houston | 1.1% | 2.4% | 1.6% | |
| San Antonio | 0.2% | 1.3% | 0.5% | |
| Texas | 1.5% | 4.5% | 2.7% | |
| Indianapolis | 2.6% | 8.9% | 7.0% | |
| Midwest United States | 2.6% | 8.9% | 7.0% | |
| Total/Weighted average | 3.0% | 11.3% | 5.6% | |

(1) Represents quarterly rent growth on Tricon's proportionate share of the same home portfolio, calculated as the percentage difference between the monthly rent from an expiring lease and the monthly rent from the next lease and reflecting the impact of rent concessions amortized over the life of the related lease.

Rental properties

The table below presents the change in Tricon's proportionate share of the fair value of rental properties for the twelve months ended December 31, 2020 and December 31, 2019.

| For the years ended December 31 | | |
|--|--------------|--------------|
| (in thousands of U.S. dollars) | 2020 | 2019 |
| Cost basis of rental properties, beginning of period | \$ 2,913,716 | \$ 2,563,505 |
| Acquisition of rental properties | 124,907 | 281,229 |
| Disposition of rental properties | (16,714) | (18,924) |
| Renovation capital expenditures | 34,354 | 53,538 |
| Recurring capital expenditures | 22,462 | 23,165 |
| Value-enhancing capital expenditures | 10,053 | 11,203 |
| Total cost basis of rental properties | 3,088,778 | 2,913,716 |
| Cumulative fair value adjustment | 852,307 | 634,897 |
| Fair value of rental properties | \$ 3,941,085 | \$ 3,548,613 |

For the twelve months ended December 31, 2020, Tricon acquired 1,836 homes compared to 3,787 for the same period in the prior year. This variance is attributable to the acquisition program being paused for approximately six months in 2020 in light of the COVID-19 pandemic and Tricon's decision to focus on its existing residents and the safety of its operating team. Of the homes acquired during the year, 1,816 were acquired in SFR JV-1 compared to 3,622 in the prior year. As a result of the lower acquisition volume, renovation capital expenditures declined by \$19.2 million or 36% year-over-year. Lower resident turnover during the year also led to a reduction in recurring capital expenditures of \$0.7 million or 3% compared to the prior year.



The following table presents details regarding Tricon's proportionate share of cost to maintain for the single-family rental portfolio.

| (in thousands of U.S. dollars, except | | | | | | | | | | | |
|--|----|---------|----|--------|----|---------|-------------|-------------|-------------|-------------|-------------|
| cost to maintain per home and cost to maintain per square foot) | C | 24 2020 | C | 2020 | (| 22 2020 | Q1 2020 | Q4 2019 | Q3 2019 | Q2 2019 | Q1 2019 |
| Recurring operating expense | | | | | | | | | | | |
| Repairs and maintenance | | | | | | | | | | | |
| operating expense | \$ | 4,057 | \$ | 4,023 | \$ | 3,680 | \$ 3,655 | \$ 3,673 | \$ 3,550 | \$ 3,459 | \$ 3,108 |
| Turnover operating expense | | 986 | | 1,368 | | 1,504 | 1,337 | 1,877 | 1,731 | 1,458 | 1,555 |
| Total recurring operating expense | | 5,043 | | 5,391 | | 5,184 | 4,992 | 5,550 | 5,281 | 4,917 | 4,663 |
| Recurring capital expenditures | | | | | | | | | | | |
| Repairs and maintenance | | | | | | | | | | | |
| capital expense | \$ | 5,129 | \$ | 5,666 | \$ | 4,330 | \$ 4,136 | \$ 2,818 | \$ 4,848 | \$ 5,020 | \$ 4,104 |
| Turnover capital expense | | 421 | | 726 | | 628 | 1,426 | 1,299 | 1,587 | 2,063 | 1,426 |
| Total recurring capital | | | | | | | | | | | |
| expenditures | | 5,550 | | 6,392 | | 4,958 | 5,562 | 4,117 | 6,435 | 7,083 | 5,530 |
| Total cost to maintain | | 10,593 | | 11,783 | | 10,142 | 10,554 | 9,667 | 11,716 | 12,000 | 10,193 |
| Annualized recurring operating | | | | | | | | | | | |
| expense per home | | 1,152 | | 1,238 | | 1,193 | 1,160 | 1,317 | 1,277 | 1,204 | 1,381 |
| Annualized recurring capital | | | | | | | | | | | |
| expense per home | | 1,267 | | 1,468 | | 1,141 | 1,293 | 977 | 1,555 | 1,734 | 1,164 |
| Total annualized cost | | | | | | | | | | | |
| to maintain per home | \$ | 2,419 | \$ | 2,706 | \$ | 2,334 | \$ 2,453 | \$ 2,294 | \$ 2,832 | \$ 2,938 | \$ 2,545 |
| Total annualized cost to | | | | | | | | | | | |
| maintain per square foot | \$ | 1.50 | \$ | 1.68 | \$ | 1.45 | \$ 1.52 | \$ 1.42 | \$ 1.76 | \$ 1.83 | \$ 1.59 |

Total cost to maintain was \$10.6 million for the three months ended December 31, 2020, an increase of \$0.9 million compared to the same period in the prior year, driven by higher repairs and maintenance expense on a larger portfolio of homes and an increase in repair activities from severe weather conditions in the Atlanta and Southeast Florida markets. This higher repair expense was partially offset by lower turnover expense as fewer residents moved out during the quarter, as well as improved cost discipline and controlled scoping of work when turning homes.

The following table provides details regarding Tricon's proportionate share of total capital expenditures incurred for the singlefamily rental portfolio.

| (in thousands of U.S. dollars) | Q4 2020 | Q3 2020 | Q2 2020 | Q1 2020 | Q4 2019 | Q3 2019 | Q2 2019 | Q1 2019 |
|--|-----------|-----------|-----------|-----------|-----------|-----------|-----------|-----------|
| Renovation capital expenditures ⁽¹⁾ | \$ 13,376 | \$ 6,020 | \$ 5,952 | \$ 9,006 | \$ 11,745 | \$ 17,952 | \$ 12,561 | \$ 11,282 |
| Recurring capital expenditures ⁽²⁾ | 5,550 | 6,392 | 4,958 | 5,562 | 4,117 | 6,435 | 7,083 | 5,530 |
| Value-enhancing | | | | | | | | |
| capital expenditures ⁽³⁾ | 2,141 | 2,525 | 2,728 | 2,659 | 3,014 | 2,983 | 2,549 | 2,656 |
| Total capital expenditures | \$ 21,067 | \$ 14,937 | \$ 13,638 | \$ 17,227 | \$ 18,876 | \$ 27,370 | \$ 22,193 | \$ 19,468 |

(1) Renovation capital expenditures are incurred in order to prepare the property for rental use in accordance with Tricon's standards. These expenditures are either incurred shortly after acquisition on vacant homes or deferred until the resident moves out if homes are occupied when acquired.

(2) Recurring capital expenditures represent ongoing costs associated with maintaining and preserving the quality of a property after the home has been renovated. (3) Value-enhancing capital expenditures are defined as capital expenditures that go above and beyond maintaining the quality of a property and are incurred for the purpose of increasing expected future returns.

Total capital expenditures were \$21.1 million for the three months ended December 31, 2020, an increase of \$2.2 million compared to the same period in the prior year. The variance was primarily attributable to an increase in renovation capital expenditures as a higher number of homes were acquired through organic channels during the quarter and required upfront renovations (708 of the home acquisitions in the comparative period were part of a stabilized portfolio transaction).


4.2 Multi-Family Rental

Tricon's multi-family rental business segment consists of 24 assets, including 23 predominantly garden-style apartments in the U.S. Sun Belt and one Class A high-rise property in downtown Toronto (note that eight other properties in downtown Toronto are currently under development and are discussed in <u>Section 4.3</u>).

4.2.1 U.S. multi-family rental

Several metrics in the tables and disclosure throughout this subsection relating to periods prior to the Company's ownership of the U.S. multi-family rental portfolio are KPI measures that were reported historically (refer to Starlight U.S. Multi-Family (No. 5) Core Fund profile on SEDAR at www.sedar.com) while some are Tricon KPIs (as defined in <u>Section 7.1</u>) not previously reported. Any differences are described in the notes to the relevant tables below. Management believes this historical information is useful in understanding the performance of the acquired portfolio.

Operating results

The Company's U.S. multi-family rental business continued to experience softness in demand and pricing amid the negative economic and employment impact caused by the COVID-19 pandemic. Growing preference for detached single-family rental homes as opposed to higher density apartments weakened multi-family rental market fundamentals. While Tricon's U.S. multi-family rental business has faced near-term challenges in maintaining occupancy levels and driving rent growth throughout the pandemic, the fourth quarter performance improved sequentially compared to the third quarter of 2020.

The U.S. multi-family rental business reported a 56.6% NOI margin (a 140-bps improvement from 55.2% in Q3 2020), 93.6% occupancy rate and collections equal to 98% of rent billed during the quarter. The Company continued to adopt an occupancybiased rental strategy by offering lower rents and/or higher concessions, which led to a 0.8% increase in occupancy compared to the preceding quarter but a decline in new lease rents of 5.6% (inclusive of concessions). Renewal rent growth, however, improved meaningfully to 2.3%, and reflects a preference of existing residents to remain in place, even at higher rents, which bodes well for 2021 performance. This positive trend has continued into 2021, and average blended rent growth increased to 1.1% in January 2021, registering the first month of positive blended rent growth since February of 2020.

Annualized turnover decreased by 15.3% to 46.5% in the fourth quarter compared to 61.8% in the third quarter. This favourable change was driven by Tricon's efforts to provide a higher level of customer service to residents and a reduced desire to move during the pandemic. On a full-year basis, resident turnover was 50.8% compared to 52.6% in the prior year.

The Company's U.S. multi-family rental business continued to experience performance challenges in the Houston and Orlando markets, which make up 32% of the total suites in the portfolio. These markets have been among the hardest hit by the COVID-19 pandemic, with local employment being further impacted by weak oil prices and diminished tourism, respectively. Operating performance was also impeded by higher bad debt ascribable to slower collections, which remain volatile quarter-over-quarter. Management has made active efforts to connect residents with local rent relief programs and has also created incentive plans for residents who are habitually delinquent to get current on their rent payments.

Despite lower NOI in the fourth quarter compared to the same period in the prior year, the U.S. multi-family rental Core FFO has remained flat as the decrease in NOI was fully offset by savings in interest expense on the portfolio's floating-rate loan facilities (see Section 5).

A weak labour market and muted wage growth weighed on multi-family rental demand in 2020; however, the broader industry is expected to rebound in 2021 and 2022 as the population is vaccinated, the economy recovers, and demand increases for high-quality multi-family housing that offers a strong sense of community and first-class amenities. Meanwhile, management continues to execute an occupancy-biased rental strategy and aims to achieve a positive blended rent growth while maintaining occupancy.

4.2 Multi-Family Rental

The table below provides a summary of certain operating metrics for the U.S. multi-family rental segment that management uses to evaluate performance over time. The metrics are key drivers of revenue for the multi-family rental business and ultimately its NOI (KPI measure; refer to Section 7.1).

| | Q4 2020 | Q3 2020 | Q2 2020 | Q1 2020 | Q4 2019 | Q3 2019 | Q2 2019 | Q1 2019 |
|--|----------|----------|----------|----------|----------|----------|----------|----------|
| Number of suites | 7,289 | 7,289 | 7,289 | 7,289 | 7,289 | 7,289 | 7,289 | 7,289 |
| Occupancy ⁽¹⁾ | 93.6% | 92.8% | 93.5% | 94.4% | 94.9% | 95.2% | 94.7% | 93.1% |
| Annualized turnover rate | 46.5% | 61.8% | 46.5% | 47.5% | 51.3% | 52.9% | 53.5% | N/A |
| Average monthly $rent^{(2),(3)}$ | \$ 1,217 | \$ 1,228 | \$ 1,240 | \$ 1,244 | \$ 1,233 | \$ 1,234 | \$ 1,236 | \$ 1,232 |
| Average quarterly rent growth – renewals ^{(2),(4)} | 2.3% | 1.2% | _ | 3.4% | 4.6% | 4.5% | 4.4% | N/A |
| Average quarterly rent growth – new move-in ^{(2),(4)} | (5.6%) | (4.5%) | (5.5%) | (1.7%) | (1.7%) | _ | (0.3%) | N/A |
| Average quarterly rent growth – blended ^{(2),(4)} | (1.8%) | (2.0%) | (2.2%) | 1.1% | 1.1% | 2.1% | 1.9% | N/A |

(1) The occupancy rate from Q2 2019 to Q4 2020 represents average physical occupancy (refer to <u>Section 7.1</u> for Tricon's definition of this KPI), while the occupancy rate for Q1 2019 represents economic occupancy as previously reported by the U.S. multi-family rental portfolio under prior ownership.

(2) These metrics are Tricon's KPIs and they were not previously disclosed by the U.S. multi-family rental portfolio under prior ownership.

(3) Average monthly rent represents average monthly rental income per suite for occupied suites and reflects the impact of rent concessions amortized over the life of the related leases.

(4) Average rent growth during the period represents the percentage difference between the monthly rent from an expiring lease and the monthly rent from the next lease and reflects the impact of rent concessions amortized over the life of the related lease. Excluding the impact of concessions, the Q4 2020 rent growth was 2.2% for renewals, (4.8%) for new move-ins, and (1.4%) blended.

The table below presents a breakdown of Tricon's NOI (KPI measure; refer to <u>Section 7.1</u>) for the U.S. multi-family rental business. The financial information presented in the table includes prior-year results reported for comparability, although Tricon did not own the portfolio prior to June 11, 2019. Management believes this information is useful in understanding the performance of the acquired portfolio.

| | | Three months | | | | | Twelve months | | | | |
|--|-----------|--------------|-----------|---------|----|---------|---------------|---------|------------|---------|------------|
| For the periods ended December 31 | | % of | | % of | | | | % of | | % of | |
| (in thousands of U.S. dollars) | 2020 | revenue | 2019 | revenue | V | ariance | 2020 | revenue | 2019 | revenue | Variance |
| Rental revenue | \$ 25,159 | | \$ 25,605 | | \$ | (446) | \$ 100,938 | | \$ 101,781 | | \$ (843) |
| Concessions and abatements | (565) | | (210) | | | (355) | (1,903) | | (869) | | (1,034) |
| Fees and other revenue ⁽¹⁾ | 3,829 | | 3,491 | | | 338 | 14,700 | | 13,926 | | 774 |
| Bad debt expense | (840) | | (319) | | | (521) | (2,530) | | (1,079) | | (1,451) |
| Total revenue from rental properties | \$ 27,583 | 100% | \$ 28,567 | 100% | \$ | (984) | \$ 111,205 | 100% | \$ 113,759 | 100% | \$ (2,554) |
| Property taxes | 4,418 | 16% | 4,377 | 15% | | (41) | 18,623 | 17% | 18,354 | 16% | (269) |
| Repairs, maintenance and turnover | 1,214 | 4% | 1,119 | 4% | | (95) | 4,411 | 4% | 4,108 | 4% | (303) |
| Property management expenses | 3,092 | 11% | 2,936 | 10% | | (156) | 12,097 | 11% | 11,995 | 11% | (102) |
| Utilities and other direct costs ⁽²⁾ | 1,840 | 7% | 1,773 | 6% | | (67) | 7,514 | 7% | 6,709 | 6% | (805) |
| Property insurance | 632 | 2% | 494 | 2% | | (138) | 2,472 | 2% | 1,915 | 2% | (557) |
| Marketing and leasing | 426 | 2% | 384 | 1% | | (42) | 1,413 | 1% | 1,582 | 1% | 169 |
| Other property operating expenses | 357 | 1% | 520 | 2% | | 163 | 1,766 | 2% | 1,926 | 2% | 160 |
| Total direct operating expenses | 11,979 | | 11,603 | | | (376) | 48,296 | | 46,589 | | (1,707) |
| Net operating income (NOI) ⁽³⁾ | \$ 15,604 | | \$ 16,964 | | \$ | (1,360) | \$ 62,909 | | \$ 67,170 | | \$ (4,261) |
| Net operating income (NOI) margin ⁽³⁾ | 56.6% | | 59.4% | | | | 56.6% | | 59.0% | | |

Note: Given that the suite count did not change from 2019 to 2020, this should also be considered the "Same Property" portfolio.

(1) The comparative period has been reclassified to conform with the current period presentation. One-time insurance recoveries of \$270 for the twelve months ended December 31, 2019 have been reclassified out of NOI since they do not meet Tricon's definition of operating activities. The amount was previously included in revenue by the U.S. multi-family rental portfolio under prior ownership. No insurance recoveries were recorded in the three months ended December 31, 2019.

(2) Utilities and other direct costs include water and sewer expense, valet waste expense, electricity and gas and cable contract costs.

(3) KPI measures; see Section 7.1.

4.2 Multi-Family Rental

For the three months ended December 31, 2020, revenue decreased by \$1.0 million or 3.4% to \$27.6 million compared to \$28.6 million for the same period in 2019. The variance is primarily a result of (i) a decrease of 1.3% in occupancy attributable to weaker leasing demand during the COVID-19 pandemic, (ii) an increase of \$0.4 million in leasing concessions to attract or retain residents (equivalent to 0.6 weeks of average rent per lease signed in Q4 2020, comprised of 1 week for new leases and 0.2 weeks for renewals leases), and (iii) incremental bad debt expense of \$0.5 million as a larger percentage of accounts receivable is expected to become uncollectible as a result of the pandemic.

Tricon reserved 100% of residents' accounts receivable balances aged more than 30 days, incorporating management's conservative measurement when estimating the collectability of outstanding amounts. The decrease in revenue was partially offset by \$0.3 million in additional income from ancillary service offerings such as bundled media packages (largely cable and internet) and onsite or offsite parcel storage, resulting in total fees and other revenue of \$3.8 million in the quarter.

Total operating expenses increased by \$0.4 million or 3.2% to \$12.0 million. Notable operating expense variances for the quarter include:

- Property taxes Fourth quarter property tax expense remained flat year-over-year at \$4.4 million, as successful appeal efforts in the fourth quarter of 2020 resulted in tax recoveries that offset normal course increases. On a full-year basis, property tax expense growth of \$0.3 million or 1.5% was relatively muted and reflected strong efforts in contesting property tax assessments.
- Property management expenses Property management expenses increased by \$0.2 million to \$3.1 million driven by the Company's use of additional employee incentive programs in the current quarter to encourage higher occupancy targets, higher rent growth, and lower turnover in the challenging pandemic leasing environment.
- **Repairs, maintenance and turnover** Expenses increased by \$0.1 million to \$1.2 million driven by increased security costs during the pandemic.
- Property insurance Property insurance costs increased by \$0.1 million year-over-year to \$0.6 million, reflecting significantly
 higher insurance premiums across the industry and comparable to similarly-sized portfolios in Tricon's markets. In the new year,
 insurance premiums are expected to increase by approximately 10% compared to 2020 based on the Company's recently
 negotiated insurance pricing.
- Other property operating expenses Other property operating expenses decreased by \$0.2 million to \$0.4 million as a result of cost containment efforts to reduce property-level general and administration expenses.

As a result of these drivers, the portfolio NOI decreased by \$1.4 million or 8.0% to \$15.6 million in the fourth quarter of 2020 compared to \$17.0 million in the fourth quarter of 2019. NOI margin decreased to 56.6% in the fourth quarter of 2020 from 59.4% for the same period in the prior year. Of note, the Company expensed all rent concessions instead of amortizing them over the expected life of the lease term. NOI in the fourth quarter of 2020 would have been \$15.8 million, a \$1.2 million or 7.1% decrease year-over-year, if leasing concessions were reflected on an amortized basis.



Market-level results

The U.S. multi-family rental business is diversified across 13 markets. Market-level details on all the properties owned by the Company are presented below.

| | | | Average cost | | Average suite |
|-------------------------|------------|-----------------|--------------|--------|-----------------|
| Geography | Properties | Average vintage | per property | Suites | size (sq. feet) |
| Austin | 4 | 2010 | \$ 61,884 | 1,454 | 941 |
| Houston | 3 | 2009 | 55,260 | 1,098 | 942 |
| Dallas | 2 | 2012 | 52,335 | 640 | 922 |
| San Antonio | 1 | 2013 | 39,575 | 276 | 874 |
| Texas | 10 | 2010 | \$ 55,756 | 3,468 | 932 |
| Orlando | 4 | 2012 | \$ 69,534 | 1,215 | 1,059 |
| Tampa | 1 | 2014 | 64,967 | 304 | 998 |
| Florida | 5 | 2012 | \$ 68,620 | 1,519 | 1,047 |
| Atlanta | 2 | 2012 | \$ 61,169 | 607 | 860 |
| Charlotte | 1 | 2015 | 59,014 | 320 | 973 |
| Nashville | 1 | 2015 | 47,625 | 288 | 1,085 |
| Raleigh | 1 | 2014 | 51,280 | 265 | 996 |
| Southeast United States | 5 | 2014 | \$ 56,051 | 1,480 | 953 |
| Las Vegas | 1 | 2012 | \$ 62,169 | 320 | 1,042 |
| Phoenix | 1 | 2012 | 54,398 | 274 | 966 |
| Denver | 1 | 2014 | 56,423 | 228 | 930 |
| Western United States | 3 | 2013 | \$ 57,664 | 822 | 986 |
| Total/Weighted average | 23 | 2012 | \$ 58,866 | 7,289 | 966 |

Year-over-year comparison

| | | | | NOI | | NOI margin | | | |
|-------------------------|--------|----------|---|-----------|------------|------------|---------|------------|--|
| Geography | Suites | Q4 20 | 0 | Q4 2019 | Change (%) | Q4 2020 | Q4 2019 | Change (%) | |
| Austin | 1,454 | \$ 3,06 | 4 | \$ 3,296 | (7.0%) | 54.5% | 59.7% | (5.2%) | |
| Houston | 1,098 | 1,79 | 8 | 2,176 | (17.4%) | 50.4% | 54.8% | (4.4%) | |
| Dallas | 640 | 98 | 9 | 1,017 | (2.8%) | 46.2% | 45.7% | 0.5% | |
| San Antonio | 276 | 45 | 6 | 514 | (11.3%) | 48.5% | 52.0% | (3.5%) | |
| Texas | 3,468 | \$ 6,30 | 7 | \$ 7,003 | (9.9%) | 51.4% | 55.1% | (3.7%) | |
| Orlando | 1,215 | \$ 3,03 | 1 | \$ 3,364 | (9.9%) | 60.6% | 62.3% | (1.7%) | |
| Tampa | 304 | 8 | 1 | 828 | 5.2% | 68.2% | 65.5% | 2.7% | |
| Florida | 1,519 | \$ 3,90 | 2 | \$ 4,192 | (6.9%) | 62.1% | 62.9% | (0.8%) | |
| Atlanta | 607 | \$ 9 | 3 | \$ 1,337 | (27.2%) | 43.2% | 55.2% | (12.0%) | |
| Charlotte | 320 | 78 | 2 | 768 | 1.8% | 64.7% | 62.8% | 1.9% | |
| Nashville | 288 | 74 | 3 | 732 | 1.5% | 65.7% | 65.5% | 0.2% | |
| Raleigh | 265 | 60 | 7 | 644 | (5.7%) | 59.2% | 60.0% | (0.8%) | |
| Southeast United States | 1,480 | \$ 3,10 | 5 | \$ 3,481 | (10.8%) | 55.3% | 59.7% | (4.4%) | |
| Las Vegas | 320 | \$ 8! | 8 | \$ 819 | 4.8% | 70.3% | 68.7% | 1.6% | |
| Phoenix | 274 | 72 | 6 | 715 | 1.5% | 64.9% | 66.2% | (1.3%) | |
| Denver | 228 | 70 | 6 | 754 | (6.4%) | 65.8% | 69.4% | (3.6%) | |
| Western United States | 822 | \$ 2,29 | 0 | \$ 2,288 | 0.1% | 67.1% | 68.1% | (1.0%) | |
| Total/Weighted average | 7,289 | \$ 15,60 | 4 | \$ 16,964 | (8.0%) | 56.6% | 59.4% | (2.8%) | |

The Houston and Orlando markets are among the economies hardest hit by COVID-19, driven by elevated unemployment in the oil and hospitality-related industries, respectively. While the NOI and NOI margin for the Company's portfolio have decreased year-over-year, they have improved from the third quarter of 2020, as shown in the table below. Management expects NOI will continue to improve into 2021 (see "Non-IFRS measures and forward-looking statements" on page 24).

MANAGEMENT'S DISCUSSION AND ANALYSIS

for the year ended December 31, 2020



Quarter-over-quarter comparison

| | | | | | NOI | | NOI margin | | | |
|-------------------------|--------|----|---------|----|---------|------------|------------|---------|------------|--|
| Geography | Suites | C | 24 2020 | (| 23 2020 | Change (%) | Q4 2020 | Q3 2020 | Change (%) | |
| Austin | 1,454 | \$ | 3,064 | \$ | 3,046 | 0.6% | 54.5% | 54.1% | 0.4% | |
| Houston | 1,098 | | 1,798 | | 1,782 | 0.9% | 50.4% | 48.7% | 1.7% | |
| Dallas | 640 | | 989 | | 903 | 9.5% | 46.2% | 42.0% | 4.2% | |
| San Antonio | 276 | | 456 | | 418 | 9.1% | 48.5% | 46.7% | 1.8% | |
| Texas | 3,468 | \$ | 6,307 | \$ | 6,149 | 2.6% | 51.4% | 49.8% | 1.6% | |
| Orlando | 1,215 | \$ | 3,031 | \$ | 2,828 | 7.2% | 60.6% | 57.1% | 3.5% | |
| Tampa | 304 | | 871 | | 798 | 9.1% | 68.2% | 63.4% | 4.8% | |
| Florida | 1,519 | \$ | 3,902 | \$ | 3,626 | 7.6% | 62.1% | 58.3% | 3.8% | |
| Atlanta | 607 | \$ | 973 | \$ | 1,131 | (14.0%) | 43.2% | 50.2% | (7.0%) | |
| Charlotte | 320 | | 782 | | 763 | 2.5% | 64.7% | 64.1% | 0.6% | |
| Nashville | 288 | | 743 | | 687 | 8.2% | 65.7% | 64.1% | 1.6% | |
| Raleigh | 265 | | 607 | | 599 | 1.3% | 59.2% | 59.2% | - | |
| Southeast United States | 1,480 | \$ | 3,105 | \$ | 3,180 | (2.4%) | 55.3% | 57.6% | (2.3%) | |
| Las Vegas | 320 | \$ | 858 | \$ | 766 | 12.0% | 70.3% | 67.2% | 3.1% | |
| Phoenix | 274 | | 726 | | 676 | 7.4% | 64.9% | 62.0% | 2.9% | |
| Denver | 228 | | 706 | | 717 | (1.5%) | 65.8% | 66.3% | (0.5%) | |
| Western United States | 822 | \$ | 2,290 | \$ | 2,159 | 6.1% | 67.1% | 65.2% | 1.9% | |
| Total/Weighted average | 7,289 | \$ | 15,604 | \$ | 15,114 | 3.2% | 56.6% | 55.2% | 1.4% | |

Market-level details for average monthly rent and physical occupancy for the fourth quarter of 2020 and applicable comparative periods are shown below.

Year-over-year comparison

| | | Av | erage r | nonthly ren | t ⁽¹⁾ | Occupancy | | | |
|-------------------------|--------|-------------|---------|-------------|------------------|-----------|---------|------------|--|
| Geography | Suites | Q4 2020 | | Q4 2019 | Change (%) | Q4 2020 | Q4 2019 | Change (%) | |
| Austin | 1,454 | \$ 1,176 | \$ | 1,170 | 0.5% | 93.4% | 95.5% | (2.1%) | |
| Houston | 1,098 | 1,112 | | 1,143 | (2.7%) | 93.1% | 95.9% | (2.8%) | |
| Dallas | 640 | 1,103 | | 1,127 | (2.1%) | 93.5% | 92.4% | 1.1% | |
| San Antonio | 276 | 1,076 | | 1,124 | (4.3%) | 93.8% | 95.6% | (1.8%) | |
| Texas | 3,468 | \$ 1,134 | \$ | 1,150 | (1.4%) | 93.4% | 95.1% | (1.7%) | |
| Orlando | 1,215 | \$ 1,342 | \$ | 1,414 | (5.1%) | 92.6% | 94.2% | (1.6%) | |
| Tampa | 304 | 1,338 | | 1,321 | 1.3% | 95.3% | 96.2% | (0.9%) | |
| Florida | 1,519 | \$ 1,341 | \$ | 1,395 | (3.9%) | 93.2% | 94.6% | (1.4%) | |
| Atlanta | 607 | \$ 1,303 | \$ | 1,301 | 0.2% | 93.5% | 93.5% | - | |
| Charlotte | 320 | 1,183 | | 1,171 | 1.0% | 92.9% | 95.6% | (2.7%) | |
| Nashville | 288 | 1,204 | | 1,207 | (0.2%) | 95.1% | 95.4% | (0.3%) | |
| Raleigh | 265 | 1,180 | | 1,188 | (0.7%) | 93.8% | 95.5% | (1.7%) | |
| Southeast United States | 1,480 | \$ 1,236 | \$ | 1,234 | 0.2% | 93.8% | 94.7% | (0.9%) | |
| Las Vegas | 320 | \$ 1,223 | \$ | 1,209 | 1.2% | 95.8% | 94.5% | 1.3% | |
| Phoenix | 274 | 1,247 | | 1,227 | 1.6% | 95.3% | 94.4% | 0.9% | |
| Denver | 228 | 1,475 | | 1,470 | 0.3% | 94.4% | 96.0% | (1.6%) | |
| Western United States | 822 | \$ 1,301 | \$ | 1,287 | 1.1% | 95.3% | 94.9% | 0.4% | |
| Total/Weighted average | 7,289 | \$ 1,217 | \$ | 1,233 | (1.3%) | 93.6% | 94.9% | (1.3%) | |

(1) Average monthly rent represents average monthly rental income per suite for occupied suites and reflects the impact of rent concessions amortized over the life of the related leases. The year-over-year change in average monthly rent does not equal the average quarterly rent growth, which is calculated as the percentage difference between the monthly rent from an expiring lease and the monthly rent from the next lease.

MANAGEMENT'S DISCUSSION AND ANALYSIS

for the year ended December 31, 2020



Quarter-over-quarter comparison

| | | Ave | erage monthly ren | t ⁽¹⁾ | Occupancy | | | |
|-------------------------|--------|----------|-------------------|------------------|-----------|---------|------------|--|
| Geography | Suites | Q4 2020 | Q3 2020 | Change (%) | Q4 2020 | Q3 2020 | Change (%) | |
| Austin | 1,454 | \$ 1,176 | \$ 1,178 | (0.2%) | 93.4% | 93.9% | (0.5%) | |
| Houston | 1,098 | 1,112 | 1,135 | (2.0%) | 93.1% | 92.2% | 0.9% | |
| Dallas | 640 | 1,103 | 1,128 | (2.2%) | 93.5% | 91.6% | 1.9% | |
| San Antonio | 276 | 1,076 | 1,090 | (1.3%) | 93.8% | 93.3% | 0.5% | |
| Texas | 3,468 | \$ 1,134 | \$ 1,148 | (1.2%) | 93.4% | 92.9% | 0.5% | |
| Orlando | 1,215 | \$ 1,342 | \$ 1,381 | (2.8%) | 92.6% | 90.0% | 2.6% | |
| Tampa | 304 | 1,338 | 1,332 | 0.5% | 95.3% | 95.2% | 0.1% | |
| Florida | 1,519 | \$ 1,341 | \$ 1,371 | (2.2%) | 93.2% | 91.0% | 2.2% | |
| Atlanta | 607 | \$ 1,303 | \$ 1,313 | (0.8%) | 93.5% | 92.4% | 1.1% | |
| Charlotte | 320 | 1,183 | 1,174 | 0.8% | 92.9% | 93.7% | (0.8%) | |
| Nashville | 288 | 1,204 | 1,191 | 1.1% | 95.1% | 94.0% | 1.1% | |
| Raleigh | 265 | 1,180 | 1,194 | (1.2%) | 93.8% | 94.0% | (0.2%) | |
| Southeast United States | 1,480 | \$ 1,236 | \$ 1,238 | (0.2%) | 93.8% | 93.3% | 0.5% | |
| Las Vegas | 320 | \$ 1,223 | \$ 1,207 | 1.3% | 95.8% | 95.0% | 0.8% | |
| Phoenix | 274 | 1,247 | 1,233 | 1.1% | 95.3% | 95.4% | (0.1%) | |
| Denver | 228 | 1,475 | 1,473 | 0.1% | 94.4% | 94.0% | 0.4% | |
| Western United States | 822 | \$ 1,301 | \$ 1,289 | 0.9% | 95.3% | 94.8% | 0.5% | |
| Total/Weighted average | 7,289 | \$ 1,217 | \$ 1,228 | (0.9%) | 93.6% | 92.8% | 0.8% | |

(1) Average monthly rent represents average monthly rental income per suite for occupied suites and reflects the impact of rent concessions amortized over the life of the related leases. The quarter-over-quarter change in average monthly rent does not equal the average quarterly rent growth, which is calculated as the percentage difference between the monthly rent from an expiring lease and the monthly rent from the next lease.

Rent growth by market for the U.S. multi-family rental portfolio for the fourth quarter of 2020 is presented below.

| | | Rent growth ⁽¹⁾ | | | | | | |
|-------------------------|---------|----------------------------|---------|--|--|--|--|--|
| Geography | Renewal | New move-in | Blended | | | | | |
| Austin | 3.3% | (5.0%) | (2.0%) | | | | | |
| Houston | 0.1% | (8.2%) | (4.2%) | | | | | |
| Dallas | 2.1% | (5.2%) | (2.4%) | | | | | |
| San Antonio | 1.5% | (11.1%) | (5.7%) | | | | | |
| Texas | 1.9% | (6.5%) | (3.1%) | | | | | |
| Orlando | 1.6% | (9.4%) | (5.4%) | | | | | |
| Tampa | 4.4% | 4.0% | 4.2% | | | | | |
| Florida | 2.1% | (6.7%) | (3.5%) | | | | | |
| Atlanta | 2.5% | (2.9%) | (0.7%) | | | | | |
| Charlotte | 2.8% | (0.3%) | 1.6% | | | | | |
| Nashville | 1.4% | 3.4% | 2.0% | | | | | |
| Raleigh | 2.8% | (4.8%) | 0.1% | | | | | |
| Southeast United States | 2.4% | (1.4%) | 0.5% | | | | | |
| Las Vegas | 3.8% | 6.0% | 4.7% | | | | | |
| Phoenix | 4.5% | 7.6% | 5.9% | | | | | |
| Denver | 1.8% | (8.2%) | (4.4%) | | | | | |
| Western United States | 3.5% | 2.6% | 2.5% | | | | | |
| Total/Weighted average | 2.3% | (5.6%) | (1.8%) | | | | | |

(1) Average rent growth during the quarter represents the percentage difference between the monthly rent from an expiring lease and the monthly rent from the next lease and reflects the impact of rent concessions amortized over the life of the related lease.



Rental properties

The table below presents the change in the fair value of rental properties for the twelve months ended December 31, 2020.

| For the year ended December 31 | |
|--|--------------|
| (in thousands of U.S. dollars) | 2020 |
| Cost basis of rental properties, beginning of year | \$ 1,344,844 |
| Recurring capital expenditures | 5,373 |
| In-suite value-enhancing capital expenditures | 1,399 |
| Common area value-enhancing capital expenditures | 2,295 |
| Total cost basis of rental properties | \$ 1,353,911 |
| Cumulative fair value adjustment ⁽¹⁾ | (22,535) |
| Fair value of rental properties | \$ 1,331,376 |

(1) The Company determined the fair value of each investment property using the direct income capitalization approach.

For the twelve months ended December 31, 2020, recurring capital expenditures of \$5.4 million were incurred predominately for exterior remodelling and turnover activities such as in-suite appliance replacements and full carpet replacements.

In-suite value-enhancing capital expenditures for the year totalled \$1.4 million related to flooring and kitchen upgrades, installation of smart home technology and upgraded stainless steel washer and dryer units.

In addition, \$2.3 million was invested to improve common area amenities as several large capital projects resumed in the fourth quarter of 2020 after a temporary suspension during the COVID-19 pandemic. Projects included clubhouse renovations, pet facility improvements, balcony resurfacing and water-efficient irrigation upgrades. Management believes these projects will provide a more enjoyable resident experience, improve the properties' curb appeal and reduce future utility expenses.

The following table provides details regarding costs to maintain for the three months ended December 31, 2020 and applicable comparative periods. The financial information presented in the two tables below includes prior-year results reported for comparability, although Tricon did not own the portfolio prior to June 11, 2019. Management believes this information is useful in understanding the performance of the acquired portfolio.

| to maintain per square foot | \$ 1.75 | \$ 1.51 | \$ 1.06 | \$ 1.23 | \$ 1.61 | \$ 1.45 | \$ 1.27 | \$ 1.10 |
|--|----------|----------|----------|----------|----------|----------|----------|----------|
| Total annualized cost | + 1,000 | ÷ 1,101 | ÷ 1,020 | ÷ 1,107 | ÷ 1,001 | ÷ 1,105 | ÷ 1,220 | ÷ 1,007 |
| Total annualized cost to maintain per suite | \$ 1,695 | \$ 1.461 | \$ 1,026 | \$ 1,187 | \$ 1,551 | \$ 1,403 | \$ 1,228 | \$ 1,067 |
| expense per suite | 1,029 | 638 | 508 | 589 | 937 | 782 | 695 | 581 |
| Annualized recurring capital | | | | | | | | |
| Annualized recurring operating expense per suite | 666 | 823 | 518 | 598 | 614 | 621 | 533 | 486 |
| Total cost to maintain | \$ 3,089 | \$ 2,662 | \$ 1,870 | \$ 2,163 | \$ 2,826 | \$ 2,556 | \$ 2,238 | \$ 2,324 |
| | | , | | , | , - | , - | , | , |
| Total recurring capital expenditures | 1,875 | 1.499 | 926 | 1.073 | 1.707 | 1.425 | 1.266 | 1.058 |
| Turnover capital expense | 934 | 961 | 462 | 628 | 962 | 946 | 758 | 771 |
| capital expense | 941 | 538 | 464 | 445 | 745 | 479 | 508 | 287 |
| Repairs and maintenance | | 500 | 46.4 | 4.45 | 745 | 470 | 500 | 207 |
| Recurring capital expenditures | | | | | | | | |
| Total recurring operating expense | 1,214 | 1,163 | 944 | 1,090 | 1,119 | 1,131 | 972 | 886 |
| Turnover operating expense | 197 | 112 | 70 | 155 | 143 | 150 | 92 | 52 |
| operating expense | \$ 1,017 | \$ 1,051 | \$ 874 | \$ 935 | \$ 976 | \$ 981 | \$ 880 | \$ 834 |
| Recurring operating expense Repairs and maintenance | | | | | | | | |
| (in thousands of U.S. dollars, except cost to maintain per suite and cost to maintain per square foot) | Q4 2020 | Q3 2020 | Q2 2020 | Q1 2020 | Q4 2019 | Q3 2019 | Q2 2019 | Q1 2019 |

Total cost to maintain increased by \$0.3 million to \$3.1 million for the three months ended December 31, 2020 compared to the same period in the prior year. Additional work was performed at the properties for repairs and maintenance to replace in-suite appliances and restore common areas in order to attract and retain residents.



The following table provides details regarding recurring and non-recurring capital expenditures for the U.S. multi-family rental portfolio.

| (in thousands of U.S. dollars) | Q4 2020 | Q3 2020 | Q2 2020 | Q1 2020 | Q4 2019 | Q3 2019 | Q2 2019 | Q1 2019 |
|--|----------|----------|----------|----------|----------|----------|----------|----------|
| Recurring capital expenditures ⁽¹⁾ | \$ 1,875 | \$ 1,499 | \$ 926 | \$ 1,073 | \$ 1,707 | \$ 1,425 | \$ 1,266 | \$ 1,058 |
| In-suite value-enhancing capital expenditures ⁽²⁾ | 333 | 189 | 393 | 484 | 709 | 585 | 502 | 592 |
| Common area value-enhancing capital expenditures ⁽²⁾ | 1,336 | 250 | 256 | 453 | 870 | 594 | 578 | 679 |
| Total capital expenditures | \$ 3,544 | \$ 1,938 | \$ 1,575 | \$ 2,010 | \$ 3,286 | \$ 2,604 | \$ 2,346 | \$ 2,329 |

(1) Recurring capital expenditures represent ongoing costs associated with maintaining and preserving the quality of a property including significant work performed during the turnover of a suite.

(2) Value-enhancing capital expenditures are defined as capital expenditures that go above and beyond maintaining the quality of a property and are incurred for the purpose of increasing expected future returns. These costs for the multi-family portfolio are split between work performed in-suite and that performed on common area spaces and amenities.

Total capital expenditures were \$3.5 million for the three months ended December 31, 2020, an increase of \$0.3 million compared to the same period in the prior year attributable to the delay of several large-scale common area projects planned which resumed construction in the fourth quarter of 2020.

4.2.2 Canadian multi-family rental – The Selby

As at December 31, 2020, Tricon's Canadian multi-family rental portfolio included its first operating building, The Selby, located in downtown Toronto. The Selby was substantially completed and nearly stabilized and was therefore reclassified from the Residential Development segment (Section 4.3) to the Multi-Family Rental segment for internal and external reporting purposes during the first quarter of 2020.

Operating results

Similar to Tricon's U.S. multi-family rental business, the Company's Canadian multi-family rental business continued to experience softening demand fundamentals during the quarter as a result of the challenges presented by the COVID-19 pandemic. The confluence of border restrictions, curtailed immigration, elevated unemployment, work-from-home orders, and temporary migration to suburban regions for more space continued to place downward pressure on occupancy and rents in downtown Toronto. This impact was further exacerbated by increased supply, attributable to increased rental listings of vacant condominium units (recently completed buildings and units withdrawn from the short-term rental market) and new purpose-built rental buildings coming online. The slowdown in demand and additional supply increased competition among rental housing providers resulting in lower rental rates and heightened concessions across the rental market. Despite these challenges, purpose-built rental properties have performed better than condominium rental properties in terms of average rent, occupancy and renewal rates as residents pursue a flight to quality and the benefits of professionally managed buildings, which are now available at a lower price point.

During the fourth quarter, the Canadian multi-family rental business reported an NOI margin of 55.6%, representing a 1.6% increase over the prior quarter. The Company continued with several of its marketing initiatives from the previous quarter to attract and retain residents, such as online resident engagement programs, targeted digital advertising campaigns and customer service training for on-site staff. Tricon proactively transitioned to an occupancy bias in the fourth quarter by offering pricing adjustments for suites on lower floors of The Selby. The strategic use of incentives and flexible leasing options to secure and retain residents helped maintain occupancy at 87.0% and reduced turnover to 41.6% but negatively impacted blended rent growth which decreased to (5.1%).

While these metrics remain well below long-term expectations, the Company firmly believes that the downtown Toronto rental market will bounce back as the COVID-19 vaccination program is accelerated, businesses re-open and higher education students return to in-person learning (see "<u>Non-IFRS measures and forward-looking statements</u>" on page 24).



The tables in this section provide a summary of certain operating metrics for the Canadian multi-family rental portfolio that management uses to evaluate the performance of this business segment over time and relative to industry peers. Many of the metrics referenced in these tables are KPI measures that are defined in <u>Section 7.1</u> and are key drivers of revenue and ultimately NOI (KPI measure; refer to <u>Section 7.1</u>).

All dollar amounts in this subsection are expressed in Canadian dollars and represent Tricon's share of the operating results unless otherwise indicated. Tricon currently owns a 15% equity interest in The Selby.

| | Q4 2020 | Q3 2020 | Q2 2020 | Q1 2020 |
|--|----------|----------|----------|----------|
| Number of suites | 500 | 500 | 500 | 500 |
| Occupancy | 87.0% | 87.1% | 88.2% | 85.8% |
| Annualized turnover rate | 41.6% | 52.8% | 27.2% | 10.4% |
| Average monthly rent ⁽¹⁾ | \$ 2,648 | \$ 2,664 | \$ 2,675 | \$ 2,666 |
| Average quarterly rent growth – renewals ⁽²⁾ | 1.3% | (0.7%) | 0.8% | 2.2% |
| Average quarterly rent growth – new move-in ⁽²⁾ | (11.3%) | (3.8%) | - | 4.2% |
| Average quarterly rent growth – blended ⁽²⁾ | (5.1%) | (2.0%) | 0.7% | 2.4% |

(1) Average monthly rent represents average monthly rental income per suite for occupied suites and reflects the impact of rent concessions amortized over the life of the related leases.

(2) Average rent growth during the period represents the percentage difference between the monthly rent from an expiring lease and the monthly rent from the next lease and reflects the impact of rent concessions amortized over the life of the related lease. Excluding the impact of concessions, the Q4 2020 rent growth was (0.4%) for renewals, (5.2%) for new move-ins and (2.8%) blended.

The table below presents a breakdown of Tricon's NOI (KPI measure; refer to <u>Section 7.1</u>) for the Canadian multi-family rental business. Comparative period results are not shown as The Selby was in the initial lease-up phase during 2019.

| (in thousands of Canadian dollars, unless otherwise indicated) | three months December 3 | | % of revenue | For the twelve months ended December 31, 2020 | % of revenue |
|---|----------------------------|-------|--------------|---|--------------|
| Rental revenue | \$ | 530 | | \$ 2,121 | |
| Concessions and abatements | | (39) | | (94) | |
| Fees and other revenue ⁽¹⁾ | | 30 | | 124 | |
| Bad debt expense | | (5) | | (33) | |
| Total revenue from rental properties | \$ | 516 | 100% | \$ 2,118 | 100% |
| Property taxes | | 50 | 10% | 203 | 10% |
| Repairs, maintenance and turnover | | 33 | 6% | 116 | 5% |
| Property management expenses | | 52 | 10% | 233 | 11% |
| Utilities | | 33 | 6% | 121 | 6% |
| Property insurance | | 19 | 4% | 73 | 3% |
| Marketing and leasing | | 22 | 4% | 78 | 4% |
| Other property operating expense | | 20 | 4% | 52 | 2% |
| Total direct operating expenses | | 229 | | 876 | |
| Net operating income (NOI) | \$ | 287 | | \$ 1,242 | |
| Net operating income (NOI) margin | | 55.6% | | 58.6% | |
| Net operating income (NOI) ⁽²⁾ | US\$ | 220 | | US\$ 927 | |

(1) Fees and other revenue include commercial rental revenue, ancillary income earned on usage of facilities, parking services and storage usage fees as well as utility recovery from residents.

(2) The weighted average USD/CAD exchange rate used to present the multi-family rental portfolio NOI was 1.3030 and 1.3398 for the three and twelve months ended December 31, 2020, respectively.

NOI for the three months ended December 31, 2020 was C\$0.3 million, remaining relatively in line with the third quarter of 2020 and reflecting an NOI margin of 55.6%. Tricon's share of revenue generated by The Selby was C\$0.5 million, attributable to average monthly rent of C\$2,648 per suite and occupancy of 87.0%, while operating expenses were C\$0.2 million.



4.3 Residential Development

Tricon's Residential Development business segment currently includes (i) new Class A multi-family rental apartments in Canada that are in the development and construction stages, and which Tricon intends to own long term following completion, and (ii) legacy investments in for-sale housing development projects predominantly in the U.S. Once construction is complete and lease-up stabilization occurs, newly built Canadian multi-family rental apartments will transition from the Residential Development segment to Tricon's multi-family rental business segment.

The table below presents the components of Tricon's net assets in residential developments, including Canadian multi-family developments which are classified as either investments or consolidated development properties according to their legal and ownership structure.

| As at (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|---|----------------------|----------------------|
| Investments in Canadian multi-family developments ⁽¹⁾ – See <u>Section 4.3.1</u> | \$ 74,955 | \$ 75,141 |
| Canadian development properties, net of debt ⁽²⁾ – See Section 4.3.1 | 53,161 | 22,021 |
| Investments in for-sale housing – See Section 4.3.2 | 164,842 | 300,653 |
| Net investments in residential developments | 292,958 | 397,815 |
| Other net assets ⁽³⁾ | 10,002 | 3,471 |
| Net assets attributable to Tricon – see <u>Section 5</u> | \$ 302,960 | \$ 401,286 |

(1) Includes Tricon's investment in The Taylor, The Ivy, West Don Lands and 7 Labatt, where Tricon is a co-investor in each project alongside institutional partners. The comparative balance also includes The Selby.

(2) Refers to the net assets of The James (Scrivener Square) and The Shops of Summerhill, which are wholly-owned by Tricon as of June 23, 2020. As of December 31, 2020, the net assets of \$53,161 include development properties of \$110,018 less debt of \$60,018 and other net assets of \$3,161.

(3) Other net assets include deferred income tax assets and other working capital items.



4.3.1 Canadian Class A multi-family developments

Tricon is focused on developing, owning and operating the leading portfolio of Class A rental apartments in the Greater Toronto Area. The Company is one of the most active rental developers in downtown Toronto with eight projects totalling 3,720 units (including select condominium units) under construction or in pre-construction as at December 31, 2020, in addition to 500 units at The Selby, which is now essentially stabilized. The Company's portfolio also includes an existing commercial property, The Shops of Summerhill, adjacent to one of its multi-family development properties.

As at December 31, 2020, the carrying value of Tricon's net assets in its Canadian multi-family development portfolio was \$128.1 million. The following table summarizes the net assets by project.

| | | | Decembe | 20 | | | December 31, 2019 | | | | | | | | | |
|--|------|-------------------------------------|---------|--|-----------------|---|-------------------|---------------------------------------|-------|------------------------------------|-------|--|-----------------|---|----|--------------------------------------|
| (in thousands of U.S. dollars) | | con's share of property value | | Tricon's are of debt and lease obligations ⁽¹⁾ | of net v cap | 's share vorking ital and er items | I | Tricon's net assets ⁽²⁾ | | on's share of property value | i | Tricon's re of debt and lease bligations ⁽¹⁾ | of net v cap | s share vorking ital and r items | n | Tricon's et assets ⁽²⁾ |
| Projects in pre-construction | | | | | | | | | | | | | | | | |
| The James (Scrivener Square) | \$ | 5 73,299 | \$ | (47,555) | \$ | 1,514 | \$ | 27,258 | \$ | 25,170 | \$ | (10,779) | \$ | 289 | \$ | 14,680 |
| 7 Labatt 🤇 | | 24,941 | | (8,814) | | 53 | | 16,180 | | 23,593 | | (8,640) | | 66 | | 15,019 |
| West Don Lands – Blocks 3/4/7 | | 23,639 | | (11,818) | | (2,246) | | 9,575 | | 6,121 | | (5,075) | | 201 | | 1,247 |
| West Don Lands – Block 20 | | 15,232 | | (14,551) | | 256 | | 937 | | 3,117 | | (2,963) | | 129 | | 283 |
| West Don Lands – Block 10 ⁽³⁾ | | 850 | | - | | 2,144 | | 2,994 | | - | | - | | 1,689 | | 1,689 |
| Subtotal – Projects in pre-construction | \$ | 5 137,961 | \$ | (82,738) | \$ | 1,721 | \$ | 56,944 | \$ | 58,001 | \$ | (27,457) | \$ | 2,374 | \$ | 32,918 |
| Projects under construction | | | | | | | | | | | | | | | | |
| The Taylor (57 Spadina) | \$ | 33,972 | \$ | (11,920) | \$ | (664) | \$ | 21,388 | \$ | 27,088 | \$ | (7,297) | \$ | 18 | \$ | 19,809 |
| West Don Lands – Block 8 | | 37,496 | | (29,545) | | (468) | | 7,483 | | 13,047 | | (16,954) | | 5,784 | | 1,877 |
| The Ivy (8 Gloucester) | | 19,175 | | (3,138) | | 361 | | 16,398 | | 15,046 | | _ | | 438 | | 15,484 |
| Subtotal – Projects under | | | | | | | | | | | | | | | | |
| construction | \$ | 90,643 | \$ | (44,603) | \$ | (771) | \$ | 45,269 | \$ | 55,181 | \$ | (24,251) | \$ | 6,240 | \$ | 37,170 |
| Projects in lease-up | | | | | | | | | | | | | | | | |
| The Selby (592 Sherbourne) ⁽⁴⁾ | \$ | ; – | \$ | _ | \$ | _ | \$ | - | \$ | 37,167 | \$ | (17,645) | \$ | 211 | \$ | 19,733 |
| Subtotal – Projects in lease-up | \$ | ; – | \$ | - | \$ | - | \$ | - | \$ | 37,167 | \$ | (17,645) | \$ | 211 | \$ | 19,733 |
| Stabilized projects | | | | | | | | | | | | | | | | |
| The Shops of Summerhill | \$ | 36,719 | \$ | (12,463) | \$ | 1,647 | \$ | 25,903 | \$ | 10,455 | \$ | (3,149) | \$ | 35 | \$ | 7,341 |
| Subtotal – Stabilized projects | \$ | 36,719 | \$ | (12,463) | \$ | 1,647 | \$ | 25,903 | \$ | 10,455 | \$ | (3,149) | \$ | 35 | \$ | 7,341 |
| Total | \$ | 5 265,323 | \$ | (139,804) | \$ | 2,597 | \$ | 128,116 | \$ | 160,804 | \$ | (72,502) | \$ | 8,860 | \$ | 97,162 |
| Investments in Canadian | •••• | ••••• | ••••• | | | ••••• | ••••• | •••••• | ••••• | | ••••• | | | •••••• | | |
| multi-family developments | \$ | 5 155,305 | \$ | (79,786) | \$ | (564) | \$ | 74,955 | \$ | 125,179 | \$ | (58,574) | \$ | 8,536 | \$ | 75,141 |
| Canadian development | | | | , | | , / | | | · | | | / | | | | · |
| properties, net of debt ⁽⁵⁾ | | 110,018 | | (60,018) | | 3,161 | | 53,161 | | 35,625 | | (13,928) | | 324 | | 22,021 |
| Total | \$ | 6 265,323 | \$ | (139,804) | \$ | 2,597 | \$ | 128,116 | \$ | 160,804 | \$ | (72,502) | \$ | 8,860 | \$ | 97,162 |

 Tricon's share of debt and lease obligations includes land and construction loans (net of deferred financing fees), vendor take-back loans and lease obligations under ground leases.

(2) Represents Tricon's share of development properties and other working capital items, net of debt and lease obligations.

(3) Tricon's share of net assets of DKT B10 LP includes the purchase price paid to third-party partners for a one-third ownership interest in the partnership.

(4) The Selby was reclassified from property under development to income-producing property during the first quarter of 2020, and therefore removed from the Residential Development segment disclosure.

(5) On June 23, 2020, Tricon acquired the remaining 50% and 75% of The James and The Shops of Summerhill, respectively, for cash of \$7,643 and recognized an additional \$65,861 of property value, \$53,340 of debt and net working deficit of \$4,878.



Project details and projections

The tables in this subsection provide a summary of certain details and projections for Canadian Class A multi-family development projects that management uses to evaluate the ongoing performance of these projects over time and relative to industry peers. The Canadian multi-family development segment targets a development yield spread (net operating income/project cost) of approximately 100 basis points over the yield available on core assets, and is expected to deliver a 4.75% yield at the stabilization of the portfolio. Projected units, rentable area, costs and timelines are estimated based on current project plans which are subject to change. Refer to page 24, "Non-IFRS measures and forward-looking statements".

As at December 31, 2020, the Canadian multi-family development portfolio consisted of 3,720 projected rental and condominium units. The current status of these units is presented below:

| | Projected rental and condominium units | | | | |
|------------------|--|-------------------|--|--|--|
| As at | December 31, 2020 | December 31, 2019 | | | |
| Pre-construction | 2,433 | 2,541 | | | |
| Construction | 1,287 | 286 | | | |
| Lease-up | - | 500 | | | |
| Total | 3,720 | 3,327 | | | |

| | Neighbourhood/ Major intersections in Toronto | Fee simple interest/ ground lease | Tricon's percentage interest | Projected o units ⁽¹⁾ | Estimated commercial area (sq. feet) |
|---|---|--------------------------------------|------------------------------------|-------------------------------------|--|
| Projects in pre-construction | | | | | |
| The James (Scrivener Square) | Rosedale | Fee simple interest | 100% | 120 | 31,000 |
| 7 Labatt | Downtown East – | | | | |
| | Corktown | Fee simple interest | 30% | 558 | 51,000 |
| West Don Lands – Blocks 3/4/7 | Downtown East – | | | | |
| | Distillery District | Ground lease | 33% | 855 | 39,000 |
| West Don Lands – Block 20 | Downtown East – | | | | |
| | Distillery District | Ground lease | 33% | 661 | 250,000 |
| West Don Lands – Block 10 | Downtown East – | | | | |
| | Distillery District | Ground lease ⁽²⁾ | 33% | 239 | TBD |
| Subtotal – Projects in pre-construction | | | | 2,433 | 371,000 |
| Projects under construction | | | | | |
| The Taylor (57 Spadina) | Entertainment District | Fee simple interest | 30% | 286 | 44,000 |
| West Don Lands – Block 8 | Downtown East – | | | | |
| | Distillery District | Ground lease | 33% | 770 | 4,000 |
| The Ivy (8 Gloucester) | Yonge & Bloor | Fee simple interest | 47% | 231 | 1,600 |
| Subtotal – Projects under construction | | | | 1,287 | 49,600 |
| Total | | | | 3,720 | 420,600 |

(1) Includes 3,419 projected rental units and 301 projected condominium units.

(2) The ground lease for West Don Lands – Block 10 is under contract and will be in force upon the severance of the leased premises from a broader land parcel.

MANAGEMENT'S DISCUSSION AND ANALYSIS

for the year ended December 31, 2020

4.3 Residential Development

| (in thousands of U.S. dollars) | Cost to date | Projected remaining costs | Projected total cost | Percentage completed ⁽¹⁾ | Tricon's unfunded equity commitment |
|---|--------------|---------------------------|-------------------------|--|---|
| Projects in pre-construction | | | | | |
| The James (Scrivener Square) ⁽²⁾ | \$ 70,000 | \$ 195,000 | \$ 265,000 | 8% | \$ 31,653 |
| 7 Labatt | 63,000 | 223,000 | 286,000 | 3% | 8,088 |
| West Don Lands – Blocks 3/4/7 | 6,000 | 395,000 | 401,000 | 1% | 13,275 |
| West Don Lands – Block 20 | 2,000 | 371,000 | 373,000 | _ | 704 |
| West Don Lands – Block 10 | 3,000 | 91,000 | 94,000 | 3% | 7,058 |
| Subtotal – Projects in pre-construction | 144,000 | 1,275,000 | 1,419,000 | | 60,778 |
| Projects under construction | | | | | |
| The Taylor (57 Spadina) | 73,000 | 55,000 | 128,000 | 46% | - |
| West Don Lands – Block 8 ⁽³⁾ | 64,000 | 217,000 | 281,000 | 22% | - |
| The Ivy (8 Gloucester) | 34,000 | 79,000 | 113,000 | 10% | - |
| Subtotal – Projects under construction | 171,000 | 351,000 | 522,000 | | - |
| Total | \$ 315,000 | \$ 1,626,000 | \$ 1,941,000 | | \$ 60,778 |

(1) Percentage completed is calculated by taking cost to date as a percentage of projected total cost, excluding the cost of land.

Tricon's unfunded equity commitment for The James excludes \$25,564 of committed capital set aside to repay the vendor take-back (VTB) loan due in 2021.
 West Don Lands – Block 8 has a construction loan facility of \$280,000 and therefore Tricon does not expect to fund its remaining equity commitment of \$16,873.

The projected timelines for construction and lease-up of Tricon's Canadian multi-family development projects are presented below (see "Non-IFRS measures and forward-looking statements" on page 24).



Construction period Initial occupancy

Performance overview – projects in pre-construction

At **The James**, the project received the final form zoning by-law and demolition permits and erected site hoarding during the quarter. Subsequent to year-end, Tricon commenced demolition of the existing office building, and shoring and excavation are expected to start in the second quarter of 2021.

The West Don Lands master-plan consists of four projects – Block 8 (currently under construction), Blocks 3/4/7, Block 20, and Block 10. During the quarter, Blocks 3/4/7 received a Ministerial Zoning Order to accelerate entitlements. Tricon and its joint venture partners have received a commitment for a construction loan backed by the Canada Mortgage and Housing Corporation, and the partnership expects to satisfy all outstanding funding conditions and start drawing on the construction loan in mid-2021. Block 20 also received a Ministerial Zoning Order to accelerate entitlements, and is in preparation for site plan submission. At Block 10, formal registration of the distinct rental development land parcel is expected in the first quarter of 2021. The project is fully zoned and will include a multi-family rental building with 239 rental units, along with Toronto's first purpose-built Indigenous hub. Once part lot control is completed, the joint venture partners together will control over eleven acres of prime land within the West Don Lands, one of the largest and most significant rental communities in Canada that involves all three levels of government.

At **7** Labatt, the project continues to progress with the site plan approval process, which includes refinement of the architectural design. Given the current COVID-19 environment and restrictions in place, the scheduled condominium sales launch has been pushed back, which in turn has delayed the scheduled construction commencement to the second half of 2021.



Performance overview – projects under construction

As a result of the Province of Ontario deeming residential construction an essential service, the COVID-19 pandemic has only had a nominal impact on the development timelines of Tricon's projects that are already under construction.

At **The Taylor**, forming has progressed to the 15th floor, installation of the building's precast masonry façade progressed through the 9th floor, and window installation commenced during the quarter. The project is expected to "top off" in the coming months, with first resident move-ins slated for early 2022 and construction completion in mid-2022.

At **Block 8** of the West Don Lands, forming of the below-grade structure is now complete and forming of the above-grade structure projected to be completed in early 2022. Installation of the precast brick façade is expected to commence in mid-2021.

At **The Ivy**, construction progressed as planned during the quarter. Subsequent to year-end, a crane was erected following the completion of excavation. The project is expected to close on a construction loan in the first quarter of 2021.

4.3.2 Investments in for-sale housing

The Company's legacy for-sale housing business provides equity or equity-type financing to local and regional developers and homebuilders for housing development, primarily in the U.S. The investments are typically made through Investment Vehicles which hold an interest in for-sale residential land, homebuilding and condominium development projects.

INVESTMENTS IN FOR-SALE HOUSING BY LOCATION



Investment performance of for-sale housing

As part of its strategic shift towards becoming a rental housing company, Tricon intends to decrease its balance sheet investments in its for-sale housing business over time through natural liquidation, and where possible, the strategic disposition of assets. Tricon's for-sale housing investments are carried at \$164.8 million, representing 2.3% of Tricon's total balance sheet assets as at December 31, 2020.

While the Company has decreased its balance sheet exposure, investments in for-sale housing continue to be a significant source of cash generation for the Company and distributed \$14.4 million to Tricon during the quarter, including \$1.7 million in performance fees (see Sections 3.2 and 4.4).

In the first quarter of 2020, Tricon recognized a significant write-down of its investments in the for-sale housing business under the context of a precipitous drop in sales in late March and uncertainty over the timing of future cash flows brought on by the pandemic. This resulted in a loss to Tricon of \$61.8 million for the year ended December 31, 2020. However, since the onset of the pandemic, sales in several projects have recovered to pre-COVID-19 levels, as historically low mortgage rates, coupled with strong de-urbanization trends and work-from-home mandates, have encouraged home buying in the suburbs. As a result of this acceleration in housing demand, the Company was able to recover some of its previously recognized losses in the fourth quarter. While the for-sale housing market outlook for 2021 appears favourable, management is also mindful of rising labour and material costs, which could partially offset rising home prices and high absorption rates.



Project details and projections

The table below presents Tricon's share of key financial metrics and projections in its for-sale housing investments.

| | | Distributions | | Projected distributions net |
|--------------------------------|------------------|---|--------------------------------------|---|
| (in thousands of U.S. dollars) | Advances to date | Distributions to date ⁽¹⁾ | Tricon's fair value of investment | of advances remaining ⁽²⁾ |
| For-sale housing investments | \$ 520,066 | \$ 451,986 | \$ 164,842 | \$ 322,580 |

(1) Distributions include repayments of preferred return and capital.

(2) Projected distributions are based on current project plans which are subject to change. Refer to page 24, "Non-IFRS measures and forward-looking statements".

For-sale housing investments are structured as self-liquidating investments generally with cash flows generated as land, lots or homes are sold to third-party buyers (typically large homebuilders or commercial developers in the case of land and end consumers for homebuilding). For-sale housing investments now represent 2.3% of total assets but are still expected to generate approximately \$322.6 million of net cash flow to Tricon over the next ten years.

The scheduled time frame for Tricon to receive the projected net distributions remaining is as follows:

| (in thousands of U.S. dollars) | 1 to 2 years | 3 to 5 years | More than 5 years | Total |
|---|--------------|--------------|----------------------|------------|
| Projected distributions net of advances remaining | \$ 83,080 | \$ 120,587 | \$ 118,913 | \$ 322,580 |



4.4 Private Funds and Advisory

Through its private funds and advisory ("PF&A") business, Tricon earns fees from managing third-party capital co-invested in its real estate assets. Activities of this segment include providing asset management, property management and development management services. In aggregate, Tricon manages \$2.6 billion of third-party capital across its business segments and intends to continue raising and managing third-party capital to generate scale and drive operational synergies, diversify its investor base, capitalize on opportunities that would otherwise be too large for the Company, reduce its balance sheet exposure to development activities, and enhance Tricon's return on equity by earning asset management and other fees.

During the quarter, the Company made significant progress in negotiations with institutional investors to syndicate an 80% interest in its U.S. multi-family rental portfolio to a new Investment Vehicle. Subsequent to year-end, the Company announced that this syndication is expected to close in March of 2021 (see Section 3.3). The creation of this new Investment Vehicle will act as a catalyst to further enhance the Company's private funds and advisory revenue and will provide greater flexibility over capital reallocation. The Company intends to use the net syndication proceeds to reduce debt as part of its deleveraging plan.

Looking forward, the Company intends to launch several new Investment Vehicles in 2021. They include but are not limited to: (i) forming a separate growth-oriented joint venture to acquire additional multi-family properties in the U.S. Sun Belt; (ii) creating a second single-family rental joint venture ("SFR JV-2"), a successor joint venture to SFR JV-1, which is on track to be fully invested by mid-2021; (iii) organizing an additional Investment Vehicle to purchase single-family rental homes directly from homebuilders; and (iv) establishing a joint venture focused on developing and owning Class A rental apartments in the Greater Toronto Area. With third-party equity capital commitments of over \$1.2 billion estimated to close in 2021, Tricon is projected to earn an incremental \$10 million of annualized asset management fees from these Investment Vehicles in the future.

Performance overview

The following table provides details of revenue from private funds and advisory services for the three and twelve months ended December 31, 2020.

| For the periods ended December 31 | Three months | | | | | | Twelve months | | | | |
|---|--------------|-------|----|--------|----|---------|---------------|--------|---------|-----|------------|
| (in thousands of U.S. dollars) | | 2020 | | 2019 | V | ariance | | 2020 | 2 | 019 | Variance |
| Asset management fees ⁽¹⁾ | \$ | 2,815 | \$ | 3,355 | \$ | (540) | \$ | 12,061 | \$ 15,0 | 99 | \$ (3,038) |
| Performance fees ⁽²⁾ | | 1,691 | | 2,565 | | (874) | | 2,836 | 7,4 | 48 | (4,612) |
| Development fees ⁽³⁾ | | 5,653 | | 5,876 | | (223) | | 18,298 | 17,7 | 36 | 562 |
| Property management fees ⁽⁴⁾ | | 180 | | 342 | | (162) | | 895 | 7 | 77 | 118 |
| Revenue from private funds | | | | | | | | | | | |
| and advisory services | \$ 1 | 0,339 | \$ | 12,138 | \$ | (1,799) | \$ | 34,090 | \$ 41,0 | 60 | \$ (6,970) |

(1) Ranges typically from 1–2% of committed or invested capital throughout the lives of the Investment Vehicles under management.

(2) Calculated as approximately 20% (in most cases) of net cash flow after investors' capital has been returned, together with a pre-tax preferred return on capital of, typically, between 8% and 10%.

(3) Calculated as 2–5% of the sales price of single-family lots, residential land parcels and commercial land within master-planned communities, and 4–5% of overall development costs of Canadian multi-family rental apartments.

(4) Includes property management fees of 4% of rental revenue from Canadian multi-family rental properties and other ancillary fees.

Revenue from private funds and advisory services for the three and twelve months ended December 31, 2020 decreased by \$1.8 million and \$7.0 million, respectively, compared to the same periods in the prior year. Refer to the variance commentary in Section 3.1 for more details.

4.4 Private Funds and Advisory



(46,115)

\$ 41,060

(46,115)

777

\$

The Company also earns significant fees from managing the single-family rental homes and Canadian multi-family developments held in controlled subsidiaries, which are eliminated upon consolidation. The tables below provide an overview of the gross fees earned, followed by consolidation eliminations to arrive at the net fees earned in the three and twelve months ended December 31, 2020 as well as the comparative periods.

| (in thousands of U.S. dollars) | Asset management fees | Performance fees | Development fees | Property management fees ⁽¹⁾ | Total |
|--|---|---|--|---|--|
| For the three months ended | | | | | |
| December 31, 2020 | | | | | |
| Gross management fees | \$ 2,815 | \$ 1,691 | \$ 6,027 | \$ 12,597 | \$ 23,130 |
| Less fees eliminated upon consolidation: | | | | | |
| Development fees eliminated | - | - | (374) | - | (374) |
| Property management | | | | | |
| fees eliminated | - | - | - | (12,417) | (12,417) |
| Total revenue from private funds | | | | | |
| and advisory services | \$ 2,815 | \$ 1,691 | \$ 5,653 | \$ 180 | \$ 10,339 |
| For the three months ended | | | | | |
| December 31, 2019 | | | | | |
| Gross management fees | \$ 3,355 | \$ 2,565 | \$ 5,876 | \$ 13,692 | \$ 25,488 |
| Less fees eliminated upon consolidation: | | | | | |
| Property management | | | | | |
| fees eliminated | - | _ | - | (13,350) | (13,350) |
| Total revenue from private funds | | | | | |
| and advisory services | \$ 3,355 | \$ 2,565 | \$ 5,876 | \$ 342 | \$ 12,138 |
| | | + _, | | \$ 5.2 | ÷, |
| | | + _, | | | +, |
| | Asset | | | Property | • • • • • • • • • • • • • • • • • • • |
| (in thousands of LLS, dollars) | management | Performance | Development | Property management | |
| (in thousands of U.S. dollars) | | | | Property | Total |
| For the twelve months ended | management | Performance | Development | Property management | |
| For the twelve months ended December 31, 2020 | management fees | Performance fees | Development fees | Property management fees ⁽¹⁾ | Total |
| For the twelve months ended December 31, 2020 Gross management fees | management | Performance | Development | Property management | |
| For the twelve months ended December 31, 2020 Gross management fees Less fees eliminated upon consolidation: | management fees | Performance fees | Development fees \$ 19,038 | Property management fees ⁽¹⁾ | Total \$ 79,399 |
| For the twelve months ended December 31, 2020 Gross management fees Less fees eliminated upon consolidation: Development fees eliminated | management fees | Performance fees | Development fees | Property management fees ⁽¹⁾ | Total |
| For the twelve months ended December 31, 2020 Gross management fees Less fees eliminated upon consolidation: Development fees eliminated Property management | management fees | Performance fees | Development fees \$ 19,038 | Property management fees ⁽¹⁾ \$ 45,464 – | Total \$ 79,399 (740) |
| For the twelve months ended December 31, 2020 Gross management fees Less fees eliminated upon consolidation: Development fees eliminated Property management fees eliminated | management fees | Performance fees | Development fees \$ 19,038 | Property management fees ⁽¹⁾ | Total \$ 79,399 |
| For the twelve months ended December 31, 2020 Gross management fees Less fees eliminated upon consolidation: Development fees eliminated Property management | management fees | Performance fees | Development fees \$ 19,038 | Property management fees ⁽¹⁾ \$ 45,464 – | Total \$ 79,399 (740) |
| For the twelve months ended December 31, 2020 Gross management fees Less fees eliminated upon consolidation: Development fees eliminated Property management fees eliminated Total revenue from private funds and advisory services | management fees \$ 12,061 _ _ | Performance fees \$ 2,836 - - | Development fees \$ 19,038 (740) _ | Property management fees ⁽¹⁾ \$ 45,464 – (44,569) | Total \$ 79,399 (740) (44,569) |
| For the twelve months ended December 31, 2020 Gross management fees Less fees eliminated upon consolidation: Development fees eliminated Property management fees eliminated Total revenue from private funds and advisory services For the twelve months ended | management fees \$ 12,061 _ _ | Performance fees \$ 2,836 - - | Development fees \$ 19,038 (740) _ | Property management fees ⁽¹⁾ \$ 45,464 – (44,569) | Total \$ 79,399 (740) (44,569) |
| For the twelve months ended December 31, 2020 Gross management fees Less fees eliminated upon consolidation: Development fees eliminated Property management fees eliminated Total revenue from private funds and advisory services | management fees \$ 12,061 _ _ | Performance fees \$ 2,836 - - | Development fees \$ 19,038 (740) _ | Property management fees ⁽¹⁾ \$ 45,464 – (44,569) | Total \$ 79,399 (740) (44,569) |

_

\$ 17,736

\$ 7,448

(1) Property management fees also include leasing, acquisition and construction management fee revenue.

_

\$ 15,099

Less fees eliminated upon consolidation:

Property management fees eliminated

Total revenue from private funds and advisory services



Future performance fees

The table below provides a summary of projected performance fees by business that Tricon could earn over time based on current business plans (forward-looking information; see page 24). Projected performance fees are based on Tricon's analysis of projected cash flows over the expected life of existing projects and Investment Vehicles in each business. Projected cash flows are determined based on detailed quarterly and/or annual budgets prepared by management or third-party developers or in certain cases based on third-party appraisals performed in the current quarter.

| | 41.2 | | More than | Tatal | |
|---|--------------|--------------|-----------|------------|--|
| (in thousands of U.S. dollars) | 1 to 2 years | 3 to 5 years | 5 years | Total | |
| Single-family rental | \$ - | \$ - | \$ 60,000 | \$ 60,000 | |
| Canadian multi-family rental developments | - | - | 18,891 | 18,891 | |
| For-sale housing | 5,224 | 9,786 | 6,760 | 21,770 | |
| Estimated future performance fees | \$ 5,224 | \$ 9,786 | \$ 85,651 | \$ 100,661 | |

Assets under management

Third-party AUM increased by \$118.7 million or 5% to \$2.6 billion as at December 31, 2020, from \$2.4 billion as at December 31, 2019. Refer to Section 7.2 for a definition of AUM.



The primary changes in third-party AUM since December 31, 2019 were:

- An increase of \$279.2 million in single-family rental AUM, driven primarily by a new securitization transaction used to finance a growing portfolio of single-family rental homes in SFR JV-1. This transaction increased the outstanding property-level debt, which is a component of single-family rental AUM (see Section 7.2).
- An increase of \$150.7 million in multi-family rental AUM, as The Selby approached stabilization and thus was reclassified from the residential development business segment to the multi-family rental business segment.
- An offsetting decrease of \$147.9 million in Canadian multi-family rental development AUM, primarily driven by the reclassification of The Selby discussed above, along with the reclassification of The James and The Shops of Summerhill, given the Company has acquired its former partners' interests in those properties. These decreases were partially offset by additional funded debt as construction progressed at various projects in the portfolio.
- An additional decrease of \$163.3 million in for-sale housing AUM, primarily attributable to \$102.5 million of distributions from commingled funds and separate accounts to third-party investors, and the disposition by Tricon and its partners of the separate account investment at Fulshear Farms (Houston, Texas), among other items.



The following table provides a further breakdown of the components of third-party AUM.

| (in thousands of U.S. dollars) | December 31, 2020 ⁽¹⁾ | September 30, 2020 ⁽¹⁾ | June 30, 2020 ⁽¹ | March 31,) 2020 ⁽¹ | December 31, 2019 ⁽¹⁾ | September 30, 2019 ⁽¹⁾ | June 30, 2019 ⁽¹⁾ | March 31, 2019 ⁽¹⁾ |
|--------------------------------|-------------------------------------|--------------------------------------|--------------------------------|-----------------------------------|-------------------------------------|--------------------------------------|---------------------------------|----------------------------------|
| Single-family rental | \$ 1,137,936 | \$ 1,042,386 | \$ 933,947 | \$ 935,134 | \$ 858,723 | \$ 738,717 | \$ 673,754 | \$ 609,957 |
| Multi-family rental | 150,659 | 134,527 | 132,666 | 127,780 | - | - | - | - |
| Residential development | | | | | | | | |
| Canadian multi-family rental | | | | | | | | |
| developments | 231,945 | 208,933 | 226,812 | 242,244 | 379,812 | 369,078 | 364,062 | 345,576 |
| For-sale housing | 1,032,818 | 1,089,535 | 1,100,417 | 1,175,016 | 1,196,075 | 1,224,623 | 804,686 | 836,330 |
| Residential development | 1,264,763 | 1,298,468 | 1,327,229 | 1,417,260 | 1,575,887 | 1,593,701 | 1,168,748 | 1,181,906 |
| Third-party AUM | \$ 2,553,358 | \$ 2,475,381 | \$ 2,393,842 | \$ 2,480,174 | \$ 2,434,610 | \$ 2,332,418 | \$ 1,842,502 | \$ 1,791,863 |

USD/CAD exchange rates used at each balance sheet date are: at December 31, 2020: 1.2732; September 30, 2020: 1.3339; June 30, 2020: 1.3628; March 31, 2020: 1.4187; December 31, 2019: 1.2988; September 30, 2019: 1.3243; June 30, 2019: 1.3087; March 31, 2019: 1.3363.

The table below provides a reconciliation, by business, of the outstanding third-party capital investment balances to AUM (KPI measure; refer to Section 7.2).

| (in thousands of U.S. dollars) | Outstanding invested capital (at cost) | Share of outstanding project debt | Unfunded equity commitment ⁽¹⁾ | Third-party AUM as at December 31, 2020 |
|--|--|---|--|---|
| Single-family rental ⁽²⁾ | \$ 341,471 | \$ 729,266 | \$ 67,199 | \$ 1,137,936 |
| Multi-family rental ⁽³⁾ | 40,368 | 110,291 | - | 150,659 |
| Canadian multi-family rental developments ⁽⁴⁾ | 75,427 | 95,069 | 61,449 | 231,945 |
| For-sale housing ⁽⁵⁾ | 588,813 | - | 444,005 | 1,032,818 |
| Total | \$ 1,046,079 | \$ 934,626 | \$ 572,653 | \$ 2,553,358 |

 Commitments to projects include guarantees made under loan agreements plus reserves. Project commitments can exceed total capitalization as a result of reinvestment rights.

(2) Single-family rental includes SFR JV-1. Limited partners' share of the outstanding debt includes their share of the JV-1 warehouse credit facility, JV-1 securitization debt 2019-1, JV-1 securitization debt 2020-1, and the JV-1 subscription facility, the latter of which is a substitute for invested capital and can be replaced by equity funding at management's discretion. As at December 31, 2020, Tricon's share of outstanding invested capital and its unfunded equity commitment for SFR JV-1 were \$151,590 and \$77,185, respectively.

(3) Multi-family rental includes The Selby commencing in Q1 2020, as construction was substantially completed.

(4) Canadian multi-family rental developments include The Taylor, West Don Lands, The Ivy and 7 Labatt. Comparative periods also include The Selby, which was reclassified to income-producing multi-family rental properties in Q1 2020, and The James (Scrivener Square) and The Shops of Summerhill, which are wholly-owned by Tricon effective June 23, 2020. Other than in respect of The Selby, The Taylor and 7 Labatt, Tricon has partnered with strategic partners that do not pay asset management or performance fees to the Company for management of their invested capital (but for clarity do pay development and property management fees). Refer to the AIF for a description of these Investment Vehicles.

(5) For-sale housing includes THP1 US, THP2 US, THP2 Canada, THP3 Canada, Mahogany, Cross Creek Ranch, Grand Central Park, Trilogy at Verde River, Viridian, THP US SP1, THP US SP2, Trilogy at Vistancia West, Trilogy Lake Norman, Arantine Hills and THPAS JV-1 (including single-family rental build-to-rent communities). Refer to the AIF for a description of these Investment Vehicles.





SUMMARY OF NON-IFRS SEGMENT INFORMATION

5. SUMMARY OF NON-IFRS SEGMENT INFORMATION

Management considers Core FFO and AFFO to be key measures of the Company's operating performance (refer to Section 7.1 for KPI definitions). These are metrics commonly used by securities analysts, investors and other interested parties in the evaluation of real estate entities, particularly those that own and operate income-producing properties. Management believes that providing these performance measures on a supplemental basis is helpful to investors in assessing the overall performance of the Company's business. Refer to the discussion of non-IFRS measures on page 24, including FFO, Core FFO and AFFO.

The discussion and presentation of non-IFRS measures throughout this section reflect Tricon's proportionate share of the business, unless otherwise stated.

The following table reconciles FFO, Core FFO and AFFO to the net income reflected in the Company's income statement for the three and twelve months ended December 31, 2020. Comparative periods have been reclassified to conform with the current period presentation.

| For the periods ended December 31 | | Three months | | | Twelve months | |
|---|-----------|--------------|-------------|------------|---------------|-----------|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance |
| Net income attributable | | | | | | |
| to Tricon's shareholders | \$ 79,678 | \$ 42,354 | \$ 37,324 | \$ 113,322 | \$ 107,762 | \$ 5,560 |
| Fair value gain on rental properties ⁽¹⁾ | (106,995) | (32,025) | (74,970) | (198,314) | (116,548) | (81,766) |
| Loss from investments in for-sale housing ⁽²⁾ | _ | _ | _ | 79,579 | - | 79,579 |
| Fair value loss (gain) on derivative financial | | | | | | |
| instruments and other liabilities ⁽¹⁾ | 16,418 | 1,462 | 14,956 | 7,461 | (2,357) | 9,818 |
| Other adjustments ⁽³⁾ | 12,204 | 3,029 | 9,175 | 30,388 | 5,585 | 24,803 |
| FFO attributable to Tricon's shareholders | \$ 1,305 | \$ 14,820 | \$ (13,515) | \$ 32,436 | \$ (5,558) | \$ 37,994 |
| Other income ⁽¹⁾ | (1,774) | _ | (1,774) | (1,774) | - | (1,774) |
| Transaction costs ⁽¹⁾ | 2,491 | 6,532 | (4,041) | 14,016 | 36,415 | (22,399) |
| Deferred tax expense ⁽¹⁾ | 33,133 | 3,228 | 29,905 | 40,425 | 11,934 | 28,491 |
| Amortization and depreciation expense ⁽¹⁾ | 2,614 | 2,733 | (119) | 10,848 | 10,543 | 305 |
| Foreign exchange (gain) loss ⁽¹⁾ | (948) | (178) | (770) | 166 | (42) | 208 |
| Interest incurred on convertible debentures ⁽⁴⁾ | 2,506 | 2,492 | 14 | 9,927 | 9,902 | 25 |
| Interest on Due to Affiliate ⁽⁴⁾ | 4,312 | _ | 4,312 | 5,654 | - | 5,654 |
| Amortization of deferred financing costs, | | | | | | |
| discounts and lease obligations ⁽⁴⁾ | 3,730 | 2,055 | 1,675 | 10,922 | 7,081 | 3,841 |
| Gain on sale of U.S. multi-family developments ⁽¹⁾ | - | (1,113) | 1,113 | - | (9,718) | 9,718 |
| Non-cash compensation ⁽⁵⁾ | 702 | 704 | (2) | 4,979 | 3,979 | 1,000 |
| Non-recurring compensation | - | 27 | (27) | 107 | 1,184 | (1,077) |
| Other adjustments ⁽⁶⁾ | (7,452) | (9,060) | 1,608 | (15,559) | (9,072) | (6,487) |
| Limited partners' share of | | | | | | |
| Core FFO adjustments ⁽⁷⁾ | (709) | (492) | (217) | (2,563) | (1,637) | (926) |
| Core FFO attributable to Tricon's shareholders | \$ 39,910 | \$ 21,748 | \$ 18,162 | \$ 109,584 | \$ 55,011 | \$ 54,573 |
| Recurring capital expenditures | (7,445) | (5,825) | (1,620) | (27,875) | (26,623) | (1,252) |
| AFFO attributable to Tricon's shareholders | \$ 32,465 | \$ 15,923 | \$ 16,542 | \$ 81,709 | \$ 28,388 | \$ 53,321 |

(1) Refer to consolidated income statement in Section 3.1.

(2) Relates to a one-time unrealized fair value loss taken on the for-sale housing assets in Q1 2020.

(3) Relates to limited partners' share of FFO adjustments for fair value gains/(losses).

(4) Refer to the breakdown of interest expense in Section 3.1.

(5) Comprised of equity-settled AIP and LTIP expense, which is presented in Section 3.1.

(6) Comprised of amortization, unrealized foreign exchange and deferred taxes within income from equity-accounted investments and investments held at FVTPL, non-controlling interests' share of amortization and depreciation and other income from government assistance, other non-recurring expenses and lease payments related to the Company's right-of-use assets. Fair value gains from investments in Canadian multi-family developments are also included as eliminations.

(7) Comprised of limited partners' share of transaction costs and amortization of deferred financing fees.

The following table provides a breakdown of Core FFO by business segment, AFFO, Core FFO per share and AFFO per share. Core FFO and AFFO per share amounts are calculated based on the weighted average common shares outstanding in the period, assuming the conversion of all potentially dilutive shares (including convertible debt and exchangeable preferred units of Tricon PIPE LLC).

| For the periods ended December 31 | | | | | | | | | | | |
|--|------------|---------|----------|-------------|---------|-----|----------|-------|-----------|------|----------|
| (in thousands of U.S. dollars, except | | Three | months | | | | | Twelv | ve months | | |
| per share amounts which are in U.S. dollars) | 202 | 0 | 2019 | V | ariance | | 2020 | | 2019 | , | Variance |
| Single-family rental Core FFO ⁽¹⁾ | \$ 28,67 | 8 \$ | 23,031 | \$ | 5,647 | \$ | 110,685 | \$ | 84,663 | \$ | 26,022 |
| Multi-family rental Core FFO | 7,19 | 6 | 7,056 | | 140 | | 27,988 | | 15,470 | | 12,518 |
| Residential development Core FFO ⁽²⁾ | 11,53 | 2 | 3,076 | | 8,456 | | 18,913 | | 8,240 | | 10,673 |
| Private funds and advisory Core FFO ⁽¹⁾ | 5,81 | 5 | 8,154 | | (2,339) | | 20,813 | | 28,141 | | (7,328) |
| | \$ 53,22 | 1 \$ | 41,317 | \$ 1 | 11,904 | \$ | 178,399 | \$ | 136,514 | \$ | 41,885 |
| Corporate overhead | (19,62 | 7) | (15,586) | | (4,041) | | (60,295) | | (62,230) | | 1,935 |
| Corporate interest expense 🛛 🕒 | (1,01 | 8) | (5,297) | | 4,279 | | (13,032) | | (18,173) | | 5,141 |
| Current income tax recovery C | 7,33 | 4 | 1,314 | | 6,020 | | 4,512 | | (1,100) | | 5,612 |
| Core funds from | | | | | | | | | | | |
| operations (Core FFO) | \$ 39,91 | 0\$ | 21,748 | \$ 1 | 18,162 | \$ | 109,584 | \$ | 55,011 | \$ | 54,573 |
| Recurring capital expenditures | (7,44 | 5) | (5,825) | | (1,620) | | (27,875) | | (26,623) | | (1,252) |
| Adjusted funds from | | | | | | | | | | | |
| operations (AFFO) | \$ 32,46 | 5 \$ | 15,923 | \$ 1 | 16,542 | \$ | 81,709 | \$ | 28,388 | \$ | 53,321 |
| Core FFO per share | \$ 0.1 | 6 \$ | 0.10 | \$ | 0.06 | \$ | 0.49 | \$ | 0.29 | \$ | 0.20 |
| Core FFO per share (CAD) ⁽³⁾ | \$ 0.2 | 1 \$ | 0.13 | \$ | 0.08 | \$ | 0.66 | \$ | 0.38 | \$ | 0.28 |
| AFFO per share | \$ 0.1 | 3 \$ | 0.07 | \$ | 0.06 | \$ | 0.36 | \$ | 0.15 | \$ | 0.21 |
| AFFO per share (CAD) ⁽³⁾ | \$ 0.1 | 7 \$ | 0.09 | \$ | 0.08 | \$ | 0.48 | \$ | 0.20 | \$ | 0.28 |
| Core FFO payout ratio ⁽⁴⁾ | 27 | % | 48% | | (21%) | | 37% | | 70% | | (33%) |
| AFFO payout ratio ⁽⁴⁾ | 33 | % | 66% | | (33%) | | 49% | | 136% | | (87%) |
| Weighted average shares | | | | | | | | | | | |
| outstanding – diluted | 248,247,01 | 8 213,6 | 82,237 | 34,56 | 54,781 | 224 | ,015,498 | 191, | 081,128 | 32,9 | 34,370 |

(1) Certain fees earned from limited partners totalling \$1,489 in the first quarter of 2020 (Q1 2019 – \$1,295) have been reclassified from single-family rental Core FFO to private funds and advisory Core FFO to conform with the current period presentation. This change in classification did not result in any changes to total Core FFO and AFFO.

(2) Certain fair value gains recognized on equity-accounted investments totalling \$5,099 in the first quarter of 2020 have been removed from residential development Core FFO to conform with the current period presentation. This change resulted in a \$5,099 decrease to Core FFO for the year ended December 31, 2020.

(3) USD/CAD exchange rates used are 1.3030 and 1.3415 for the three and twelve months ended December 31, 2020, respectively. For the three and twelve months ended December 31, 2019, USD/CAD exchange rates used are 1.3200 and 1.3269, respectively.

(4) Core FFO and AFFO payout ratios are computed by dividing dividends declared for the period by Core FFO and AFFO, respectively.

For the three months ended December 31, 2020, Core FFO increased by \$18.2 million to \$39.9 million compared to \$21.7 million in the same period of the prior year. The variance is explained by:

- An increase in single-family rental Core FFO of \$5.6 million, primarily attributable to higher NOI on a larger rental portfolio. This increase in NOI was bolstered by strong rent growth and higher occupancy, alongside successful cost containment aided by a lower turnover rate.
- An increase in multi-family rental Core FFO of \$0.1 million from the inclusion of The Selby as the property was stabilized in 2020. The U.S. multi-family rental Core FFO remained relatively flat as a decrease in NOI, reflecting the negative impact of the COVID-19 pandemic, was completely offset by savings in interest expense associated with a lower LIBOR and a reduced debt balance.
- An increase in residential development Core FFO of \$8.5 million, driven largely by Tricon's investments in for-sale housing. Despite the negative impact of the COVID-19 pandemic in early 2020, sales in several projects have recovered to pre-COVID-19 levels. For-sale housing investments distributed \$12.7 million of cash to Tricon during the quarter, excluding performance fees.
- A decrease in Core FFO from private funds and advisory of \$2.3 million, driven by lower performance fees and asset management fees as a result of lower investment balances for maturing for-sale housing investments, along with a decrease in Johnson development fees.

- An increase in corporate overhead of \$4.0 million, driven by higher cash compensation arising from performance-based AIP awards, which were finalized in the fourth quarter of 2020, as well as payroll costs that correspond with the Company's ongoing growth plans. This increase was partially offset by a decrease in general and administration expense that was higher in 2019 attributable to various consulting fees related to the formal implementation of the Company's ESG roadmap.
- A decrease in corporate interest expense of \$4.3 million, resulting from a significant repayment of the corporate credit facility using the proceeds of the Due to Affiliate (see Section 3.2) as well as lower average effective interest rates. As at December 31, 2020, the corporate credit facility balance was \$26.0 million compared to \$297.0 million at the end of the prior year.
- An increase in the current income tax recovery of \$6.0 million from the Company's application of a loss carryback provision which enabled the Company to recover current year losses in certain corporate entities from taxes paid in prior years.

For the twelve months ended December 31, 2020, Core FFO increased by \$54.6 million to \$109.6 million compared to \$55.0 million in the prior year, primarily driven by the same reasons noted above. In addition, the current year includes a full twelve months of U.S. multi-family rental operating results compared to approximately seven months of results in 2019. Corporate overhead also decreased as more property management personnel costs were allocated to property direct operating costs, as the rental portfolio under management continued to expand.

AFFO for the three and twelve months ended December 31, 2020 increased by \$16.5 million and \$53.3 million, respectively, from the same periods in the prior year. These variances reflect the increase in Core FFO noted above, along with higher recurring capital expenditures attributable to the full-year inclusion of the U.S. multi-family rental portfolio results. While Tricon's single-family rental portfolio has expanded in 2020, the Company was able to lower recurring capital expenditures as a result of lower turnover and a targeted reduction in elective capital projects during the COVID-19 pandemic.

Core FFO per share increased by \$0.06 and \$0.20 to \$0.16 and \$0.49, respectively, for the three and twelve months ended December 31, 2020 compared to the same periods in the prior year for the reasons discussed above. AFFO per share increased by \$0.06 and \$0.21 to \$0.13 and \$0.36, respectively, for the three and twelve months ended December 31, 2020 compared to the same periods in the prior year for the reasons discussed above. These increases were partially offset by a higher number of weighted average dilutive shares, which includes the impact of the exchangeable preferred units of Tricon PIPE LLC.

The following table provides reconciliations of the items marked "A" and "C" above, from the corporate-level expenses, as shown in the *Corporate* column of the proportionate income statement by business segment. Refer to the proportionate income statement below for a reconciliation of corporate-level costs to proportionate and consolidated results per IFRS.

The breakdown of recurring capital expenditures by business segment is also presented and reconciled to the item marked as "D" in the table above.

| For the periods ended December 31 | | Three months | | | Twelve months | |
|---|------------|--------------|------------|------------|---------------|------------|
| (in thousands of U.S. dollars) | 2020 | 2019 | Variance | 2020 | 2019 | Variance |
| Property management overhead | \$ 5,872 | \$ 5,675 | \$ (197) | \$ 22,654 | \$ 25,875 | \$ 3,221 |
| Cash compensation expense ⁽¹⁾ | 10,468 | 6,296 | (4,172) | 23,748 | 22,954 | (794) |
| General and administration expense ⁽²⁾ | 3,287 | 3,615 | 328 | 13,893 | 13,401 | (492) |
| Corporate overhead | \$ 19,627 | \$ 15,586 | \$ (4,041) | \$ 60,295 | \$ 62,230 | \$ 1,935 |
| Interest expense | \$ 10,212 | \$ 8,974 | \$ (1,238) | \$ 35,048 | \$ 32,648 | \$ (2,400) |
| Convertible debentures | (2,506) | (2,492) | 14 | (9,927) | (9,902) | 25 |
| Interest on Due to Affiliate | (4,312) | - | 4,312 | (5,654) | _ | 5,654 |
| Amortization of deferred financing costs, | | | | | | |
| discounts and lease obligations | (2,376) | (1,185) | 1,191 | (6,435) | (4,573) | 1,862 |
| Corporate interest expense B | \$ 1,018 | \$ 5,297 | \$ 4,279 | \$ 13,032 | \$ 18,173 | \$ 5,141 |
| Current income tax expense | \$ (7,334) | \$ (1,997) | \$ 5,337 | \$ (4,512) | \$ 5,201 | \$ 9,713 |
| Tax on sale of U.S. multi-family | | | | | | |
| developments | - | 683 | 683 | - | (4,101) | (4,101) |
| Total income tax expense C | \$ (7,334) | \$ (1,314) | \$ 6,020 | \$ (4,512) | \$ 1,100 | \$ 5,612 |
| Single-family rental | \$ 5,550 | \$ 4,117 | \$ (1,433) | \$ 22,462 | \$ 23,165 | \$ 703 |
| U.S. multi-family rental | 1,875 | 1,708 | (167) | 5,373 | 3,458 | (1,915) |
| Canadian multi-family rental | 20 | _ | (20) | 40 | _ | (40) |
| Recurring capital expenditures D | \$ 7,445 | \$ 5,825 | \$ (1,620) | \$ 27,875 | \$ 26,623 | \$ (1,252) |

(1) Compensation expense for Core FFO purposes excludes equity-settled, non-cash compensation and non-recurring compensation. The table below reconciles salaries and benefits and cash-settled AIP and LTIP expense to total compensation expense in the *Corporate* segment.

| For the periods ended December 31 | Three months | | | | | Twelve months | | | | | |
|-----------------------------------|------------------|----|-------|----|----------|---------------|----|--------|----|----------|--|
| (in thousands of U.S. dollars) | 2020 | | 2019 | | Variance | 2020 | | 2019 | | Variance | |
| Salaries and benefits | \$ 2,738 | \$ | 2,402 | \$ | (336) | \$ 10,493 | \$ | 8,539 | \$ | (1,954) | |
| Cash-settled AIP | 6,233 | | 2,577 | | (3,656) | 15,393 | | 11,572 | | (3,821) | |
| Cash-settled LTIP | 1,497 | | 1,317 | | (180) | (2,138) | | 2,843 | | 4,981 | |
| Compensation expense for FFO | \$ 10,468 | \$ | 6,296 | \$ | (4,172) | \$ 23,748 | \$ | 22,954 | \$ | (794) | |
| Non-cash compensation | \$ 558 | \$ | 335 | \$ | (223) | \$ 4,593 | \$ | 3,508 | \$ | (1,085) | |
| Non-recurring compensation | - | | 27 | | 27 | 107 | | 1,184 | | 1,077 | |
| Total compensation expense | | | | | | | | | | | |
| in Corporate segment | \$ 11,026 | \$ | 6,658 | \$ | (4,368) | \$ 28,448 | \$ | 27,646 | \$ | (802) | |

(2) General and administration expense for Core FFO purposes refers to the general and administration expense in the Corporate segment, plus \$569 and \$2,415 related to lease payments on right-of-use assets for the three and twelve months ended December 31, 2020, respectively (2019 – \$933 and \$2,291, respectively). The twelve months ended December 31, 2019 also includes an add-back of \$20 related to gain on disposition of fixed assets.

Proportionate income statement by business segment

The tables below present Tricon's proportionate share of the consolidated financial results for the three and twelve months ended December 31, 2020 by deducting third-party interests' share of each income statement line item in the single-family rental business segment. Third-party interests and inter-segment eliminations are adjusted for in the *IFRS reconciliation* column to arrive at the consolidated results under IFRS. Net income attributable to non-controlling interests is deducted in the Private Funds and Advisory segment to arrive at net income attributable to Tricon's shareholders.

Tricon's proportionate share of results by business segment

| | | mons | hioho | | | silless segment | | | | |
|---|---------------|----------|-------|----------------|---------------|-------------------|-----------|------------------|----------------|----------------|
| For the three months ended December 31, 2020 | Single-Family | Multi-Fa | | | Private Funds | | proportio | | IFRS | Tricon results |
| (in thousands of U.S. dollars) | Rental | Re | ental | Development | and Advisory | Corporate | res | sults | reconciliation | as reported |
| | Section 4.1 | Section | 1 4.2 | Section 4.3 | Section 4.4 | | | | | Section 3.1 |
| Revenue from rental properties | \$ 75,254 | \$27, | 583 | \$ - | \$ - | \$ - | \$ 102, | 837 | \$ 19,146 | \$ 121,983 |
| Direct operating expenses | (24,778) | (11, | ,979) | - | - | - | (36, | 757) | (5,903) | (42,660) |
| Net operating income from rental properties | 50,476 | 15 | ,604 | - | - | - | 66, | 080 | 13,243 | 79,323 |
| Revenue from private funds and advisory services | - | | - | - | 10,339 | - | 10,3 | 339 | - | 10,339 |
| Income from investments in | | | | | | | | | | |
| Canadian multi-family developments | - | | 427 | 8,293 | - | - | 8, | 720 | - | 8,720 |
| Other income from Canadian development properties | | | - | 309 | - | - | | 309 | - | 309 |
| Income from investments in for-sale housing | - | | - | 10,191 | - | - | 10, | 191 | - | 10,191 |
| Other income | _ | | - | - | 1,774 | - | 1, | 774 | - | 1,774 |
| Property management overhead | - | | - | - | - | (5,872) | (5, | 872) | - | (5,872) |
| Compensation expense | - | | _ | - | (3,914) | (11,026) | (14, | 940) | _ | (14,940) |
| General and administration expense | (1,826) | | (406) | - | (172) | (2,718) | (5, | 122) | (626) | (5,748) |
| Other expense | _ | | _ | _ | - | - | . , | _ | (791) | (791 |
| Interest expense | (20,365) | (8. | ,077) | (226) | - | (10,212) | (38. | 880) | (5,541) | (44,421 |
| Fair value gain on rental properties | 94,791 | | _ | _ | _ | - | | , 791 | 12,204 | 106,995 |
| Fair value loss on derivative | | | | | | | | | | |
| financial instruments and other liabilities | (11) | | _ | _ | _ | (16,407) | (16. | 418) | _ | (16,418 |
| Transaction costs | (24) | | (505) | _ | _ | (1,962) | | 491) | _ | (2,491) |
| Amortization and depreciation expense | (, | | (6) | _ | (747) | , | | 614) | _ | (2,614 |
| Foreign exchange gain | _ | | (0) | _ | (,) | 948 | | 948 | _ | 948 |
| Leasing commission income (expense) | _ | | _ | _ | 709 | | | 709 | (709) | 510 |
| Net change in fair value of limited partners' | | | | | 705 | | | /05 | (703) | |
| interests in rental business | _ | | _ | _ | _ | _ | | _ | (17,780) | (17,780) |
| Current income tax (expense) recovery | (249) | | 5 | _ | (3) | | 7 | 087 | (17,700) | 7,087 |
| Deferred income tax recovery (expense) | (2 13) | | _ | 287 | (3) | (33,420) | | 133) | _ | (33,133) |
| Non-controlling interest | _ | | _ | | (1,800) | | | 800) | _ | (1,800) |
| Net income (loss) attributable | | | | | (1,000) | | (1, | 000) | | (1,000) |
| to Tricon's shareholders | \$ 122,792 | \$7, | 042 | \$ 18,854 | \$ 6,186 | \$ (75,196) | \$ 79,0 | 678 | \$ - | \$ 79,678 |
| Fair value gain on rental properties | (94,791) | | _ | _ | _ | _ | (94. | 791) | (12,204) | (106,995) |
| Fair value loss on derivative | (,) | | | | | | (, | , | (,, | (,, |
| financial instruments and other liabilities | 11 | | _ | _ | _ | 16,407 | 16 | 418 | _ | 16,418 |
| Other adjustments | _ | | _ | _ | _ | | , | _ | 12,204 | 12,204 |
| FFO attributable to Tricon's shareholders | \$ 28,012 | \$7, | 042 | \$ 18,854 | \$ 6,186 | \$ (58,789) | \$ 1,3 | 305 | \$ - | \$ 1,305 |
| Other income | _ | | _ | _ | (1,774) | _ | (1. | 774) | _ | (1,774) |
| Transaction costs | 24 | | 505 | _ | | 1,962 | | 491 [′] | _ | 2,491 |
| Deferred tax (recovery) expense | _ | | _ | (287) | _ | 33,420 | | 133 | _ | 33,133 |
| Amortization and depreciation expense | _ | | 6 | (207) | 747 | 1,861 | | 614 | _ | 2,614 |
| Foreign exchange gain | _ | | _ | _ | - | (948) | | 948) | _ | (948) |
| Interest incurred on convertible debentures | _ | | _ | _ | _ | 2,506 | , | 506 | _ | 2,506 |
| Interest on Due to Affiliate | _ | | _ | _ | _ | 4,312 | | 312 | _ | 4,312 |
| Amortization of deferred financing costs, | | | | | | 1,012 | ч, | 2.2 | | 1,012 |
| discounts and lease obligations | 642 | | _ | 3 | _ | 2,376 | 3 | 021 | 709 | 3,730 |
| Non-cash compensation ⁽¹⁾ | - | | _ | - | 144 | 558 | | 702 | | 702 |
| Other adjustments ⁽²⁾ | _ | | (357) | (7,038) | 512 | (569) | | 452) | _ | (7,452 |
| Limited partners' share of Core FFO adjustments | _ | | - | (,,000) | | (505) | (7, | | (709) | (7,432) |
| Core FFO attributable to Tricon's shareholders | \$ 28,678 | \$7. | 196 | \$ 11,532 | \$ 5,815 | \$ (13,311) | \$ 39,9 | | \$ - | \$ 39,910 |
| Recurring capital expenditures | (5,550) | | | \$ 11,552 _ | ÷ 5,815 | - + (13,311) - | | | • - • | (7,445) |
| | | | ,895) | | | | | 445) | | |
| AFFO attributable to Tricon's shareholders | \$ 23,128 | \$5, | 301 | \$ 11,532 | \$ 5,815 | \$ (13,311) | \$ 32,4 | 405 | \$ - | \$ 32,465 |

(1) Comprised of equity-settled AIP and LTIP expense, which is presented in Section 3.1.

(2) Comprised of amortization, unrealized foreign exchange and deferred taxes within income from equity-accounted investments and investments held at FVTPL, non-controlling interests' share of amortization and depreciation and other income from government assistance, other non-recurring expenses and lease payments related to the Company's right-of-use assets. Fair value gains from investments in Canadian multi-family developments are also included as eliminations.

MANAGEMENT'S DISCUSSION AND ANALYSIS

for the year ended December 31, 2020

| | | Tri | con's prop | ortionate sh | are o | of results by bus | sines | s segment | | | |
|--|-------------------------------|-----|----------------------------|---------------|-------|-------------------|-------|-----------|--------------------------------|-----------------------------|-------------|
| For the twelve months ended December 31, 2020 | Single-Family | Мι | ulti-Family | Residen | | Private Funds | | Corporato | Total proportionate | - IFRS | |
| (in thousands of U.S. dollars) | Rental | 50 | Rental | Developm | | and Advisory | | Corporate | results | reconciliation | • |
| | Section 4.1 | | ction 4.2 | Section | 4.3 | Section 4.4 | | | | | Section 3.1 |
| Revenue from rental properties Direct operating expenses | \$ 296,940 (99,412) | \$ | 111,205 (48,296) | \$ | - | \$ – | \$ | - | \$ 408,145 (147,708) | \$ 70,042 (21,830 | |
| Net operating income from rental properties | 197,528 | | 62,909 | | - | - | | - | 260,437 | 48,212 | 308,649 |
| Revenue from private funds and advisory services ⁽¹ |) – | | - | | - | 34,090 | | - | 34,090 | - | 34,090 |
| Income from investments in | | | | | | | | | | | |
| Canadian multi-family developments | - | | 746 | 13,3 | 378 | - | | - | 14,124 | - | 14,124 |
| Other income from Canadian development properties | s – | | - | 7 | 791 | - | | - | 791 | - | 791 |
| Loss from investments in for-sale housing | - | | - | (61,7 | 776) | - | | - | (61,776) | - | (61,776) |
| Other income | - | | - | | _ | 1,774 | | - | 1,774 | - | 1,774 |
| Property management overhead | _ | | _ | | _ | - | | (22,654) | (22,654) | - | (22,654) |
| Compensation expense | _ | | _ | | _ | (11,652) | | (28,448) | (40,100) | - | (40,100) |
| General and administration expense | (6,878) | | (2,111) | | _ | (879) | | (11,478) | (21,346) | (2,223 |) (23,569) |
| Other expense | - | | - | | _ | - | | - | _ | (3,173 | |
| Interest expense | (81,564) | | (33,464) | (5 | 524) | _ | | (35,048) | (150,600) | (20,010 | |
| Fair value gain (loss) on rental properties | 190,461 | | (22,535) | 1 | _ | _ | | - | 167,926 | 30,388 | |
| Fair value loss on derivative financial instruments | | | ,,000) | | | | | | , | 00,000 | |
| and other liabilities | (39) | | _ | | _ | _ | | (7,422) | (7,461) | _ | (7,461) |
| Transaction costs | (24) | | (2,409) | | _ | _ | | (11,583) | (14,016) | _ | |
| Amortization and depreciation expense | (2-!) | | (22) | | _ | (3,079) | | (7,747) | (10,848) | _ | (10,848) |
| Foreign exchange gain (loss) | _ | | (22) | | _ | (0,070) | | (170) | (166) | _ | |
| Leasing commission income (expense) ⁽¹⁾ | _ | | - | | _ | 2,613 | | (170) | 2,613 | (2,613 | . , |
| Net change in fair value of limited partners' | | | | | | 2,013 | | | 2,015 | (2,015 |) – |
| interests in rental business | | | _ | | | _ | | _ | _ | (50,581 |) (50,581) |
| Current income tax (expense) recovery | (319) | | 5 | 11 | 145) | (3) | | 4,512 | 4,050 | (50,50 | |
| Deferred income tax recovery (expense) | (515) | | - | , | 118 | (3) | | (48,543) | (40,425) | _ | (40,425) |
| Non-controlling interest | _ | | _ | 0, | | (3,091) | | (40,545) | (40,423) | - | (0,004) |
| Net income (loss) attributable | | | | | _ | (5,051) | | | (5,051) | | (5,051) |
| to Tricon's shareholders | \$ 299,165 | \$ | 3,123 | \$ (40,1 | 58) | \$ 19,773 | \$ | (168.581) | \$ 113,322 | \$ - | \$ 113,322 |
| Fair value (gain) loss on rental properties | (190,461) | | 22,535 | | _ | _ | | _ | (167,926) | (30,388 | |
| Loss from investments in for-sale housing | (100,101) | | | 79,5 | 579 | _ | | _ | 79,579 | (00,000 | |
| Fair value loss on derivative | | | | 70,0 | ,, , | | | | 75,575 | | 73,073 |
| financial instruments and other liabilities | 39 | | _ | | _ | _ | | 7,422 | 7,461 | _ | 7,461 |
| Other adjustments | - 55 | | | | _ | _ | | | | 30,388 | |
| FFO attributable to Tricon's shareholders | \$ 108,743 | \$ | 25,658 | \$ 39,4 | | \$ 19,773 | ¢ | (161,159) | \$ 32,436 | \$ - | |
| | | φ | | р 39,4 | | | φ | | | - | |
| Other income | - | | - | | - | (1,774) | | - | (1,774) | - | (.,, |
| Transaction costs | 24 | | 2,409 | (0.1 | - | - | | 11,583 | 14,016 | - | 14,016 |
| Deferred tax (recovery) expense | - | | - | (8,1 | 118) | _ | | 48,543 | 40,425 | - | 40,425 |
| Amortization and depreciation expense | - | | 22 | | - | 3,079 | | 7,747 | 10,848 | - | 10,848 |
| Foreign exchange (gain) loss | - | | (4) | | - | - | | 170 | 166 | - | 166 |
| Interest incurred on convertible debentures | - | | - | | - | - | | 9,927 | 9,927 | - | 9,927 |
| Interest on Due to Affiliate | - | | - | | - | - | | 5,654 | 5,654 | - | 5,654 |
| Amortization of deferred financing costs, | | | | | | | | | | | |
| discounts and lease obligations | 1,918 | | - | | 6 | - | | 6,435 | 8,359 | 2,563 | |
| Non-cash compensation ⁽²⁾ | - | | - | | - | 386 | | 4,593 | 4,979 | - | 4,979 |
| Non-recurring compensation | - | | - | | - | - | | 107 | 107 | - | 107 |
| Other adjustments ⁽³⁾ | - | | (97) | (12,3 | 396) | (651) | | (2,415) | (15,559) | - | (15,559) |
| Limited partners' share of Core FFO adjustments | - | | - | | _ | - | | - | | (2,563 |) (2,563) |
| Core FFO attributable to Tricon's shareholders | \$ 110,685 | \$ | 27,988 | \$ 18,9 | 13 | \$ 20,813 | \$ | (68,815) | \$ 109,584 | \$ - | \$ 109,584 |
| Recurring capital expenditures | (22,462) | | (5,413) | | - | - | | - | (27,875) | - | (27,875) |
| AFFO attributable to Tricon's shareholders | \$ 88,223 | \$ | 22,575 | \$ 18,9 | 13 | \$ 20,813 | \$ | (68,815) | \$ 81,709 | \$ - | \$ 81,709 |

Tricon's proportionate share of results by business segment

(1) Certain fees earned from limited partners totalling \$1,489 in the first quarter of 2020 have been reclassified from single-family rental Core FFO to private funds and advisory Core FFO to conform with the current period presentation. This change in classification did not result in any changes to total Core FFO and AFFO.

(2) Comprised of equity-settled AIP and LTIP expense, which is presented in Section 3.1.

(3) Comprised of amortization, unrealized foreign exchange and deferred taxes within income from equity-accounted investments and investments held at FVTPL, non-controlling interests' share of amortization and depreciation and other income from government assistance, other non-recurring expenses and lease payments related to the Company's right-of-use assets. Fair value gains from investments in Canadian multi-family developments are also included as eliminations.

Proportionate balance sheet by business segment

The table below presents Tricon's proportionate share of net assets as at December 31, 2020 by deducting third-party interests' share of each balance sheet line item in the single-family rental business segment. Third-party interests and inter-segment eliminations are adjusted for in the *IFRS reconciliation* column to arrive at the consolidated net assets under IFRS. Net assets attributable to non-controlling interests are deducted in the Private Funds and Advisory segment to arrive at net assets attributable to Tricon's shareholders.

| Net debt to assets ⁽³⁾ | | | | | | | | | | | 55.3% | | | | |
|---|-------------------------|----|------------------------|-------|------------------------|--------|----------------------|-------|-----------|----|----------------------------------|-----|----------------------|------------|---------------------------|
| Tricon's shareholders | \$ 1,557,740 | \$ | 446,287 | \$ | 302,960 | \$ | 9,893 | \$ | (581,784) | \$ | 1,735,096 | \$ | - | \$1 | ,735,096 |
| Non-controlling interest | _ | | - | | - | | 8,142 | | - | | 8,142 | | - | | 8,142 |
| Total liabilities | 2,481,177 | | 933,363 | | 64,055 | | 2,322 | | 856,870 | | 4,337,787 | 1 | ,093,809 | 5 | ,431,596 |
| Other liabilities ⁽²⁾ | 68,967 | | 23,023 | | 4,037 | | 2,322 | | 86,177 | | 184,526 | | 375,960 | | 560,486 |
| Deferred income tax liabilities | - | | - | | - | | - | | 298,071 | | 298,071 | | - | | 298,071 |
| Long-term incentive plan | - | | - | | - | | - | | 17,930 | | 17,930 | | - | | 17,930 |
| Due to Affiliate | - | | - | | - | | - | | 251,647 | | 251,647 | | - | | 251,647 |
| Convertible debentures | - | | - | | - | | - | | 165,956 | | 165,956 | | - | | 165,956 |
| Debt | \$ 2,412,210 | \$ | 910,340 | \$ | 60,018 | \$ | - | \$ | 37,089 | \$ | 3,419,657 | \$ | 717,849 | \$ 4 | 4,137,506 |
| Liabilities | φ 1 ,050,517 | Ψ | 1,575,050 | ÷ | 507,015 | Ψ | 20,337 | Ψ | 275,000 | Ψ | 0,001,023 | Ψ. | ,055,005 | Υ γ | , 174,054 |
| Total assets | \$ 4,038,917 | ¢ | 1,379,650 | ¢ | 367,015 | ¢ | 20,357 | ¢ | 275,086 | ¢ | 6,081,025 | ¢ 1 | .093,809 | ¢ 7 | ,174,834 |
| Other working capital items ⁽¹⁾ | 9,758 | | 6,411 | | 1,811 | | 3,204 | | 17,530 | | 38,714 | | 538 | | 39,252 |
| Cash | 14,909 | | 3,582 | | 637 | | 7,178 | | 5,713 | | 32,019 | | 23,139 | | 55,158 |
| Derivative financial instruments | _ | | | | 10,550 | | 1,052 | | 841 | | 841 | | | | 841 |
| Deferred income tax assets | _ | | - | | 10,558 | | 1,052 | | 90,834 | | 102,444 | | | | 102,444 |
| Goodwill and other assets | _ | | 100 | | _ | | 324 | | 156,404 | | 156,828 | | _ | | 156,828 |
| Intangible assets | 73,105 | | 10,200 | | 4,134 | | - 8,599 | | 3,764 | | 12,363 | | 20,075 | | 12,363 |
| Restricted cash | 73,165 | | 18,268 | | 4,194 | | _ | | - | | 95,627 | | 20,675 | | 116,302 |
| Investments in for-sale housing | - | | _ | | 164,842 | | _ | | _ | | 164,842 | | - | | 164,842 |
| Investments in Canadian multi-family developments Canadian development properties | - | | 19,913 | | 74,955 110,018 | | - | | - | | 94,868 110.018 | | - | | 94,868 110,018 |
| Rental properties | \$ 3,941,085 | \$ | 1,331,376 | \$ | - | \$ | - | \$ | - | \$ | 5,272,461 | \$ | 1,049,457 | \$ (| 6,321,918 |
| Assets | Section 4.1 | | Section 4.2 | Se | ction 4.3 | See | ction 4.4 | | | | | | | S | ection 3.2 |
| (in thousands of U.S. dollars) | Single-Family Rental | | Multi-Family Rental | Dev | esidential elopment | and | te Funds Advisory | | Corporate | p | Total roportionate results | ree | IFRS conciliation | a | con results s reported |
| | | Tr | icon's proporti | onate | share of ba | lances | heet by bu | sines | s segment | | | - | | | |

(1) Other working capital items include amounts receivable and prepaid expenses and deposits.

(2) Other liabilities include derivative financial instruments, other liability, limited partners' interests, dividends payable, resident security deposits and amounts payable and accrued liabilities.

(3) Calculated by dividing net debt by total assets (net of cash and restricted cash).

Summary of selected income statement, balance sheet and operating items

Management considers net assets (book value) per share to be an important component of the Company's value, and it reflects the IFRS value of its rental and development businesses. The Company also creates additional franchise value through its Private Funds and Advisory business and vertically integrated, technology-enabled operating platform which allows it to generate various forms of contractual fees that are not reflected in the IFRS book values disclosed below.

| | | | | Renta | al portfolio ⁽¹⁾ | | | |
|---|-------------------------|------------|-----------------------------|-------|---|---------------------------|-----|-------------------------|
| (in thousands of U.S. dollars, except units, average monthly rent, percentages and per share amounts) | Single-Family Rental | | Multi-Family Rental B | pr | Tricon oportionate results (A) + (B) | Consolidat reconciliat | | Consolidated results |
| Total rental units managed | | 22,766 | 7,289 | | | | | 30,055 |
| Tricon's proportionate share of rental units | | 17,859 | 7,289 | | 25,148 | 4,9 | 07 | 30,055 |
| Average monthly rent | \$ | 1,464 | \$ 1,217 | | | | | |
| Occupancy | | 96.4% | 93.6% | | | | | |
| NOI margin | | 67.1% | 56.6% | | | | | |
| Quarterly NOI | | 50,476 | 15,604 | | 66,080 | 13,2 | 43 | 79,323 |
| Annualized NOI | | 201,904 | 62,416 | | 264,320 | 52,9 | 72 | 317,292 |
| Rental properties | : | 3,941,085 | 1,331,376 | | 5,272,461 | 1,049,4 | 57 | 6,321,918 |
| Investments in Canadian multi-family | | | | | | | | |
| developments (The Selby) | | - | 19,913 | | 19,913 | | - | 19,913 |
| Net debt | (1 | 2,324,136) | (888,490) | | (3,212,626) | (674,0 | 35) | (3,886,661) |
| Other liabilities | | (59,209) | (16,512) | | (75,721) | (375,4 | 22) | (451,143) |
| Net assets attributable to | | | | | | | | |
| Tricon's shareholders | \$ 1 | ,557,740 | \$ 446,287 | \$ | 2,004,027 | \$ | - | \$ 2,004,027 |
| Net assets per share ⁽²⁾ | \$ | 8.06 | \$ 2.31 | \$ | 10.37 | | | |
| Net assets per share (CAD) ⁽²⁾ | \$ | 10.26 | \$ 2.94 | \$ | 13.20 | | | |

(1) Figures presented exclude Canadian multi-family rental (The Selby) except for investments in Canadian multi-family developments (The Selby) of \$19,913.

(2) As at December 31, 2020, common shares outstanding were 193,175,802 and the USD/CAD exchange rate was 1.2732.

| | | Development portfolio | | | | | | | | | | | |
|---|----|---|----------------------|---------|---|----------|----------------------|---|--------------------------------------|--|--|--|--|
| (in thousands of U.S. dollars, except per share amounts) | | Canadian Aulti-Family Rental A | lti-Family Rental | | Tricon proportionate results (A) + (B) | | Consolio reconcil | | Consolidated results A + B + C | | | | |
| Estimated annual NOI upon stabilization ⁽¹⁾ | \$ | 39,203 | | | | | | | | | | | |
| Projected distributions | | | | | | | | | | | | | |
| net of advances remaining | | | \$ | 322,580 | | | | | | | | | |
| Property value ⁽²⁾ | \$ | 184,973 | \$ | 164,842 | \$ | 349,815 | \$ | _ | \$ 349,815 | | | | |
| Net debt | | (55,187) | | - | | (55,187) | | _ | (55,187) | | | | |
| Other (liabilities) assets | | (1,670) | | 10,002 | | 8,332 | | _ | 8,332 | | | | |
| Net assets attributable | | | | | | | | | | | | | |
| to Tricon's shareholders | \$ | 128,116 | \$ | 174,844 | \$ | 302,960 | \$ | - | \$ 302,960 | | | | |
| Net assets per share ⁽³⁾ | \$ | 0.66 | \$ | 0.91 | \$ | 1.57 | | | | | | | |
| Net assets per share (CAD) ⁽³⁾ | \$ | 0.84 | \$ | 1.16 | \$ | 2.00 | | | | | | | |

(1) Calculated on a total portfolio basis excluding The Selby, and based on current project development plans assuming a target development yield of 4.75% on cost.

(2) Includes investments in Canadian multi-family developments, investments in for-sale housing and Canadian development properties.

(3) As at December 31, 2020, common shares outstanding were 193,175,802 and the USD/CAD exchange rate was 1.2732.

MANAGEMENT'S DISCUSSION AND ANALYSIS

for the year ended December 31, 2020

| | Corpora | ate assets and liabilities | (1) |
|---|------------------------------------|------------------------------|-------------------------|
| (in thousands of U.S. dollars, except per share amounts) | Tricon proportionate results | Consolidation reconciliation | Consolidated results |
| Intangible assets and other assets | \$ 169,091 | \$ - | \$ 169,091 |
| Deferred income tax liabilities | (206,185) | - | (206,185) |
| Net debt | (24,198) | - | (24,198) |
| Convertible debentures | (165,956) | - | (165,956) |
| Due to Affiliate | (251,647) | - | (251,647) |
| Other liabilities | (92,996) | - | (92,996) |
| Net assets attributable to Tricon's shareholders | \$ (571,891) | \$ - | \$ (571,891) |
| Net assets per share ⁽²⁾ | \$ (2.96) | | |
| Net assets per share (CAD) ⁽²⁾ | \$ (3.77) | | |

....

(1) Includes the assets and liabilities of the Private Funds and Advisory and Corporate segments.

(2) As at December 31, 2020, common shares outstanding were 193,175,802 and the USD/CAD exchange rate was 1.2732.

| | | Future performance fees | | | | | | | | |
|--|-------------------------|------------------------------------|---------------------|------|---------------------------------|--|--|--|--|--|
| (in thousands of U.S. dollars, except per share amounts) | Single-Family Rental | Canadian Multi-Family Rental | For-Sale Housing | prop | Tricon portionate results | | | | | |
| | Δ | в | G | A | + B + C | | | | | |
| Estimated future performance fees ⁽¹⁾ | \$ 60,000 | \$ 18,891 | \$ 21,770 | \$ | 100,661 | | | | | |
| Net assets per share ⁽²⁾ | | | | \$ | 0.52 | | | | | |
| Net assets per share (CAD) ⁽²⁾ | | | | \$ | 0.66 | | | | | |

(1) Includes estimated future performance fees before the deduction of any amounts paid to employees under the LTIP.

(2) As at December 31, 2020, common shares outstanding were 193,175,802 and the USD/CAD exchange rate was 1.2732.

| | Summary of net assets (book value) per share | | | | | | | | | |
|--|--|---------|-------------------------------|--------|---|----------|--|---------|-------------------------------|------|
| (in thousands of U.S. dollars, except per share amounts) | Rental portfolio A | | Development portfolio B | | Corporate assets and liabilities © | | Tricon proportionate results (A + B + C | | Future performance fees | |
| Net assets attributable | ¢ | 004 007 | * ~ | 00.000 | ¢ (| E74 004) | ¢ 4 | 705 000 | | |
| to Tricon's shareholders | \$ 2,004,027 | | \$ 302,960 | | \$ (571,891) | | \$ 1,735,096 | | | |
| Net assets per share ⁽¹⁾ | \$ | 10.37 | \$ | 1.57 | \$ | (2.96) | \$ | 8.98 | \$ | 0.52 |
| Net assets per share (CAD) ⁽¹⁾ | \$ | 13.20 | \$ | 2.00 | \$ | (3.77) | \$ | 11.43 | \$ | 0.66 |

(1) As at December 31, 2020, common shares outstanding were 193,175,802 and the USD/CAD exchange rate was 1.2732.

Corporate overhead

The cost of the Company's integrated, technology-enabled operating platform and other overhead costs are presented below (based on the activities within Tricon's Private Funds and Advisory and Corporate segments). While Tricon is intent on reducing its overhead as a percentage of revenue from rental properties and of total assets over time by growing its rental business and managing more third-party capital, it should be noted that the Company employs talented and relatively well-paid investment, development and computer engineering professionals that add to its cost structure but which position Tricon for future growth and longer-term operating efficiencies.

| For the years ended | | |
|---|-------------------|-------------------|
| (in thousands of U.S. dollars, except percentages) | December 31, 2020 | December 31, 2019 |
| Property management overhead | \$ 22,654 | \$ 25,875 |
| Compensation expense | 40,100 | 37,681 |
| General and administration expense | 12,357 | 12,144 |
| Total overhead costs ⁽¹⁾ | 75,111 | 75,700 |
| Net of revenue from private funds and advisory services | (34,090) | (41,060) |
| Net overhead costs | \$ 41,021 | \$ 34,640 |
| As a % of revenue from rental properties | 8.6% | 9.6% |
| As a % of total assets | 0.6% | 0.5% |

(1) Includes the sum of the corporate overhead of the Private Funds and Advisory and Corporate segments.



7

LIQUIDITY AND CAPITAL RESOURCES

OPERATIONAL KEY PERFORMANCE INDICATORS



ACCOUNTING ESTIMATES AND POLICIES, CONTROLS AND PROCEURES, AND RISK ANALYSIS

9

HISTORICAL FINANCIAL INFORMATION

6. LIQUIDITY AND CAPITAL RESOURCES

6.1 Financing strategy

The Company seeks to maintain financial strength and flexibility by lowering its cost of debt and equity capital and minimizing interest rate fluctuations over the long term. Some key elements of Tricon's financing strategy are:

- Using various forms of debt such as floating-rate bank financing and unsecured debentures with conversion features, and attempting to stagger the maturity of its obligations.
- Using convertible or exchangeable securities where the principal can be redeemed by the issuance of common shares at the Company's option.
- Where appropriate, raising equity through the public or private markets to finance its growth and strengthen its financial position.

6.2 Liquidity

Tricon generates substantial liquidity through:

- · Stable cash flow received from our single-family rental and multi-family rental businesses.
- Cash distributions from land, lot and home sales in our legacy for-sale housing business.
- Fee income from our PF&A business.
- Repatriation of capital extracted through refinancings.
- · Cash distributions generated from the turnover of assets with shorter investment horizons.
- · Syndicating investments to private investors and thereby extracting Tricon's invested capital.

To enable us to react to attractive acquisition or investment opportunities and deal with contingencies when they arise, we typically maintain sufficient liquidity at the corporate level and within our key operating platforms. Our primary sources of liquidity consist of cash and a corporate credit facility.

Working capital

As at December 31, 2020, Tricon had a net working capital deficit of \$333.9 million, reflecting current assets of \$94.4 million, offset by current liabilities of \$428.3 million. The working capital deficit is driven primarily by debt coming due in 2021. Of this debt, \$116,000 relates to the SFR JV-1 subscription facility that is expected to be partially repaid with limited partners' capital contributions and \$109,890 relates to the U.S. multi-family credit facility that is expected be repaid upon syndication of a majority interest in the portfolio. The Company has determined that its current financial obligations and working capital deficit are adequately funded from the available borrowing capacity and from operating cash flows.

6.3 Capital resources

Debt structure

Management mitigates interest rate risk by maintaining the majority of its debt at fixed rates. The impact of variable interest rate increases or decreases is discussed in the Company's consolidated financial statements. Management also mitigates its exposure to fixed-rate interest risk by staggering maturities with the objective of achieving even, annual maturities over a ten-year time horizon to reduce Tricon's exposure to interest rate fluctuations in any one period. The Company's long-term debt structure is summarized in <u>Section 3.2</u>.

The Company provides limited financial guarantees for land loans and construction loans in its Canadian multi-family developments.

As at December 31, 2020, the Company was in compliance with all of its financial covenants.

Equity issuance and cancellations

The Company's Dividend Reinvestment Plan ("DRIP") provides eligible holders of common shares with the opportunity to reinvest their cash dividends paid on the Company's common shares to purchase additional common shares at a price equal to the average market price (as defined in the DRIP) on the applicable dividend payment date, less an applicable discount of up to 5% determined by the Board from time to time.

As of December 31, 2020, there were 193,544,915 common shares issued by the Company, of which 193,175,802 were outstanding and 369,113 were reserved to settle restricted share awards in accordance with the Company's Restricted Share Plan.

As of December 31, 2020, there was \$172.4 million in outstanding aggregate principal amount of 5.75% extendible convertible unsecured subordinated debentures (the "2022 convertible debentures"). The 2022 convertible debentures bear interest at 5.75% per annum and are convertible into 16,481,837 common shares of the Company at a conversion rate of 95.6023 common shares per \$1,000 principal amount, or a conversion price of approximately \$10.46 per common share.

As of December 31, 2020, there was \$300.0 million in outstanding aggregate principal amount of Due to Affiliate in connection with the exchangeable preferred units issued by Tricon PIPE LLC (see <u>Section 3.2</u>). Pursuant to the Transaction Documents, the Investors have rights to exchange the Preferred Units into common shares of the Company at an initial exchange price of \$8.50 (C\$11.18 as of August 26, 2020) per common share, as may be adjusted from time to time in accordance with the terms of the Transaction Documents. As at December 31, 2020, this equates to 35,801,471 common shares of the Company.

7. OPERATIONAL KEY PERFORMANCE INDICATORS

7.1 Key performance indicators

The KPIs discussed throughout this MD&A for each of the Company's business segments are calculated based on Tricon's proportionate share of each portfolio or business and are defined below. These measures are commonly used by entities in the real estate industry as useful metrics for measuring performance; however, they do not have any standardized meaning prescribed by IFRS and are not necessarily comparable to similar measures presented by other publicly-traded entities. These measures should be considered as supplemental in nature and not as a substitute for the related financial information prepared in accordance with IFRS. See "Non-IFRS measures and forward-looking statements" on page 24.

Single-family and multi-family rental

- Net operating income ("NOI") represents total revenue from rental properties, less direct operating expenses and property
 management expenses. NOI excludes non-property specific and indirect overhead expenses, interest expense and non-core
 income or expenses such as gains or losses on the disposition of rental properties. Tricon believes NOI is a helpful metric to
 evaluate the performance of its rental business and compare it to industry peers.
- Net operating income ("NOI") margin represents net operating income as a percentage of total revenue from rental properties.
- Occupancy rate represents the total number of days that units were occupied during the measurement period, divided by
 the total number of days that the units were owned during the measurement period (excluding units held for sale). Management
 believes occupancy is a main driver of rental revenues and that comparing occupancy across different periods is helpful in
 evaluating changes in rental revenues.
- Annualized turnover rate during the period represents the number of resident move-outs divided by the weighted average number of rental units (excluding units held for sale) in the period, annualized for a twelve-month period. Management believes the annualized turnover rate impacts occupancy and therefore revenue, as well as the cost to maintain the rental portfolios.
- Average monthly rent represents average monthly rental income per unit for occupied units and reflects the impact of rent concessions amortized over the life of the related leases. Tricon believes average monthly rent reflects pricing trends which impact rental revenue over time.
- Average rent growth during the period represents the percentage difference between the monthly rent from an expiring lease and the monthly rent from the next lease and reflects the impact of rent concessions amortized over the life of the related lease. Leases are either renewal leases, where a current resident chooses to stay for a subsequent lease term, or a new lease, where a previous resident moves out and a new resident signs a lease to occupy the same unit. Average rent growth drives average monthly rent and management finds it is useful to evaluate changes in rental revenue across periods.

Residential Development

- Development yield represents the estimated stabilized net operating income of a property following its completion as a percentage of its estimated total development cost.
- Core funds from operations, specifically for residential developments, presents net income as a normalized figure, adjusting for transaction costs and non-recurring and non-cash items, and is a metric that management believes to be helpful in evaluating Tricon's residential development business and comparing its performance to industry peers. Core funds from operations as a metric used in measuring Company performance is described below.

Private Funds and Advisory

 Total fee revenue represents total asset management, property management, development management and performance fees earned, excluding inter-company fees earned.

Company operating performance

Funds from operations ("FFO"), core funds from operations ("Core FFO") and adjusted funds from operations ("AFFO") are metrics that management believes to be helpful in evaluating the Company's operating performance, considering the recent expansion of its residential rental portfolio. These are metrics commonly used by securities analysts, investors and other interested parties in the evaluation of real estate entities, particularly those that own and operate income-producing properties. Management believes that providing these performance measures on a supplemental basis is helpful to investors in assessing the overall performance of the Company's business.

- FFO represents net income excluding the impact of fair value adjustments and amortization of intangibles arising from business
 combinations. The Company's definition of FFO reflects all adjustments that are specified by the National Association of Real
 Estate Investment Trusts ("NAREIT"). In addition to the adjustments prescribed by NAREIT, Tricon excludes any fair value gains
 that arise as a result of reporting under IFRS, except for fair value gains arising from Tricon's for-sale housing business which
 are intended to act as a proxy for cash generation.
- Core FFO presents FFO as a normalized figure, adjusting for transaction costs, convertible debentures interest, interest on Due to Affiliate, non-recurring and non-cash items.
- AFFO represents Core FFO less recurring capital expenditures.

Core FFO and AFFO per share amounts are calculated based on the weighted average common shares outstanding in the period, assuming the conversion of all potentially dilutive shares (including convertible debt and exchangeable preferred units).

7.2 Assets under management

Management believes that monitoring changes in the Company's AUM is key to evaluating trends in fee revenue. Growth in AUM is driven by principal investments and capital commitments to our managed Investment Vehicles by private investors.

For reporting purposes, AUM includes balance sheet capital invested in the Company's principal investments and capital managed on behalf of third-party investors in the Private Funds and Advisory business, and is calculated as follows:

| ASSETS UNDER MANAGEMENT Principal Assets Under Management | | | | |
|---|--|--|--|--|
| | | | | |
| For-sale housing | Fair value of invested capital plus unfunded commitment | | | |
| Third-Party Assets Under Management | | | | |
| Single-family rental, multi-family rental and multi-family developments | Outstanding invested capital and project-level funded debt plus unfunded commitment less return of capital | | | |
| For-sale housing | Commingled funds During the investment period, AUM = capital commitment After the investment period, AUM = outstanding invested capital Separate accounts/side-cars/syndicated investments/joint ventures Outstanding invested capital and unfunded commitment less return of capital | | | |

8. ACCOUNTING ESTIMATES AND POLICIES, CONTROLS AND PROCEDURES, AND RISK ANALYSIS

8.1 Revenue and income recognition

The following table summarizes the revenue earned from the Company's business segments.

| TOTAL REVENUE | | | | | |
|---|--|--|--|--|--|
| Revenue | | | | | |
| Revenue from rental properties | Lease revenue is primary rental revenue from a lease contract, earned directly from leasing the homes within the single-family rental business and the apartment suites within the multi-family rental business. | | | | |
| | Ancillary revenue is income earned from residents that is not primary rental revenue from a lease contract. Ancillary revenue includes pet fees, early termination fees and other service fees. | | | | |
| | Non-lease revenue includes property management services, such as repairs and maintenance performed on the properties. | | | | |
| Revenue from private funds and advisory services | Asset management fees from managing third-party capital invested through Investment Vehicles within the single-family rental, multi-family rental and residential development businesses. | | | | |
| | Performance fees from Investment Vehicles. | | | | |
| | Development management and advisory fees generated from residential development projects. Property management fees from managing single-family rental homes and multi-family rental properties. | | | | |

Revenue from rental properties

Revenue recognition under a lease commences when a resident has a right to use the leased asset, which is typically when the resident takes possession of, or controls the physical use of, the leased property. Generally, this occurs on the lease commencement date.

Lease contracts with residents normally include lease and non-lease components, which may be bundled into one fixed gross lease payment. Lease revenue earned directly from leasing the homes and apartment suites is recognized and measured on a straight-line basis over the lease term in accordance with IFRS 16, *Leases* ("IFRS 16"). Leases for single-family rental homes and multi-family rental properties are generally for a term of one to two years.

Ancillary revenue is income the Company generates from providing services that is not primary rental revenue from a lease contract. Ancillary revenue includes pet fees, early termination fees and other service fees. Ancillary revenue is measured at the amount of consideration which the Company expects to receive in exchange for providing services to a resident. Ancillary revenue is included in revenue from rental properties in the consolidated statements of comprehensive income.

Revenue from private funds and advisory services

The Company's vertically integrated management platform provides asset management, property management and development management services.

The Company provides asset management services to joint venture partners and third-party investors for which it earns marketbased fees in connection with its portfolio of properties and equity investments in the U.S. and Canada. These contractual fees are typically 1–2% of committed or invested capital throughout the lives of the Investment Vehicles managed. Contractual fees earned in exchange for providing asset management services are billed on a quarterly basis, provided that the Company's services are rendered as per the contract over the project period.

The Company also earns performance fees and they are earned once targeted returns are achieved. The Company recognizes performance fee revenue only to the extent that it is highly probable that a significant amount of the cumulative revenue recognized will not reverse. Consideration for these services is variable as it is dependent upon the occurrence of a future event that is the repayment of investor capital and a predetermined rate of return. Revenue from performance fees is typically earned and recognized towards the end of the life of an Investment Vehicle.

The Company also earns fees for development management and advisory services provided to third parties and/or related parties. Development management and advisory services are satisfied over time. Revenues are recognized based on the best estimate of the amounts earned for those services, which typically reflects contractual fees of 2–5% of the sales price of single-family lots, residential land parcels and commercial land within master-planned communities, and 4–5% of overall development costs of Canadian multi-family rental apartments. The Company includes variable consideration in revenue only to the extent that it is highly probable that a significant amount of the cumulative revenue recognized will not reverse. Specifically for Johnson, consideration for these services is variable as it is dependent upon the occurrence of a future event that is the sale of the developed property. Revenue is typically recognized as the development of the property is completed, and control has been transferred to the respective buyer. Contractual fees earned in exchange for providing development management and advisory services are billed upon the sale of the property.

The Company earns property management fees, leasing fees, acquisition and disposition fees, and construction management fees from the rental portfolio through its technology-enabled rental platform. These management services are satisfied over time and revenues for such services are recognized as services are provided in accordance with IFRS 15, *Revenue from Contracts with Customers*.

Income from investments in for-sale housing

The Company also earns income from investments in for-sale housing, which is calculated based on its share of the changes in the fair value of the net assets of each of the Investment Vehicles in which it invests. The fair value of each Investment Vehicle's net assets is determined by the waterfall distribution calculations specified in the relevant governing agreements. The inputs into the waterfall distribution calculations include the fair values of the land development and homebuilding projects and working capital held by the Investment Vehicles. The fair values of the land development and homebuilding projects are based on appraisals prepared by external third-party valuators or on internal valuations using comparable methodologies and assumptions.

Income from investments in Canadian multi-family developments

The Company recognizes income from investments in Canadian multi-family developments under the equity method. The Company's investments in Canadian multi-family developments are initially recognized at cost, and adjusted thereafter to recognize the Company's share of profit or loss of the investee in accordance with Tricon's accounting policies, which are discussed in Note 3 to the consolidated financial statements.

8.2 Accounting estimates and policies

The Company makes estimates and assumptions concerning the future. The resulting accounting estimates will, by definition, seldom equal the related actual results. Refer to the notes to the consolidated financial statements for details on critical accounting estimates.

Transition to a rental housing company

In January 2020, the Company completed its previously announced transition to an owner and operator of diversified rental housing, resulting in the Company determining that it no longer meets the criteria for being an investment entity ("Investment Entity Accounting") under IFRS 10, *Consolidated Financial Statements* ("IFRS 10"). The exact timing of the transition from an investment entity to a rental housing company is highly judgmental and the Company concluded that this transition occurred in January 2020. As a result, effective January 1, 2020 (the "Transition Date"), the Company was required to apply the acquisition method of accounting as per IFRS 3, *Business Combinations* ("IFRS 3"), to all subsidiaries that were previously measured at fair value through profit or loss ("FVTPL").

Consequently, the Company began consolidating the financial results of controlled subsidiaries including those holding its investments in single-family rental homes and U.S. multi-family rental properties, resulting in the inclusion of these subsidiaries' assets, liabilities and non-controlling interests on the balance sheet of the Company. Similarly, these subsidiaries' income and expenses have been reported on the Company's consolidated statement of comprehensive income together with the non-controlling interests' share of income.

Concurrently, the Company's investments in Canadian multi-family developments are accounted for in one of two ways: (i) proportionate consolidation for joint operations for the period between January 1, 2020 and June 22, 2020, during which time the Company owned 50% and 25% interests in The James and The Shops of Summerhill, respectively; and (ii) equity accounting for associates and joint ventures, in accordance with IFRS 11, *Joint Arrangements* and IAS 28, *Investments in Associates and Joint Ventures*. The remaining investments in for-sale housing in the U.S. will continue to be accounted for as portfolio investments (financial assets) measured at FVTPL in accordance with IFRS 9, *Financial Instruments*.
The accounting impact of the Company's businesses and their presentation in the Company's consolidated financial statements on the Transition Date are summarized in the table below.

| | ACCO | UNTING | PRESENTATION | | | | |
|--|--|---|---|--|---|--|--|
| Business segment | Accounting assessment | Accounting methodology | Presentation in Balance Sheet | Presentation in Statement of Income | Presentation of Non-controlling interest | | |
| Single-Family Rental | | | | | | | |
| Tricon wholly-owned | Controlled subsidiary | Consolidation | Rental properties | Revenue from | N/A | | |
| SFR JV-1 | Controlled subsidiary | Consolidation | - | rental properties | Limited partners' interests (Component of liabilities) | | |
| Multi-Family Rental | | | | | | | |
| U.S. multi-family | Controlled subsidiary | Consolidation | Rental properties | Revenue from rental properties | N/A | | |
| Canadian multi-family: 592 Sherbourne (The Selby) | Investments in associate | Equity method | Investments in Canadian multi-family developments | Income from investments in Canadian multi-family developments | N/A | | |
| Canadian Multi-Famil | y Developments | | | | | | |
| The Shops of Summerhill ⁽¹⁾ | Joint operation for the period between | Proportionate consolidation | Canadian development | Other income from Canadian | N/A | | |
| The James (Scrivener Square) ⁽¹⁾ | January 1, 2020 and June 22, 2020, and controlled subsidiary from June 23, 2020 | between January 1, 2020 and June 22, 2020, and consolidation from June 23, 2020 | properties | development properties | N/A | | |
| 57 Spadina (The Taylor) | Investments in associate | Equity method | Investments in Canadian multi-family | Income from investments in | N/A | | |
| WDL – Block 8 | Joint venture | Equity method | developments | Canadian multi-family | N/A | | |
| WDL – Block 20 | Joint venture | Equity method | - | developments | N/A | | |
| WDL – Blocks 3/4/7 | Joint venture | Equity method | | | N/A | | |
| WDL – Block 10 | Joint venture | Equity method | - | | N/A | | |
| 6–8 Gloucester (The lvy) | Joint venture | Equity method | - | | N/A | | |
| 7 Labatt | Joint venture | Equity method | - | | N/A | | |
| Private Funds and Ad | lvisory | | | | | | |
| Private funds GP entities | Controlled subsidiary | Consolidation | Consolidated | Revenue from private funds | N/A | | |
| Johnson development management | Controlled subsidiary | Consolidation | Consolidated | and advisory | Component of equity | | |
| For-Sale Housing | | | | | | | |
| Commingled funds | Portfolio investments | FVTPL | Investments in Income fr | Income from | N/A | | |
| Separate accounts, side-cars and joint ventures | Portfolio investments | FVTPL | for-sale housing | investments in for-sale housing | N/A | | |

(1) On June 23, 2020, Tricon acquired the remaining ownership interests of 50% and 75% in The James and The Shops of Summerhill, respectively. As a result, these investees ceased to be accounted for as joint operations, and the Company began to consolidate these subsidiaries on a prospective basis.

These financial reporting changes are material to the Company and have been applied on a prospective basis in accordance with the relevant guidance of IFRS 10 and, as such, the comparative period presentation reflects Investment Entity Accounting as previously reported.

Significant estimates

Income taxes

The determination of the Company's income and other tax liabilities requires interpretation of complex laws and regulations often involving multiple jurisdictions. Significant estimates are required in determining the Company's consolidated income tax provision. There are many transactions and calculations for which the ultimate tax determination is uncertain. The Company recognizes liabilities for anticipated tax audit issues based on estimates of whether additional taxes will be due. Where the final tax outcome of these matters is different from the amounts that were initially recorded, such differences will impact the current tax and deferred tax provisions. Furthermore, deferred income tax balances are recorded using enacted or substantively enacted future income tax rates. Changes in enacted income tax amounts that may differ significantly from estimates recorded in deferred tax balances.

Valuation of rental properties

Fair value is determined using independent external valuations prepared by management's specialists or detailed internal valuations prepared by management using market-based assumptions, each in accordance with recognized valuation techniques as set out in Note 6 to the consolidated financial statements. Significant estimates used in determining the fair value of the Company's rental properties include estimating, among other things, future stabilized net operating income, capitalization rates, discount rates, and other future cash flows applicable to rental properties (all considered Level 3 inputs) as well as market comparables based on recent transaction prices. A change to any one of these inputs could significantly alter the fair value of a rental property. In addition, the novel coronavirus ("COVID-19") pandemic and related market and economic uncertainty that occurred in 2020 had a significant impact on estimates used in the valuation of the rental properties and this impact may continue into 2021. Management will continue to monitor the situation and its impact on the Company.

Fair value and impairment of financial instruments

Certain financial instruments are recorded in the Company's consolidated balance sheets at values that are representative of or approximate fair value.

The fair values of the Company's investments in for-sale housing are determined using the valuation methodologies described in Note 9 to the consolidated financial statements. By their nature, these valuation techniques require the use of assumptions that are mainly based on market conditions existing at the end of each reporting period. Changes in the underlying assumptions could materially impact the determination of the fair value of a financial instrument. Imprecision in determining fair value using valuation techniques may affect the investment income recognized in a particular period. Any significant changes to the inputs and assumptions owing to the COVID-19 pandemic as discussed above could further impact the valuation of the for-sale housing investments in future periods.

Fair value of incentive plans

Management is required to make certain assumptions and to estimate future financial performance in order to estimate the fair value of incentive plans at each consolidated balance sheet date. Significant estimates and assumptions relating to such incentive plans are disclosed in Notes 3 and 30 to the consolidated financial statements. The LTIP requires management to estimate future non-IFRS earnings measures, namely future performance fees relative to each Investment Vehicle. Future non-IFRS measures are estimated based on current projections, and are updated at least annually, taking into account actual performance since inception.

Goodwill impairment

Assessment of impairment is based on management's judgment of whether there are internal and external factors that would indicate that an asset or cash-generating unit ("CGU") is impaired. The determination of the Company's goodwill impairment involves management's significant estimates and assumptions with respect to future cash flows, growth rates and discount rates of the underlying CGU. The risk premiums expected by market participants related to uncertainties about the industry and assumptions relating to future cash flows may differ, depending on economic conditions and other events. Changes in any of these underlying assumptions could materially affect the assessment of the recoverable value of a CGU.

Due to Affiliate

In connection with the Due to Affiliate transaction, the Company made certain key assumptions about the structure, cash flow and terms of the issued instruments. In addition, management was required to make significant estimates in determining the initial recognition and measurement of the Due to Affiliate and related derivative instruments.

Significant judgments

Acquisition of rental properties

The Company's accounting policies relating to rental properties are described in Note 3 to the consolidated financial statements. In applying these policies, judgment is exercised in determining whether certain costs are additions to the carrying amount of a rental property and whether properties acquired are considered to be asset acquisitions or business combinations. Should the purchase meet the criteria of a business combination then transaction costs such as appraisal and legal fees are expensed immediately and included in the consolidated statements of comprehensive income. If the purchase is an asset acquisition, transaction costs form part of the purchase price and earnings are not immediately affected.

Basis of consolidation

The consolidated financial statements of the Company include the accounts of Tricon and its wholly-owned subsidiaries, as well as entities over which the Company exercises control on a basis other than majority ownership of voting interests within the scope of IFRS 10. Judgment is applied in determining if an entity meets the criteria of control as defined in the accounting standard.

Investments in joint ventures and joint arrangements

The Company makes judgments in determining the appropriate accounting for investments in other entities. These judgments include determining the significant relevant activities and assessing the level of influence Tricon has over the activities through contractual arrangements. In addition, the Company also determines whether Tricon's rights and obligations are directly related to the assets and liabilities of the arrangement or to the net assets of the joint arrangement.

CGU determination for goodwill impairment assessment

The determination of CGUs is based on management's judgment and is an assessment of the smallest group of assets that generate cash inflows independently of other assets. Factors considered include whether an active market exists for the output produced by the asset or group of assets as well as how management monitors and makes decisions about the Company's operations.

8.3 Controls and procedures

Pursuant to National Instrument 52-109 – *Certification of Disclosure in Issuers' Annual and Interim Filings* released by the Canadian Securities Administrators, the Company's CEO and CFO have evaluated the design and operating effectiveness of the Company's disclosure controls and procedures and the Company's internal controls over financial reporting for the year ended December 31, 2020. The CEO and CFO did not identify any material weaknesses in the Company's system of internal controls over financial reporting.

During the year ended December 31, 2020, there were no changes to policies, procedures and processes that comprise the system of internal controls over financial reporting that may have affected, or are reasonably likely to materially affect, the Company's internal controls over financial reporting. Such controls and procedures are subject to continuous review and changes to such controls and procedures may require management resources and systems in the future.

Management does not expect that the disclosure controls or internal controls over financial reporting of the Company will prevent or detect all errors and all fraud or will be effective under all potential future conditions. A control system is subject to inherent limitations and, no matter how well designed and operated, can provide only reasonable, not absolute, assurance that the control system's objectives will be met. Further, the design of a control system must reflect that there are resource constraints, and the benefits of controls must be considered relative to their costs. Inherent limitations include the realities that judgments in decisionmaking can be faulty, and that breakdowns can occur because of simple errors or mistakes. Controls can also be circumvented by individual acts of some persons, by collusion of two or more people or by management override of the controls. Due to the inherent limitations in a cost-effective control system, misstatements due to error or fraud may occur and not be detected. The design of any control system is also based in part upon certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential conditions. Projections of any evaluations of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

8.4 Transactions with related parties

Senior management of the Company own units, directly or indirectly, in the various Tricon private funds, as well as common shares and debentures of the Company.

Refer to the related party transactions and balances note in the consolidated financial statements for further details concerning the Company's transactions with related parties.

8.5 Dividends

On March 2, 2021, the Board of Directors of the Company declared a dividend of seven cents per common share in Canadian dollars payable on or after April 15, 2021 to shareholders of record on March 31, 2021.

8.6 Compensation incentive plans

The Company's annual compensation incentive plans include an annual incentive plan ("AIP") and a long-term incentive plan ("LTIP").

The Company's AIP provides for an aggregate bonus pool based on the sum of all employees' individual AIP targets. The portion of the pool attributable to senior executive management is market-benchmarked and subject to an adjustment factor, as approved by the Board, of between 50% and 150%, based on achievement of Company performance objectives determined by the Board at the beginning of each year. The final pool is then allocated among employees based on individual and collective performance. AIP awards will be made in cash and equity-based grants, with the proportion of equity-based awards being correlated to the seniority of an individual's role within the Company.

LTIP expense is generated from two sources: (i) 50% of the Company's share of performance fees or carried interest from Investment Vehicles, paid in cash when received; and (ii) 15% of income earned from THP1 US (a for-sale housing Investment Vehicle), payable in deferred share units which vest in equal tranches over a three-year period (previously a five-year period) pursuant to the LTIP. Amounts under the LTIP are allocated among employees in accordance with the plan.

Complete details concerning the Company's compensation plans are set out in the Company's most recent Management Information Circular, available on SEDAR at www.sedar.com and on the Company's website at www.triconresidential.com.

8.7 Risk definition and management

There are certain risks inherent in the Company's activities and those of its investees, including the ones described below, which may impact the Company's financial and operating performance, the value of its investments and the value of its securities. The risks described below are not the only ones facing the Company and holders of common shares. Additional risks not currently known to us or that we currently consider to be immaterial may also affect our activities.

General economic conditions

The success of our business is highly dependent upon conditions in the Canadian and United States real estate markets (and in particular the residential sector) and economic conditions throughout North America that are outside our control and difficult to predict. Factors such as interest rates, housing prices, availability of credit, inflation rates, energy prices, economic uncertainty, changes in laws (including laws relating to taxation), trade barriers, currency exchange rates and controls, and national and international political circumstances (including wars, terrorist acts or security operations) could have a material negative impact on our financial performance and the value of our investments.

Unpredictable or unstable market conditions, adverse economic conditions, or volatility in the capital markets may result in reduced opportunities to find suitable risk-adjusted investments to deploy capital, may reduce the market value of our assets, and may make it more difficult for the Company and its investment vehicles to exit and realize value from existing real estate holdings, any of which could materially adversely affect our revenues, the value of our investments, and our ability to raise and deploy new capital and sustain our profitability and growth.

Real estate industry conditions

The residential real estate industry is cyclical and is significantly affected by changes in general and local economic and industry conditions, such as employment levels, availability of financing for homebuyers, interest rates, consumer confidence, levels of new and existing homes for sale, demographic trends and housing demand. In addition, an oversupply of new homes or alternatives to new homes, such as resale homes, including homes held for sale by investors and speculators, foreclosed homes and rental properties, may reduce the ability to rent or sell residential properties, depress prices and reduce margins from the rental and sale of residential properties. Conversely, if property prices in target markets increase at a rate faster than rents, this could result in downward pressure on gross rental yields and impact the ability to make acquisitions. Any of these factors could negatively impact the value of the Company's financial condition and performance.

Builders, developers and renovators are also subject to risks related to the availability and cost of materials and labour, and adverse weather conditions that can cause delays in construction schedules and cost overruns. Furthermore, the market value of undeveloped land, buildable lots and housing inventories can fluctuate significantly as a result of changing economic and real estate market conditions and may result in impairment charges. If there are significant adverse changes in economic or real estate market conditions, residential properties may have to be sold at a loss, rented at less than expected rates, or held longer than planned. These circumstances can result in losses in a poorly performing investment or market. If market conditions deteriorate, the Company's financial condition and performance may be adversely impacted.

Portfolio concentration

Although our real estate holdings span numerous markets across North America, real estate is a local business, and our revenues are directly and indirectly derived from residential real estate located in our primary geographic markets. A prolonged downturn in the economies of these markets, or the impact that a downturn in the overall national economies of the United States or Canada may have upon these markets, could negatively impact the value of our financial condition and performance.

Furthermore, because we are focused on residential real estate (as compared to a more diversified real estate portfolio), a decrease in demand specifically for residential real estate could adversely affect the value of our investments and our financial condition and performance.

Competition

The residential real estate business is competitive and each segment of our business is subject to competition in varying degrees. We compete on the basis of a number of factors, including, but not limited to: the quality of our employees, transaction execution, innovation, reputation and above all, our rental operations. Numerous developers, managers and owners of properties compete with the Company in seeking attractive residents and home purchasers, in the efficiency of their operations, and in the quality of their service offering. In addition, there is significant competition for suitable real property investments, with other operators and investors seeking similar assets to those targeted by the Company. A number of these investors may have greater financial resources than those of the Company, or operate without the same investment or operating restrictions. An increase in competition for real property investments may increase purchase prices, diminish the number of suitable investments available, and reduce the ability to achieve optimal portfolio size or expected yields, which could impact the Company's investments and financial performance.

Furthermore, we compete in pursuit of investor capital to be invested in our securities and investment vehicles. Competition for investor capital, in particular, is intense and investors are increasingly seeking to manage their own assets or reduce their management fees. Further, our competitors may have certain competitive advantages, including greater financial, technical, marketing and other resources, more personnel, less onerous regulatory requirements, or a lower cost of capital, and access to funding sources or other resources that are not available to us.

These pressures, or an increase in competition, could impact our revenues and operating margins and negatively affect our overall financial condition.

The residential development, homebuilding, renovation and rental industries are themselves highly competitive. Residential developers, homebuilders, renovators and operators compete not only for homebuyers and/or tenants on the basis of price and product offering, but also for desirable properties, building materials, labour and capital. Competitive conditions in the industry could result in: difficulty in acquiring suitable properties at acceptable prices; increased selling or rental incentives; lower sales volumes and prices; higher vacancy; lower profit margins and development yields; impairments in the value of inventory and other assets; increased construction costs; and delays in construction. These factors may negatively impact the Company's financial condition and performance.

Investment pipeline

An important component of the Company's growth strategy is the ongoing availability of attractive real estate acquisition or investment opportunities. If we are not able to find sufficient residential real estate investments in a timely manner, our performance could be adversely affected. Furthermore, if we do not have sufficient investment opportunities, we may elect to limit our growth and reduce the rate at which we attract third-party capital, which could impact our growth plans and revenues. Finally, a scarcity of desirable investment opportunities may lead us to make investments with lower expected returns than those we have historically targeted. Any of these factors could negatively impact our financial condition and performance.

Liquidity risk

Residential real estate assets generally cannot be sold quickly, particularly if local market conditions are poor. As a result, the Company may not be able to acquire or sell assets promptly in response to economic or other conditions. This inability to promptly reallocate capital or exit the market in a timely manner could adversely affect the Company's financial condition and performance. Additionally, financial difficulties of other property owners resulting in distressed sales could depress real estate values in the markets in which we invest. These restrictions could reduce our ability to respond to changes in the performance of our portfolios and could adversely affect our financial condition and performance.

Transaction execution

Before making investments, we conduct extensive due diligence reviews that we deem reasonable and appropriate based on the facts and circumstances applicable to each asset. Our due diligence process includes in-depth reference checks of developers (where applicable), environmental audits, market analysis, site analysis, financial and construction cost analysis and legal review. When conducting due diligence, we may be required to evaluate important and complex business, financial, tax, accounting, environmental and legal issues. Outside consultants, legal advisors, accountants and investment banks may be involved in the due diligence process in varying degrees depending on the asset class and size of transaction. Nevertheless, when conducting due diligence, we rely on the resources available to us, including information provided by the developer or operating partner (where applicable) and, in some circumstances, third-party investigations. The due diligence investigation that we carry out with respect to any investment opportunity may not reveal or highlight all relevant facts that may be necessary or helpful in evaluating such opportunity. Moreover, such an investigation will not necessarily result in the investment being successful. Unknown factors or unforeseen risks may cause performance to fall short of expectations and may negatively impact our financial condition and performance.

Indebtedness and rising interest rates

The degree to which the Company is leveraged could have important consequences to the Company, including: (i) the Company's future ability to obtain additional financing for working capital, capital expenditures or other purposes may be limited; (ii) the Company may be unable to refinance indebtedness on terms acceptable to the Company or at all; (iii) a significant portion of the Company's cash flow could be dedicated to the payment of the principal of and interest on its indebtedness, thereby reducing funds available for future operations, capital expenditures and/or dividends on its common shares and increasing the risk of default on the Company's debt obligations; (iv) the Company may be negatively impacted by rising interest rates; and (v) the Company may be more vulnerable to economic downturns and be limited in its ability to withstand competitive pressures.

Moreover, rising interest rates, decreased availability of mortgage financing or of certain mortgage programs, higher down payment requirements or increased monthly mortgage costs may increase the cost of capital for the Company and may lead to reduced demand for new home sales and resales and mortgage loans, which could negatively impact our financial condition and performance.

Benchmark interest rate reform risk

Regulators in the United Kingdom and elsewhere have recommended and are seeking to implement broad changes to benchmark interest rates, such as LIBOR. It is expected that a transition away from the widespread use of LIBOR and such other benchmark rates to alternative reference rates and other potential interest rate benchmark reforms will occur over the course of the next few years. For example, the United Kingdom's Financial Conduct Authority has announced that LIBOR is to be phased out by the end of 2021. As a result, there is near-term uncertainty about how the currently dominant benchmarks will be phased out, the speed at which modified or replacement benchmarks will take their place, the acceptance of such alternatives, and the ultimate effect any such changes may have on markets for financial instruments and the access to and cost of debt. Abandonment of or modifications to such benchmarks could have adverse impacts on the Company's newly-issued financial instruments and existing financial instruments that reference such benchmarks. While some of the Company's debt instruments may contemplate a scenario where LIBOR or another applicable benchmark is no longer available by providing for an alternative rate-setting methodology, not all of our instruments may have such provisions, and the impact of any such alternative methodologies is unclear. Abandonment of or modifications to LIBOR or another relevant benchmark could lead to market instability, and could adversely impact the pricing, liquidity, value or return of the Company's debt instruments, affect the Company's ability to meet its payment obligations thereunder, require extensive changes to documentation, result in disputes, or cause the Company to incur additional costs. Depending on these and several other factors, many of which are beyond the Company's control, the Company's business, financial condition and results of operations could be materially adversely impacted by any such market transition or reform of benchmark interest rates. It remains uncertain how such changes would be implemented and the effects such changes may have on the Company, its business, financial condition and results of operations, its investees and financial markets generally. The Company continues to actively monitor these potential changes and to include alternative rate-setting methodologies in its newly-issued debt instruments.

Sustaining growth

Our continuing growth has caused, and if it continues will continue to cause, significant demands on our legal, accounting and operational infrastructure, and increased expenses. In addition, we are required to continuously develop our systems and infrastructure in response to the increasing sophistication of the residential real estate investment industry, the investment management market, and legal, accounting and regulatory developments.

Our future growth will depend, among other things, on our ability to maintain an operating platform and management systems sufficient to address our growth, and will require us to incur additional expenses and to commit additional senior management and operational resources. There can be no assurance that we will be able to manage our expanding operations effectively or that we will be able to continue to grow, and any failure to do so could adversely affect our ability to generate revenue and control our expenses.

Insurance

We have various types of insurance, including errors and omissions insurance and general commercial liability insurance, as well as relevant insurance obtained to protect the value of our assets. The adequacy of insurance coverage is evaluated on an ongoing basis, including the cost relative to the benefits. However, there can be no assurance that potential claims or losses will not exceed the limits, or fall outside the scope, of available insurance coverage or that any claim or claims will be ultimately satisfied by an insurer. A loss or judgment in excess of available insurance or in respect of which insurance is not available could have a material adverse effect on our financial condition and the value of our assets. There can be no assurance that insurance coverage on favourable economic terms will continue to be available in the future.

Environmental risk

Our real estate portfolios are subject to various Canadian and United States federal, provincial, state and municipal laws relating to environmental matters. These laws could hold developers or property owners liable for the costs of removal and remediation of certain hazardous substances or wastes released or deposited on properties or disposed of at other locations. The failure to remove or remediate such substances, if any, could adversely affect the developer's or owner's ability to sell the properties or to borrow using real estate as collateral, and could potentially result in claims or other proceedings. We are not aware of any material non-compliance with environmental laws in respect of our assets or those in which our investment vehicles invest. We are also not aware of any material pending or threatened claims relating to environmental conditions, in connection with any of the residential real estate in which we or our investment vehicles invest. Environmental laws and regulations can change rapidly and may impose more stringent environmental laws and regulations, or compliance with more stringent legislative frameworks, could have an adverse effect on our financial condition and performance.

Disease outbreak risks

A local, regional, national or international outbreak of a contagious disease, including the current COVID-19 pandemic, Middle East Respiratory Syndrome, Severe Acute Respiratory Syndrome, H1N1 influenza virus, avian flu, or any other similar illness could result in: a general or acute decline in economic activity in the regions the Company holds assets and conducts business, a decrease in the willingness or ability of the general population to travel, staff and labour shortages, diversion of management attention, reduced tenant and customer traffic and demand, reduced employment and financial wherewithal of our residents, mobility restrictions and other quarantine measures, supply shortages, increased government regulation (including regulations impacting property operations, limiting rent increases or limiting eviction actions), and the quarantine or contamination of one or more of the Company's rental properties. These and other related consequences could negatively impact: rental revenue, the ability to collect rent and enforce leases, fee income and other revenue sources, rental rates and for-sale housing prices, property values, bad debt expense, liquidity, the Company's ability to grow and expand its portfolios, development timelines, project cash flows, compliance with debt covenants and default risk, and the Company's ability to achieve its financial and strategic goals and targets. In addition, the Company's response to such a crisis may be made in the context of economic and epidemiological uncertainty and changing legal regulations which may increase the risk of legal or regulatory liability to the Company.

Climate change risks

To the extent that significant changes in the climate occur in areas where our properties are located, increasingly extreme weather, changes in precipitation, flooding, wildfires, hurricanes and rising temperatures in those areas may result in physical damage to, or a decrease in demand for, properties located in those areas or affected by those conditions. Should the impact of climate change be material in nature, including significant property damage to or destruction of our properties, or occur for lengthy periods of time, our financial condition and performance may be adversely affected. Climate change, to the extent it causes changes in weather patterns, could also increase the cost of property insurance and utilities at our properties and impact demographic trends in ways that result in decreased demand for our properties. In addition, changes in federal, provincial, state and local laws based on concerns about climate change could result in reduced operational flexibility and/or increased expenses on our existing properties (for example, to improve their energy efficiency and/or resistance to inclement weather or to reduce their carbon footprint) without a corresponding increase in revenue, which could adversely affect our performance.

Conflicts of interest

Some of the parties in which and with which we currently invest may have competing interests in the markets in which Tricon invests. While the Company takes precautions and negotiates contractual restrictions in definitive legal documentation in order to avoid such conflicts, conflicts of interest may nonetheless arise and may have an adverse effect on the Company's financial condition and performance.

Certain of the directors and officers of the Company may also serve as directors and/or officers of other companies and consequently the possibility exists for such directors and officers to be in a position of conflict. Any decision made by any such director or officer involving the Company is to be made in accordance with their duties and obligations to deal fairly and in good faith with a view to the best interests of the Company, but there can be no assurance that a conflict of interest will not have an adverse effect on the Company or its financial condition.

Management team

The Company's executive officers and other senior management have a significant role in our success and oversee the execution of our strategy. Our continued ability to respond promptly to opportunities and challenges as they arise depends on cooperation across our organization and our team-oriented management structure, which benefits greatly from management continuity. Our ability to retain our management group or attract suitable replacements, should any members of the management group leave, is dependent on, among other things, the competitive nature of the employment market and the career opportunities that we can offer. Ensuring that we continue to pay market compensation in order to retain key professionals may lead to increasing costs. We have experienced departures of key professionals in the past and may do so in the future, and we cannot predict the impact that any such departures will have on our ability to achieve our objectives. Competition for the best people is intense and the loss of services from key members of the management group or a limitation in their availability could adversely impact our financial performance. Furthermore, such a loss could be negatively perceived in the capital markets.

Government regulation

The Company's activities are subject to numerous regulations across various jurisdictions in North America. Changes in legislation and regulation could result in increased costs and increased risk of non-compliance, which could adversely affect the Company's financial condition and performance.

Certain jurisdictions have enacted residential tenancy legislation which imposes, among other things, rent control guidelines that limit the ability to raise rental rates at residential properties. In addition to limiting the ability to raise rental rates, residential tenancy legislation in some jurisdictions prescribes certain limitations on terminations of residential tenancies. Certain jurisdictions have enacted rent control regulations and/or eviction moratoria in response to the COVID-19 pandemic. Any limits on the Company's ability to raise rental rates at its properties, or to terminate defaulting tenancies, may adversely affect our financial condition and performance.

Acquisitions and development projects undertaken by the Company may require zoning and other approvals from local government agencies. The process of obtaining such approvals may take months or years, and there can be no assurance that the necessary approvals for any particular project will be obtained. Holding costs accrue while regulatory approvals are being sought, and delays could negatively impact performance.

Construction industry risks

Our success is very often dependent on stability in the construction industry. This industry may from time to time experience significant difficulties in the supply of materials and services, including with respect to: shortages of qualified tradespeople; labour disputes; shortages of building materials; unforeseen environmental and engineering problems; and increases in the cost of certain materials. When any of these difficulties occur, it may cause delays and increase anticipated costs, which could adversely affect the Company's financial condition and performance.

Taxation risks

We endeavour to structure our holdings and operations to be efficient under the prevailing U.S. and Canadian tax frameworks. Changes in tax legislation or policy could adversely affect the after-tax return we can earn on our investments and activities, capital available for growth and investment (including from our institutional investors), and the willingness of investors to acquire our securities or invest in our investment vehicles. A number of other factors may increase our effective tax rates, which would have a negative impact on our net income. These include, but are not limited to, changes in the valuation of our deferred tax assets and liabilities, and any reassessment of taxes by a taxation authority.

Furthermore, tax changes (such as rising property and franchise tax rates) could impact the efficiency of our operations and could also impact the overall economic conditions relevant to the success of our business. For example, in the United States, the significant expenses of owning a home, including mortgage interest and state and property tax, are generally deductible for tax purposes (subject to various limitations). Any changes to modify these benefits could increase the after-tax cost of owning a new home, which could adversely impact housing demand and/or sales prices.

Cybersecurity risk

Cyberattacks are increasingly common and sophisticated, leading to unauthorized access and fraudulent activities threatening the confidentiality, integrity or availability of our information resources. Cyberattacks could cause disruption of operations, data corruption or theft of confidential information. The consequences of cybersecurity risk may include remediation costs, additional regulatory scrutiny, litigation and reputational damage, any of which could negatively impact our financial condition and performance. We have security procedures and measures in place to protect our systems and information from cyberattacks and we monitor our systems for malicious threats in an effort to ensure we maintain high privacy and security standards.

Lease renewal and turnover risk

If a tenant decides to vacate a rental property, whether as a result of deciding not to renew their lease or by vacating prior to the expiry of the lease, the Company may not be able to re-let that property in a short amount of time or at all. Additionally, even if we are successful in renewing a lease or re-letting a property, the terms of the renewal or re-letting may be less favourable than the original terms.

The ability to rent residential properties is affected by many factors, including changes in general economic conditions (such as the availability and cost of mortgage funds, vacancy rates, the availability of suitable potential tenants and the job market for prospective tenants), local conditions (such as an oversupply of space or a reduction in demand for real estate in the area), government regulations, changing demographics or social preferences, competition from other available properties, and various other factors.

If the Company is unable to promptly renew leases or re-let properties, or if the rental rates upon renewal or re-letting are significantly lower than expected rates, our financial condition and performance may be negatively impacted.

Furthermore, if a significant number of tenants are unable to meet their obligations under their leases or if a significant number of properties become vacant and cannot be re-let on economically favourable terms, the Company may not generate revenues sufficient to meet operating expenses, including debt service and capital expenditures.

Resident default

The success of the Company's rental operations depends in large part upon the ability to attract and retain qualified residents. This will depend, in turn, upon the ability to screen applicants, identify qualified residents, and avoid residents who may default. The Company relies on information supplied by prospective residents in their rental applications to make leasing decisions, and this information may not be accurate. The Company may not successfully screen applicants, and as a result, may rent to residents who default on leases or fail to comply with the terms of the lease or applicable homeowners' association regulations, which may negatively affect financial performance, reputation, and the quality and value of our properties.

In the event of a resident default or bankruptcy, we may experience delays in enforcing our rights as landlord and obtaining possession of the premises and may incur legal, maintenance and other costs in protecting the value of our assets. In addition, we will incur turnover costs associated with re-letting the property such as marketing and brokerage commissions, will not collect revenue while the property sits vacant, and may be unable to re-let the property at the rental rate previously received.

Reliance on vendors

The Company relies on local vendors and service providers, including house renovation professionals, maintenance providers, leasing agents, and property management companies in situations where it is cost-effective to do so or if our internal staff is unable to perform these functions. We generally do not have exclusive or long-term contractual relationships with any of these providers, and can provide no assurance that we will have uninterrupted or unlimited access to their services. Furthermore, selecting, managing and supervising these service providers requires significant management resources and expertise. Poor performance by service providers, especially those who interact with residents at our properties, will reflect poorly on the Company, could significantly damage our reputation among desirable residents and potentially impact financial performance. Moreover, notwithstanding efforts to implement and enforce strong policies and practices regarding service providers, we may not successfully detect and prevent fraud, incompetence or theft by service providers, which could expose us to liability or responsibility for associated damages and cause us to incur fines or penalties. In addition, any delay in identifying a service providers would require the Company to seek new vendors or providers, which could create delays and adversely affect financial and operating results.

Increased expenses

The failure to maintain stable or increasing average monthly rental rates combined with acceptable occupancy levels would likely have a material adverse effect on our business, cash flows, financial condition and results of operations. Certain significant expenditures, including property taxes, maintenance costs, mortgage payments, insurance costs and related charges, must be made throughout the period of ownership of real property regardless of whether a property is producing any income. There is a risk that property taxes may be increased as a result of revaluations of properties and their adherent tax rates. In some instances, enhancements to properties may result in significant increases in property assessments following a revaluation. Additionally, utility expenses have been subject to considerable price fluctuations over the past several years and any significant increase in these costs that we cannot charge back to our residents may have an adverse effect on our business, cash flows, financial condition and results of operations.

Substitutions for rental properties

Demand for rental properties is impacted by and inversely related to the relative cost of home ownership. The cost of home ownership depends upon, among other things, interest rates offered by financial institutions on mortgages and similar home financing transactions. If the interest rates offered by financial institutions for home ownership financing remain low or fail to rise, demand for rental properties may be adversely affected.

An economic downturn may also impact job markets and the ability of tenants to afford the rents associated with certain rental properties, which may result in increased demand for lower-cost rental options. Such a reduction in demand may have an adverse effect on rental revenues.

Tenant relief laws

As the landlord of numerous properties, the Company is involved from time to time in evicting residents who are not paying their rent or who are otherwise in material violation of the terms of their lease. Eviction activities impose legal and managerial expenses that increase costs and expose us to potential negative publicity. The eviction process is typically subject to legal barriers, mandatory "cure" policies, internal policies and procedures and other sources of expense and delay, each of which may delay our ability to gain possession and stabilize the property. Additionally, state, provincial and local landlord-tenant laws may impose legal duties to assist residents in relocating to new housing, or restrict the landlord's ability to remove the resident on a timely basis or to recover certain costs or charge residents for damage residents cause to the landlord's premises. Because such laws vary by state, province and locality, the Company must be familiar with and take all appropriate steps to comply with all applicable landlord-tenant laws, and needs to incur supervisory and legal expenses to ensure such compliance. To the extent that we do not comply with state, provincial or local laws, we may be subjected to civil litigation filed by individuals, in class actions or actions by state or local law enforcement and the Company's reputation and financial results may suffer. The Company may be required to pay adversaries' litigation fees and expenses if judgment is entered against us in such litigation or if we settle such litigation.

Title risk

The Company's acquisition of single-family rental homes is often completed through a title company with an owner's title insurance policy being obtained. However, U.S. distressed single-family homes may also be acquired through trustee auctions. Although the Company conducts due diligence and employs a title company to review title on target housing assets prior to purchasing such homes, title on the homes purchased through foreclosure sales and auctions is occasionally only assumed weeks after the purchase. Furthermore, an owner's title insurance policy is not available to protect against the inherent title risk arising through the foreclosure auction process. In the event that the Company fails to independently and properly assess a title risk or fails to assume one or more homes because of such failed analysis, it may not achieve its expected financial performance.

Homeowners' association issues

A number of our properties are located within homeowners associations ("HOAs"), which are private entities that regulate the activities of and levy assessments on properties in a residential subdivision. HOAs in which we own properties may have or enact onerous or arbitrary rules that restrict our ability to renovate, market or lease our properties or require us to renovate or maintain such properties at standards or costs that are in excess of our planned operating budgets. Such rules may include requirements for landscaping, limitations on signage promoting a property for lease or sale, or the use of specific construction materials in renovations. Some HOAs also impose limits on the number of property owners who may rent their homes, which if met or exceeded, would cause us to incur additional costs to resell properties within the HOA and may also result in opportunity costs of lost rental income. Many HOAs impose restrictions on the conduct of occupants of homes and the use of common areas, and we may have residents who violate HOA rules and for which we may be liable as the properties or may block our access to HOA records, initiate litigation, restrict our ability to sell our properties, impose assessments, or arbitrarily change the HOA rules. We may be unaware of or unable to review or comply with HOA rules before purchasing a property and any such excessively restrictive or arbitrary regulations may cause us to sell such property at a loss, prevent us from renting such property, or otherwise reduce our cash flow from such property, which would have an adverse effect on our financial condition and performance.

Government subsidies

Some of our rental income is derived from government subsidized rental support programs, such as the Section 8 program operated by the U.S. Department of Housing and Urban Development. A reduction or elimination of government funding of such programs could result in higher rental turnover and downward pressure on rental rates, which could negatively impact our financial performance.

Guarantees of project debt

The Company may agree to provide financial assistance to the subsidiary entities through which it carries on its activities. Such financial assistance may include the provision of payment guarantees to a project entity's lenders of acquisition financing, construction debt or long-term financing, and the provision of construction completion guarantees. Such guarantees may be joint or several with other partners in a particular investment. The Company's and its partners' guarantees of project-level obligations may not be in proportion to their respective investments in the project entity. The provision of such guarantees may reduce the Company's capacity to borrow funds under its separate credit facilities, which may impact its ability to finance its operations. If such guarantees are called upon for payment or performance, they may have a negative impact on the Company's cash position and financial performance. If the Company provides a joint guarantee with an investment partner, a default by the partner in its payment or performance obligation under the guarantee could cause the Company to pay a disproportionate amount in satisfaction of the guarantee, which may have a negative impact on the Company cash position and financial performance.

Operational and credit risks

On a strategic and selective basis, we and our for-sale housing investment vehicles provide financing to develop properties. The residential real estate development business involves significant risks that could adversely affect performance, including: the developer may not be able to complete construction on schedule or within budget, resulting in increased debt service expense and construction costs and delays in selling the properties; the developer may not be able to obtain, or may experience delays in obtaining, all necessary zoning, land-use, building, occupancy and other governmental permits and authorizations for the properties; the developer may not be able to sell properties on favourable terms or at all; construction costs, total investment amounts and the Company's or investment vehicle's share of remaining funding may exceed our estimates; and projects may not be completed and delivered as planned.

Our for-sale housing investments are made through the financing of local developers, including Johnson, and, consequently, we rely to a great extent on those developers to successfully manage their development projects. Furthermore, given the Company's majority interest in Johnson, we rely on Johnson's ability to execute on portions of our for-sale housing business strategy. Investments in partnerships, joint ventures or other entities may involve risks not present were a third party not involved, including the possibility that the development partners might become bankrupt or otherwise fail to fund their share of required capital contributions. Additionally, the development partners might at any time have economic or other business interests or goals which are inconsistent with our business interests or goals. In addition, we do not have sole control of certain important decisions relating to these development properties, including decisions relating to: the sale of the development properties; refinancing; timing and amount of distributions of cash from such development properties; and capital improvements. Any of these factors could negatively impact the value of our investments and our financial condition and performance.

Long investment periods

The investment horizons in our for-sale housing assets are relatively long and these extended timelines increase the risk that circumstances will arise which delay investment realization, and that markets may deteriorate between the time of our initial investment and our exit. This may be the result of many factors that present themselves over the duration of an investment, including local and overall market and economic conditions, increasing competition over time, market value fluctuation and changing interest rates. Delays or market deterioration over time could have an adverse effect on the returns from our investments, our fee revenue, and our financial condition and performance.

Formation of future investment vehicles

The ability to raise capital for any future investment vehicles remains subject to various conditions which Tricon cannot control, including the negotiation and execution of definitive legal documentation and commitments made by third-party investors. There can be no assurance that any capital will be raised through future investment vehicles or that any future warehoused investments of the Company will be acquired by any other future vehicles. A failure to raise sufficient capital through other investment vehicles could impair our future revenues and growth.

Structure of future investment vehicles

There can be no assurance that the manner in which our private funds and advisory revenues and/or investment income are calculated in respect of future investment vehicles will be the same as the active investment vehicles. Any such changes could result in the Company earning lesser fees from investment vehicles of the same nature and size as the active investment vehicles and could expose the Company's co-investments in such future investment vehicles to increased risk, including, but not limited to, the risk of reduced income (at comparable investment performance levels) and the increased risk of loss of capital of the Company.

Ongoing investment performance

We believe that our ongoing investment performance is one of the most important factors for the success and growth of Private Funds and Advisory activities. Poor investment performance could impair our ability to raise future private capital, which could impact our ability to earn private funds and advisory revenue. In addition, our ability to earn performance fees is directly related to our investment performance and therefore poor investment performance may cause us to earn less or no performance fees.

Investment vehicle governance

The governing agreements for certain active investment vehicles provide that the general partner or manager of the investment vehicle may be removed by the investors in certain prescribed circumstances, including in some cases (and with the approval of a prescribed number of investors) without cause. These agreements may not provide for termination payments to the general partner or manager in the event of removal without cause. The removal of the general partner or the manager of an active investment vehicle prior to the termination of such investment vehicle could materially adversely affect the reputation of Tricon, reduce our private funds and advisory revenue, and have a negative impact on our financial condition and performance.

Capital commitment

The third-party investors in Tricon's investment vehicles comprise a relatively small group of reputable, primarily institutional, investors. To date, each of these investors has met its commitments on called capital and we have received no indications that any investor will be unable to meet its capital commitments in the future. While our experience with our investors suggests that commitments will be honoured, and notwithstanding the adverse consequences to a defaulting investor under the terms of the applicable investment vehicle, no assurances can be given that an investor will meet its entire commitment over the life of an investment vehicle. A failure by one or more investors to satisfy a drawdown request could impair an investment vehicle's ability to fully finance its investment, which could have a material adverse effect on the performance and value of that investment, which in turn could negatively impact the Company's financial condition and performance.

Stock exchange prices

The market price of our common shares could fluctuate significantly as a result of many factors, including the following:

- · economic and stock market conditions generally and specifically as they may impact participants in the real estate industry;
- our earnings and results of operations and other developments affecting our business;
- · changes in financial estimates and recommendations by securities analysts following our common shares;
- · earnings and other announcements by, and changes in market evaluations of, participants in the real estate industry;
- · changes in business or regulatory conditions affecting participants in the real estate industry;
- addition or departure of the Company's executive officers and other key personnel;
- · sales or perceived sales of additional common shares; and
- trading volume of the common shares.

In addition, the financial markets may experience significant price and volume fluctuations that affect the market prices of equity securities of companies and that are unrelated to the operating performance, underlying asset value or prospects of such companies. Accordingly, the market price of our common shares may decline even if our operating results or prospects have not changed. The value of the common shares is also subject to market fluctuations based upon factors which influence the Company's operations, such as legislative or regulatory developments, competition, technological change and global capital market activity. As well, certain institutional investors may base their investment decisions on consideration of the Company's environmental, social and governance practices and performance against such institutions' respective investment guidelines and criteria, and failure to meet such criteria may result in limited or no investment in the common shares by those institutions, which could adversely affect the trading price of the common shares.

Additional capital

The Company's ability to carry on its business generally, and in particular to take advantage of investment opportunities, may require it to raise additional capital. Additional capital may be sought through public or private debt or equity financings by Tricon or another Tricon entity and may result in dilution to or otherwise may have a negative effect on existing Tricon shareholders. Further, there can be no assurances that additional financing will be available to Tricon when required or desired by Tricon, on advantageous terms or at all, which may adversely affect Tricon's ability to carry on its business.

Dividends

Holders of common shares do not have a right to dividends on such shares unless declared by the Board of Directors. Although the Board has established a dividend policy authorizing the declaration and payment of dividends to holders of common shares on a quarterly basis, the declaration of dividends is at the discretion of the Board of Directors even if the Company has sufficient funds, net of its liabilities, to pay such dividends.

The Company may not declare or pay a dividend if there are reasonable grounds to believe that (i) the Company is, or would after the payment be, unable to pay its liabilities as they become due, or (ii) the realizable value of the Company's assets would thereby be less than the aggregate quantum of its liabilities. Liabilities of the Company will include those arising in the ordinary course of business and indebtedness.

Future sales and dilution

The Company's articles permit the issuance of an unlimited number of common shares, and shareholders have no pre-emptive rights in connection with such further issuances. The Board has the discretion to determine the price and the terms of issue of further issuances of common shares and securities convertible into common shares. Any future issuances of common shares could be dilutive to shareholder interests at the time of issuance.

Holding company

Tricon Residential Inc. is a holding company and a substantial portion of its assets are the equity interests in its subsidiaries. As a result, investors are subject to the risks attributable to its subsidiaries. As a holding company, the Company conducts substantially all of its business and makes its investments through its subsidiaries, which generate substantially all of its revenues. Consequently, the Company's performance and growth are dependent on the earnings of its subsidiaries and the distribution of those earnings to the Company. The ability of these entities to pay distributions will depend on their operating results and may be subject to applicable laws and regulations and to contractual restrictions contained in the instruments governing their debt. In the event of a bankruptcy, liquidation or reorganization of any of the Company's subsidiaries, holders of indebtedness and trade creditors will generally be entitled to payment of their claims from the assets of those subsidiaries before any assets are made available for distribution to the Company.

Financial reporting and other public company requirements

The Company is subject to reporting and other obligations under applicable Canadian securities laws and Toronto Stock Exchange rules, including National Instrument 52-109 – *Certification of Disclosure in Issuers' Annual and Interim Filings*. These reporting and other obligations place significant demands on Tricon's management, administrative, operational and accounting resources. Moreover, any failure to maintain effective internal controls could cause the Company to fail to meet its reporting obligations or result in material misstatements in its consolidated financial statements. If the Company cannot provide reliable financial reports or prevent fraud, its reputation and operating results could be materially harmed, which could also cause investors to lose confidence in the Company's reported financial information, and could result in a lower trading price of its common shares.

Management does not expect that Tricon's disclosure controls and procedures and internal controls over financial reporting will prevent all errors and all fraud. A control system, no matter how well designed and implemented, can provide only reasonable, not absolute, assurance that its objectives will be met. Further, the design of a control system must reflect the fact that there are resource constraints, and the benefits of controls must be considered relative to their costs. Due to the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that all control issues within a company are detected. The inherent limitations include the realities that judgments in decision-making can be faulty, and that breakdowns can occur because of simple errors or mistakes. Controls can also be circumvented by individual acts of some persons, by collusion of two or more people or by management override of the controls. Due to the inherent limitations in a cost-effective control system, misstatements due to error or fraud may occur and not be detected in a timely manner or at all.

9. HISTORICAL FINANCIAL INFORMATION

The following table shows selected IFRS measures for the past eight quarters.

Effective January 1, 2020, the Company commenced consolidation of the financial statements of single-family rental and multi-family rental entities that are considered controlled subsidiaries. On the date of transition, the Company applied the requirements of IFRS 3 to all subsidiaries that were previously measured at fair value through profit or loss. As the requirements of IFRS 3 are applied prospectively, the IFRS measures below for all quarters prior to January 1, 2020 have not been recast and are presented under investment entity accounting in accordance with IFRS 10.

| For the three months ended | | | | |
|--|----------------------|-----------------------|------------------|-------------------|
| (in thousands of U.S. dollars, except per share amounts which are in U.S. dollars) | December 31, 2020 | September 30, 2020 | June 30, 2020 | March 31, 2020 |
| Financial statement results | | | | |
| Net operating income from rental properties | \$ 79,323 | \$ 77,867 | \$ 77,000 | \$ 74,459 |
| Total revenue | 132,322 | 128,934 | 127,034 | 123,987 |
| Net income (loss) | 81,478 | 58,099 | 17,341 | (40,505) |
| Basic earnings (loss) per share | 0.41 | 0.30 | 0.09 | (0.21) |
| Diluted earnings (loss) per share | 0.39 | 0.23 | 0.09 | (0.21) |

| For the three months ended | | | | | | | | |
|--|----------------------|--------|-----------------------|--------|------------------|--------|-------------------|--------|
| (in thousands of U.S. dollars, except per share amounts which are in U.S. dollars) | December 31, 2019 | | September 30, 2019 | | June 30, 2019 | | March 31, 2019 | |
| Financial statement results | | | | | | | | |
| Total revenue | \$ | 11,716 | \$ | 11,323 | \$ | 9,367 | \$ | 7,489 |
| Net income | | 45,259 | | 32,457 | | 12,356 | | 24,063 |
| Basic earnings per share | | 0.23 | | 0.16 | | 0.08 | | 0.17 |
| Diluted earnings per share | | 0.22 | | 0.15 | | 0.04 | | 0.16 |

The following tables show selected IFRS measures for the past three years.

| (in thousands of U.S. dollars, except | De | December 31, | | December 31, | | December 31, | |
|--|-----|--------------|------|--------------|-----|--------------|--|
| per share amounts which are in U.S. dollars) | | 2020 | 2019 | | | 2018 | |
| Financial statement results | | | | | | | |
| Total revenue ⁽¹⁾ | \$ | 512,277 | \$ | 39,895 | \$ | 30,347 | |
| Net income | | 116,413 | | 114,135 | | 216,355 | |
| Basic earnings per share | | 0.58 | | 0.65 | | 1.57 | |
| Diluted earnings per share | | 0.58 | | 0.63 | | 1.28 | |
| Dividends per share | C\$ | 0.28 | C\$ | 0.28 | C\$ | 0.28 | |

(1) For the years ended December 31, 2019 and 2018, excludes investment income recognized under Investment Entity accounting.

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 | December 31, 2018 |
|--------------------------------|----------------------|----------------------|----------------------|
| Total Assets | \$ 7,174,834 | \$ 2,302,289 | \$ 1,687,662 |
| Debt | 4,137,506 | 470,553 | 374,716 |

The following factors have caused material changes to the Company's financial results over the past three years:

- On June 27, 2018, the Company entered into a joint venture arrangement ("SFR JV-1") with two leading institutional investors to acquire and manage a portfolio of 10,000–12,000 single-family rental homes, thus introducing third-party capital to the single-family business segment and growing Tricon's total managed homes by 7,439 homes or 47% to date. Since the launch of the joint venture, the value of Tricon's single-family rental portfolio has grown by \$1.6 billion.
- On June 29, 2018, Tricon completed the sale of its 14 manufactured housing communities in an effort to simplify the Company's overall business model and focus on housing sectors where it can achieve scale and industry leadership. As a result of the sale, in 2018, Tricon recognized a gain of \$21.2 million and a reduction in total assets of \$83.5 million.
- On June 11, 2019, the Company completed the acquisition of Starlight U.S. Multi-Family (No. 5) Core Fund (the "U.S. multi-family rental portfolio"), thus establishing a new U.S. multi-family platform for Tricon. The acquisition of the portfolio, which consists of 23 properties totalling 7,289 suites in 13 major markets, increased the value of Tricon's rental portfolio by \$1.3 billion and contributed \$111.2 million of rental revenue for the year ended December 31, 2020.
- In January 2020, the Company completed its transition to an owner and operator of diversified rental housing, resulting in the Company determining that it no longer meets the criteria for being an investment entity under IFRS 10, *Consolidated Financial Statements* ("IFRS 10"). As a result, the Company began consolidating the financial results of controlled subsidiaries including those holding its investments in single-family rental homes and U.S. multi-family rental properties, resulting in the inclusion of these subsidiaries' assets, liabilities and non-controlling interests in the balance sheet of the Company on a prospective basis in accordance with the relevant guidance of IFRS 10.

Consolidated Financial Statements

for the year ended December 31, 2020



Independent auditor's report

To the Shareholders of Tricon Residential Inc.

Our opinion

In our opinion, the accompanying consolidated financial statements present fairly, in all material respects, the financial position of Tricon Residential Inc. and its subsidiaries (together, the Company) as at December 31, 2020 and 2019, and its financial performance and its cash flows for the years then ended in accordance with International Financial Reporting Standards (IFRS).

What we have audited

The Company's consolidated financial statements comprise:

- the consolidated balance sheets as at December 31, 2020 and 2019;
- the consolidated statements of comprehensive income for the years then ended;
- the consolidated statements of changes in equity for the years then ended;
- the consolidated statements of cash flows for the years then ended; and
- the notes to the consolidated financial statements, which include significant accounting policies and other explanatory information.

Basis for opinion

We conducted our audit in accordance with Canadian generally accepted auditing standards. Our responsibilities under those standards are further described in the *Auditor's responsibilities for the audit of the consolidated financial statements* section of our report.

We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our opinion.

Independence

We are independent of the Company in accordance with the ethical requirements that are relevant to our audit of the consolidated financial statements in Canada. We have fulfilled our other ethical responsibilities in accordance with these requirements.

Key audit matters

Key audit matters are those matters that, in our professional judgment, were of most significance in our audit of the consolidated financial statements for the year ended December 31, 2020. These matters were addressed in the context of our audit of the consolidated financial statements as a whole, and in forming our opinion thereon, and we do not provide a separate opinion on these matters.

PricewaterhouseCoopers LLP PwC Tower, 18 York Street, Suite 2600, Toronto, Ontario, Canada M5J 0B2 T: +1 416 863 1133, F: +1 416 365 8215

"PwC" refers to PricewaterhouseCoopers LLP, an Ontario limited liability partnership.



Valuation of rental properties, Canadian development properties and investments in for-sale housing

Refer to note 4 – Critical Accounting Estimates and Judgments, note 6 – Rental Properties, note 8 – Canadian Development Properties and note 9 – Investments in For-Sale Housing to the consolidated financial statements.

As at December 31, 2020, the Company had \$6,322 million of rental properties, \$110 million of Canadian development properties and \$165 million of investments in for-sale housing.

Rental properties (single-family rental homes and multi-family rental properties), Canadian development properties and investments in forsale housing are recorded at fair value. The valuation techniques and models used to determine the fair value of rental properties, Canadian development properties and investments in for-sale housing involve assumptions (observable and unobservable) that require significant judgment and estimation by management. The Company's Valuation Committee is responsible for reviewing and approving the valuation results every quarter.

The valuation techniques and models used include the following:

1) Broker Price Opinion (BPO) and Home Price Index (HPI) methodologies for singlefamily rental homes. BPOs are quoted by independent brokers who hold an active real estate license and have market experience in the locations and segments of the properties being valued. The brokers value

How our audit addressed the key audit matter

Our approach to addressing the matter included the following procedures, among others:

- For a sample of rental properties, Canadian development properties and investments in for-sale housing, tested the fair value determined by management, by performing the following:
 - Read the minutes of the quarterly Valuation Committee meetings to understand the Company's valuation estimates.
 - Evaluated the appropriateness of the valuation techniques and models used by management to determine the fair value of rental properties, Canadian development properties and investments in for-sale housing.
 - Tested the underlying data used in each valuation technique and model, including the BPO/HPI methodologies, direct income capitalization method, waterfall distribution calculations and asset purchase model.
 - Professionals with specialized skill and knowledge in the field of valuation further assisted us in assessing the appropriateness of management's valuation techniques, methodologies and models and assessed the reasonableness of key assumptions used, including BPO/HPI, discount rates, capitalization rates, projected stabilized NOI, estimate of future cash flows and project-specific construction/development costs and selling prices, by benchmarking them to market data.



each property based on recent comparable sales and active comparable listings in the area. HPI is used to update the value, on a quarterly basis, of homes that were most recently valued using a BPO, as well as homes held for more than six months following initial acquisition. The HPI is calculated based on a repeat-sales model using large real estate information databases compiled from public records.

2) Direct income capitalization method for multi-family rental properties. This method requires that a projected stabilized net operating income (NOI) for each property is divided by the appropriate capitalization rate to determine a property's fair value. Key assumptions used in the method included capitalization rates and projected stabilized NOI.

3) Asset purchase model for Canadian development properties. The fair value of these properties is determined based on the property's transaction price and any directly attributable expenditures, including transaction costs.

4) Waterfall distribution calculations for investments in for-sale housing. The fair value of these investments is determined by the waterfall distribution calculations specified in the relevant governing agreements. The inputs into the waterfall distribution calculations include the fair values of the land development and homebuilding projects and working capital. Key assumptions used in the calculations included discount rate, estimate of future

How our audit addressed the key audit matter

 Tested the disclosures made in the consolidated financial statements, particularly with regard to the sensitivity of the key assumptions.



How our audit addressed the key audit matter

cash flows and project-specific construction/development costs and selling prices.

Given the unpredictable long-term economic impact due to the global COVID-19 pandemic and corresponding reductions in real estate transactions in the markets in which the Company operates, the uncertainty inherent in any valuation model and technique is heightened, requiring greater levels of management judgment in the estimation of fair value.

We considered this a key audit matter due to the significant judgments made by management in determining the key assumptions used in the valuation techniques and models. This has resulted in a high degree of subjectivity and audit effort in performing audit procedures to test the key assumptions. Professionals with specialized skill and knowledge in the field of valuation assisted us in performing our procedures.

Transition to consolidation of controlled investments

Refer to note 2 – Basis of Presentation and note 5 – Business Combinations to the consolidated financial statements.

In January 2020, the Company completed its previously announced transition to an owner and operator of diversified rental housing, resulting in the Company determining that it no longer meets the criteria to apply Investment Entity Accounting. As a result, effective January 1, 2020, the Company was required to apply the acquisition method of accounting to all Our approach to addressing the matter included the following procedures, among others:

- Assessed the transition and timing thereof by reading meeting minutes of the Board of Directors and public announcements made by the Company leading up to its planned transition.
- Obtained an understanding of management's consolidation process and the accounting policies to be applied to underlying assets and liabilities and transactions, and considered their compliance with relevant accounting standards and guidelines.



| Key audit matter | How our audit addressed the key audit matter |
|---|--|
| subsidiaries that were previously measured at fair value through profit or loss. Consequently, the Company began consolidating the financial results of controlled subsidiaries including those holding its investments in single-family rental homes and U.S. multi-family rental properties, resulting in the inclusion of these subsidiaries' assets, liabilities and non-controlling interests in the consolidated balance sheet of the Company. Similarly, these subsidiaries' revenues and expenses have been reported in the Company's consolidated statement of comprehensive income together with the non-controlling interests' share of income. The indicators that the Company no longer met the definition of an investment company and the timing thereof, as well as the determination of which entities are controlled by the Company, represent significant judgment made by management. | Evaluated management's assessment of control over each entity, with reference to relevant accounting standards and guidelines, by considering whether: the decisions over relevant activities as set out in the executed agreements required consent of the Company without the need for consent of other parties; the Company has exposure to variable returns from its investment; the Company has the ability to use its power to affect the amount of returns; and the structure of the arrangement as outlined in the executed agreements includes any significant rights held by other parties. Assessed the appropriateness of the change in basis of presentation, the impacts on the presentation of the primary financial statements and the related disclosures made in the consolidated financial statements. |
| We considered this a key audit matter due to the significant judgment applied by management in assessing the transition and its timing, including the significant impact on the presentation of the consolidated financial statements and | |

presentation.

disclosures. This has resulted in a high degree of subjectivity and audit effort in performing audit

procedures to test the change in basis of



How our audit addressed the key audit matter

Initial recognition and measurement of Tricon PIPE LLC arrangement

Key audit matter

Refer to note 3 – Summary of Significant Accounting Policies, note 19 – Due to Affiliate and note 20 – Derivative Financial Instruments to the consolidated financial statements.

As at December 31, 2020, the Company reported a \$252 million amount as a promissory note due to affiliate and a related \$45 million of derivative financial liabilities in connection with the issuance of Tricon PIPE LLC preferred units.

On August 26, 2020, the Company and its affiliate, Tricon PIPE LLC (the Affiliate) entered into subscription agreements with each investor in a syndicate of investors (the Investors), pursuant to which the Investors subscribed for preferred units of the Affiliate for an aggregate subscription price of \$300 million. The Affiliate is an unconsolidated structured entity as it was created for the sole purpose of issuing its preferred units to investors and offering financing to the Company, and the Company does not have exposure to variable returns or make the relevant decisions for the entity. Through the agreements that the Company and the Affiliate entered into with the Investors, holders of preferred units have the right to exchange the preferred units into common shares of the Company at any time, which is accounted for as a derivative.

The Company borrowed the subscription proceeds of \$300 million from the Affiliate, which is evidenced by a promissory note with a maturity of September 3, 2032. The promissory note contains certain mandatory prepayment Our approach to addressing the matter included the following procedures, among others:

- Obtained an understanding of the structure of the financing arrangement by reading the transaction documents.
- Assessed management's determination that Tricon PIPE LLC is an unconsolidated structured entity by considering the nature and scope of its operations and related exposure to variable returns.
- With the assistance of professionals with specialized skill and knowledge in the field of valuation, developed independent point estimates of the initial fair values of the promissory note, derivative and embedded derivative. This involved the use of available market data to independently develop assumptions related to volatility of the underlying equity, expected life of the investment horizon of the Investors and expected cash flows of the promissory note. The independent point estimates were compared to management's estimates to evaluate the reasonableness of the fair values of the promissory note, derivative and embedded derivative.



provisions that are measured separately from the promissory note and classified as an embedded derivative. This embedded derivative factors in certain assumptions regarding the exchange provisions of the underlying preferred units, as prepayment of the promissory note effectively terminates such exchange rights.

As a result, the values of the embedded derivative and derivative resulting from the prepayment provisions and the exchange provisions are determined on a combined basis using an option pricing model. Key assumptions included in the model are implied volatility of the underlying equity and expected life of the investment horizon of the Investors.

The promissory note, which management valued using a discounted cash flow model, also contains step-up interest clauses that require significant estimates to be made about the expected cash flows and term of the note for purposes of determining the initial fair value of the promissory note. For this purpose, the Company used an amortization period of nine years and nine months determined by using a probability weighted methodology.

Significant judgment was made by management in determining the key assumptions related to the initial recognition and measurement of the promissory note, the derivative and the embedded derivative.

We considered this a key audit matter due to the complexity of the Tricon PIPE LLC arrangement and the significant judgment applied by management in determining the initial

How our audit addressed the key audit matter



How our audit addressed the key audit matter

recognition and measurement of the promissory note, the derivative and the embedded derivative. This has resulted in a high degree of subjectivity and audit effort in performing audit procedures to assess control and test the initial recognition and measurement of the promissory note, the derivative and the embedded derivative. Professionals with specialized skill and knowledge in the field of valuation assisted us in performing our procedures.

Other information

Management is responsible for the other information. The other information comprises the Management's Discussion and Analysis, which we obtained prior to the date of this auditor's report and the information, other than the consolidated financial statements and our auditor's report thereon, included in the annual report, which is expected to be made available to us after that date.

Our opinion on the consolidated financial statements does not cover the other information and we do not and will not express an opinion or any form of assurance conclusion thereon.

In connection with our audit of the consolidated financial statements, our responsibility is to read the other information identified above and, in doing so, consider whether the other information is materially inconsistent with the consolidated financial statements or our knowledge obtained in the audit, or otherwise appears to be materially misstated.

If, based on the work we have performed on the other information that we obtained prior to the date of this auditor's report, we conclude that there is a material misstatement of this other information, we are required to report that fact. We have nothing to report in this regard. When we read the information, other than the consolidated financial statements and our auditor's report thereon, included in the annual report, if we conclude that there is a material misstatement therein, we are required to communicate the matter to those charged with governance.

Responsibilities of management and those charged with governance for the consolidated financial statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with IFRS, and for such internal control as management determines is necessary to enable the preparation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.



In preparing the consolidated financial statements, management is responsible for assessing the Company's ability to continue as a going concern, disclosing, as applicable, matters related to going concern and using the going concern basis of accounting unless management either intends to liquidate the Company or to cease operations, or has no realistic alternative but to do so.

Those charged with governance are responsible for overseeing the Company's financial reporting process.

Auditor's responsibilities for the audit of the consolidated financial statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditor's report that includes our opinion. Reasonable assurance is a high level of assurance, but is not a guarantee that an audit conducted in accordance with Canadian generally accepted auditing standards will always detect a material misstatement when it exists. Misstatements can arise from fraud or error and are considered material if, individually or in the aggregate, they could reasonably be expected to influence the economic decisions of users taken on the basis of these consolidated financial statements.

As part of an audit in accordance with Canadian generally accepted auditing standards, we exercise professional judgment and maintain professional skepticism throughout the audit. We also:

- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to fraud or error, design and perform audit procedures responsive to those risks, and obtain audit evidence that is sufficient and appropriate to provide a basis for our opinion. The risk of not detecting a material misstatement resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control.
- Obtain an understanding of internal control relevant to the audit in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control.
- Evaluate the appropriateness of accounting policies used and the reasonableness of accounting estimates and related disclosures made by management.
- Conclude on the appropriateness of management's use of the going concern basis of accounting and, based on the audit evidence obtained, whether a material uncertainty exists related to events or conditions that may cast significant doubt on the Company's ability to continue as a going concern. If we conclude that a material uncertainty exists, we are required to draw attention in our auditor's report to the related disclosures in the consolidated financial statements or, if such disclosures are inadequate, to modify our opinion. Our conclusions are based on the audit evidence obtained up to the date of our auditor's report. However, future events or conditions may cause the Company to cease to continue as a going concern.



- Evaluate the overall presentation, structure and content of the consolidated financial statements, including the disclosures, and whether the consolidated financial statements represent the underlying transactions and events in a manner that achieves fair presentation.
- Obtain sufficient appropriate audit evidence regarding the financial information of the entities or business activities within the Company to express an opinion on the consolidated financial statements. We are responsible for the direction, supervision and performance of the group audit. We remain solely responsible for our audit opinion.

We communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit and significant audit findings, including any significant deficiencies in internal control that we identify during our audit.

We also provide those charged with governance with a statement that we have complied with relevant ethical requirements regarding independence, and to communicate with them all relationships and other matters that may reasonably be thought to bear on our independence, and where applicable, related safeguards.

From the matters communicated with those charged with governance, we determine those matters that were of most significance in the audit of the consolidated financial statements of the current period and are therefore the key audit matters. We describe these matters in our auditor's report unless law or regulation precludes public disclosure about the matter or when, in extremely rare circumstances, we determine that a matter should not be communicated in our report because the adverse consequences of doing so would reasonably be expected to outweigh the public interest benefits of such communication.

The engagement partner on the audit resulting in this independent auditor's report is Derek Hatoum.

/s/PricewaterhouseCoopers LLP

Chartered Professional Accountants, Licensed Public Accountants

Toronto, Ontario March 2, 2021

CONSOLIDATED BALANCE SHEETS

(in thousands of U.S. dollars)

| | Notes | December 31, 2020 | December 31, 2019 |
|---|-------------|-------------------|-------------------|
| ASSETS | | | |
| Non-current assets | | | |
| Rental properties | <u>6</u> | \$ 6,321,918 | \$ – |
| Investments in Canadian multi-family developments | <u> </u> | 94,868 | - |
| Canadian development properties | 3, 8 | 110,018 | - |
| Investments in for-sale housing | <u> </u> | 164,842 | 300,653 |
| Investments – Tricon American Homes | <u> </u> | _ | 1,365,007 |
| Investments – Tricon Lifestyle Rentals | 2 | - | 525,932 |
| Restricted cash | _ | 116,302 | - |
| Goodwill | 5, 12 | 108,838 | 219 |
| Intangible assets | 24 | 12,363 | 16,396 |
| Other assets | 25 | 47,990 | 30,677 |
| Deferred income tax assets | 13 | 102,444 | 44,749 |
| Derivative financial instruments | 20 | 841 | _ |
| Total non-current assets | | 7,080,424 | 2,283,633 |
| | | | |
| Current assets | | | |
| Cash | | 55,158 | 8,908 |
| Amounts receivable | <u>16</u> | 25,593 | 8,952 |
| Prepaid expenses and deposits | | 13,659 | 796 |
| Total current assets | | 94,410 | 18,656 |
| Total assets | | \$ 7,174,834 | \$ 2,302,289 |
| LIABILITIES | | | |
| Non-current liabilities | | | |
| Long-term debt | <u>17</u> | \$ 3,863,316 | \$ 307,869 |
| Convertible debentures | <u>18</u> | 165,956 | 161,311 |
| Due to Affiliate | <u>19</u> | 251,647 | - |
| Derivative financial instruments | 20 | 45,494 | 657 |
| Limited partners' interests in rental business | <u>5</u> | 356,305 | - |
| Long-term incentive plan | <u>30</u> | 17,930 | 21,409 |
| Other liabilities | 26 | 4,599 | 14,329 |
| Deferred income tax liabilities | 13 | 298,071 | 98,584 |
| Total non-current liabilities | | 5,003,318 | 604,159 |
| | | | |
| Current liabilities | | 00.000 | 00.400 |
| Amounts payable and accrued liabilities | <u>11</u> | 98,290 | 26,190 |
| Resident security deposits | | 45,157 | - |
| Dividends payable | <u>27</u> | 10,641 | 10,474 |
| Current portion of long-term debt | <u>17</u> | 274,190 | 284 |
| Total current liabilities Total liabilities | | 428,278 | 36,948 |
| | | 5,431,596 | 641,107 |
| Equity | | | |
| Share capital | <u>28</u> | 1,192,963 | 1,201,061 |
| Share capital reserve | | - | (13,057) |
| Contributed surplus | | 19,738 | 20,223 |
| Cumulative translation adjustment | | 23,395 | 19,396 |
| Retained earnings | | 499,000 | 425,515 |
| Total shareholders' equity | | 1,735,096 | 1,653,138 |
| Non-controlling interest | | 8,142 | 8,044 |
| Total equity | | 1,743,238 | 1,661,182 |
| Total liabilities and equity | | \$ 7,174,834 | \$ 2,302,289 |

The accompanying notes are an integral part of these consolidated financial statements.

Approved by the Board of Directors

CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME

(in thousands of U.S. dollars, except per share amounts which are in U.S. dollars, unless otherwise indicated)

| For the years ended | Notes | December 31, 2020 | December 31, 2019 |
|--|---------------|-------------------|-------------------|
| Revenue from rental properties | 14 | \$ 478,187 | \$ - |
| Direct operating expenses | 22 | (169,538) | _ |
| Net operating income from rental properties | | 308,649 | - |
| Revenue from private funds and advisory services | <u>15</u> | \$ 34,090 | \$ 39,895 |
| Income from investments in Canadian multi-family developments | <u>7</u> | 14,124 | - |
| Other income from Canadian development properties | <u>8</u> | 791 | - |
| (Loss) income from investments in for-sale housing | <u>9</u> | (61,776) | 9,646 |
| Property management overhead | 22 | (22,654) | - |
| Compensation expense | <u>30</u> | (40,100) | (37,681) |
| General and administration expense | | (23,569) | (11,683) |
| Other income (expense) | <u>23</u> | (1,399) | _ |
| Interest expense | <u>21</u> | (170,610) | (32,439) |
| Fair value gain on rental properties | 6 | 198,314 | _ |
| Fair value (loss) gain on derivative financial instruments and other liabilities | 20 | (7,461) | 2,961 |
| Transaction costs | | (14,016) | (32,626) |
| Amortization and depreciation expense | <u>24, 25</u> | (10,848) | (6,274) |
| Realized and unrealized foreign exchange (loss) gain | | (166) | 42 |
| Net change in fair value of limited partners' interests in rental business | 5 | (50,581) | _ |
| Investment income – Tricon American Homes | | - | 162,193 |
| Investment income – Tricon Lifestyle Rentals | | - | 34,980 |
| | | (189,951) | 89,119 |
| Income before income taxes | | \$ 152,788 | \$ 129,014 |
| Income tax recovery (expense) – current | <u>13</u> | 4,050 | (5,410) |
| Income tax expense – deferred | <u>13</u> | (40,425) | (9,469) |
| Net income | | \$ 116,413 | \$ 114,135 |
| Attributable to: | | | |
| Shareholders of Tricon | | 113,322 | 111,562 |
| Non-controlling interest | | 3,091 | 2,573 |
| Net income | | \$ 116,413 | \$ 114,135 |
| Other comprehensive income | | | |
| Items that will be reclassified subsequently to net income | | | |
| Cumulative translation reserve | | 3,999 | (129) |
| Comprehensive income for the year | | \$ 120,412 | \$ 114,006 |
| Attributable to: | | | |
| Shareholders of Tricon | | 117,321 | 111,433 |
| Non-controlling interest | | 3,091 | 2,573 |
| Comprehensive income for the year | | \$ 120,412 | \$ 114,006 |
| Basic earnings per share attributable to shareholders of Tricon | <u>29</u> | \$ 0.58 | \$ 0.65 |
| Diluted earnings per share attributable to shareholders of Tricon | 29 | \$ 0.58 | \$ 0.63 |
| Weighted average shares outstanding – basic | <u>29</u> | 194,627,127 | 172,735,776 |
| Weighted average shares outstanding – diluted | 29 | 195,795,473 | 191,081,128 |

The accompanying notes are an integral part of these consolidated financial statements.

CONSOLIDATED STATEMENTS OF CHANGES IN EQUITY

(in thousands of U.S. dollars)

| | Notes | Share capital | Share capital reserve | Contributed surplus | Cumulative translation adjustment | Retained earnings | Total shareholders' equity | Non- controlling interest | Total |
|---|-----------|------------------|-----------------------------|------------------------|---|----------------------|----------------------------------|---------------------------------|--------------|
| Balance at | | | | | | | | | |
| January 1, 2020 | | \$ 1,201,061 | \$ (13,057) | \$ 20,223 | \$ 19,396 | \$ 425,515 | \$ 1,653,138 | \$ 8,044 | \$ 1,661,182 |
| Net income | | - | _ | _ | _ | 113,322 | 113,322 | 3,091 | 116,413 |
| Shares repurchased | | | | | | | | | |
| under put rights | | | | | | | | | |
| on common shares | | | | | | | | | |
| issued to acquire | | | | | | | | | |
| Starlight U.S. | | | | | | | | | |
| Multi-Family (No. 5) Core Fund | 20 | (14022) | 13,057 | | | | (1 965) | | (1 965) |
| Core Fund Cumulative translation | <u>28</u> | (14,922) | 13,057 | - | - | - | (1,865) | - | (1,865) |
| reserve | | _ | _ | _ | 3,999 | _ | 3,999 | _ | 3,999 |
| Distributions to | | | | | 3,333 | | 3,333 | | 3,333 |
| non-controlling | | | | | | | | | |
| interest | | _ | _ | _ | _ | _ | _ | (2,993) | (2,993) |
| Dividends/Dividend | | | | | | | | | |
| reinvestment plan | 27 | 4,388 | _ | _ | _ | (40,192) | (35,804) | - | (35,804) |
| Stock options | <u>30</u> | 1,615 | - | (2,394) | - | 355 | (424) | - | (424) |
| Shares reserved | | | | | | | | | |
| for restricted | | | | | | | | | |
| share awards | <u>30</u> | (541) | - | 276 | - | - | (265) | - | (265) |
| Deferred share units | <u>30</u> | 1,362 | - | 1,633 | - | - | 2,995 | - | 2,995 |
| Balance at | | | | | | | | | |
| December 31, 2020 | | \$ 1,192,963 | \$ - | \$ 19,738 | \$ 23,395 | \$ 499,000 | \$ 1,735,096 | \$ 8,142 | \$ 1,743,238 |
| Balance at | | | | | | | | | |
| January 1, 2019 | | \$ 793,521 | \$ - | \$ 17,468 | \$ 19,525 | \$ 353,220 | \$ 1,183,734 | \$ 8,864 | \$ 1,192,598 |
| Net income | | _ | - | - | _ | 111,562 | 111,562 | 2,573 | 114,135 |
| Shares issued to | | | | | | | | | |
| acquire Starlight | | | | | | | | | |
| U.S. Multi-Family | | | | | | | | | |
| (No. 5) Core Fund | 28 | 405,491 | (13,057) | - | - | - | 392,434 | - | 392,434 |
| Cumulative translation | | | | | | | | | |
| reserve | | - | - | - | (129) | - | (129) | - | (129) |
| Distributions to | | | | | | | | | |
| non-controlling | | | | | | | | (2 202) | (2 20 2) |
| interest | | - | _ | _ | - | - | - | (3,393) | (3,393) |
| Dividends/Dividend reinvestment plan | 27 | 3,793 | | | _ | (38,575) | (34,782) | _ | (34,782) |
| Repurchase of | <u>27</u> | 5,795 | - | - | _ | (30,373) | (34,782) | - | (34,782) |
| common shares | 28 | (3,067) | _ | _ | _ | (692) | (3,759) | _ | (3,759) |
| Debentures conversion | 28 | 100 | _ | _ | _ | (002) | 100 | _ | (3,733) |
| Stock options | 30 | 258 | _ | 579 | _ | _ | 837 | _ | 837 |
| Shares repurchased | | | | 2.9 | | | / | | / |
| and reserved | | | | | | | | | |
| for restricted | | | | | | | | | |
| share awards | <u>30</u> | (590) | - | 225 | - | - | (365) | - | (365) |
| Deferred share units | | 1,555 | - | 1,951 | - | - | 3,506 | - | 3,506 |
| Balance at | | | | | | | | | |
| December 31, 2019 | | \$ 1,201,061 | \$ (13,057) | \$ 20,223 | \$ 19,396 | \$ 425,515 | \$ 1,653,138 | \$ 8,044 | \$ 1,661,182 |

The accompanying notes are an integral part of these consolidated financial statements.

CONSOLIDATED STATEMENTS OF CASH FLOWS

(in thousands of U.S. dollars)

| For the years ended | Notes | December 31, 2020 | December 31, 2019 |
|--|---------------|-------------------|-------------------|
| CASH PROVIDED BY (USED IN) | | | |
| Operating activities | | | |
| Net income | | \$ 116,413 | \$ 114,135 |
| Adjustments for non-cash items | | | |
| Fair value gain on rental properties | 6 | (198,314) | - |
| Fair value loss (gain) on derivative financial instruments | | | |
| and other liabilities | 20 | 7,461 | (2,961) |
| Loss (income) from investments in for-sale housing | <u>9</u> | 61,776 | (9,646) |
| Income from investments in Canadian multi-family developments | <u>7</u> | (14,124) | - |
| Amortization and depreciation expense | <u>24, 25</u> | 10,848 | 6,274 |
| Deferred income taxes | <u>13</u> | 40,425 | 9,469 |
| Net change in fair value of limited partners' interests in rental business | <u>5</u> | 50,581 | - |
| Other non-cash items | 35 | 22,340 | (176,291) |
| Cash paid for AIP and LTIP | | (16,733) | (14,083) |
| Distributions to non-controlling interests | | (2,993) | (3,393) |
| Advances made to investments | <u>7, 9</u> | (7,702) | (197,067) |
| Distributions received from investments | <u>7, 9</u> | 78,378 | 200,631 |
| Changes in non-cash working capital items | 35 | (5,343) | 28,631 |
| Net cash (used in) provided by operating activities | | \$ 143,013 | \$ (44,301) |
| Investing activities | | | |
| Cash acquired in deemed acquisitions | 5 | 22,199 | _ |
| Acquisition of remaining interest of Canadian development properties | 8 | (7,643) | _ |
| Acquisition of rental properties | <u>6</u> | (356,514) | _ |
| Capital additions to rental properties | <u>6</u> | (102,635) | - |
| Disposition of rental properties | <u>6</u> | 18,070 | - |
| Additions to fixed assets and other non-current assets | <u>8, 25</u> | (13,025) | (10,017) |
| Net cash (used in) provided by investing activities | | \$ (439,548) | \$ (10,017) |
| Financing activities | | | |
| Lease payments | 26 | (2,415) | (180) |
| Repurchase of common shares | 28 | (14,922) | (3,759) |
| Equity issuance costs | | - | (223) |
| Proceeds from corporate borrowing | <u>36</u> | 163,500 | 547,000 |
| Repayments of corporate borrowing | 36 | (434,775) | (455,683) |
| Proceeds from rental and development properties borrowing | 36 | 1,361,458 | _ |
| Repayments of rental and development properties borrowing | <u>36</u> | (969,979) | _ |
| Proceeds from Due to Affiliate | <u>19</u> | 287,798 | _ |
| Dividends paid | <u>27</u> | (35,637) | (31,725) |
| Change in restricted cash | | (32,220) | _ |
| Advances from limited partners | 5 | 66,112 | - |
| Distributions to limited partners | 5 | (46,162) | - |
| Net cash (used in) provided by financing activities | | \$ 342,758 | \$ 55,430 |
| Effect of foreign exchange rate difference on cash | | 27 | 23 |
| Change in cash during the year | | 46,250 | 1,135 |
| Cash – beginning of year | | 8,908 | 7,773 |
| Cash – end of year | | \$ 55,158 | \$ 8,908 |
| Supplementary information | | | |
| Cash paid on | | | |
| Income taxes | | \$ 226 | \$ 1,224 |
| | | \$ 155,053 | \$ 27,824 |

The accompanying notes are an integral part of these consolidated financial statements.

1. NATURE OF BUSINESS

Tricon Residential Inc. ("Tricon" or the "Company"), formerly Tricon Capital Group Inc., is a residential real estate company primarily focused on owning and operating rental housing in North America. Tricon currently owns and operates approximately 31,000 single-family rental homes and multi-family rental units in 21 markets across the United States and Canada. Through its fully integrated operating platform, the Company earns rental income and ancillary revenue from single-family and multi-family rental properties as well as fees from managing third-party capital associated with its businesses.

Tricon was incorporated on June 16, 1997 under the Business Corporations Act (Ontario) and its head office is located at 7 St. Thomas Street, Suite 801, Toronto, Ontario, M5S 2B7. The Company is domiciled in Canada. Tricon became a public company on May 20, 2010, and its common shares are listed on the Toronto Stock Exchange ("TSX") (symbol: TCN).

These consolidated financial statements were approved for issue on March 2, 2021 by the Board of Directors of Tricon.

2. BASIS OF PRESENTATION

Transition to a rental housing company

In January 2020, the Company completed its previously announced transition to an owner and operator of diversified rental housing, resulting in the Company determining that it no longer meets the criteria for being an investment entity ("Investment Entity Accounting") under IFRS 10, *Consolidated Financial Statements* ("IFRS 10"). The exact timing of the transition from an investment entity to a rental housing company is highly judgmental and the Company concluded that this transition occurred in January 2020. As a result, effective January 1, 2020 (the "Transition Date"), the Company was required to apply the acquisition method of accounting as per IFRS 3, *Business Combinations* ("IFRS 3"), to all subsidiaries that were previously measured at fair value through profit or loss ("FVTPL") (Note 5).

Consequently, the Company began consolidating the financial results of controlled subsidiaries including those holding its investments in single-family rental homes and U.S. multi-family rental properties, resulting in the inclusion of these subsidiaries' assets, liabilities and non-controlling interests in the consolidated balance sheet of the Company. Similarly, these subsidiaries' revenues and expenses have been reported in the Company's consolidated statement of comprehensive income together with the non-controlling interests' share of income.

Concurrent with the consolidation of the single-family and multi-family rental properties, the Company's investments in Canadian multi-family developments are accounted for as follows: (i) proportionate consolidation for joint operations in accordance with IFRS 11, *Joint Arrangements* ("IFRS 11") for the period between January 1, 2020 and June 22, 2020, during which time the Company owned 50% and 25% interests in The James and The Shops of Summerhill, respectively; and (ii) equity accounting for associates and joint ventures under IAS 28, *Investments in Associates and Joint Ventures* ("IAS 28"). The Company's legacy investments in for-sale housing in the U.S. will continue to be accounted for as portfolio investments (financial assets) measured at FVTPL in accordance with IFRS 9, *Financial Instruments* ("IFRS 9").

The accounting impact of the Company's businesses and their presentation in the Company's consolidated financial statements on the Transition Date are summarized in the table below.

| | ACCO | UNTING | PRESENTATION | | | | |
|--|--|---|---|--|---|--|--|
| Business segment | Accounting assessment | Accounting methodology | Presentation in Balance Sheet | Presentation in Statement of Income | Presentation of Non-controlling interest | | |
| Single-Family Rental | | | | | | | |
| Tricon wholly-owned | Controlled subsidiary | Consolidation | Rental properties | Revenue from | N/A | | |
| SFR JV-1 | Controlled subsidiary | Consolidation | - | rental properties | Limited partners' interests (Component of liabilities) | | |
| Multi-Family Rental | | | | | | | |
| U.S. multi-family | Controlled subsidiary | Consolidation | Rental properties | Revenue from rental properties | N/A | | |
| Canadian multi-family: 592 Sherbourne (The Selby) | Investments in associate | Equity method | Investments in Canadian multi-family developments | Income from investments in Canadian multi-family developments | N/A | | |
| Canadian Multi-Family | y Developments | | | · | | | |
| The Shops of Summerhill ⁽¹⁾ | Joint operation for the period between | Proportionate consolidation | Canadian development | Other income from Canadian | N/A | | |
| The James (Scrivener Square) ⁽¹⁾ | January 1, 2020 and June 22, 2020, and controlled subsidiary from June 23, 2020 | between January 1, 2020 and June 22, 2020, and consolidation from June 23, 2020 | properties | development properties | N/A | | |
| 57 Spadina (The Taylor) | Investments in associate | Equity method | Investments in Canadian multi-family | Income from investments in | N/A | | |
| WDL – Block 8 | Joint venture | Equity method | developments | Canadian multi-family | N/A | | |
| WDL – Block 20 | Joint venture | Equity method | - | developments | N/A | | |
| WDL – Blocks 3/4/7 | Joint venture | Equity method | - | | N/A | | |
| WDL – Block 10 | Joint venture | Equity method | - | | N/A | | |
| 6–8 Gloucester (The lvy) | Joint venture | Equity method | - | | N/A | | |
| 7 Labatt | Joint venture | Equity method | - | | N/A | | |
| Private Funds and Ad | visory | 1 | 1 | 1 | 1 | | |
| Private funds GP entities | Controlled subsidiary | Consolidation | Consolidated | Revenue from private funds | N/A | | |
| Johnson development management | Controlled subsidiary | Consolidation | Consolidated | and advisory | Component of equity | | |
| For-Sale Housing | | | | | | | |
| Commingled funds | Portfolio investments | FVTPL | Investments in Income from | | N/A | | |
| Separate accounts, side-cars and joint ventures | Portfolio investments | FVTPL | for-sale housing | investments in for-sale housing | N/A | | |

(1) On June 23, 2020, Tricon acquired the remaining ownership interests of 50% and 75% in The James and The Shops of Summerhill, respectively (see Note 8). As a result, these investees ceased to be accounted for as joint operations, and the Company began to consolidate these subsidiaries on a prospective basis.

These financial reporting changes are material to the Company and have been applied on a prospective basis in accordance with the relevant guidance of IFRS 10 and, as such, the comparative period presentation reflects Investment Entity Accounting as previously reported.

Preparation of consolidated financial statements

The consolidated financial statements are prepared on a going-concern basis and have been presented in U.S. dollars, which is also the Company's functional currency. All financial information is presented in thousands of U.S. dollars except where otherwise indicated.

These consolidated financial statements have been prepared in accordance with International Financial Reporting Standards ("IFRS") as set out in the CPA Canada Handbook – Accounting. The preparation of consolidated financial statements in accordance with IFRS requires the use of certain critical accounting estimates. It also requires management to exercise judgment in applying Tricon's accounting policies. The estimates involving a high degree of judgment or complexity, or estimates where assumptions are significant to the consolidated financial statements, are disclosed in Note 4.

These consolidated financial statements have been prepared under the historical cost convention, except for:

- (i) Rental properties, which are recorded at fair value with changes in fair value recorded in the consolidated statements of comprehensive income;
- (ii) Canadian development properties, which are recorded at fair value with changes in fair value recorded in the consolidated statements of comprehensive income;
- (iii) Investments in for-sale housing, which are accounted for as portfolio investments (financial assets) and are recorded at fair value through profit or loss;
- (iv) Derivative financial instruments, which are recorded at fair value through profit or loss; and
- (v) Limited partners' interests, which are recorded at fair value through profit or loss.

3. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

The following is a summary of the significant accounting policies applied in the preparation of these consolidated financial statements.

Consolidation

The consolidated financial statements include the financial statements of the Company and its controlled subsidiaries. The Company controls an entity when it is exposed to, or has rights to, variable returns from its involvement with the entity and has the ability to affect those returns through its power over the entity.

The accounting policies of subsidiaries have been modified where necessary to align them with the policies adopted by the Company. When the Company does not own all of the equity in a subsidiary, the non-controlling equity interest is disclosed in the consolidated balance sheet as a separate component of total equity. A non-controlling interest may also be classified as a financial liability if the non-controlling interest contains an option or a redemption feature, which is the case for SFR JV-1. All intra-group balances and transactions are fully eliminated upon consolidation.

The Company currently consolidates Tricon Single-Family Rental REIT LLC and its wholly-owned subsidiaries, along with SFR JV-1 (collectively, the "single-family rental" business), Tricon US Multi-Family REIT Inc. and its wholly-owned subsidiaries (collectively, the "multi-family rental" business), and The James (Scrivener Square) and The Shops of Summerhill (collectively, the "Canadian development properties"). The single-family and multi-family rental businesses were previously held through Tricon SF Home Rental ULC and TLR Saturn Master LP until the Company reorganized and simplified its legal structure in May 2020.

Joint arrangements and interests in associates

Investments in joint arrangements are classified as either joint operations or joint ventures depending on the contractual rights and obligations of each investor. Joint operations are accounted for using proportionate consolidation as per IFRS 11 while joint ventures apply the equity method in accordance with IAS 28.

Joint operations – proportionate consolidation

A joint operation is a joint arrangement under which the investors involved have joint control and usually results from the investors holding direct interests in the assets and liabilities of an investee (without establishing a separate legal entity). At the Transition Date, the Company had interests in one development project (The James) and an adjacent commercial property (The Shops of Summerhill) in Toronto that were accounted for as joint operations. On June 23, 2020, Tricon acquired the remaining ownership interests of 50% and 75% in The James and The Shops of Summerhill, respectively, and as a result, the Company began to consolidate these subsidiaries on a prospective basis (Note 8).

Joint ventures - equity method of accounting

A joint venture is a joint arrangement under which the investors have joint control through a separate legal entity established and hold an interest in the net assets (as opposed to a direct interest in the underlying project). The Company accounts for its joint ventures using the equity method. The Company currently has six active Canadian multi-family developments that are governed by joint venture arrangements.

Interests in associates - equity method of accounting

An associate is an entity over which the Company has significant influence, but not control (or joint control), in accordance with IAS 28. Generally, the Company is considered to exert significant influence when it holds, directly or indirectly, 20% or more of the voting power of the investee. However, determining significant influence is a matter of judgment and specific circumstances. The Company's interests in 592 Sherbourne LP (The Selby) and 57 Spadina LP (The Taylor) are accounted for using the equity method.

Under the equity method, a contribution to an investee is initially recognized at cost and adjusted thereafter to recognize the Company's share of profit or loss of the investee in accordance with Tricon's accounting policies. Distributions received from an investee reduce the carrying amount of the investment.

The Company's associates and joint ventures that are equity-accounted include:

| Name | Туре | Principal place of business | Country of incorporation | Ownership interest % | Voting rights % ⁽¹⁾ |
|-------------------------------|---------------------|--------------------------------|--------------------------|-------------------------|-----------------------------------|
| Associates | | | | | |
| 592 Sherbourne LP (The Selby) | Limited Partnership | Canada | Canada | 15% | 50% |
| 57 Spadina LP (The Taylor) | Limited Partnership | Canada | Canada | 30% | 50% |
| Joint ventures | | | | | |
| WDL 3/4/7 LP | Limited Partnership | Canada | Canada | 33% | 33% |
| WDL 8 LP | Limited Partnership | Canada | Canada | 33% | 33% |
| WDL 20 LP | Limited Partnership | Canada | Canada | 33% | 33% |
| DKT B10 LP | Limited Partnership | Canada | Canada | 33% | 33% |
| 6–8 Gloucester LP (The Ivy) | Limited Partnership | Canada | Canada | 47% | 50% |
| Labatt Village Holding LP | Limited Partnership | Canada | Canada | 38% | 50% |

(1) In respect of major decisions only.

Structured entity – unconsolidated

A structured entity is an entity created to accomplish a narrow and well-defined objective. Those entities' activities are restricted to the extent that they are, in essence, not directed by voting or similar rights. The Company concluded that Tricon PIPE LLC is a structured entity as it was created for the sole purpose of issuing its preferred units to investors and offering financing to the Company (Note 19), and the Company does not have exposure to variable returns related to its involvement in the entity or make the relevant decisions for the entity. Under IFRS 10, such a structured entity does not meet the criteria for control and is not required to be consolidated.

Investments in for-sale housing

Investments that are held as part of the Company's for-sale housing portfolio are carried on the consolidated balance sheets at fair value even though the Company may have significant influence over those companies. This treatment is permitted by IAS 28, which allows portfolio investments that are held by the Company to be recognized and measured at FVTPL and accounted for in accordance with IFRS 9 and IFRS 13, *Fair Value Measurement* ("IFRS 13"), with changes in fair value recognized in the consolidated statements of comprehensive income.

The Company invests in for-sale housing by providing equity or equity-type financing to experienced local or regional developers and builders primarily in the United States. The investments are typically made through co-investments in commingled funds, separate accounts, side-cars and joint ventures ("Investment Vehicles") which hold interests in land development and homebuilding projects. The Company's investments in for-sale housing include:

| Name | Туре | Principal place of business | Country of incorporation | Ownership interest % | Voting rights % ⁽¹⁾ |
|--|---------------------|--------------------------------|--------------------------|-------------------------|-----------------------------------|
| Tricon Housing Partners US LP ⁽²⁾ | Limited Partnership | USA | USA | 68% | 68% |
| Tricon Housing Partners US Syndicated Pool I LP | Limited Partnership | USA | USA | 20% | 50% |
| Tricon Housing Partners US Syndicated Pool II LP | Limited Partnership | USA | USA | 20% | 50% |
| Tricon Housing Partners US II LP ⁽²⁾ | Limited Partnership | USA | USA | 8% | >50% |
| Tricon Housing Partners Canada III LP ⁽²⁾ | Limited Partnership | Canada | Canada | 10% | >50% |
| CCR Texas Equity LP | Limited Partnership | USA | USA | 10% | 50% |
| Vistancia West Equity LP | Limited Partnership | USA | USA | 7% | 50% |
| Conroe CS Texas Equity LP | Limited Partnership | USA | USA | 10% | 50% |
| Tegavah Equity LP | Limited Partnership | USA | USA | 10% | 50% |
| Lake Norman Equity LP | Limited Partnership | USA | USA | 7% | 50% |
| Arantine Hills Equity LP | Limited Partnership | USA | USA | 7% | 50% |
| Viridian Equity LP | Limited Partnership | USA | USA | 18% | 50% |
| THPAS Holdings JV-1 LLC | Limited Partnership | USA | USA | 11% | 50% |
| McKinney Project Equity LLC | Limited Partnership | USA | USA | 44% | 50% |

(1) In respect of major decisions only.

(2) For the purposes of analysis under IFRS, it was determined that Tricon acts primarily as an agent for the benefit of its investors in these partnership entities, and thus Tricon does not control these entities in accordance with the criteria set out in IFRS 10.

Business combination

The Company assesses whether an acquisition transaction should be accounted for as an asset acquisition or a business combination under IFRS 3. A business combination is defined as an acquisition of assets and liabilities that constitute a business that is an integrated set of activities consisting of inputs (such as assets), and processes that when applied to those inputs have the ability to create outputs that provide a return to the Company and its shareholders.

The Company applies the acquisition method to account for business combinations in accordance with IFRS 3. The consideration transferred for the acquisition of the business is the fair value of the assets transferred net of the liabilities assumed, any non-controlling interest in the acquiree, as well as any goodwill or bargain purchase gain recognized and measured by the Company. These identifiable assets acquired and liabilities assumed in a business combination are measured initially at their fair values at the acquisition date. All acquisition costs associated with a transaction identified as a business combination are expensed as incurred.

Goodwill

Goodwill arises on the acquisition (or deemed acquisition) of subsidiaries and represents the excess of the consideration transferred over the Company's interest in the net fair value of the net identifiable assets, liabilities and contingent liabilities of the acquiree and the fair value of any non-controlling interest in the acquiree. Upon initial recognition, goodwill is allocated to the cash-generating unit to which it relates. The Company identifies a cash-generating unit ("CGU") as the smallest identifiable group of assets that generates cash inflows that are largely independent of the cash inflows from other assets or group of assets. For example, a CGU can be an individual property or a group of properties. Goodwill acquired in business combinations is allocated to the CGUs that are expected to benefit from the synergies of that business combination.

Goodwill impairment reviews are undertaken annually or more frequently if events or changes in circumstances indicate a potential impairment. The Company's goodwill impairment test is performed at the CGU level as it is the lowest level within the Company at which goodwill is monitored for internal management purposes. Any goodwill impairment is recognized immediately as an expense in the consolidated statements of comprehensive income in the period in which it arises and is not subsequently reversed.

Rental properties

The Company's rental properties consist of single-family rental homes and multi-family rental properties held to earn rental income.

At the time of the acquisition of a property, the Company applies judgment when determining if the acquisition is an asset acquisition or a business combination. The Company classifies its acquisitions as asset acquisitions when it acquires a single asset (or a group of similar assets) and it has not assumed any employees or acquired an operating platform. Where the Company has concluded that it has acquired an asset, the Company uses the asset purchase model whereby the initial cost of a rental property is comprised of its purchase price and any directly attributable expenditures. Directly attributable expenditures include transaction costs such as due diligence costs, appraisal fees, environmental fees, legal fees, land transfer taxes and brokerage fees.

Subsequent to initial recognition, rental properties are recorded at fair value in accordance with IAS 40, *Investment Property* ("IAS 40"). Fair value is determined based on a combination of internal and external processes and valuation techniques according to the valuation policy discussed in Note 6. Gains or losses arising from changes in the fair value and capitalized costs of rental properties are recorded in the consolidated statements of comprehensive income in the period in which they arise.

In determining whether certain costs are additions to the carrying amount of rental properties or period expenses, management applies judgment based on whether these costs are incurred to enhance the service potential of the property. All costs associated with upgrading and extending the economic life of the existing properties, including internal amounts that are directly attributable to a specific rental property, other than ordinary repairs and maintenance, are capitalized to rental property.

Rental income and operating expenses from rental properties are reported within rental revenue and direct operating expenses incurred for rental properties, respectively, in the consolidated statements of comprehensive income.

Foreign currency translation

Currency translation

Foreign currency transactions (Canadian dollar) are translated into U.S. dollars using exchange rates in effect at the date of the transaction. Monetary assets and liabilities denominated in Canadian dollars are translated into U.S. dollars using the exchange rate in effect at the measurement date. Non-monetary assets and liabilities denominated in Canadian dollars are translated into U.S. dollars are translated into U.S. dollars using the historical exchange rate or the exchange rate in effect at the measurement date for items recognized at FVTPL. Gains and losses arising from foreign exchange are included in the consolidated statements of comprehensive income.

Consolidated entities

For subsidiaries that are required to be consolidated, the results and financial position of those subsidiaries with a functional currency different from the presentation currency are translated into the presentation currency as follows:

- (i) assets and liabilities are translated at the closing rate at the date of the balance sheet;
- (ii) income and expenses are translated at average exchange rates. The Company uses monthly average exchange rates due to the volume of transactions each month; and

(iii) all resulting exchange differences are recognized in other comprehensive income.

Other assets

Other assets include fixed assets, leasehold improvements and right-of-use assets.

Fixed assets and leasehold improvements

Fixed assets (building, property-related systems software, vehicles, furniture and office equipment and computer equipment) and leasehold improvements are accounted for at cost less accumulated depreciation and impairment. Leasehold improvements are amortized on a straight-line basis over their useful lives, which are typically their lease terms. All other depreciation expense is recorded on a straight-line basis over the estimated useful lives of the fixed assets, as follows:

| Building | 30 years | |
|-----------------------------------|-----------|--|
| Property-related systems software | 15 years | |
| Vehicles | 5 years | |
| Furniture and office equipment | 2–7 years | |
| Computer equipment | 2–7 years | |
| Computer software | 3 years | |

The estimated useful lives of fixed assets are reviewed and adjusted, if appropriate, at each financial year-end. As described below under Impairment of non-financial assets, fixed assets are also reviewed at each balance sheet date to determine whether there is an indication of impairment.

Right-of-use assets and lease liabilities

At the lease commencement date, a right-of-use asset and lease liability are recognized on the consolidated balance sheets for all leases, with the exception of short-term and low-value leases. The right-of-use assets and lease liabilities are initially measured at the present value of the lease payments.

Lease payments are apportioned between the implicit finance charge and the implicit repayment of the lease liability so as to achieve a constant rate of interest on the remaining balance of the liability. Finance charges are recognized in the consolidated statements of comprehensive income using the effective interest method.

Right-of-use assets are amortized on a straight-line basis over their lease terms and are accounted for at cost less accumulated amortization and reviewed at each balance sheet date to determine whether there is an indication of impairment.
Intangible assets

Intangible assets include capitalized placement fees, customer relationship and contractual development fees.

Placement fees represent costs incurred to secure investment management contracts. Performance fee rights represent costs incurred to obtain rights to receive future performance fees from joint venture projects. These are accounted for as intangible assets carried at cost less accumulated amortization. Amortization is recorded using the straight-line method and is based on the estimated useful lives of the associated joint ventures, which are generally eight years.

The customer relationship intangible relates to the Company's ownership of The Johnson Companies LP ("Johnson"), in which Tricon owns a 50.1% interest, and represents an estimate of the potential management fees, development fees and commissions that Tricon could collect, based on potential future projects resulting from Johnson's existing customer relationships at the time of the acquisition of Johnson, and as such are considered to be definite-life intangibles. Similarly, the contractual development fee intangibles from Johnson represent an estimate of the future lot development fees and commissions that Tricon expects to collect over the lives of the projects that Johnson managed at the time of acquisition. They are amortized by project over the estimated periods that the Company expects to collect these fees, which is approximately seven years for both management fees and lot development fees.

Impairment of non-financial assets

Assets that are subject to amortization and depreciation are reviewed for impairment whenever events or changes in circumstances indicate that the carrying amount may not be recoverable. An impairment loss is recognized for the amount by which the asset's carrying amount exceeds its recoverable amount. The recoverable amount is the higher of an asset's fair value less costs to sell and value in use. For the purposes of assessing impairment, assets are grouped at the lowest CGU level. Non-financial assets are reviewed for possible impairment or reversal of a previously recorded impairment as at each reporting date.

Financial instruments

Financial assets

The Company's financial assets are comprised of cash, restricted cash, amounts receivable, derivative financial instruments and investments in for-sale housing accounted for as portfolio investments. Financial assets within the scope of IFRS 9 are initially measured at fair value and subsequently classified and measured in one of three categories in accordance with IFRS 9: amortized cost, fair value through other comprehensive income ("FVOCI") or FVTPL.

Transaction costs related to derivative financial instruments and investments in for-sale housing are expensed as incurred and charged to income within the consolidated statements of comprehensive income.

Gains and losses arising from changes in the fair value of investments in for-sale housing are presented in the consolidated statements of comprehensive income within income from investments in for-sale housing. Gains and losses arising from changes in the fair value of derivative financial instruments are presented in the consolidated statements of comprehensive income together with gains and losses arising from changes in the fair value of other liabilities.

Financial assets and liabilities classified and measured at FVTPL are presented within changes in operating assets and liabilities in the consolidated statements of cash flows.

Financial assets are derecognized only when the contractual rights to the cash flows from the financial assets expire or the Company transfers substantially all of the risks and rewards of ownership.

The Company assesses, at each balance sheet date, whether or not there is an expected credit loss with respect to amounts receivable. If in a subsequent period the amount of the impairment loss decreases and the decrease can be related objectively to an event occurring after the impairment was recognized, the previously recognized impairment loss is reversed, to the extent that the carrying value of the receivable does not exceed its amortized cost at the reversal date. Any subsequent reversal of an impairment loss is recognized in net income.

Financial liabilities

Financial liabilities within the scope of IFRS 9 are initially measured at fair value and subsequently classified and measured at FVTPL or amortized cost, as appropriate.

A financial liability is derecognized when the obligation under the liability is discharged or cancelled, or expires.

The Company's financial liabilities consist of amounts payable and accrued liabilities, resident security deposits, dividends payable, debt, convertible debentures, Due to Affiliate, derivative financial instruments, limited partners' interests in rental business and other liabilities.

Interest expense is accounted for using the effective interest rate method.

The effective interest rate method is a method of calculating the amortized cost of a financial asset or financial liability and of allocating the interest income or interest expense over the expected life of the instrument. The effective interest rate is the rate that discounts estimated future cash payments or receipts throughout the expected life of the financial instrument, or a shorter period where appropriate, to the net carrying amount of the financial asset or financial liability. When calculating the effective interest rate, the Company estimates cash flows considering all contractual terms of the financial instrument but does not consider future credit losses. The calculation includes all fees paid or received between parties to the contract that are an integral part of the effective interest rate, transaction costs and all other premiums or discounts.

Gains or losses from the modification of borrowing terms during the year are recognized over the remaining life of the borrowing by adjusting the effective interest rate, on the basis that the terms and conditions of the liability remained largely unchanged. Should the modification be considered substantial, the original financial liability is derecognized and a new financial liability is recognized at fair value.

Convertible debentures

Convertible debentures issued by the Company are comprised of convertible unsecured subordinated debentures that can be converted to share capital at the option of the holder. The Company may settle the conversion right in cash in lieu of common shares unless the holder has explicitly indicated that they do not wish to receive cash. The cash settlement amount depends on the weighted average trading price of the common shares of the Company. This settlement option requires the Company to record the conversion option as a derivative financial instrument measured at fair value at each reporting period, with changes in fair value recorded in the consolidated statements of comprehensive income.

In addition, the debentures contain a redemption option, subject to several conditions, which allows the Company to redeem the debentures, in whole or in part, and the Company may settle the redemption option either in cash at par plus accrued and unpaid interest or in common shares, with the number of common shares to be issued depending on the weighted average trading price of the common shares of the Company. The redemption option is recorded as a derivative financial instrument measured at fair value at each reporting period, with changes in fair value recorded in the consolidated statements of comprehensive income.

The host liability component of a compound financial instrument is recognized initially at the fair value of a similar liability that does not have an equity conversion option. The conversion and redemption options are considered to be interrelated and therefore are treated as a single compound embedded derivative which is recognized at fair value.

Any directly attributable transaction costs are allocated entirely to the host liability component.

Derivative financial instruments

Derivative financial instruments are initially recognized at fair value on the date a derivative contract is entered into and are subsequently remeasured at fair value with the resulting gain or loss reflected in net income. The Company has two derivative financial instruments: (i) the conversion and redemption options related to its outstanding convertible debentures; and (ii) the mandatory prepayment provision related to the Due to Affiliate, along with the exchange and redemption provisions of the underlying preferred units (Note 20). Derivatives are valued using model calibration. Inputs to the valuation model are determined from observable market data wherever possible, including prices available from exchanges, over-the-counter markets and consensus pricing. Certain inputs may not be observable in the market directly, but can be determined from observable prices via model calibration procedures or estimated from historical data or other sources. Any directly attributable transaction costs are allocated between the derivative and the host liability component, and the portion attributed to the derivative is expensed in the consolidated statements of comprehensive income.

Offsetting financial instruments

Financial assets and liabilities are offset and the net amount is reported on the consolidated balance sheets when there is a legally enforceable right to offset the recognized amounts and there is an intention to settle on a net basis or realize the asset and settle the liability simultaneously. As of December 31, 2020, the Company does not have any assets or liabilities that are subject to an offsetting agreement.

Limited partners' interests in rental business

The interests of the limited partners in SFR JV-1 Holdings LP, SFR JV-1 REIT 1 LLC, SFR JV-1 REIT 2 LLC, SFR JV-1 Equity LLC and SFR JV-1 LP (collectively, "SFR JV-1") are recognized as financial liabilities in accordance with IAS 32, *Financial Instruments: Presentation* ("IAS 32"). Limited partners' interests in rental business are recorded at fair value through profit or loss and reflect the fair value of the underlying investments in SFR JV-1, along with any contributions by and distributions to limited partners during the period. Changes in the fair value of the limited partners' interests in rental business are reflected in the consolidated statements of comprehensive income.

(in thousands of U.S. dollars, except per share amounts and percentage amounts)

Cash

Cash includes cash deposited in banks. The Company maintains its cash in financial institutions with high credit quality in order to minimize its credit loss exposure.

Restricted cash

Restricted cash primarily consists of resident security deposits held by the Company in separate bank accounts, as well as property tax reserves, capital reserves, and collateralized rent payment receipts held in bank accounts controlled by lenders.

Share capital

Common shares are classified as equity. Incremental costs directly attributable to the issuance of new shares are shown as a deduction, net of tax, from the proceeds.

Where the Company purchases its equity share capital to settle restricted share awards or for cancellation, the consideration paid, including any directly attributable incremental costs, is deducted from equity attributable to the Company's equity holders.

Earnings per share

Basic

Basic earnings per share is determined using the weighted average number of shares outstanding including vested deferred share units, taking into account on a retrospective basis any increases or decreases caused by share splits or reverse share splits occurring after the reporting period, but prior to the consolidated financial statements being authorized for issue.

Diluted

The Company considers the effects of stock compensation, convertible debentures and exchange rights in connection with the preferred unit issuance of Tricon PIPE LLC in calculating diluted earnings per share. Diluted earnings per share is calculated by adjusting net income attributable to shareholders of the Company and the weighted average number of shares outstanding based on the assumption of the conversion of all potentially dilutive shares on a weighted average basis from the beginning of the year or, if later, the date the stock compensation, convertible debentures or conversion rights were issued to the balance sheet date.

Dividends

Dividends on common shares are recognized in the consolidated financial statements in the period in which the dividends are approved by Tricon's Board of Directors.

Current and deferred income taxes

Income tax expense includes current and deferred income taxes. Income tax expense is recognized in the consolidated statements of comprehensive income, except to the extent that it relates to items recognized directly in equity, in which case the tax is also recognized directly in equity.

Current income taxes are the expected taxes payable on the taxable income for the period, using income tax rates enacted, or substantively enacted, at the end of the reporting period, and any adjustment to income taxes payable in respect of previous years. The Company uses the liability method to recognize deferred income taxes on temporary differences arising between the tax bases of assets and liabilities and their carrying amounts. Deferred income tax assets are only recorded if it is probable that they will be realized. Enacted or substantively enacted rates in effect at the consolidated balance sheet date that are expected to apply when the deferred income tax asset is realized or the deferred tax liability is settled are used to calculate deferred income taxes.

Deferred income tax assets and liabilities are offset when there is a legally enforceable right to offset current tax assets against current tax liabilities and when the deferred income tax assets and liabilities relate to income taxes levied by the same taxation authority on either the same taxable entity or different taxable entities where there is an intention to settle the balances on a net basis.

Revenue

Revenue from rental properties

Revenue recognition under a lease commences when a resident has a right to use the leased asset, which is typically when the resident takes possession of, or controls the physical use of, the leased property. Generally, this occurs on the lease commencement date.

Lease contracts with residents normally include lease and non-lease components, which may be bundled into one fixed gross lease payment. Lease revenue earned directly from leasing the homes and apartment suites is recognized and measured on a straight-line basis over the lease term in accordance with IFRS 16, *Leases* ("IFRS 16"). Leases for single-family rental homes and multi-family rental properties are generally for a term of one to two years.

Ancillary revenue is income the Company generates from providing services that are not primary rental revenue from a lease contract. Ancillary revenue includes pet fees, early termination fees and other service fees. Ancillary revenue is measured at the amount of consideration which the Company expects to receive in exchange for providing services to a resident. Ancillary revenue is included with revenue from rental properties in the consolidated statements of comprehensive income, and the details of revenue, including ancillary income, are discussed in Note 14.

In addition to revenue generated from the lease component, revenue from rental properties includes a non-lease component earned from the residents, which is recognized under IFRS 15, *Revenue from Contracts with Customers* ("IFRS 15"). Non-lease revenue includes property management services, such as repairs and maintenance performed on the properties. These services represent a single performance obligation and revenue is recognized over time as the services are provided, regardless of when the payment is received. Revenue from rental properties is allocated to non-lease components using a cost-plus margin approach whereby the Company separates the operating costs that pertain to the services provided to the residents and applies a reasonable profit margin.

The Company has concluded that it is the principal in all of its revenue arrangements since it is the primary obligor in all of the revenue arrangements, it has pricing latitude and it is also exposed to credit risks.

Revenue from private funds and advisory services

The Company's vertically integrated management platform provides asset management, property management and development management services.

The Company provides asset management services to joint venture partners and third-party investors for which it earns marketbased fees in connection with its businesses in the U.S. and Canada. These contractual fees are typically 1–2% of committed or invested capital throughout the lives of the Investment Vehicles under management. The Company may also earn performance fees once targeted returns are achieved by an Investment Vehicle. The Company recognizes performance fees only to the extent that it is highly probable that a significant amount of the cumulative revenue recognized will not reverse. Consideration for these services is variable as it is dependent upon the occurrence of a future event that includes the repayment of investor capital and a predetermined rate of return. Revenue from performance fees is typically earned and recognized towards the end of the life of an Investment Vehicle.

The Company also earns development management and advisory service fees from third parties and/or related parties. Development management and advisory services are satisfied over time. Revenues are recognized based on the best estimate of the amounts earned for those services, which typically reflects contractual fees of 2–5% of the sales price of single-family lots, residential land parcels and commercial land within master-planned communities, and 4–5% of overall development costs of Canadian multi-family rental apartments. The Company includes variable consideration in the revenues only to the extent that it is highly probable that a significant amount of the cumulative revenue recognized will not reverse. Specifically for Johnson, consideration for these services is variable as it is dependent upon the occurrence of a future event that is the sale of the developed property. Revenue is typically recognized as the development of the property is completed, and control has been transferred to the respective buyer. These management fees earned in exchange for providing development management and advisory services are billed upon the sale of the property.

The Company earns property management fees, leasing fees, acquisition and disposition fees, and construction management fees through its rental operating platform. These management services are satisfied over time and revenues are recognized as services are provided in accordance with IFRS 15.

Compensation arrangements

Stock option plan

The Company accounts for its stock option plan by calculating the fair value of the options as of the grant date using a Black-Scholes option pricing model and observable market inputs. This fair value is recognized as compensation cost using the graded vesting method over the vesting period of the options.

Annual Incentive Plan ("AIP")

The Company's AIP provides for an aggregate bonus pool based on the sum of all employees' individual AIP targets. The portion of the pool attributable to senior executive management is market-benchmarked and subject to an adjustment factor, as approved by the Board, of between 50% and 150%, based on the achievement of Company performance objectives determined by the Board at the beginning of each year. The final pool is then allocated among employees based on individual and collective performance. AIP awards will be made in cash and equity-based grants, with the proportion of equity-based awards being correlated to the seniority of an individual's role within the Company. Equity-based AIP awards are granted in a combination of deferred share units ("DSUs"), performance share units ("PSUs"), stock options and restricted shares, pursuant to the Company's Deferred Share Unit Plan ("DSUP"), Performance Share Unit Plan ("PSUP"), stock option plan and Restricted Share Plan, respectively.

Long-term incentive plan ("LTIP")

LTIP expense is generated from two sources: (i) 50% of the Company's share of performance fees or carried interest from Investment Vehicles, paid in cash when received; and (ii) 15% of the income from THP1 US (a for-sale housing Investment Vehicle), payable in DSUs which vest in equal tranches over a three-year period (previously a five-year period) pursuant to the LTIP as amended on May 6, 2019. Amounts under the LTIP are allocated among employees in accordance with the plan.

For the LTIP generated from the Company's share of performance fees or carried interest from certain Investment Vehicles, the Company estimates the LTIP liability by determining the performance fees at each reporting date based on the estimated fair value of the underlying investments. Changes in the LTIP liability are recognized in the consolidated statements of comprehensive income.

Directors' fees

One-half of each independent Director's base annual retainer is paid in DSUs which vest immediately upon their grant. An independent Director may also elect each year to receive a portion of the balance of his or her fees (including his or her base annual retainer and any additional retainer) in DSUs, which also vest on the date of their grant. Any remaining balance of such fees not payable in DSUs is paid in cash. The DSUs granted to Directors are governed by the DSUP.

Reportable segments

Tricon is comprised of four operating segments: Single-Family Rental, Multi-Family Rental, Residential Development and Private Funds and Advisory. Including the Company's corporate activities, there are five reportable segments for internal and external reporting purposes. The reportable segments are business units offering different products and services, and are managed separately due to their distinct operating natures. These five reportable segments have been determined by the Company's chief operating decision-makers (Note 31).

Accounting standards and interpretations adopted

Effective January 1, 2020, the Company has adopted amendments to IFRS 3, *Business Combinations*. The amendments provide further guidance on the determination of whether a transaction should be accounted for as a business combination or as an asset acquisition. The Company has also adopted amendments to IAS 1, *Presentation of Financial Statements* ("IAS 1"), and IAS 8, *Accounting Policies, Changes in Accounting Estimates and Errors*, which provide further clarification on the definition of materiality, specifying that materiality will depend on the nature or magnitude of information. The adoption of these standards did not have a significant impact on the Company's consolidated financial statements.

Accounting standards and interpretations issued but not yet adopted

In August 2020, the International Accounting Standards Board ("IASB") issued amendments to IFRS 9, *Financial Instruments*, IAS 39, *Financial Instruments: Recognition and Measurement*, IFRS 7, *Financial Instruments: Disclosure*, IFRS 4, *Insurance Contracts*, and IFRS 16, *Leases*, as part of phase 2 of its project related to interest rate benchmark reform. The amendments address issues that might affect financial reporting after the reform of an interest rate benchmark, including its replacement with alternative benchmark rates. The amendments are effective for annual periods beginning on or after January 1, 2021, with earlier application permitted.

The Company is currently finalizing its assessment of the impact of the adoption of the phase 2 amendments. As a result of the adoption, it is not expected that interest rate benchmark reform will have a material impact on the Company's financial statements.

In January 2020, the IASB issued amendments to IAS 1 to provide a more general approach to the classification of liabilities under IAS 1 based on the contractual arrangements in place at the reporting date. The amendments to IAS 1 are effective for annual reporting periods beginning on or after January 1, 2023.

There are no other standards, interpretations or amendments to existing standards that are not yet effective that are expected to have a material impact on the consolidated financial statements of the Company.

for the year ended December 31, 2020 (in thousands of U.S. dollars, except per share amounts and percentage amounts)

4. CRITICAL ACCOUNTING ESTIMATES AND JUDGMENTS

The Company makes estimates and assumptions concerning the future. The resulting accounting estimates will, by definition, seldom equal the related actual results. The following are the accounting policies subject to judgments and estimation uncertainty that management believes could have a significant risk of causing material adjustments to the amounts recognized in the consolidated financial statements. Actual results could differ from these estimates and the differences may be material.

Significant estimates

Income taxes

The determination of the Company's income and other tax liabilities requires interpretation of complex laws and regulations often involving multiple jurisdictions. Significant estimates are required in determining the Company's consolidated income tax provision. There are many transactions and calculations for which the ultimate tax determination is uncertain. The Company recognizes liabilities for anticipated tax audit issues based on estimates of whether additional taxes will be due. Where the final tax outcome of these matters is different from the amounts that were initially recorded, such differences will impact the current tax and deferred tax provisions. Furthermore, deferred income tax balances are recorded using enacted or substantively enacted future income tax rates. Changes in enacted income tax rates are not within the control of management. However, any such changes in income tax rates may result in actual income tax amounts that may differ significantly from estimates recorded in deferred tax balances.

Valuation of rental properties

Fair value is determined using independent external valuations prepared by management's specialists or detailed internal valuations prepared by management using market-based assumptions, each in accordance with recognized valuation techniques as set out in Note 6. Significant estimates used in determining the fair value of the Company's rental properties include estimating, among other things, future stabilized net operating income, capitalization rates, discount rates, and other future cash flows applicable to rental properties (all considered Level 3 inputs), as well as market comparables based on recent transaction prices. A change to any one of these inputs could significantly alter the fair value of a rental property. In addition, the novel coronavirus ("COVID-19") pandemic and related market and economic uncertainty that occurred in 2020 has had a significant impact on estimates used in the valuation of the rental properties and this impact may continue into 2021. Management will continue to monitor the situation and its impact on the Company.

Fair value and impairment of financial instruments

Certain financial instruments are recorded in the Company's consolidated balance sheets at values that are representative of or approximate fair value.

The fair values of the Company's investments in for-sale housing are determined using the valuation methodologies described in Note 9. By their nature, these valuation techniques require the use of assumptions that are mainly based on market conditions existing at the end of each reporting period. Changes in the underlying assumptions could materially impact the determination of the fair value of a financial instrument. Imprecision in determining fair value using valuation techniques may affect the investment income recognized in a particular period. Any significant changes to the inputs and assumptions owing to the COVID-19 pandemic as discussed above could further impact the valuation of the for-sale housing investments in future periods.

Fair value of incentive plans

Management is required to make certain assumptions and to estimate future financial performance in order to estimate the fair value of incentive plans at each consolidated balance sheet date. Significant estimates and assumptions relating to such incentive plans are disclosed in Notes 3 and 30. The LTIP requires management to estimate future non-IFRS earnings measures, namely future performance fees relative to each Investment Vehicle. Future non-IFRS measures are estimated based on current projections, and are updated at least annually, taking into account actual performance since inception.

Goodwill impairment

Assessment of impairment is based on management's judgment of whether there are internal and external factors that would indicate that an asset or CGU is impaired. The determination of the Company's goodwill impairment involves management's significant estimates and assumptions with respect to future cash flows, growth rates and discount rates of the underlying CGU. The risk premiums expected by market participants related to uncertainties about the industry and assumptions relating to future cash flows may differ, depending on economic conditions and other events. Changes in any of these underlying assumptions could materially affect the assessment of the recoverable value of a CGU (Note 12).

Due to Affiliate

In connection with the Due to Affiliate transaction, the Company made certain key assumptions about the structure, cash flow and terms of the issued instruments. In addition, management was required to make significant estimates in determining the initial recognition and measurement of the Due to Affiliate and related derivative instruments (Notes 19 and 20).

NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

for the year ended December 31, 2020 (in thousands of U.S. dollars, except per share amounts and percentage amounts)

Significant judgments

Acquisition of rental properties

The Company's accounting policies relating to rental properties are described in Note 3. In applying these policies, judgment is exercised in determining whether certain costs are additions to the carrying amount of a rental property and whether properties acquired are considered to be asset acquisitions or business combinations. Should the purchase meet the criteria of a business combination, then transaction costs such as appraisal and legal fees are expensed immediately and included in the consolidated statements of comprehensive income. If the purchase is an asset acquisition, transaction costs form part of the purchase price and earnings are not immediately affected.

Basis of consolidation

The consolidated financial statements of the Company include the accounts of Tricon and its wholly-owned subsidiaries, as well as entities over which the Company exercises control on a basis other than majority ownership of voting interests within the scope of IFRS 10. Judgment is applied in determining if an entity meets the criteria of control as defined in the accounting standard.

Investments in joint ventures and joint arrangements

The Company makes judgments in determining the appropriate accounting for investments in other entities. These judgments include determining the significant relevant activities and assessing the level of influence Tricon has over the activities through contractual arrangements. In addition, the Company also determines whether Tricon's rights and obligations are directly related to the assets and liabilities of the arrangement or to the net assets of the joint arrangement.

CGU determination for goodwill impairment assessment

The determination of CGUs is based on management's judgment and is an assessment of the smallest group of assets that generate cash inflows independently of other assets. Factors considered include whether an active market exists for the output produced by the asset or group of assets as well as how management monitors and makes decisions about the Company's operations.

5. BUSINESS COMBINATIONS

As discussed in Note 2, the Company successfully completed its transition to a rental housing company effective January 1, 2020, and as a result, it was required to apply the acquisition method of accounting in accordance with IFRS 3 to all subsidiaries that were previously measured at FVTPL (the "Deemed Acquisition"), as discussed in further detail below.

Deemed acquisition of single-family and multi-family rental businesses

On the Transition Date, Tricon SF Home Rental ULC and its wholly-owned subsidiaries, along with SFR JV-1 (collectively, the "single-family rental" business), and TLR Saturn Master LP and its wholly-owned subsidiaries (collectively, the "multi-family rental" business) were deemed to have been acquired by the Company and were accounted for as business combinations in accordance with IFRS 3.

The following table summarizes the deemed consideration paid and the estimates of the fair values of identified assets acquired and liabilities assumed from both businesses on the Transition Date. The deemed consideration paid reflects the fair value of the Company's interests in the single-family and multi-family rental businesses as portfolio investments immediately prior to the Transition Date.

| (in thousands of U.S. dollars) | Single-Family Rental ⁽¹⁾ | Multi-Family Rental | | |
|--|-------------------------------------|---------------------|-----------|--|
| Deemed consideration transferred | \$ 1,270,293 | \$ | 429,060 | |
| Recognized amounts of assets acquired | | | | |
| Cash | \$ 18,948 | \$ | 2,537 | |
| Restricted cash | 67,519 | | 16,563 | |
| Amounts receivable | 1,033 | | 3,436 | |
| Derivative financial instruments | 28 | | - | |
| Prepaid expenses and deposits | 9,829 | | 720 | |
| Rental properties | 4,337,681 | | 1,344,844 | |
| Deferred income tax assets | 40,000 | | - | |
| Other assets | 11,255 | | 90 | |
| Total identifiable assets | \$ 4,486,293 | \$ [.] | 1,368,190 | |
| Recognized amount of liabilities assumed | | | | |
| Amounts payable and accrued liabilities | \$ 49,623 | \$ | 20,759 | |
| Resident security deposits | 30,094 | | 2,031 | |
| Other liabilities | 5,435 | | - | |
| Debt | 2,716,840 | | 916,340 | |
| Deferred income tax liabilities | 157,741 | | 79,112 | |
| Limited partners' interests in rental business | 285,774 | | - | |
| Total identifiable liabilities | 3,245,507 | | 1,018,242 | |
| Total identifiable assets and liabilities | \$ 1,240,786 | \$ | 349,948 | |
| Goodwill | 29,507 | | 79,112 | |
| Total | \$ 1,270,293 | \$ | 429,060 | |

(1) The deemed consideration transferred reflects the fair value of the Company's interests in the single-family rental business as a portfolio investment immediately prior to the Transition Date, net of the Company's deferred tax liabilities associated with the investment of \$94,714.

The purchase price allocation resulted in \$29,507 and \$79,112 of goodwill being recognized from the Deemed Acquisition of the single-family rental and multi-family rental businesses, respectively, due to the recognition of deferred tax liabilities because the tax bases of the net assets are lower than their acquisition date fair values.

Ownership interests in SFR JV-1 are in the form of limited partnership interests which are classified as liabilities under the provisions of IAS 32. Limited partners' interests in rental business are measured as a percentage of net assets acquired.

The following table presents the changes in the limited partners' interests in rental business balance for the year ended December 31, 2020, representing third-party limited partners' 66.33% ownership interests in the net assets of SFR JV-1.

| Balance, end of year | \$ 356,305 |
|--|------------|
| Net change in fair value of limited partners' interests in rental business | 50,581 |
| Distributions | (46,162) |
| Contributions | 66,112 |
| Balance, beginning of year ⁽¹⁾ | \$ 285,774 |
| For the year ended December 31 | 2020 |
| (in thousands of U.S. dollars) | |

(1) The initial balance was recognized as a result of the Deemed Acquisition of the single-family rental business.

Deemed acquisition of Canadian multi-family development business under joint operations (proportionate consolidation)

In the Company's Canadian multi-family development business, TLR Investment LP through its wholly-owned subsidiaries (collectively, the "Canadian multi-family development" business) held a 50% and 25% direct ownership interest, respectively, of the properties known as The James (Scrivener Square) and The Shops of Summerhill, which were classified as joint operations under IFRS 11. As at the Transition Date, the Company's proportionate interests in these properties were deemed to be acquired by the Company and were treated as business combinations in accordance with IFRS 3.

The following table summarizes the deemed consideration paid for the Canadian multi-family development business and the estimates of the fair values of identified assets acquired and liabilities assumed, on a proportionate basis, from the Canadian multi-family development business on the Transition Date.

| (in thousands of U.S. dollars) | The James (Scrivener Square) | The Shops of Summerhill | Other entities ⁽¹⁾ | Canadian multi-family developments |
|--|---------------------------------|----------------------------|----------------------------------|--|
| Deemed consideration transferred | \$ 14,682 | \$ 7,339 | \$ 74,851 | \$ 96,872 |
| Recognized amounts of assets acquired | | | | |
| Cash | \$ 420 | \$ 65 | \$ 229 | \$ 714 |
| Amounts receivable | 131 | 51 | 248 | 430 |
| Prepaid expenses and deposits | 12 | 1 | - | 13 |
| Other assets | - | - | 49 | 49 |
| Investments in Canadian multi-family developments ⁽²⁾ | - | - | 75,141 | 75,141 |
| Canadian development properties | 25,170 | 10,455 | - | 35,625 |
| Total identifiable assets | \$ 25,733 | \$ 10,572 | \$ 75,667 | \$ 111,972 |
| Recognized amount of liabilities assumed | | | | |
| Amounts payable and accrued liabilities | \$ 272 | \$ 84 | \$ 816 | \$ 1,172 |
| Debt | 10,779 | 3,149 | _ | 13,928 |
| Total identifiable liabilities | 11,051 | 3,233 | 816 | 15,100 |
| Total identifiable assets and liabilities – proportionate basis | \$ 14,682 | \$ 7,339 | \$ 74,851 | \$ 96,872 |

(1) Other entities include Tricon Lifestyle Rentals LP and its wholly-owned subsidiaries.

(2) Includes Tricon's investment in The Selby.

On June 23, 2020, Tricon acquired the remaining 50% and 75% ownership interests in The James and The Shops of Summerhill, respectively (Note 8). The acquisition of the remaining ownership is considered to be an asset acquisition as it does not meet the definition of a business combination as prescribed by IFRS 3.

6. RENTAL PROPERTIES

The Company's Valuation Committee is responsible for fair value measurements included in the financial statements, including Level 3 measurements. The valuation processes and results are reviewed and approved by the Valuation Committee once every quarter, in line with the Company's quarterly reporting dates. The Valuation Committee consists of individuals who are knowledgeable and have experience in the fair value techniques for the real estate properties held by the Company. The Valuation Committee decides on the appropriate valuation methodologies for new real estate properties and contemplates changes in the valuation methodology for existing real estate holdings. Additionally, the Valuation Committee analyzes the movements in each property's (or group of properties') value, which involves assessing the validity of the inputs applied in the valuation.

The following table presents the changes in the rental property balances for the year ended December 31, 2020.

| | December 31, 2020 | | | | | | |
|--|----------------------|---------------------|--------------|--|--|--|--|
| (in thousands of U.S. dollars) | Single-Family Rental | Multi-Family Rental | Total | | | | |
| Initial recognition on Deemed Acquisition (Note 5) | \$ 4,337,681 | \$ 1,344,844 | \$ 5,682,525 | | | | |
| Acquisitions ⁽¹⁾ | 356,514 | - | 356,514 | | | | |
| Capital expenditures | 93,568 | 9,067 | 102,635 | | | | |
| Dispositions | (18,070) | - | (18,070) | | | | |
| Fair value adjustments | 220,849 | (22,535) | 198,314 | | | | |
| Balance, end of year | \$ 4,990,542 | \$ 1,331,376 | \$ 6,321,918 | | | | |

(1) The total purchase price includes \$1,913 of capitalized transaction costs in relation to the acquisitions.

The Company used the following techniques to determine the fair value measurements included in the consolidated financial statements categorized under Level 3.

Single-family rental homes

Valuation methodology

The fair value of single-family rental homes is typically determined by using a combination of Broker Price Opinion ("BPO") and the Home Price Index ("HPI") methodologies. In addition, homes that were purchased in the last three to six months (or properties purchased in the year that are not yet stabilized) from the reporting date are recorded at their purchase price plus the cost of capital expenditures as the home values typically do not change materially in the short term, and capital expenditures generally do not significantly impact values in those periods.

BPOs are quoted by independent brokers who hold active real estate licenses and have market experience in the locations and segments of the properties being valued. The brokers value each property based on recent comparable sales and active comparable listings in the area, assuming the properties were all renovated to an average standard in their respective areas. The Company typically obtains a BPO for a property once every three years or when a home is included in a new debt facility.

The HPI methodology is used to update the value, on a quarterly basis, of single-family rental homes that were most recently valued using a BPO as well as single-family rental homes held for more than six months following initial acquisition. The HPI is calculated based on a repeat-sales model using large real estate information databases compiled from public records. The Company uses the twelve-month trailing average HPI change to update the value of its single-family rental homes. The quarterly HPI change is then applied to the previously recorded fair value of the rental homes. The data used to determine the fair value of the Company's single-family rental homes is specific to the zip code in which the property is located.

The Company performed a valuation at November 30, 2020 for rental homes acquired prior to October 1, 2020, according to its valuation policy and based on the best information available. HPI growth continued across all markets during the year at 4.2% (net of capital expenditures) compared to 3.0% during the prior year. There were 6,980 homes valued using the BPO method during the year. The combination of the HPI and BPO methodologies resulted in a fair value gain of \$220,849 for the year ended December 31, 2020. Management has assessed the impact of any market changes that occurred subsequent to the date of the valuation and has determined that the values were valid as of December 31, 2020.

Sensitivity

The weighted average of the quarterly HPI change was 1.5%. If the change in the quarterly HPI increased or decreased by 0.5%, the impact on the rental properties at December 31, 2020 would be \$19,294 and (\$19,294), respectively.

Multi-family rental properties

Valuation methodology

Fair value is determined using independent valuations prepared by management's specialists or detailed internal valuations prepared by management using market-based assumptions, each in accordance with recognized valuation techniques. The Company utilizes the direct income capitalization approach to determine the fair value of its multi-family rental properties. This method requires that a projected stabilized net operating income ("NOI") for each property is divided by the appropriate capitalization rate to determine a property's fair value. NOI is calculated as a one-year income forecast based on rental income from current leases and key assumptions about rental income, vacancies and inflation rates, among other factors, less property operating costs. Fair value also considers any forecasted capital expenditures within the year to maintain the property in good condition. Given the short-term nature of residential leases (typically one to two years), revenue and costs are not discounted. The capitalization rate is determined for each property based on location, size and quality/vintage of the property and takes into account market information related to recent sales of comparable buildings within a similar geographic location.

In applying the Company's valuation policies, external valuations are obtained from third-party valuation professionals on a rotational basis based on a cross-section of properties from different geographic locations and markets across the Company's multi-family rental portfolio, as determined by management and approved by the Valuation Committee. The fair value of the remainder of the Company's rental properties is determined internally by management using the same assumptions and valuation techniques as those used by the external valuation professionals.

Management assessed changes in capitalization rates in each of the markets in which it owns multi-family rental properties by consulting third-party data based on market transactions. In contrast to the single-family rental market, multi-family rental market conditions were negatively impacted by the COVID-19 pandemic. A decline in demand for multi-family living contributed to a downward adjustment in the stabilized NOI assumptions, which led to a fair value loss of \$22,535 for the year ended December 31, 2020. Management will continue to monitor rental market conditions that could adversely affect the valuation of the Company's rental properties. The key valuation assumptions for the Company's multi-family rental properties are set out below.

| | December 31, 2020 | December 31, 2019 |
|--|-------------------|-------------------|
| Capitalization rates – range | 4.00% to 5.50% | 4.50% to 5.00% |
| Capitalization rate – weighted average | 4.76% | 4.71% |

Sensitivity

Any fluctuations in either NOI from rental operations or the capitalization rate could significantly alter the fair value of the properties. Generally, an increase in stabilized NOI will result in an increase to the fair value of a rental property. An increase in the capitalization rate will result in a decrease to the fair value of a rental property. The capitalization rate magnifies the effect of a change in NOI, with a lower capitalization rate causing more change in fair value than a higher capitalization rate when applied to NOI. The table below summarizes the impact of changes in both the capitalization rates and NOI on the fair value of the Company's multi-family rental properties.

| | | Net operating income | | | | | | | | |
|---------------------|-----------|----------------------|--------------|-----------|------------|--|--|--|--|--|
| Capitalization rate | -3% | -1% | As projected | +1% | +3% | | | | | |
| -0.25% | \$ 31,497 | \$ 59,472 | \$ 73,459 | \$ 87,447 | \$ 115,422 | | | | | |
| As reported | (39,759) | (13,253) | - | 13,253 | 39,759 | | | | | |
| +0.25% | (103,903) | (78,720) | (66,129) | (53,537) | (28,354) | | | | | |

7. INVESTMENTS IN CANADIAN MULTI-FAMILY DEVELOPMENTS

The Company has entered into certain arrangements in the form of jointly controlled entities and investments in associates for various Canadian multi-family rental developments, as well as The Selby, an income-producing multi-family rental property in Toronto. Joint ventures represent development properties held in partnership with third parties where decisions relating to the relevant activities of the joint venture require the unanimous consent of the partners. These arrangements are accounted for under the equity method.

The following table presents the change in the balance of investments in joint ventures and associates for the year ended December 31, 2020.

| n thousands of U.S. dollars) Dece | |
|---|-----------|
| Initial recognition on Deemed Acquisition (Note 5) | \$ 75,141 |
| Advances | 4,294 |
| Distributions | (935) |
| Income from investments in Canadian multi-family developments | 14,124 |
| Translation adjustment | 2,244 |
| Total investments in joint ventures and associates | \$ 94,868 |

The following tables present the ownership interests and carrying values of the Company's equity-accounted investments. The financial information below discloses each investee at 100% and at Tricon's ownership interests in the net assets of the investee.

| | December 31, 2020 | | | | | | | | | | | | |
|--------------------------------|-------------------|----------------------------|----|-------------------|----|----------------------|----|------------------------|----|---------------------------|----|------------|---------------------------------------|
| (in thousands of U.S. dollars) | Location | Tricon's ownership % | | Current assets | N | on-current assets | | Current liabilities | N | on-current liabilities | 1 | Net assets | n's share et assets ⁽¹⁾ |
| Joint ventures | | | | | | | | | | | | | |
| WDL 3/4/7 LP | Toronto, ON | 33% | \$ | 1,050 | \$ | 70,918 | \$ | 7,813 | \$ | 35,454 | \$ | 28,701 | \$ 9,575 |
| WDL 8 LP | Toronto, ON | 33% | | 6,659 | | 112,488 | | 8,083 | | 88,635 | | 22,429 | 7,483 |
| WDL 20 LP | Toronto, ON | 33% | | 770 | | 45,697 | | 24 | | 43,653 | | 2,790 | 937 |
| DKT B10 LP ⁽²⁾ | Toronto, ON | 33% | | 2,683 | | 2,551 | | 966 | | _ | | 4,268 | 2,994 |
| 6–8 Gloucester LP | | | | | | | | | | | | | |
| (The Ivy) | Toronto, ON | 47% | | 3,587 | | 40,799 | | 3,091 | | 6,676 | | 34,619 | 16,398 |
| Labatt Village | | | | | | | | | | | | | |
| Holding LP ⁽³⁾ | Toronto, ON | 38% | | - | | 43,160 | | 16 | | - | | 43,144 | 16,180 |
| | | | | 14,749 | | 315,613 | | 19,993 | | 174,418 | | 135,951 | 53,567 |
| Associates | | | | | | | | | | | | | |
| 592 Sherbourne LP | | | | | | | | | | | | | |
| (The Selby) | Toronto, ON | 15% | | 12,988 | | 252,065 | | 2,201 | | 126,008 | | 136,844 | 19,913 |
| 57 Spadina LP | | | | | | | | | | | | | |
| (The Taylor) | Toronto, ON | 30% | | 448 | | 113,215 | | 3,419 | | 39,724 | | 70,520 | 21,388 |
| | | | | 13,436 | | 365,280 | | 5,620 | | 165,732 | | 207,364 | 41,301 |
| Total | | | \$ | 28,185 | \$ | 680,893 | \$ | 25,613 | \$ | 340,150 | \$ | 343,315 | \$ 94,868 |

(1) Tricon's share of net assets of \$94,868 is comprised of \$93,541 as per the investees' financial statements plus \$1,327 of fair value differences arising from the initial recognition on January 1, 2020 and foreign exchange translation adjustments.

(2) Tricon's share of net assets of DKT B10 LP includes the purchase price paid to third-party partners for a one-third ownership interest in the partnership.

(3) Labatt Village Holding LP has an 80% ownership interest in the Labatt Village LP project partnership, and therefore Tricon has a 30% effective interest in the project.

| | For the year ended December 31, 2020 | | | | | | | | | | | |
|--------------------------------|--------------------------------------|----------------------------|-----------|------------|---------------------|--|------------------------------|--|--|--|--|--|
| (in thousands of U.S. dollars) | Location | Tricon's ownership % | Revenue | Expenses | Fair value gains | Net and other comprehensive income | Tricon's share of net income | | | | | |
| Joint ventures | | | | | | | | | | | | |
| WDL 3/4/7 LP | Toronto, ON | 33% | \$ 198 | \$ (104) | \$ 21,742 | \$ 21,836 | \$ 7,279 | | | | | |
| WDL 8 LP | Toronto, ON | 33% | - | (75) | 15,299 | 15,224 | 5,074 | | | | | |
| WDL 20 LP | Toronto, ON | 33% | _ | (2) | _ | (2) | (1) | | | | | |
| DKT B10 LP | Toronto, ON | 33% | _ | (16) | _ | (16) | (5) | | | | | |
| 6–8 Gloucester LP | | | | | | | | | | | | |
| (The lvy) | Toronto, ON | 47% | _ | (2) | _ | (2) | (1) | | | | | |
| Labatt Village | | | | | | | | | | | | |
| Holding LP | Toronto, ON | 38% | - | (34) | (345) | (379) | (142) | | | | | |
| | | | 198 | (233) | 36,696 | 36,661 | 12,204 | | | | | |
| Associates | | | | | | | | | | | | |
| 592 Sherbourne LP | | | | | | | | | | | | |
| (The Selby) | Toronto, ON | 15% | 10,763 | (5,791) | _ | 4,972 | 746 | | | | | |
| 57 Spadina LP | | | | | | | | | | | | |
| (The Taylor) | Toronto, ON | 30% | - | (20) | 3,933 | 3,913 | 1,174 | | | | | |
| | | | 10,763 | (5,811) | 3,933 | 8,885 | 1,920 | | | | | |
| Total | | | \$ 10,961 | \$ (6,044) | \$ 40,629 | \$ 45,546 | \$ 14,124 | | | | | |

Based on the assessment of current economic conditions, there are no indicators of impairment of the Company's equity-accounted investments in Toronto as of December 31, 2020. Management will continue to monitor the situation as market conditions may change rapidly which could adversely affect the Company's underlying valuation of such investments.

8. CANADIAN DEVELOPMENT PROPERTIES

The Company's Canadian development properties include one development project (The James) and an adjacent commercial property (The Shops of Summerhill) in Toronto that were previously accounted for as joint operations (Note 2).

The following table presents the changes in the Canadian development properties balance for the year ended December 31, 2020.

| (in thousands of U.S. dollars) Dec | | 31, 2020 |
|--|------|----------|
| Initial recognition on Deemed Acquisition (Note 5) | \$ | 35,625 |
| Acquisitions | | 65,861 |
| Development expenditures | | 2,998 |
| Translation adjustment | | 5,534 |
| Balance, end of year | \$ 1 | 110,018 |

On June 23, 2020, Tricon acquired the remaining ownership interests of 50% and 75% in The James and The Shops of Summerhill, respectively, and began consolidating these entities. These two properties are recorded at fair value with changes in fair value recorded in the consolidated statements of comprehensive income in accordance with IAS 40, with the acquired portion added to the fair value of the properties based on the purchase price paid. The Company utilized the asset purchase model whereby the initial cost of a development property is comprised of its purchase price and any directly attributable expenditures, including transaction costs.

The following table summarizes the purchase price paid for each of the properties.

| (in thousands of U.S. dollars) | T۲ Scrivene) | ne James r Square) | Shops of ummerhill | Total | | |
|---|-----------------|-----------------------|--------------------|-------|----------|--|
| Canadian development properties | \$ | 40,669 | \$ 25,192 | \$ | 65,861 | |
| Net working capital | | (3,689) | (1,189) | | (4,878) | |
| Assumed debt and vendor take-back loans | | (34,156) | (19,184) | | (53,340) | |
| Cash paid | \$ | 2,824 | \$ 4,819 | \$ | 7,643 | |

Property values typically do not change materially in the short term, and development expenditures generally do not significantly impact values in the first twelve months after purchase. Accordingly, Canadian development properties acquired within the past twelve months are recorded at their purchase price plus the cost of development expenditures.

The Company earned \$791 of commercial rental income from The Shops of Summerhill for the year ended December 31, 2020, which is classified as Other income from Canadian development properties.

9. INVESTMENTS IN FOR-SALE HOUSING

The Company makes investments in for-sale housing via equity investments and loan advances. Advances made to investments are added to the carrying value when paid; distributions from investments are deducted from the carrying value when received.

In the year ended December 31, 2020, the Company recorded a cumulative fair value loss of \$61,776, primarily related to the risk of both extended timelines and a reduction in expected future cash flows from these investments brought on by the COVID-19 pandemic.

The following table presents the changes in the investments in for-sale housing for the years ended December 31, 2020 and December 31, 2019.

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|--|-------------------|-------------------|
| Balance, beginning of year | \$ 300,653 | \$ 307,564 |
| Advances | 3,408 | 35,389 |
| Distributions | (77,443) | (51,946) |
| (Loss) income from investments in for-sale housing | (61,776) | 9,646 |
| Balance, end of year | \$ 164,842 | \$ 300,653 |
| Internal debt instruments | \$ 13,937 | \$ 16,757 |
| Equity | 150,905 | 283,896 |
| Total investments in for-sale housing | \$ 164,842 | \$ 300,653 |

The investments are measured at fair value as determined by the Company's proportionate share of the fair value of each Investment Vehicle's net assets at each measurement date. The fair value of each Investment Vehicle's net assets is determined by the waterfall distribution calculations specified in the relevant governing agreements. The inputs into the waterfall distribution calculations include the fair values of the land development and homebuilding projects and working capital held by the Investment Vehicles. The fair values of the land development and homebuilding projects are based on appraisals prepared by external third-party valuators or on internal valuations using comparable methodologies and assumptions.

The residential real estate development business involves significant risks that could adversely affect the fair value of Tricon's investments in for-sale housing, especially in times of economic uncertainty. Quantitative information about fair value measurements of the investments uses the following significant unobservable inputs (Level 3):

| | | | December | 31, 2020 | December | 31, 2019 | |
|--|---|--|---------------------------------|----------------------------------|----------------------------------|----------------------------------|---|
| Description | Valuation technique(s) | Significant unobservable input | Range of inputs | Weighted average of inputs | Range of inputs | Weighted average of inputs | Other inputs and key information |
| Commingled funds | | | | | | | |
| Equity investments | Net asset value, determined using discounted cash flow | a) Discount rate ⁽¹⁾ b) Future cash flow ⁽²⁾ | 8.0% – 15.0% 1 – 7 years | 12.9% 3.3 years | 8.0% – 20.0% 1 – 9 years | 14.4% 2.3 years | Entitlement risk, sales risk and construction risk are taken into account in determining the discount rate. |
| Separate accounts/ side-cars/syndicated investments/ joint ventures | | | | | | | |
| Equity investments ^{(3),(4)} | Waterfall distribution model | a) Discount rate ⁽¹⁾ b) Future cash flow ⁽²⁾ c) Appraised value ⁽³⁾ | 12.5% – 20.0% 1 – 7 years | 17.1% 6.2 years | 12.5% – 24.0% 1 – 16 years | 17.2% 13.0 years | Entitlement risk, sales risk and construction risk are taken into account in determining the discount rate. Price per acre of land, timing of project funding requirements and distributions. |
| Debt investments ⁽³⁾ | Net asset value, determined using discounted cash flow | a) Discount rate ⁽¹⁾ b) Future cash flow ⁽²⁾ | 20.0% 6 years | 20.0% 6 years | 15.0% – 20.0% 3 – 9 years | 17.1% 7.2 years | Estimated probability of default. |

(1) Generally, an increase in future cash flow will result in an increase in the fair value of debt instruments and fund equity investments. An increase in the discount rate will result in a decrease in the fair value of debt instruments and fund equity investments. The same percentage change in the discount rate will result in a greater change in fair value than the same absolute percentage change in flow.

(2) Estimating future cash flows involves modelling developers' cash flows to determine the quantum and timing of project funding requirements and cash distributions to the Investment Vehicle. Estimates of developers' cash flows are based on detailed quarterly and annual budgets and include estimates of construction and development costs, anticipated selling prices and absorption rates for each project.

- (3) On an annual basis, the Company normally obtains external valuations for its separate account equity and side-car investments. As at December 31, 2020, the external valuations for Tricon's interest in four separate account equity and side-car investments totalled \$41,595. The Company's investment team and finance team verify all major inputs to the valuation and review the results with the independent appraiser prior to seeking Valuation Committee approval. The significant input within the appraised value is the value of land per acre. The separate account and side-car investments that were not appraised were valued utilizing an expected sales price calibration method. As at December 31, 2020, only one debt investment remained valued using the discounted cash flow methodology.
- (4) On January 22, 2020, the Company completed the syndication of 50% of its direct investment in Trinity Falls to THPAS JV-1, subsequent to which Tricon's investment in Trinity Falls was remeasured based on the transaction price. As a result, there was a significant change in the range of inputs and weighted average inputs disclosed compared to December 31, 2019 driven by the exclusion of Trinity Falls from the discounted cash flow model.

Sensitivity

For those investments valued using discounted cash flows, an increase of 2.5% in the discount rate results in a decrease in fair value of \$4,144 and a decrease of 2.5% in the discount rate results in an increase in fair value of \$4,568 (December 31, 2019 – (\$10,656) and \$11,541, respectively).

10. FAIR VALUE ESTIMATION

Fair value measurement

Fair value is the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date, regardless of whether that price is directly observable or estimated using another valuation technique. In estimating the fair value of an asset or a liability, the Company takes into account the characteristics of the asset or liability if market participants would take those characteristics into account when pricing the asset or liability at the measurement date. Fair value for measurement and/or disclosure purposes in these consolidated financial statements is determined on this basis, unless otherwise noted.

Inputs to fair value measurement techniques are disaggregated into three hierarchical levels, which are based on the degree to which inputs to fair value measurement techniques are observable by market participants:

Level 1 - Inputs are unadjusted, quoted prices in active markets for identical assets or liabilities at the measurement date.

Level 2 – Inputs (other than quoted prices included in Level 1) are either directly or indirectly observable for the asset or liability through correlation with market data at the measurement date and for the duration of the asset's or liability's anticipated life.

Level 3 – Inputs are unobservable and reflect management's best estimate of what market participants would use in pricing the asset or liability at the measurement date. Consideration is given to the risk inherent in the valuation technique and the risk inherent in the inputs in determining the estimate.

Fair value measurements are adopted by the Company to calculate the carrying amounts of various assets and liabilities.

Acquisition costs, other than those related to financial instruments classified as FVTPL which are expensed as incurred, are capitalized to the carrying amount of the instrument and amortized using the effective interest method.

The following table provides information about assets and liabilities measured at fair value on the balance sheet and categorized by level according to the significance of the inputs used in making the measurements:

| | December 31, 2020 | | | December 31, 2019 | | | |
|----------------------------------|-------------------|-----------|--------------|-------------------|---------|--------------|--|
| (in thousands of U.S. dollars) | Level 1 | Level 2 | Level 3 | Level 1 | Level 2 | Level 3 | |
| Assets | | | | | | | |
| Rental properties (Note 6) | \$ - | \$ – | \$ 6,321,918 | \$ - | \$ - | \$ – | |
| Canadian development properties | | | | | | | |
| (Note 8) | - | - | 110,018 | - | - | - | |
| Investments in for-sale housing | | | | | | | |
| (Note 9) | - | - | 164,842 | - | _ | 300,653 | |
| Investments – | | | | | | | |
| Tricon American Homes | - | - | - | - | - | 1,365,007 | |
| Investments – | | | | | | | |
| Tricon Lifestyle Rentals | - | - | - | - | _ | 525,932 | |
| Derivative financial instruments | | | | | | | |
| (Note 20) | - | 841 | - | - | - | - | |
| | \$ - | \$ 841 | \$ 6,596,778 | \$ - | \$ - | \$ 2,191,592 | |
| Liabilities | | | | | | | |
| Derivative financial instruments | | | | | | | |
| (Note 20) | \$ - | \$ 45,494 | \$ – | \$ - | \$ 657 | \$ – | |
| Limited partners' interests | | | | | | | |
| in rental business (Note 5) | - | - | 356,305 | - | _ | _ | |
| | \$ - | \$ 45,494 | \$ 356,305 | \$ - | \$ 657 | \$ - | |

There have been no transfers between levels for the year ended December 31, 2020.

Cash, restricted cash, amounts receivable, amounts payable and accrued liabilities, lease liabilities (included in other liabilities), resident security deposits and dividends payable are measured at amortized cost, which approximates fair value because they are short-term in nature.

11. AMOUNTS PAYABLE AND ACCRUED LIABILITIES

Amounts payable and accrued liabilities consist of the following:

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|--|-------------------|-------------------|
| Trade payables and accrued liabilities | \$ 31,182 | \$ 17,789 |
| Accrued property taxes | 37,987 | - |
| AIP liability (Note 30) | 7,120 | 2,742 |
| Income taxes payable | 337 | 1,947 |
| Interest payable | 18,566 | 3,577 |
| Deferred income | 1,294 | - |
| Current portion of lease obligations (Note 26) | 1,804 | 135 |
| Total amounts payable and accrued liabilities | \$ 98,290 | \$ 26,190 |

12. GOODWILL

In connection with the Company's deemed acquisitions (Note 5), the Company has allocated material amounts of the related deemed purchase prices to goodwill. Such goodwill is tested for impairment at least annually on December 31, using estimates and assumptions affected by factors such as economic and industry conditions and changes in operating performance.

The goodwill recorded in the consolidated financial statements relates to three groups of CGUs: the single-family rental group CGU, the multi-family rental group CGU and the Johnson CGU. The net carrying amount of goodwill is as follows.

| (in thousands of U.S. dollars) | December | 31, 2020 December 3 | | er 31, 2019 |
|--------------------------------|----------|---------------------|----|-------------|
| Johnson | \$ | 219 | \$ | 219 |
| Multi-Family Rental | | 79,112 | | _ |
| Single-Family Rental | | 29,507 | | _ |
| Total goodwill | \$ | 108,838 | \$ | 219 |

The Company's assumptions used in goodwill impairment testing are affected by current market conditions and the expected net operating income in each of the CGUs. The Company compared the aggregate recoverable amount of the group of assets included in the relevant CGUs to their respective carrying amounts. The recoverable amount was determined based on the fair value less costs of disposal of the CGUs. This fair value measurement is categorized as Level 3 in the fair value hierarchy and requires assumptions about revenue and operating expense growth rates as well as discount rates, which are discussed below.

| | December 3 | 31, 2020 |
|--|----------------------|---------------------|
| | Single-Family Rental | Multi-Family Rental |
| Weighted average growth rate – Years 1–5 | 3.3% | 3.5% |
| Long-term growth rate | 1.0% | 1.0% |
| Discount rate | 5.0% | 6.0% |

Growth rates

Growth rates over the five-year period are a combination of management's estimate of annual growth for the next fiscal year based on historical growth rates achieved for the two preceding years, where appropriate. Management also used available market forecasts and data for the growth rate for the next two to five years based on industry reports. The projections also take into account future expected capital expenditures to maintain the condition of the rental properties to drive future revenue growth.

Long-term growth rates

Cash flows beyond the five-year period are based largely on management's estimate of the ability of the CGU to grow in a mature and stable market.

Discount rates

Discount rates represent the current market assumption of the risks specific to each CGU regarding the time value of money and individual risks of the underlying assets, rather than the Company's specific discount rates.

Based on the assessment of current economic conditions and of the underlying cash flows at the CGU level, management concluded that there was no impairment of goodwill as at December 31, 2020, as the recoverable amounts of the individual CGUs exceeded their carrying values.

Sensitivity

The fair value less costs of disposal model utilized in calculating recoverable value is sensitive to changes in the discount rate and long-term growth rate, especially for the multi-family rental group of CGUs. If the discount rate increased by 0.1% or if the perpetual growth rate decreased by 0.1%, the carrying amount of the multi-family rental group of CGUs would exceed the recoverable amount by approximately \$20,000 to \$25,000; hence, it would trigger an impairment. For the single-family rental group of CGUs, no reasonable change in assumptions would cause the recoverable amounts to fall below the carrying values.

Management will continue to monitor the market and economic uncertainty related to the COVID-19 pandemic that could impact the significant estimates used in the discounted cash flow for annual impairment testing.

13. INCOME TAXES

| Income tax expense | \$ (36,375) | \$ (14,879) |
|---|----------------|----------------|
| Income tax expense – deferred | (40,425) | (9,469) |
| Income tax recovery (expense) – current | \$ 4,050 | \$ (5,410) |
| For the years ended December 31 | 2020 | 2019 |
| (in thousands of U.S. dollars) | | |

The tax on the Company's income differs from the theoretical amount that would arise using the weighted average tax rate applicable to income of the consolidated entities as follows:

| (in thousands of U.S. dollars) | | |
|--|------------|------------|
| For the years ended December 31 | 2020 | 2019 |
| Income before income taxes | \$ 152,788 | \$ 129,014 |
| Combined statutory federal and provincial income tax rate | 26.50% | 26.50% |
| Expected income tax expense | 40,489 | 34,189 |
| Non-taxable gains on investments | (1,460) | (24,684) |
| Non-taxable losses (gains) on derivative financial instruments | 1,912 | (785) |
| Foreign tax rate differential ⁽¹⁾ | (8,854) | 77 |
| Other, including permanent differences ⁽²⁾ | 4,288 | 6,082 |
| Income tax expense | \$ 36,375 | \$ 14,879 |

(1) Effective January 1, 2020, the Company's single-family rental and U.S. multi-family rental businesses are subject to the U.S. ordinary income tax rate of 21%, resulting in a reduction in Tricon's effective tax rate from the Canadian combined statutory income tax rate of 26.5%.

(2) Other permanent differences are comprised of non-deductible share compensation, non-deductible debentures discount amortization and non-deductible interest expense. The expected realization of deferred income tax assets and deferred income tax liabilities is as follows:

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|--|-------------------|-------------------|
| Deferred income tax assets | | |
| Deferred income tax assets to be recovered after more than 12 months | \$ 102,444 | \$ 41,049 |
| Deferred income tax assets to be recovered within 12 months | - | 3,700 |
| Total deferred income tax assets | \$ 102,444 | \$ 44,749 |
| Deferred income tax liabilities | | |
| Deferred income tax liabilities reversing after more than 12 months | \$ 298,071 | \$ 98,360 |
| Deferred income tax liabilities reversing within 12 months | - | 224 |
| Total deferred income tax liabilities | \$ 298,071 | \$ 98,584 |
| Net deferred income tax liabilities | \$ 195,627 | \$ 53,835 |

The movement of the deferred income tax accounts was as follows:

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|--|-------------------|-------------------|
| Change in net deferred income tax liabilities | | |
| Net deferred income tax liabilities, beginning of year | \$ 53,835 | \$ 45,091 |
| Initial recognition on Deemed Acquisition (Note 5) | 196,853 | - |
| Reversal of deferred income tax liabilities related to Deemed Acquisition (Note 5) | (94,714) | - |
| Charge to the statement of comprehensive income | 40,425 | 9,469 |
| Other | (772) | (725) |
| Net deferred income tax liabilities, end of year | \$ 195,627 | \$ 53,835 |

The tax effects of the significant components of temporary differences giving rise to the Company's deferred income tax assets and liabilities were as follows:

| At December 31, 2020 | \$ - | \$ 297,057 | \$ 175 | \$ 839 | \$ - | \$ 298,071 |
|------------------------------------|-------------|--|---------------------------|----------------------------|----------|------------|
| (Reversal)/addition ⁽¹⁾ | (97,338) | 297,057 | (12) | (220) | - | 199,487 |
| At December 31, 2019 | \$ 97,338 | \$ - | \$ 187 | \$ 1,059 | \$ - | \$ 98,584 |
| Deferred income tax liabilities | | | | | | |
| (in thousands of U.S. dollars) | Investments | Rental properties | Convertible debentures | Deferred placement fees | Other | Total |
| At December 31, 2020 | \$ 16,677 | \$ 6,211 | \$ 1,702 | \$ 72,292 | \$ 5,562 | \$ 102,444 |
| Addition/(reversal) ⁽¹⁾ | 16,677 | (245) | 634 | 40,492 | 137 | 57,695 |
| At December 31, 2019 | \$ - | \$ 6,456 | \$ 1,068 | \$ 31,800 | \$ 5,425 | \$ 44,749 |
| Deferred income tax assets | | | | | | |
| (in thousands of U.S. dollars) | Investments | Long-term incentive plan accrual | lssuance costs | Net operating losses | Other | Total |
| | | | | | | |

(1) Includes \$40,000 and \$142,139 of deferred income tax assets and deferred income tax liabilities, respectively, recognized as part of the business combinations (Note 5).

The Company believes it will have sufficient future income to realize the deferred income tax assets.

14. REVENUE FROM RENTAL PROPERTIES

The components of the Company's revenue from rental properties are described as follows:

| For the years ended December 31 | | 2019 | | |
|--------------------------------------|----------------------|--|------------|---------|
| (in thousands of U.S. dollars) | Single-Family Rental | Single-Family Rental Multi-Family Rental | | Total |
| Base rent | \$ 301,538 | \$ 90,290 | \$ 391,828 | \$ _ |
| Other revenue ⁽¹⁾ | 13,946 | 14,700 | 28,646 | - |
| Non-lease component | 51,498 | 6,215 | 57,713 | - |
| Total revenue from rental properties | \$ 366,982 | \$ 111,205 | \$ 478,187 | \$ - |

(1) Other revenue includes revenue earned on ancillary services and amenities as well as lease administrative fees.

15. REVENUE FROM PRIVATE FUNDS AND ADVISORY SERVICES

The components of the Company's revenue from private funds and advisory services are described in the tables below. Intercompany revenues and expenses between the Company and its subsidiaries, such as property management fees, are eliminated upon consolidation. Under certain arrangements, asset-based fees that are earned from third-party investors in Tricon's subsidiary entities are billed directly to those investors and are therefore not recognized in the accounts of the applicable subsidiary. These amounts are included in the asset management fees revenue recognized in the statements of comprehensive income.

| (in thousands of U.S. dollars) | Asset management fees | Performance fees | Development fees | Property management fees | Total |
|--|-----------------------------|---------------------|---------------------|--------------------------------|-----------|
| For the year ended December 31, 2020 | | | | | |
| Gross management fees | \$ 12,061 | \$ 2,836 | \$ 19,038 | \$ 45,464 | \$ 79,399 |
| Less fees eliminated upon consolidation: | | | | | |
| Development fees eliminated | - | - | (740) | - | (740) |
| Property management fees eliminated | - | - | - | (44,569) | (44,569) |
| Total revenue from private funds | | | | | |
| and advisory services | \$ 12,061 | \$ 2,836 | \$ 18,298 | \$ 895 | \$ 34,090 |
| For the year ended December 31, 2019 | | | | | |
| Total revenue from private funds | | | | | |
| and advisory services | \$ 15,099 | \$ 7,448 | \$ 17,348 | \$ – | \$ 39,895 |

16. AMOUNTS RECEIVABLE

Amounts receivable consist of rent receivables, trade receivables, income tax recoverable and other receivables.

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|----------------------------------|-------------------|-------------------|
| Rent receivables | \$ 4,274 | \$ - |
| Trade receivables | 5,263 | 3,057 |
| Income tax recoverable | 3,282 | 152 |
| Other receivables ⁽¹⁾ | 12,774 | 5,743 |
| Total amounts receivable | \$ 25,593 | \$ 8,952 |

(1) Other receivables are comprised of amounts due from affiliates and various amounts recoverable from third parties.

The Company has \$4,274 of rent receivables from residents as at December 31, 2020 under the relevant lease arrangements. As a result of the current COVID-19 pandemic and the resulting economic uncertainty, certain residents may experience financial difficulty which may impact their ability to continue to pay rent due and in the future.

17. DEBT

The following table presents a summary of the Company's outstanding debt as at December 31, 2020:

| | | | December | [.] 31, 2020 | | | |
|--|----------------|------------------------------|--------------------------------------|--------------------------------|-------------------------------------|----------------|---------------------|
| (in thousands of U.S. dollars) | Maturity dates | Coupon/stated interest rates | Interest rate cap or floor | Effective interest rates | Extension options ⁽¹⁾ | Total facility | Outstanding balance |
| SFR JV-1 subscription facility | August 2021 | LIBOR+1.75% | N/A | 2.31% | N/A | \$ 150,000 | \$ 116,000 |
| SFR JV-1 warehouse credit facility ⁽²⁾ | October 2021 | LIBOR+2.65% | 3.25% LIBOR cap 0.25% LIBOR floor | 3.21% | One-year | 300,000 | 96,610 |
| Term Ioan 2 ⁽³⁾ | October 2021 | LIBOR+1.95% | 2.50% LIBOR cap 0.50% LIBOR floor | 2.51% | One-year | 96,077 | 96,077 |
| Warehouse credit facility ⁽⁴⁾ | November 2021 | LIBOR+2.75% | 3.00% LIBOR cap 0.25% LIBOR floor | 3.31% | One-year | 50,000 | 10,209 |
| Securitization debt 2017-1 ⁽³⁾ | September 2022 | 3.59% | N/A | 3.59% | N/A | 459,530 | 459,530 |
| Term Ioan ⁽³⁾ | October 2022 | LIBOR+2.00% | 2.50% LIBOR cap 0.50% LIBOR floor | 2.56% | N/A | 375,000 | 374,745 |
| Securitization debt 2017-2 ⁽³⁾ | January 2024 | 3.66% | N/A | 3.66% | N/A | 363,598 | 363,598 |
| Securitization debt 2018-1 ⁽³⁾ | May 2025 | 3.96% | N/A | 3.96% | N/A | 312,540 | 312,540 |
| SFR JV-1 securitization debt 2019-1 ⁽³⁾ | March 2026 | 3.12% | N/A | 3.12% | N/A | 333,358 | 333,358 |
| SFR JV-1 securitization debt 2020-1 ^{(3),(5)} | July 2026 | 2.43% | N/A | 2.43% | N/A | 553,428 | 553,428 |
| Securitization debt 2020-2 ^{(3),(6)} | November 2027 | 1.94% | N/A | 1.94% | N/A | 440,506 | 440,506 |
| Single-family rental | | | | | | | |
| properties borrowings | | | | 2.94% | | 3,434,037 | 3,156,601 |
| U.S. multi-family credit facility | December 2021 | LIBOR+3.75% | N/A | 4.39% | N/A | 109,890 | 109,890 |
| Mortgage tranche A ⁽⁷⁾ | November 2023 | LIBOR+1.15% | 5.35% cap | 1.77% | N/A | 160,090 | 160,090 |
| Mortgage tranche B ⁽⁷⁾ | November 2024 | 3.92% | N/A | 3.92% | N/A | 400,225 | 400,225 |
| Mortgage tranche C ⁽⁷⁾ | November 2025 | 3.95% | N/A | 3.95% | N/A | 240,135 | 240,135 |
| Multi-family rental | | | | | | | |
| properties borrowings | | | | 3.61% | | 910,340 | 910,340 |
| Land Ioan ⁽⁸⁾ | July 2021 | Prime+1.50% | 3.95% floor | 4.17% | N/A | 21,991 | 21,991 |
| Vendor take-back (VTB) loan 2021 | August 2021 | - | N/A | 6.00% | N/A | 25,564 | 25,564 |
| Mortgage ⁽⁸⁾ | September 2022 | 3.67% | N/A | 3.67% | N/A | 12,482 | 12,482 |
| Canadian development | | | | | | | |
| properties borrowings | | | | 4.85% | | 60,037 | 60,037 |
| Corporate credit facility ⁽⁹⁾ | July 2022 | LIBOR+2.75% | N/A | 4.48% | N/A | 500,000 | 26,000 |
| Corporate office mortgages | November 2024 | 4.25% | N/A | 4.30% | N/A | 11,089 | 11,089 |
| Corporate borrowings | | | | 4.42% | | 511,089 | 37,089 |
| | | | | | | | \$ 4,164,067 |
| Transaction costs (net of amortization) | | | | | | | (25,019) |
| Debt discount (net of amortization) | | | | | | | (1,542) |
| Total debt | | | | 3.12% | | \$ 4,915,503 | \$ 4,137,506 |
| Current portion of long-term debt ⁽¹⁾ | | | | | | | \$ 274,190 |
| Long-term debt | | | | | | | \$ 3,863,316 |
| Fixed-rate debt – principal value | | | | 3.24% | | | \$ 3,152,455 |
| Floating-rate debt – principal value | | | | 2.76% | | | \$ 1,011,612 |

(1) The Company has the ability to extend the maturity of the loans where an extension option exists and intends to exercise such options wherever available.

The current portion of long-term debt reflects the balance after all extension options have been exercised.

(2) On October 23, 2020, SFR JV-1 amended its warehouse credit facility and extended its maturity date to October 25, 2021, with a one-year extension option available. (3) The term loans and securitization debt are secured, directly and indirectly, by approximately 21,200 single-family rental homes.

(4) On November 20, 2020, Tricon amended its warehouse credit facility and extended its maturity date to November 22, 2021, with a one-year extension option available.

(5) On July 21, 2020, SFR JV-1 closed a new securitization transaction involving the issuance and sale of six classes of fixed-rate pass-through certificates with a face amount of \$553,428, a weighted average coupon of 2.43% and a term to maturity of six years. The transaction proceeds were used to repay existing short-term SFR JV-1 debt.

(6) On November 10, 2020, Tricon closed a new securitization transaction involving the issuance and sale of six classes of fixed-rate pass-through certificates with a face amount of \$440,506, a weighted average coupon of 1.94% and a term to maturity of seven years. The transaction proceeds were used to repay existing debt and to acquire single-family rental homes.

(7) The mortgages are secured by 23 multi-family properties owned by the Company.

The land loan and mortgage are secured by the land under development at The James (Scrivener Square) and The Shops of Summerhill. (8)

(9) The Company has provided a general security agreement creating a first priority security interest on the assets of the Company, excluding, among other things, single-family rental homes, multi-family rental properties and interests in for-sale housing. As part of the corporate credit facility, the Company has designated \$15,000 to issue letters of credit as security against contingent obligations related to its Canadian multi-family developments. As at December 31, 2020, the letters of credit outstanding totalled \$10,707 (C\$13,632). During the year, the Company used proceeds from the Due to Affiliate (see Note 19) to pay down the corporate credit facility, resulting in an ending balance of \$26,000 (2019 - \$297,000).

NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

for the year ended December 31, 2020

(in thousands of U.S. dollars, except per share amounts and percentage amounts)

| | | | Decemb | er 31, 2019 | | | | | |
|--------------------------------|----------------|------------------------------|-------------------------------|--------------------------------|-------------------|----|----------------|----|------------------------|
| (in thousands of U.S. dollars) | Maturity dates | Coupon/stated interest rates | Interest rate cap or floor | Effective interest rates | Extension options | - | Total facility | C | Dutstanding balance |
| Corporate credit facility | July 2022 | LIBOR+3.75% | N/A | 5.91% | N/A | \$ | 500,000 | \$ | 297,000 |
| Corporate office mortgages | November 2024 | 4.25% | N/A | 4.30% | N/A | | 11,153 | | 11,153 |
| Total debt | | | | 5.85% | | \$ | 511,153 | \$ | 308,153 |
| Current portion of debt | | | | | | | | \$ | 284 |
| Long-term debt | | | | | | | | \$ | 307,869 |

The Company was in compliance with the covenants and other undertakings outlined in all loan agreements.

The scheduled principal repayments and debt maturities are as follows, reflecting the maturity dates after all extensions have been exercised:

| (in thousands of U.S. dollars) | Single-family rental borrowings | Multi-family rental borrowings | Canadian development properties borrowings | Corporate borrowings | Total |
|--------------------------------|------------------------------------|-----------------------------------|---|-------------------------|--------------|
| 2021 | \$ 116,000 | \$ 110,255 | \$ 47,969 | \$ 302 | \$ 274,526 |
| 2022 | 1,037,171 | 4,366 | 12,068 | 26,313 | 1,079,918 |
| 2023 | - | 156,293 | - | 329 | 156,622 |
| 2024 | 363,598 | 403,760 | _ | 10,145 | 777,503 |
| 2025 | 312,540 | 235,666 | - | _ | 548,206 |
| 2026 and thereafter | 1,327,292 | - | - | _ | 1,327,292 |
| | 3,156,601 | 910,340 | 60,037 | 37,089 | 4,164,067 |
| Transaction costs | | | | | |
| (net of amortization) | | | | | (25,019) |
| Debt discount | | | | | |
| (net of amortization) | | | | | (1,542) |
| Total debt | | | | | \$ 4,137,506 |

Fair value of debt

The table below presents the fair value and the carrying value (net of unamortized deferred financing fees and debt discount) of the fixed-rate loans as at December 31, 2020.

| | December | 31, 2020 |
|-------------------------------------|--------------|----------------|
| (in thousands of U.S. dollars) | | Carrying value |
| Securitization debt 2017-1 | \$ 466,210 | \$ 459,530 |
| Securitization debt 2017-2 | 373,583 | 362,683 |
| Securitization debt 2018-1 | 329,876 | 311,913 |
| SFR JV-1 securitization debt 2019-1 | 347,177 | 326,767 |
| SFR JV-1 securitization debt 2020-1 | 567,635 | 543,803 |
| Securitization debt 2020-2 | 440,506 | 432,817 |
| Mortgage tranche B | 413,778 | 400,225 |
| Mortgage tranche C | 248,936 | 240,135 |
| Vendor take-back (VTB) Ioan 2021 | 26,356 | 25,564 |
| Mortgage | 12,641 | 12,463 |
| Corporate office mortgages | 11,728 | 11,089 |
| Total | \$ 3,238,426 | \$ 3,126,989 |

The carrying value of variable term loans approximates their fair value, since their variable interest terms are indicative of prevailing market prices.

18. CONVERTIBLE DEBENTURES

On March 17, 2017, the Company completed the offering, on a bought deal basis, of \$172,500 aggregate principal amount of 5.75% extendible convertible unsecured debentures (the "2022 convertible debentures"), including \$22,500 aggregate principal amount of 2022 convertible debentures issued pursuant to the exercise of underwriters' over-allotment options. The net offering proceeds to the Company were \$164,554 after transaction costs of \$7,946.

Upon the closing of the acquisition of Silver Bay on May 9, 2017, the 2022 convertible debentures became convertible to common shares of the Company in accordance with their terms, and their maturity date was extended to March 31, 2022.

The 2022 convertible debentures bear interest at 5.75% per annum, which is payable semi-annually in arrears in March and September, and are convertible into common shares of the Company at a conversion rate of 95.6023 common shares per \$1,000 principal amount of 2022 convertible debentures (equivalent to a conversion price of approximately \$10.46 per common share (equivalent to C\$13.32 as of December 31, 2020)).

The Company may settle the conversion right in cash in lieu of common shares unless the holder has explicitly indicated that it does not wish to receive cash.

On or after March 31, 2020 and prior to March 31, 2021, the 2022 convertible debentures may be redeemed by the Company at a price equal to the principal amount thereof plus accrued and unpaid interest, provided that the Current Market Price (as defined in the trust indenture governing the 2022 convertible debentures) of the Company's common shares on the fifth trading day immediately preceding the date on which notice of redemption is given is not less than 125% of the conversion price. On or after March 31, 2021 and prior to their final maturity date, the 2022 convertible debentures may be redeemed by the Company at a price equal to the principal amount thereof plus accrued and unpaid interest. The Company has an option to settle the redemption right, where applicable, by delivering the number of common shares determined by dividing the principal amount of the convertible debentures by 95% of the Current Market Price of the Company's common shares on the fifth trading day immediately preceding the date fixed for redemption or the maturity date. For the year ended December 31, 2020, there were no conversions of the 2022 convertible debentures (\$100 principal amount was converted into 9,560 common shares during the year ended December 31, 2019).

The host liability component of the outstanding 2022 convertible debentures recognized on the consolidated balance sheets was calculated as follows:

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|---|-------------------|-------------------|
| Principal amount outstanding | \$ 172,400 | \$ 172,400 |
| Less: Transaction costs (net of amortization) | (2,249) | (3,884) |
| Liability component on initial recognition | 170,151 | 168,516 |
| Debentures discount (net of amortization) | (4,195) | (7,205) |
| 2022 convertible debentures | \$ 165,956 | \$ 161,311 |

The above carrying values were recognized at amortized cost after discounting the future interest and principal payments using the effective interest rates. The fair value of the host liability component of the 2022 convertible debentures was \$178,412 as of December 31, 2020 and \$177,777 as of December 31, 2019. The difference between the amortized cost and implied fair value is a result of the difference between the effective interest rate and the market interest rate for debt with similar terms.

19. DUE TO AFFILIATE

On August 26, 2020, Tricon and its affiliate, Tricon PIPE LLC (the "Affiliate" or "LLC") entered into subscription agreements with each investor in a syndicate of investors (the "Investors"), pursuant to which the Investors subscribed for Preferred Units of the Affiliate (the "Preferred Units") for an aggregate subscription price of \$300,000 (the "Transaction"). The Transaction was completed on September 3, 2020, on which date the Company and the Affiliate entered into various agreements with the Investors in connection with the Transaction (together with the subscription agreements, the "Transaction Documents").

Transaction – between Tricon and Investors

Pursuant to the Transaction Documents, holders of Preferred Units have the right to exchange the Preferred Units into common shares of the Company at any time at the option of the holder (the "Exchange Right") at an initial exchange price of \$8.50 (C\$11.18 as of August 26, 2020) per common share, as may be adjusted from time to time in accordance with the terms of the Transaction Documents (the "Exchange Price"), subject to shareholder approval, where applicable. Holders of Preferred Units are also entitled to receive a cash dividend equal to 5.75% of the Liquidation Preference of the Preferred Units (as defined in the Transaction Documents), per annum, calculated and payable quarterly for the first seven years following closing of the Transaction ("Closing"), with a prescribed annual increase to the dividend rate of 1% per year thereafter, up to a maximum rate of 9.75% per year.

The Affiliate has the right to force the exchange (the "Forced Exchange Right") of the outstanding Preferred Units beginning after the fourth anniversary of Closing, provided the 20-day volume-weighted average price of Tricon's shares exceeds 135% of the Exchange Price (reducing to 115% following the fifth anniversary of Closing). These exchange rights are classified as a derivative financial instrument (Note 20). The Affiliate also has the right to redeem the Preferred Units ("Redemption Right") at any time following the fifth anniversary of Closing for cash equal to 105% of the Liquidation Preference of the Preferred Units (as defined in the Transaction Documents).

Promissory note – between Tricon entities

In connection with the Transaction, the Company borrowed the subscription proceeds of \$300,000 from the Affiliate. This indebtedness, which is evidenced by a promissory note (the "Promissory Note" or "Due to Affiliate"), has a maturity of September 3, 2032 (permitting prepayment at any time pursuant to its terms) and bears interest at a rate of 5.75% per annum, calculated and payable quarterly for the first seven years following Closing with increases thereafter matching the applicable increases of the dividend rate applicable to the Preferred Units, described above. The Company incurred \$15,192 of transaction costs in connection with the Transaction, of which \$12,202 was capitalized, which reduced the initial fair value of the Promissory Note, and the remaining portion was expensed as it was attributed to the derivative component of the Promissory Note. In determining the initial fair value of the Promissory Note, the Company used a discounted cash flow model and an amortization period of nine years and nine months determined by a probability weighted methodology under IFRS 9.

The Promissory Note payable to Tricon PIPE LLC is subsequently measured at amortized cost using the effective interest rate method. Under the effective interest rate method, the transaction costs along with the discount of the Promissory Note are amortized over the expected life and recorded as net interest expense in the consolidated statements of comprehensive income. This amortization period is subject to re-estimation on a regular basis as adjustments to the carrying value are made under the effective interest rate method. During the period from September 3, 2020 to December 31, 2020, the Company recorded interest expense of \$7,116, including accretion expense of \$1,462 with respect to the amortization of transaction costs and the discount.

The Promissory Note contains mandatory prepayment provisions ("Mandatory Prepayment") applicable in connection with certain provisions of the Transaction Documents requiring the redemption of all or a portion of the outstanding Preferred Units. This Mandatory Prepayment is a derivative, which incorporates assumptions in respect of the Exchange Right, Forced Exchange Right and Redemption Right, and is measured separately from the Promissory Note and classified as a derivative financial instrument (Note 20).

NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

for the year ended December 31, 2020 (in thousands of U.S. dollars, except per share amounts and percentage amounts)

The fair values of the Promissory Note and the derivative on September 3, 2020 are as follows:

| (in thousands of U.S. dollars) | Principal value | Fair value of derivative | Fair value of Promissory Note |
|---|-----------------|--------------------------|----------------------------------|
| Promissory Note principal value as at September 3, 2020 | \$ 300,000 | \$ 37,613 | \$ 262,387 |
| Transaction costs ⁽¹⁾ | 15,192 | (2,990) | (12,202) |
| Liability component on initial recognition | | \$ 37,613 | \$ 250,185 |

(1) Transaction costs of \$2,990 that were attributed to the derivative component were expensed in the consolidated statements of comprehensive income during the period while transaction costs of \$12,202 were capitalized into the underlying promissory note and reduced the fair value at inception.

The movement of the derivative financial liability in connection with the Promissory Note (or Due to Affiliate) from September 3, 2020 to December 31, 2020 is shown below:

| (in thousands of U.S. dollars) | |
|---|-----------|
| Fair value of derivative on initial recognition as at September 3, 2020 | \$ 37,613 |
| Fair value loss on derivative during the period | 7,881 |
| Fair value of derivative as at December 31, 2020 (Note 20) | \$ 45,494 |

The fair value loss on the derivative was primarily driven by an increase in Tricon's share price, on a USD converted basis, which served to increase the probability of exchange of the preferred units into Tricon's common shares (Note 20).

The Promissory Note payable to Tricon PIPE LLC is subsequently measured at amortized cost using the effective interest rate method and the carrying value of Due to Affiliate is shown below. The fair value of the Promissory Note was \$293,465 as of December 31, 2020. The difference between the amortized cost and the implied fair value is a result of the difference between the effective interest rate and the market interest rate for debt with similar terms.

| (in thousands of U.S. dollars) | December 31, 2020 | September 3, 2020 |
|--|-------------------|-------------------|
| Principal amount outstanding | \$ 300,000 | \$ 300,000 |
| Less: Discount and transaction costs (net of amortization) | (48,353) | (49,815) |
| Due to Affiliate | \$ 251,647 | \$ 250,185 |

Structured entity – Tricon PIPE LLC ("Affiliate" or "LLC")

Tricon PIPE LLC was incorporated on August 7, 2020 for the purpose of raising third-party capital through the issuance of preferred units for an aggregate amount of \$300,000. The Company has a 100% voting interest in this LLC; however, the Company is not required to consolidate this structured entity, as discussed in Note 3.

As of December 31, 2020, the LLC has a preferred unit liability of \$300,000 and a Promissory Note receivable of \$300,000. During the year ended December 31, 2020, the LLC earned interest income of \$5,654 from the Company and recognized dividends declared of \$5,654.

The Company's obligation with respect to its involvement with the structured entity is equal to the cash flows under the Promissory Note payable (or Due to Affiliate). The Company has not recognized any income or losses in connection with its interest in this unconsolidated structured entity from September 3, 2020 to December 31, 2020.

20. DERIVATIVE FINANCIAL INSTRUMENTS

The conversion and redemption features of the convertible debentures are combined pursuant to IFRS 9, *Financial Instruments: Recognition and Measurement*, and are measured at fair value at each reporting period using model calibration. The conversion and redemption components were valued using a binomial pricing model and then the valued amount was calibrated to the traded price of the underlying debentures. The valuation model uses market-based inputs, including the spot price of the underlying equity, implied volatility of the equity and USD/CAD foreign exchange rates, risk-free rates from the U.S. dollar swap curves and dividend yields related to the equity. The valuation of the conversion and redemption components assumes that the debentures are held to maturity.

Quantitative information about fair value measurements (Level 2) using significant observable inputs other than quoted prices included in Level 1 is as follows:

| 2022 convertible debentures | December 31, 2020 | December 31, 2019 |
|-----------------------------------|-------------------|-------------------|
| Risk-free rate ⁽¹⁾ | 0.21% | 1.70% |
| Implied volatility ⁽²⁾ | 30.69% | 20.78% |
| Dividend yield ⁽³⁾ | 2.45% | 2.63% |

(1) Risk-free rates were from the U.S. dollar swap curves matching the terms to maturity of the debentures.

(2) Implied volatility was computed from the trading volatility of the Company's stock over a comparable term to maturity and the volatility of USD/CAD exchange rates.(3) Dividend yields were from the forecast dividend yields matching the terms to maturity of the debentures.

The Company recognized a new derivative in connection with its Transaction completed during the year (Note 19). The Promissory Note contains the Mandatory Prepayment that is intermingled with other options pursuant to the Transaction, as exercising the Mandatory Prepayment effectively terminates the other options. Although the Exchange Right and Redemption Right exist at the Affiliate level, the Affiliate is unable to issue the common shares of the Company upon exercise of one or all of the rights by either party. As a result, such options, in essence, were deemed to be written by the Company and are treated as a single combined financial derivative instrument for valuation purposes in accordance with IFRS 9. The option pricing model for the derivative uses market-based inputs, including the spot price of the underlying equity, implied volatility of the equity and USD/CAD foreign exchange rates, risk-free rates from the U.S. dollar swap curves and dividend yields related to the underlying equity. The valuation of the derivative assumes a 9.75-year expected life of the investment horizon of the unitholders.

Quantitative information about fair value measurements (Level 2) using significant observable inputs other than quoted prices included in Level 1 is as follows:

| Due to Affiliate | December 31, 2020 | December 31, 2019 |
|-----------------------------------|-------------------|-------------------|
| Risk-free rate ⁽¹⁾ | 0.40% | N/A |
| Implied volatility ⁽²⁾ | 31.78% | N/A |
| Dividend yield ⁽³⁾ | 2.45% | N/A |

(1) Risk-free rates were from the U.S. dollar swap curves matching the expected maturity of the Due to Affiliate.

(2) Implied volatility was computed from the trading volatility of the Company's stock over a comparable term to maturity and the volatility of USD/CAD exchange rates.(3) Dividend yields were from the forecast dividend yields matching the expected maturity of the Due to Affiliate.

The Company also has other types of derivative financial instruments that consist of interest rate caps on the Company's floatingrate debt and are classified and measured at FVTPL. Interest rate caps are valued using model calibration. Inputs to the valuation model are determined from observable market data wherever possible, including market volatility and interest rates.

NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

for the year ended December 31, 2020

(in thousands of U.S. dollars, except per share amounts and percentage amounts)

Conversion/ Exchange/ (in thousands of U.S. dollars) Interest rate caps⁽¹⁾ redemption options prepayment options Total For the year ended December 31, 2020 (629) Derivative financial assets (liabilities), beginning of year \$ (657) \$ \$ 28 \$ Addition of derivative financial liability in connection with Due to Affiliate (37, 613)(37, 613)Addition of interest rate caps 11 11 (7,881) Fair value gain (loss) 1.498 (39) (6,422) Derivative financial assets (liabilities), end of year \$ 841 \$ (45,494) \$ _ \$ (44,653) For the year ended December 31, 2019 Derivative financial assets (liabilities), beginning of year \$ (3,936) \$ \$ \$ (3,936) 3,279 Fair value gain (loss) 3.279 Derivative financial assets (liabilities), end of year \$ (657) \$ _ \$ \$ (657) _

The values attributed to the derivative financial instruments are shown below:

(1) Initial balance was recognized as part of the Deemed Acquisition of the single-family rental business (Note 5).

As at December 31, 2020, the conversion and redemption features of the 2022 convertible debentures are presented as an asset of \$841, and the exchange and prepayment features related to the Due to Affiliate are presented as a liability of \$45,494.

For the year ended December 31, 2020, there was a fair value gain on the embedded derivative on the 2022 convertible debentures of \$1,498. For the period from September 3 to December 31, 2020, there was a fair value loss on the derivative recognized on the Due to Affiliate of \$7,881. The fair value gain on the embedded derivative on the 2022 convertible debentures was driven by an increase in the value of Tricon's redemption option relative to the holders' conversion option, resulting in the embedded derivative being an asset as at December 31, 2020 (December 31, 2019 – liability). The fair value loss on the derivative related to the Due to Affiliate was primarily driven by an increase in Tricon's share price, on a USD converted basis, which served to increase the probability of exchange of the preferred units into Tricon's common shares.

For the year ended December 31, 2020, the Company recognized a \$6,422 fair value loss on derivative financial instruments (2019 – fair value gain of \$3,279) and a \$1,039 fair value loss on the put liability (2019 – fair value loss of \$318), for a total loss of \$7,461 (2019 – fair value gain of \$2,961). The put liability was redeemed on March 4, 2020 in connection with the Company's acquisition and cancellation of 1,867,675 outstanding common shares (Note 28).

NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS for the year ended December 31, 2020 (in thousands of U.S. dollars, except per share amounts and percentage amounts)

21. INTEREST EXPENSE

Interest expense is comprised of the following:

| (in thousands of U.S. dollars) | | |
|---|------------|-----------|
| For the years ended December 31 | 2020 | 2019 |
| SFR JV-1 subscription facility | \$ 3,763 | \$ - |
| SFR JV-1 warehouse credit facility | 6,091 | - |
| Term Ioan 2 | 2,796 | - |
| Warehouse credit facility | 1,246 | - |
| Securitization debt 2017-1 | 16,612 | - |
| Term loan | 10,891 | - |
| Securitization debt 2017-2 | 13,408 | - |
| Securitization debt 2018-1 | 12,473 | - |
| SFR JV-1 securitization debt 2019-1 | 10,385 | - |
| SFR JV-1 securitization debt 2020-1 | 6,030 | - |
| Securitization debt 2020-2 | 1,221 | - |
| Securitization debt 2016-1 ⁽¹⁾ | 12,177 | - |
| Single-family rental interest expense | 97,093 | - |
| U.S. multi-family credit facility | 5,006 | - |
| Mortgage tranche A | 2,865 | - |
| Mortgage tranche B | 15,950 | - |
| Mortgage tranche C | 9,643 | - |
| Multi-family rental interest expense | 33,464 | - |
| Mortgage | 285 | _ |
| Vendor take-back (VTB) Ioan 2020 ⁽¹⁾ | 233 | - |
| Canadian development properties interest expense ⁽²⁾ | 518 | _ |
| Corporate credit facility | 12,582 | 17,819 |
| Corporate office mortgages | 450 | 354 |
| Corporate interest expense | 13,032 | 18,173 |
| Amortization of financing costs | 5,900 | 1,570 |
| Amortization of debt discounts | 4,694 | 2,729 |
| Debentures interest | 9,927 | 9,902 |
| Interest on Due to Affiliate | 5,654 | - |
| Interest on lease obligation | 328 | 65 |
| Total interest expense | \$ 170,610 | \$ 32,439 |

(1) The securitization debt 2016-1 and vendor take-back (VTB) loan 2020 were fully repaid during the year.

(2) Canadian development properties capitalized \$1,708 of interest for the year ended December 31, 2020.

for the year ended December 31, 2020 (in thousands of U.S. dollars, except per share amounts and percentage amounts)

22. EXPENSES

The Company's expenses are comprised of direct operating expense for rental properties, property management overhead, compensation, general and administration, interest and depreciation and amortization. Direct operating expense for rental properties includes all attributable expenses incurred at the property level. Property management overhead expenses are incurred in running the Company's property management platform headquartered in Orange County, California, and include all direct expenses associated with managing rental properties, acquisitions and dispositions activities and other service activities.

The following table lists details of the direct operating expenses for rental properties by type.

| (in thousands of U.S. dollars) | Single-Far | milv Multi-Family | / |
|--------------------------------------|------------|-----------------------|--------------|
| For the year ended December 31, 2020 | 5 | ntal Renta | l Total |
| Property taxes | \$ 55,6 | 515 \$ 18,623 | 3 \$ 74,238 |
| Repairs, maintenance and turnover | 24,5 | 575 4,41 ⁻ | 1 28,986 |
| Property management expenses | 25,6 | 686 12,097 | 7 37,783 |
| Property insurance | 5,3 | 306 2,472 | 2 7,778 |
| Homeowners' association (HOA) costs | 4,9 | 006 52 | 2 4,958 |
| Other direct expense ⁽¹⁾ | 5,1 | 54 10,64 | 1 15,795 |
| Direct operating expenses | \$ 121,2 | 42 \$ 48,296 | 5 \$ 169,538 |

(1) Other direct expense includes property marketing, utilities and other property operating costs.

The following table provides details of direct expenses incurred at the property management platform by nature.

| (in thousands of U.S. dollars) | |
|---|-----------|
| For the year ended December 31, 2020 | |
| Salaries and benefits ⁽¹⁾ | \$ 12,903 |
| General and administration expense ⁽²⁾ | 7,926 |
| Travel and entertainment | 373 |
| Marketing | 1,173 |
| Other expense | 279 |
| Property management overhead | \$ 22,654 |

(1) Salaries and benefits incurred at the property management platform exclude property management salaries and benefits allocated to direct operating expenses of \$25,686 for the year ended December 31, 2020.

(2) General and administration expense incurred at the property management platform includes professional fees, insurance and other miscellaneous office expenses.

23. OTHER INCOME (EXPENSE)

Other income (expense) includes government assistance received by Johnson in the amount of \$1,774, net of other expense.

24. INTANGIBLE ASSETS

The intangible assets are as follows:

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|----------------------------------|-------------------|-------------------|
| Placement fees | \$ 3,764 | \$ 4,747 |
| Customer relationship intangible | 3,215 | 3,731 |
| Contractual development fees | 5,384 | 7,918 |
| Total intangible assets | \$ 12,363 | \$ 16,396 |

Intangible assets represent future management fees, development fees and commissions that Tricon expects to receive over the life of the assets and Investment Vehicles that the Company manages. They are amortized over the estimated periods that the Company expects to collect these fees, which range from 2 to 13 years. Amortization expense for the year ended December 31, 2020 was \$4,034 (2019 – \$4,338).

| (in thousands of U.S. dollars) | | | Amortization | Translation | |
|--------------------------------------|-----------|-----------|--------------------|-------------|-----------|
| For the year ended December 31, 2020 | Opening | Additions | expense adjustment | | Ending |
| Placement fees | \$ 4,747 | \$ - | \$ (984) | \$ 1 | \$ 3,764 |
| Customer relationship intangible | 3,731 | - | (516) | - | 3,215 |
| Contractual development fees | 7,918 | - | (2,534) | - | 5,384 |
| Intangible assets | \$ 16,396 | \$ - | \$ (4,034) | \$ 1 | \$ 12,363 |

| (in thousands of U.S. dollars) | | | Amortization | Translation | |
|--------------------------------------|-----------|-----------|--------------|-------------|-----------|
| For the year ended December 31, 2019 | Opening | Additions | expense | adjustment | Ending |
| Placement fees | \$ 5,735 | \$ - | \$ (989) | \$ 1 | \$ 4,747 |
| Rights to performance fees | 65 | _ | (65) | - | _ |
| Customer relationship intangible | 4,245 | - | (514) | - | 3,731 |
| Contractual development fees | 10,688 | - | (2,770) | _ | 7,918 |
| Intangible assets | \$ 20,733 | \$ - | \$ (4,338) | \$ 1 | \$ 16,396 |

In light of the COVID-19 pandemic and the related market and economic uncertainty, the Company recognized a fair value write-down with respect to its for-sale housing investments in the first quarter of 2020 (Note 9). As a result, management has also assessed whether the write-down impacted the carrying value of the intangible assets recognized as part of the acquisition of Johnson in 2014. Specifically, contractual development fees and customer relationship intangibles were initially recognized through the purchase price allocation performed in 2014. Management has assessed the potential impact on the underlying business at Johnson and its existing contracts with developers in determining if an impairment exists on intangibles as at December 31, 2020. Management has concluded there was no impairment of the intangibles but will continue to monitor the situation closely.

NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

for the year ended December 31, 2020

(in thousands of U.S. dollars, except per share amounts and percentage amounts)

25. OTHER ASSETS

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|--|-------------------|-------------------|
| Building | \$ 30,602 | \$ 24,987 |
| Furniture, computer and office equipment | 8,015 | 4,272 |
| Right-of-use assets (Note 26) | 6,018 | 987 |
| Leasehold improvements | 1,251 | 431 |
| Property-related systems software | 1,478 | - |
| Vehicles | 626 | - |
| Total other assets | \$ 47,990 | \$ 30,677 |

| (in thousands of U.S. dollars) | Ir | itial recognition for business | | | | |
|---|-----------|-----------------------------------|-----------|-------------------------|---------------------------|-----------|
| For the year ended December 31, 2020 | Opening | combinations (Note 5) | Additions | Depreciation expense | Translation adjustment | Ending |
| Building | \$ 24,987 | \$ - | \$ 5,462 | \$ (539) | \$ 692 | \$ 30,602 |
| Furniture, computer | | | | | | |
| and office equipment | 4,272 | 2,795 | 4,072 | (3,172) | 48 | 8,015 |
| Right-of-use assets ⁽¹⁾ | 987 | 5,379 | 1,966 | (2,314) | | 6,018 |
| Leasehold improvements | 431 | 1,141 | 178 | (499) | - | 1,251 |
| Property-related | | | | | | |
| systems software | - | 1,604 | 37 | (163) | - | 1,478 |
| Vehicles | - | 475 | 278 | (127) | - | 626 |
| Other assets | \$ 30,677 | \$ 11,394 | \$ 11,993 | \$ (6,814) | \$ 740 | \$ 47,990 |

(in thousands of U.S. dollars)

| For the year ended December 31, 2019 | Opening | Additions | Depreciation expense | Translation adjustment | Ending |
|---|-----------|-----------|-------------------------|---------------------------|-----------|
| Building | \$ 15,540 | \$ 9,002 | \$ (527) | \$ 972 | \$ 24,987 |
| Furniture, computer | | | | | |
| and office equipment | 4,247 | 1,010 | (1,183) | 198 | 4,272 |
| Right-of-use assets ⁽¹⁾ | 1,140 | 5 | (158) | _ | 987 |
| Leasehold improvements | 499 | - | (68) | - | 431 |
| Other assets | \$ 21,426 | \$ 10,017 | \$ (1,936) | \$ 1,170 | \$ 30,677 |

(1) Right-of-use assets include leased space in office buildings with a carrying value of \$3,862 (December 31, 2019 – \$987) and maintenance vehicles with a carrying value of \$1,965 (December 31, 2019 – nil). The remaining balance of right-of-use assets relates to office equipment.

For the year ended December 31, 2020, the Company incurred \$5,462 related to the expansion of its head office in Toronto, comprised of \$4,774 for the acquisition of commercial condominium units and \$688 for the development and construction of the new office space. The Company also incurred \$4,072 of costs for technology hardware and software and office furniture for the new office space.

Depreciation expense for the year ended December 31, 2020 was \$6,814 (2019 - \$1,936).

26. OTHER LIABILITIES

Other liabilities consist of the non-current portion of lease obligations and a put liability, as follows:

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|---|-------------------|-------------------|
| Put liability ⁽¹⁾ | \$ - | \$ 13,375 |
| Non-current portion of lease obligations ⁽²⁾ | 4,599 | 954 |
| Total other liabilities | \$ 4,599 | \$ 14,329 |

The put liability was redeemed in full on March 4, 2020 in connection with the Company's acquisition and cancellation of 1,867,675 common shares (Note 28).
 The current portion of lease obligations is presented in amounts payable and accrued liabilities (Note 11).

The Company has multiple office leases, maintenance vehicle leases and office equipment leases. Tricon has 15 leases for office space with fixed lease terms ranging from one to eight years remaining. The Company's property management operation, located in Orange County, California, leases 108 maintenance vehicles under five-year leases in connection with its property management operations. The Company has not entered into any lease modification arrangements with its landlords as a result of the COVID-19 pandemic.

The carrying value of the Company's lease obligations is as follows:

| (in thousands of U.S. dollars) | December 31, 2020 | 1, 2019 | |
|--|-------------------|---------|-------|
| Balance, beginning of year | \$ 1,089 | \$ | 1,204 |
| Initial recognition on Deemed Acquisition (Note 5) | 5,435 | | - |
| Addition of lease obligation | 1,966 | | _ |
| Interest expense | 328 | | 65 |
| Cash payments | (2,415) | | (180) |
| Balance, end of year | \$ 6,403 | \$ | 1,089 |
| Current portion of lease obligations (Note 11) | \$ 1,804 | \$ | 135 |
| Non-current portion of lease obligations | \$ 4,599 | \$ | 954 |

As at December 31, 2020, the carrying value of the Company's lease obligations was \$6,403 (December 31, 2019 – \$1,089) and the carrying value of the right-of-use asset was \$6,018. During the year ended December 31, 2020, the Company incurred depreciation expense of \$2,314 (2019 – \$158) on the right-of-use asset.

The present value of the minimum lease payments required for the leases over the next five years and thereafter is as follows:

| (in thousands of U.S. dollars) | |
|---|-------------|
| 2021 | \$ 2,053 |
| 2022 | 1,891 |
| 2023 | 1,231 |
| 2024 | 998 |
| 2025 | 465 |
| 2026 and thereafter | 551 |
| Minimum lease payments obligation | 7,189 |
| Imputed interest included in minimum lease payments | (786) |
| Lease obligations | \$ 6,403 |

The current portion of lease obligations is included in amounts payable and accrued liabilities, and the non-current portion of lease obligations is classified as other liabilities.

NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

for the year ended December 31, 2020

(in thousands of U.S. dollars, except per share amounts and percentage amounts)

27. DIVIDENDS

| | | | | | d amount | | | Dividend re | |
|-----------------------------|-----------------------------|------------------|---------------|----------|--------------------|--------------|--------------------|---------------|--------------------|
| (in thousands of dollars | s, except per share amounts | 5) | Common shares | per | share | Total divide | end amount | plan ("DRIP") | |
| Date of declaration | Record date | Payment date | outstanding | CAD | USD ⁽¹⁾ | CAD | USD ⁽¹⁾ | CAD | USD ⁽²⁾ |
| February 24, 2020 | March 31, 2020 | April 15, 2020 | 192,772,071 | \$ 0.070 | \$ 0.049 | \$ 13,494 | \$ 9,512 | \$ 512 | \$ 369 |
| May 14, 2020 | June 30, 2020 | July 15, 2020 | 192,848,390 | 0.070 | 0.051 | 13,499 | 9,906 | 1,773 | 1,302 |
| August 4, 2020 | September 30, 2020 | October 15, 2020 | 193,082,192 | 0.070 | 0.052 | 13,516 | 10,133 | 1,978 | 1,505 |
| November 9, 2020 December 3 | December 31, 2020 | January 15, 2021 | 193,544,915 | 0.070 | 0.055 | 13,548 | 10,641 | 1,780 | 1,407 |
| | | | | | | \$ 54,057 | \$ 40,192 | \$ 6,043 | \$ 4,583 |
| February 25, 2019 | March 31, 2019 | April 15, 2019 | 143,442,251 | \$ 0.070 | \$ 0.052 | \$ 10,041 | \$ 7,514 | \$ 1,159 | \$ 870 |
| May 6, 2019 | June 30, 2019 | July 15, 2019 | 194,389,386 | 0.070 | 0.053 | 13,607 | 10,398 | 1,097 | 842 |
| August 6, 2019 | September 30, 2019 | October 15, 2019 | 194,044,544 | 0.070 | 0.053 | 13,583 | 10,257 | 1,517 | 1,148 |
| November 4, 2019 | December 31, 2019 | January 15, 2020 | 194,328,744 | 0.070 | 0.054 | 13,603 | 10,474 | 1,581 | 1,212 |
| | | | | | | \$ 50,834 | \$ 38,643 | \$ 5,354 | \$ 4,072 |

(1) Dividends are issued and paid in Canadian dollars. For reporting purposes, amounts recorded in equity are translated to U.S. dollars using the daily exchange rate on the date of record. Dividends payable of \$10,641 recorded on the Company's balance sheet are translated to U.S. dollars using the period-end exchange rate and include \$20 related to restricted shares.

(2) Dividends reinvested are translated to U.S. dollars using the daily exchange rate on the date common shares are issued.

The Company has a Dividend Reinvestment Plan ("DRIP") under which eligible shareholders may elect to have their cash dividends automatically reinvested into additional common shares. These additional shares are issued from treasury (or purchased in the open market) at a discount, in the case of treasury issuances, of up to 5% of the Average Market Price, as defined under the DRIP, of the common shares as of the dividend payment date. If common shares are purchased in the open market, they are priced at the average weighted cost to the Company of the shares purchased.

Brokerage, commissions and service fees are not charged to shareholders for purchases or withdrawals of the Company's shares under the DRIP, and all DRIP administrative costs are assumed by the Company.

For the year ended December 31, 2020, 584,974 common shares were issued under the DRIP (2019 – 491,716) for a total amount of \$4,388 (2019 – \$3,793).

28. SHARE CAPITAL

The Company is authorized to issue an unlimited number of common shares. The common shares of the Company do not have par value.

As of December 31, 2020, there were 193,544,915 common shares issued by the Company (December 31, 2019 – 194,328,744), of which 193,175,802 were outstanding (December 31, 2019 – 194,021,133) and 369,113 were reserved to settle restricted share awards in accordance with the Company's Restricted Share Plan (December 31, 2019 – 307,611) (Note 30).

| | December 31, 2020 | | | December 31, 2019 | | | |
|---|-------------------------|--------------|-------------------------|-------------------|--------------|--------------|--|
| | Number of Share capital | | Number of shares issued | Share capital | | | |
| (in thousands of dollars) | | USD | CAD | (repurchased) | USD | CAD | |
| Beginning balance | 194,021,133 | \$ 1,201,061 | \$ 1,529,568 | 143,011,130 | \$ 793,521 | \$ 988,711 | |
| Shares issued to acquire | | | | | | | |
| Starlight U.S. Multi-Family | | | | | | | |
| (No. 5) Core Fund | - | - | - | 50,779,311 | 405,491 | 537,967 | |
| Shares repurchased under | | | | | | | |
| put rights on common shares | | | | | | | |
| issued to acquire Starlight U.S. | | | | | | | |
| Multi-Family (No. 5) Core Fund ⁽¹⁾ | (1,867,675) | (14,922) | (19,797) | - | _ | - | |
| Shares issued under DRIP ⁽²⁾ | 584,974 | 4,388 | 5,844 | 491,716 | 3,793 | 5,046 | |
| Stock options exercised ⁽³⁾ | 291,832 | 1,615 | 2,133 | 73,263 | 258 | 340 | |
| Normal course issuer bid (NCIB) | - | _ | - | (495,402) | (3,067) | (3,906) | |
| Deferred share units exercised ⁽⁴⁾ | 207,040 | 1,362 | 1,791 | 223,328 | 1,555 | 2,056 | |
| Debentures conversion | - | _ | - | 9,560 | 100 | 135 | |
| Shares repurchased and reserved | | | | | | | |
| for restricted share awards ⁽⁵⁾ | (61,502) | (541) | (694) | (71,773) | (590) | (781) | |
| Ending balance | 193,175,802 | \$ 1,192,963 | \$ 1,518,845 | 194,021,133 | \$ 1,201,061 | \$ 1,529,568 | |

(1) On March 4, 2020, the Company repurchased 1,867,675 common shares as a result of holders exercising their put rights in connection with certain shares issued on June 11, 2019 in consideration for the acquisition of Starlight U.S. Multi-Family (No. 5) Core Fund. The fair value of the Company's common shares was \$7.99 (C\$10.60) per share on the acquisition date and \$8.70 (C\$11.65) per share on the exercise date.

(2) In 2020, 584,974 common shares were issued under the DRIP at an average price of \$7.50 (C\$9.99) per share.

(3) In 2020, 644,717 vested stock options were exercised and settled by issuing 291,832 common shares.

(4) In 2020, 207,040 common shares were issued for deferred share units (DSUs) redeemed at an average price of \$6.58 (C\$8.65) per share.

(5) In 2020, 61,502 shares were reserved at \$8.80 (C\$11.28) per share in order to settle restricted share awards granted to employees in 2020 and DRIP with respect to restricted share awards granted in prior years. The restricted shares granted in 2020 will vest on the 10th anniversary of the grant date.

29. EARNINGS PER SHARE

Basic

Basic earnings per share is calculated by dividing net income attributable to shareholders of Tricon by the sum of the weighted average number of shares outstanding and vested deferred share units during the period.

(in thousands of U.S. dollars, except per share amounts which are in U.S. dollars)

| For the years ended December 31 | 2020 | 2019 |
|---|-------------|-------------|
| Net income | \$ 116,413 | \$ 114,135 |
| Non-controlling interest | 3,091 | 2,573 |
| Net income attributable to shareholders of Tricon | \$ 113,322 | \$ 111,562 |
| Weighted average number of common shares outstanding | 192,973,343 | 171,427,128 |
| Adjustments for vested units | 1,653,784 | 1,308,648 |
| Weighted average number of common shares outstanding for basic earnings per share | 194,627,127 | 172,735,776 |
| Basic earnings per share | \$ 0.58 | \$ 0.65 |

Diluted

Diluted earnings per share is calculated by adjusting the weighted average number of shares outstanding to assume conversion of all potentially dilutive shares. The Company has five categories of potentially dilutive shares: stock options (Note 30), restricted shares (Note 28), deferred share units (Note 30), convertible debentures (Note 18) and the preferred units issued by the Affiliate that are exchangeable into the common shares of the Company (Note 19). For the stock options, the number of dilutive shares is based on the number of shares that could have been acquired at fair value with the assumed proceeds, if any, from their exercise (determined using the average market price of the Company's shares for the period then ended). For restricted shares and deferred share units, the number of dilutive shares is equal to the total number of unvested restricted shares and deferred shares into which the elected amount would then be convertible or exchangeable. The number of shares calculated as described above is comparable to the number of shares that would have been issued assuming the vesting of the stock compensation arrangement, the conversion of debentures and the exchange of preferred units.

Stock options, restricted shares and deferred share units

For the year ended December 31, 2020, the Company's stock compensation plans resulted in 1,168,346 dilutive share units (2019 – 1,859,639), given that it would be advantageous to the holders to exercise their associated rights to acquire common shares, as the exercise prices of these potential shares are below the Company's average market share price of \$7.41 (C\$9.94) for the period. Restricted shares and deferred share units are always considered dilutive as there is no price to the holder associated with receiving or exercising their entitlement, respectively.

Convertible debentures

For the year ended December 31, 2020, the Company's 2022 convertible debentures were anti-dilutive, as debentures interest expense, net of tax, and the fair value gain on derivative financial instruments would result in increased earnings per share upon conversion. Therefore, in computing the diluted weighted average shares outstanding and the associated earnings per share amount for the year ended December 31, 2020, the impact of the 2022 convertible debentures was excluded (2019 – included).

Preferred units issued by the Affiliate

For the year ended December 31, 2020, the impact of exchangeable preferred units of Tricon PIPE LLC (Note 19) was anti-dilutive, as the associated interest expense, net of tax, and the fair value loss on derivative financial instruments would result in increased earnings per share upon the exchange of the underlying preferred units. Therefore, in computing the diluted weighted average common shares outstanding and the associated earnings per share amount for the year ended December 31, 2020, the impact of the preferred units was excluded (2019 - N/A).

| (in thousands of U.S. dollars, except per share amounts which are in U.S. dollars) | | |
|---|-------------|-------------|
| For the years ended December 31 | 2020 | 2019 |
| Net income attributable to shareholders of Tricon | \$ 113,322 | \$ 111,562 |
| Adjustment for convertible debentures interest expense – net of tax | - | 11,161 |
| Adjustment for preferred units interest expense – net of tax | - | - |
| Fair value gain on derivative financial instruments and other liabilities | - | (3,279) |
| Adjusted net income attributable to shareholders of Tricon | \$ 113,322 | \$ 119,444 |
| Weighted average number of common shares outstanding | 194,627,127 | 172,735,776 |
| Adjustments for stock compensation | 1,168,346 | 1,859,639 |
| Adjustments for convertible debentures | - | 16,485,713 |
| Adjustments for preferred units | - | - |
| Weighted average number of common shares outstanding for diluted earnings per share | 195,795,473 | 191,081,128 |
| Diluted earnings per share | \$ 0.58 | \$ 0.63 |

30. COMPENSATION EXPENSE

The breakdown of compensation expense, including the annual incentive plan ("AIP") and long-term incentive plan ("LTIP") related to various compensation arrangements, is set out below. AIP awards include both short-term (cash and one-year DSUs) and long-term (three-year DSUs, stock options, restricted shares and PSUs) incentives.

| (in thousands of U.S. dollars) | | |
|-----------------------------------|-----------|-----------|
| For the years ended December 31 | 2020 | 2019 |
| Salaries and benefits | \$ 21,451 | \$ 19,198 |
| Annual incentive plan ("AIP") | 17,787 | 13,855 |
| Long-term incentive plan ("LTIP") | 862 | 4,628 |
| Total compensation expense | \$ 40,100 | \$ 37,681 |

The changes to transactions of the various cash-settled and equity-settled arrangements during the year are detailed in the sections below.

Annual incentive plan

 (in thousands of U.S. dollars)

 For the years ended December 31
 2020
 2019

 Cash component
 \$ 12,088
 \$ 9,806

 Restricted shares, share units and stock options
 5,699
 4,049

 Total AIP expense
 \$ 17,787
 \$ 13,855

The Company's AIP provides for an aggregate bonus pool based on the sum of all employees' individual AIP targets. The portion of the pool attributable to senior executive management is market-benchmarked and subject to an adjustment factor, as approved by the Board, of between 50% and 150%, based on achievement of Company performance objectives determined by the Board at the beginning of each year. The final pool is then allocated among employees based on individual and collective performance. AIP awards will be made in cash and equity-based grants, with the proportion of equity-based awards being correlated to the seniority of an individual's role within the Company.

Cash component

For the year ended December 31, 2020, the Company recognized \$12,088 in cash-based AIP expense (2019 – \$9,806), of which \$11,457 was paid in cash in December 2020 (2019 – \$9,786) and \$631 remained payable (2019 – \$20) as at December 31, 2020. During the year, the Company paid total cash of \$11,477 (2019 – \$10,215) in connection with the AIP, including \$20 of amounts payable from the prior year, which is reflected in the statement of cash flows.

Restricted shares, share units and stock options

For the year ended December 31, 2020, the Company recognized \$5,699 in equity-based AIP expense (2019 – \$4,049), of which \$1,631 relates to performance share units (PSUs), deferred share units (DSUs), stock options and restricted shares granted in December 2020.

The remaining \$4,068 relates to the amortization of PSUs, DSUs, stock options and restricted shares granted in prior years, along with the revaluation of PSUs at each reporting date as the total liability amount is dependent on the Company's share price.

Long-term incentive plan

| Total LTIP expense | \$ 862 | \$ 54 | 4,628 |
|---------------------------------|---------------|-----------|-------|
| Share units and stock options | 2,913 | 1 | 1,785 |
| Cash component | \$ (2,051) | \$ 5 2 | 2,843 |
| For the years ended December 31 | 2020 | | 2019 |
| (in thousands of U.S. dollars) | | | |

Cash component

A liability for cash-component LTIP awards is accrued based on expected performance fees that would be generated from the fair value of the assets within each Investment Vehicle but disbursed only when such performance fees are earned and recognized as revenue. Changes in LTIP are primarily caused by changes to fair values of investments in for-sale housing, which result from timing and cash flow changes at the project level of each Investment Vehicle, and changing business conditions.

For the year ended December 31, 2020, the Company decreased its accrual related to cash-component LTIP by \$2,051 (2019 – increase of \$2,843) as a result of a decrease in expected future performance fees from Investment Vehicles that will be paid to management when cash is received from each investment over time.

The following table summarizes the movement in the LTIP liability:

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|--------------------------------|-------------------|-------------------|
| Balance, beginning of year | \$ 21,409 | \$ 21,407 |
| LTIP (recovery) expense | (2,051) | 2,843 |
| Payments | (1,579) | (3,868) |
| Translation adjustment | 151 | 1,027 |
| Balance, end of year | \$ 17,930 | \$ 21,409 |

Share units and stock options

For the year ended December 31, 2020, the Company recorded \$2,913 in equity-based LTIP expense (2019 – \$1,785), which relates to DSUs and stock options granted in prior years. LTIP expense related to income from THP1 US (a for-sale housing investment) is paid in DSUs vesting in equal tranches over a three-year period commencing on the anniversary date of each grant, pursuant to the LTIP as amended on May 6, 2019. LTIP DSU awards prior to this LTIP amendment date vested equally over a five-year period commencing on the anniversary of each grant. Compensation expense related to the stock options is recognized on a graded vesting basis. No LTIP expense was recognized relating to current-year entitlements for the year ended December 31, 2020.
Stock option plan

For the year ended December 31, 2020, the Company recorded a stock option expense of \$2,321 (2019 – \$764), comprised of \$90 of AIP expense (2019 – \$173) and \$2,231 of LTIP expense (2019 – \$591).

The following table summarizes the movement in the stock option plan during the specified periods:

| For the years ended December 31 | 202 | 20 | 2019 | | |
|---------------------------------|----------------------|---------------------------------------|----------------------|---------------------------------------|--|
| | Number of options | Weighted average exercise price (CAD) | Number of options | Weighted average exercise price (CAD) | |
| Opening balance – outstanding | 4,572,010 | \$ 9.24 | 4,823,960 | \$ 9.18 | |
| Granted | 199,380 | 11.50 | - | - | |
| Exercised | (644,717) | 7.87 | (215,450) | 7.47 | |
| Cancelled | (1,750,334) | 8.55 | - | - | |
| Forfeited | (135,000) | 9.74 | (36,500) | 11.10 | |
| Ending balance – outstanding | 2,241,339 | \$ 10.34 | 4,572,010 | \$ 9.24 | |

The following table presents the inputs used to value the stock options granted in 2020:

| For the year ended December 31 | 2020 |
|---|-------|
| Risk-free interest rate (%) | 0.45 |
| Expected option life (years) | 4.97 |
| Expected volatility (%) | 27.11 |
| Average share price (CAD) during the year | 9.93 |
| Weighted average exercise price (CAD) | 11.50 |

The following table summarizes the stock options outstanding as at December 31, 2020:

| Grant date Expirati | | | December 31, 2020 | | |
|---------------------|-------------------|---------------------|---------------------|---|--|
| | Expiration date | Options outstanding | Options exercisable | Exercise price of outstanding options (CAD) | |
| November 14, 2016 | November 14, 2023 | 650,000 | 650,000 | \$ 8.85 | |
| December 15, 2017 | December 15, 2024 | 965,000 | 965,000 | 11.35 | |
| December 17, 2018 | December 17, 2025 | 426,959 | 284,634 | 9.81 | |
| December 15, 2020 | December 15, 2027 | 199,380 | _ | 11.50 | |
| Total | | 2,241,339 | 1,899,634 | \$ 10.34 | |

AIP liability is recorded within amounts payable and accrued liabilities, and the equity component is included in the contributed surplus. The breakdown is presented below.

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|---|-------------------|-------------------|
| Amounts payable and accrued liabilities (Note 11) | \$ 7,120 | \$ 2,742 |
| Equity – contributed surplus | 8,755 | 7,115 |
| Total AIP | \$ 15,875 | \$ 9,857 |

LTIP liability and equity components are presented on the balance sheet as follows:

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|--------------------------------|-------------------|-------------------|
| LTIP – liability | \$ 17,930 | \$ 21,409 |
| Equity – contributed surplus | 9,557 | 11,872 |
| Total LTIP | \$ 27,487 | \$ 33,281 |

31. SEGMENTED INFORMATION

In accordance with IFRS 8, *Operating Segments* ("IFRS 8"), the Company discloses information about its reportable segments based upon the measures used by management in assessing the performance of those reportable segments. The Company evaluates segment performance based on the revenue and net income of each operating segment.

Tricon is comprised of four operating segments and five reportable segments. The Company's corporate office provides support functions, and therefore, it does not represent an operating segment but rather it is included as a reportable segment. The reportable segments are business units offering different products and services, and are managed separately due to their distinct natures although they are related and complementary.

These five reportable segments have been determined by the Company's chief operating decision-makers.

- Single-Family Rental business includes owning and operating single-family rental homes primarily within major cities in the U.S. Sun Belt.
- Multi-Family Rental business includes owning and operating garden-style multi-family rental properties primarily in the U.S. Sun Belt and condominium-quality rental apartments in downtown Toronto. The Selby, a Canadian multi-family rental property, is included within this segment; however, given that it is an equity-accounted investment, its operational results are presented as a single line within this segment.
- Residential Development business includes designing and developing premier multi-family rental properties in Toronto. Canadian development properties (The James and The Shops of Summerhill) and the Company's remaining equity-accounted Canadian multi-family development activities are included in this segment. The segment also includes Tricon's legacy investments in for-sale housing developments.
- Private Funds and Advisory business includes providing asset management, property management and development
 management services. The Company's asset management services are provided to Investment Vehicles that own the singlefamily rental homes, multi-family rental properties and residential developments described above. The Company's property
 management function generates property management fees, construction management fees and leasing commissions through
 its technology-enabled platform used to operate the Company's rental portfolio. In addition, Tricon earns market-based
 development management fees from its residential developments in the U.S. and Canada.
- Corporate activities include providing support functions in the areas of accounting, treasury, credit management, information technology, legal, and human resources. Certain corporate costs such as directly identifiable compensation expense incurred on behalf of the Company's operating segments are allocated to each operating segment, where appropriate. Certain property management activities are also considered as part of corporate-level costs for the purpose of segment reporting. Those costs include salaries of employees engaged in leasing, acquisition, disposition and other property management-related activities.

Any direct property-level operating expenses are included in the net operating income of the single-family rental and multi-family rental businesses to which they belong.

The financial reporting changes to the Company's basis of preparation, effective January 1, 2020 and as outlined in Note 2, have been applied on a prospective basis in accordance with the relevant guidance of IFRS 10 and, as such, the presentation of comparative periods reflects Investment Entity Accounting as previously reported.

Inter-segment revenues adjustments

Inter-segment revenues are determined under terms that approximate market value. For the year ended December 31, 2020, the adjustment to external revenues when determining segmented revenues consists of property management revenues earned from consolidated entities totalling \$44,569 and development revenues earned from consolidated entities totalling \$740, which were eliminated on consolidation to arrive at the Company's consolidated revenues in accordance with IFRS.

| (in thousands of U.S. dollars) | Single Femily | Multi Family | Residential | Private Funds | | Consolidated |
|---|--|---------------------------------------|----------------------------|-----------------------------|--------------------------|--------------|
| For the year ended December 31, 2020 | Single-Family Rental ⁽¹⁾ | Multi-Family Rental ⁽¹⁾ | Development ⁽¹⁾ | and Advisory ⁽¹⁾ | Corporate ⁽¹⁾ | results |
| Revenue from rental properties | \$ 366,982 | \$ 111,205 | \$ - | \$ - | \$ - | \$ 478,187 |
| Direct operating expenses | (121,242) | (48,296) | - | - | - | (169,538) |
| Net operating income | | | | | | |
| from rental properties | 245,740 | 62,909 | - | - | - | 308,649 |
| Revenue from private funds | | | | | | |
| and advisory services | - | - | - | 34,090 | - | 34,090 |
| Income from investments | | | | | | |
| in Canadian multi-family | | | | | | |
| developments | - | 746 | 13,378 | - | - | 14,124 |
| Other income from Canadian | | | | | | |
| development properties | - | - | 791 | - | - | 791 |
| Loss from investments | | | | | | |
| in for-sale housing | - | - | (61,776) | - | - | (61,776) |
| Property management overhead | - | - | - | - | (22,654) | (22,654) |
| Compensation expense | - | - | - | (11,652) | (28,448) | (40,100) |
| General and administration | | | | | | |
| expense | (9,101) | (2,111) | - | (879) | (11,478) | (23,569) |
| Other (expense) income | (3,173) | - | - | 1,774 | - | (1,399) |
| Interest expense | (101,574) | (33,464) | (524) | - | (35,048) | (170,610) |
| Fair value gain (loss) | | | | | | |
| on rental properties | 220,849 | (22,535) | - | - | - | 198,314 |
| Fair value loss on derivative | | | | | | |
| financial instruments | | | | | | |
| and other liabilities | (39) | - | - | - | (7,422) | (7,461) |
| Transaction costs | (24) | (2,409) | - | - | (11,583) | (14,016) |
| Amortization and | | | | | | |
| depreciation expense | - | (22) | - | (3,079) | (7,747) | (10,848) |
| Realized and unrealized foreign | | | | | | |
| exchange gain (loss) | - | 4 | - | - | (170) | (166) |
| Net change in fair value of | | | | | | |
| limited partners' interests | (EQ EQ 1) | | | | | (50,504) |
| in rental business | (50,581) | - 5 | - | - | - | (50,581) |
| Income tax (expense) recovery | (319) | | 7,973 | (3) | (44,031) | (36,375) |
| Net income (loss) | \$ 301,778 | \$ 3,123 | \$ (40,158) | \$ 20,251 | \$ (168,581) | \$ 116,413 |

(1) Financial information for each segment is presented on a consolidated basis.

NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

for the year ended December 31, 2020

(in thousands of U.S. dollars, except per share amounts and percentage amounts)

| (in thousands of U.S. dollars) | | | | | | |
|---|-------------------------|------------------------|----------------------------|-------------------------------|--------------------------|----------------------|
| For the year ended December 31, 2019 | Single-Family Rental | Multi-Family Rental | Residential Development | Private Funds and Advisory | Corporate ⁽¹⁾ | Consolidated results |
| Revenue from private funds | | | | | | |
| and advisory services | \$ – | \$ - | \$ - | \$ 39,895 | \$ – | \$ 39,895 |
| Income from investments | | | | | | |
| in for-sale housing | - | - | 9,646 | - | - | 9,646 |
| Compensation expense | - | - | _ | - | (37,681) | (37,681) |
| General and administration | | | | | | |
| expense | - | - | - | - | (11,683) | (11,683) |
| Interest expense | _ | - | - | - | (32,439) | (32,439) |
| Fair value gain on derivative | | | | | | |
| financial instruments | | | | | | |
| and other liabilities | - | - | - | - | 2,961 | 2,961 |
| Transaction costs | - | - | - | - | (32,626) | (32,626) |
| Amortization and | | | | | | |
| depreciation expense | - | - | - | - | (6,274) | (6,274) |
| Realized and unrealized | | | | | | |
| foreign exchange gain | - | - | - | - | 42 | 42 |
| Investment income – | | | | | | |
| Tricon American Homes | 162,193 | - | - | - | - | 162,193 |
| Investment income – | | | | | | |
| Tricon Lifestyle Rentals | - | 13,508 | 11,754 | - | 9,718 | 34,980 |
| Income tax expense | - | _ | _ | - | (14,879) | (14,879) |
| Net income (loss) | \$ 162,193 | \$ 13,508 | \$ 21,400 | \$ 39,895 | \$ (122,861) | \$ 114,135 |

(1) Investment income from Tricon Lifestyle Rentals assets held for sale is included in the Corporate column.

32. RELATED PARTY TRANSACTIONS AND BALANCES

Related parties include subsidiaries, associates, joint ventures, structured entities, key management personnel, the Board of Directors ("Directors"), immediate family members of key management personnel and Directors, and entities which are directly or indirectly controlled by, jointly controlled by or significantly influenced by key management personnel, Directors or their close family members.

In the normal course of operations, the Company executes transactions on market terms with related parties that have been measured at the exchange value and are recognized in the consolidated financial statements, including, but not limited to: asset management fees, performance fees and incentive distributions; loans, interest and non-interest bearing deposits; purchase and sale agreements; capital commitments to Investment Vehicles; and development of residential real estate assets. In connection with the Investment Vehicles, the Company has unfunded capital commitments of \$199,952 as at December 31, 2020. Transactions and balances between consolidated entities are fully eliminated upon consolidation. Transactions and balances with unconsolidated structured entities are disclosed in Note 19.

Transactions with related parties

The following table lists the related party balances included within the consolidated financial statements.

| (in thousands of U.S. dollars) | | |
|---|-------------|------------|
| For the years ended December 31 | 2020 | 2019 |
| Revenue from private funds and advisory services | \$ 34,090 | \$ 39,895 |
| Income from investments in Canadian multi-family developments | 14,124 | - |
| (Loss) income from investments in for-sale housing | (61,776) | 9,646 |
| Other expense | (3,173) | - |
| Investment income – Tricon American Homes | - | 162,193 |
| Investment income – Tricon Lifestyle Rentals | - | 34,980 |
| Net (loss) income recognized from related parties | \$ (16,735) | \$ 246,714 |

Balances arising from transactions with related parties

The items set out below are included on various line items in the Company's consolidated financial statements.

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 | |
|---|-------------------|-------------------|--|
| Receivables from related parties included in amounts receivable | | | |
| Contractual fees and other receivables from investments managed | \$ 8,855 | \$ 5,404 | |
| Employee relocation housing loans ⁽¹⁾ | 2,001 | 2,065 | |
| Loan receivables from portfolio investments | 13,937 | 16,757 | |
| Annual incentive plan ⁽²⁾ | 15,875 | 9,857 | |
| Long-term incentive plan ⁽²⁾ | 27,487 | 33,281 | |
| Dividends payable | 440 | 399 | |
| Other payables to related parties included in amounts payable and accrued liabilities | 972 | 161 | |

(1) The employee relocation housing loans are non-interest bearing for a term of ten years, maturing between 2024 and 2028.

(2) Balances from compensation arrangements are due to employees deemed to be key management of the Company.

The receivables are unsecured and non-interest bearing. There are no provisions recorded against receivables from related parties at December 31, 2020 (December 31, 2019 – nil).

Key management compensation

Key management includes the Named Executive Officers ("NEOs"), who are the Chief Executive Officer, Chief Financial Officer and the three other most highly-compensated executive officers or the three most highly compensated individuals acting in a similar capacity at the end of the financial year. Compensation paid and awarded to key management for employee services is based on employment agreements and is as follows:

| Total key management compensation | \$ 10,785 | \$ 12,322 |
|-----------------------------------|-----------|-----------|
| Total LTIP | 740 | 2,590 |
| Total AIP | 7,955 | 7,725 |
| Total salaries and benefits | \$ 2,090 | \$ 2,007 |
| For the years ended December 31 | 2020 | 2019 |
| (in thousands of U.S. dollars) | | |

33. FINANCIAL RISK MANAGEMENT

The Company is exposed to the following risks as a result of holding financial instruments: market risk (i.e., interest rate risk, foreign currency risk and other price risk that may impact the fair value of financial instruments), credit risk and liquidity risk. The following is a description of these risks and how they are managed.

Market risk

Market risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market prices. Market risk includes the risk of changes in interest rates, foreign currency rates and changes in market prices due to other factors, such as changes in equity prices or credit spreads. The Company manages market risk from foreign currency assets and liabilities and the impact of changes in currency exchange rates and interest rates by funding assets with financial liabilities in the same currency and with similar interest rate characteristics, and by holding financial contracts such as interest rate derivatives to minimize residual exposures.

The sensitivities to market risks included below are based on a change in one factor while holding all other factors constant. In practice, this is unlikely to occur, and changes in some of the factors may be correlated – for example, changes in interest rates and changes in foreign currency rates.

Financial instruments held by the Company that are subject to market risk include other financial assets, borrowings and derivative instruments such as interest rate cap contracts.

Interest rate risk

Interest rate risk is the risk that the fair value or future cash flows of a financial instrument will fluctuate because of changes in market interest rates. The observable impacts on the fair values and future cash flows of financial instruments that can be directly attributable to interest rate risk include changes in the net income from financial instruments whose cash flows are determined with reference to floating interest rates and changes in the value of financial instruments whose cash flows are fixed in nature.

The Company's assets largely consist of long-term interest-sensitive physical real estate assets. Accordingly, the Company's financial liabilities consist primarily of long-term fixed-rate debt or floating-rate debt that has been swapped with interest rate derivatives. These financial liabilities are recorded at their amortized cost. The Company also holds interest rate caps to limit its exposure to increases in interest rates on floating-rate debt that has not been swapped, and sometimes holds interest rate contracts to lock in fixed rates on anticipated future debt issuances and as an economic hedge against the changes in the value of long-term interest-sensitive physical real estate assets that have not been otherwise matched with fixed-rate debt. Borrowings issued at variable rates expose the Company to cash flow interest rate risk. To limit its exposure to interest rate risk, the Company has a mixed portfolio of fixed-rate and variable-rate debt, with \$3,152,455 in fixed-rate debt and \$1,011,612 in variable-rate debt as at December 31, 2020. If interest rates had been 50 basis points higher or lower, with all other variables held constant, interest expense would have increased (decreased) by:

| For the years ended December 31 | 2020 | | 2019 | |
|---------------------------------|---------------------------------|------------|-----------------|-----------------|
| (in thousands of U.S. dollars) | 50 bps increase 50 bps decrease | | 50 bps increase | 50 bps decrease |
| Interest expense | \$ 6,791 | \$ (4,585) | \$ 1,415 | \$ (1,415) |

Foreign currency risk

Changes in foreign currency rates will impact the carrying value of financial instruments denominated in currencies other than the U.S. dollar, which is the functional and presentation currency of the Company. The Company has exposure to monetary and non-monetary foreign currency risk due to the effects of changes in foreign exchange rates related to consolidated Canadian subsidiaries, equity-accounted investments, and cash and debt in Canadian dollars held at the corporate level. The Company manages foreign currency risk by raising equity in Canadian dollars and by matching its principal cash outflows to the currency in which the principal cash inflows are denominated.

The impact of a 1% increase or decrease in the Canadian dollar exchange rate would result in the following impacts to assets and liabilities:

| | December | 31, 2020 | December 31, 2019 | |
|---|-------------|-------------|-------------------|-------------|
| (in thousands of U.S. dollars) | 1% increase | 1% decrease | 1% increase | 1% decrease |
| Assets | | | | |
| Investments in Canadian multi-family developments | \$ 954 | \$ (954) | \$ - | \$ - |
| Canadian development properties | 1,107 | (1,107) | - | - |
| Investments in for-sale housing | 6 | (6) | 54 | (54) |
| Investments – Tricon Lifestyle Rentals | - | - | 969 | (969) |
| | \$ 2,067 | \$ (2,067) | \$ 1,023 | \$ (1,023) |
| Liabilities | | | | |
| Canadian debt | \$ 715 | \$ (715) | \$ 112 | \$ (112) |
| Other liabilities | - | - | 134 | (134) |
| | \$ 715 | \$ (715) | \$ 246 | \$ (246) |

Foreign exchange volatility is already embedded in the fair value of derivative financial instruments (Note 20), and therefore is excluded from the sensitivity calculations above.

Other price risk

Other price risk is the risk of variability in fair value due to movements in equity prices or other market prices such as commodity prices and credit spreads. The Company does not hold any financial instruments that are exposed to equity price risk including equity securities and equity derivatives.

Credit risk

Credit risk is the risk that one party to a financial instrument will cause financial loss for the other party by failing to discharge an obligation. Management believes the credit risk on cash is low because the counterparties are banks with high credit ratings.

Credit risk arises from the possibility that residents may experience financial difficulty and be unable to fulfill their lease commitments. A provision for bad debt (or expected credit loss) is taken for all anticipated collectability risks. The Company also manages credit risk by performing resident underwriting due diligence during the leasing process. As at December 31, 2020, the Company had rent receivable of \$4,274 (December 31, 2019 – N/A), net of bad debt, which adequately reflects the Company's credit risk.

The Company has no significant concentrations of credit risk and its exposure to credit risk arises through loans and receivables which are due primarily from associates. The loans and receivables due from associates are subject to the risk that the underlying real estate assets may not generate sufficient cash inflows in order to recover them. The Company manages this risk by:

- Ensuring a due diligence process is conducted on each investment prior to funding;
- · Approving all loan disbursements by management;
- · Approving of total loan facilities by the Investment Committee; and
- · Actively monitoring the loan portfolio and initiating recovery procedures when necessary.

The Company assesses all counterparties, including its partners, for credit risk before contracting with them. The Company does not include any collateral or other credit risk enhancers, which may reduce the Company's exposure.

The Company provides loans to land developers, which are represented as debt investments. The credit quality of these investments is based on the financial performance of the underlying real estate assets. For those assets that are not past due, it is believed that the capital repayments and interest payments will be made in accordance with the agreed terms and conditions. No terms or conditions have been renegotiated.

At December 31, 2020, the Company's exposure to credit risk arising from its investment in debt instruments was \$13,937 (December 31, 2019 – \$16,757). Through the equity portion of its investments, the Company is also indirectly exposed to credit risk arising on loans advanced by investees to individual real estate development projects.

Liquidity risk

The real estate industry is highly capital intensive. Liquidity risk is the risk that the Company may have difficulty in meeting obligations associated with its financial liabilities as they fall due. Liquidity risk also includes the risk of not being able to liquidate assets in a timely manner at a reasonable price. The Company's liquidity risk management includes maintaining sufficient cash on hand and the availability of funding through an adequate amount of committed credit facilities, as well as performing periodic cash flow forecasts to ensure the Company has sufficient cash to meet operational and financing costs. The Company's primary source of liquidity consists of cash and other financial assets, net of deposits and other associated liabilities, and undrawn available credit facilities. Cash flow generated from operating the rental property portfolio represents the primary source of liquidity used to service the interest on the property-level debt and fund direct property operating expenses, as well as reinvest in the portfolio through capital expenditures.

The Company is subject to the risks associated with debt financing, including the ability to refinance indebtedness at maturity. The Company believes these risks are mitigated through the use of long-term debt secured by high-quality assets, by maintaining certain debt levels that are set by management, and by staggering maturities over an extended period.

Despite the Company's prudent liquidity management, the ongoing COVID-19 pandemic has introduced new challenges to the business environment which called for a necessary reassessment of its impact on the Company's cash flow, earnings and balance sheet profile. Current lending markets are re-evaluating capital allocations, and this may affect new loan originations by reducing the availability of funds or increasing the cost of interest. To date, there has not been any indication that existing credit facilities or Tricon's ability to originate new debt has been impacted by the COVID-19 pandemic. During the year ended December 31, 2020, Tricon raised \$993,934 of gross proceeds through two securitization transactions which reduced the Company's effective interest rate and extended the weighted average maturity of its debt.

The following tables present the contractual maturities of the Company's financial liabilities at December 31, 2020 and December 31, 2019, excluding remaining unamortized deferred financing fees and debt discount:

| (in thousands of U.S. dollars) | Due on demand and within | From 1 to | From 3 to | From 5 years | |
|---|-----------------------------|--------------|--------------|--------------|--------------|
| As at December 31, 2020 | the year | 2 years | 4 years | and later | Total |
| Liabilities | | | | | |
| Debt ⁽¹⁾ | \$ 274,526 | \$ 1,236,540 | \$ 1,325,709 | \$ 1,327,292 | \$ 4,164,067 |
| Other liabilities | - | 3,122 | 1,463 | 551 | 5,136 |
| Limited partners' interests in rental business | - | - | - | 356,305 | 356,305 |
| Convertible debentures | - | 172,400 | - | - | 172,400 |
| Derivative financial instruments ⁽²⁾ | - | - | - | 45,494 | 45,494 |
| Due to Affiliate | - | - | - | 300,000 | 300,000 |
| Amounts payable and accrued liabilities | 98,290 | - | - | - | 98,290 |
| Resident security deposits | 45,157 | - | - | - | 45,157 |
| Dividends payable | 10,641 | - | - | - | 10,641 |
| Total | \$ 428,614 | \$ 1,412,062 | \$ 1,327,172 | \$ 2,029,642 | \$ 5,197,490 |

(1) The contractual maturities reflect the maturity dates after all extensions have been exercised. The Company intends to exercise the extension options available on all loans.

(2) Includes the exchange/prepayment option related to Due to Affiliate (Note 20). Excludes the conversion and redemption options related to the 2022 convertible debentures as the fair value is an asset to the Company as at December 31, 2020.

| (in thousands of U.S. dollars) As at December 31, 2019 | n demand and within the year | From 1 to 2 years | From 3 to 4 years | Fro | om 5 years and later | Total |
|---|--|----------------------|----------------------|-----|-------------------------|---------------|
| Liabilities | | | | | | |
| Debt | \$ 284 | \$ 297,605 | \$ 10,264 | \$ | - | \$ 308,153 |
| Other liabilities | 13,375 | 311 | 374 | | 269 | 14,329 |
| Convertible debentures | - | 172,400 | - | | - | 172,400 |
| Derivative financial instruments | - | 657 | - | | - | 657 |
| Amounts payable and accrued liabilities | 26,190 | - | - | | - | 26,190 |
| Dividends payable | 10,474 | - | - | | - | 10,474 |
| Total | \$ 50,323 | \$ 470,973 | \$ 10,638 | \$ | 269 | \$ 532,203 |

The future repayments of principal and interest on financial liabilities are as follows, excluding remaining unamortized deferred financing fees and debt discount:

| (in thousands of U.S. dollars) | Within | From 1 to | From 3 to | From 5 years | |
|---------------------------------|------------|--------------|--------------|--------------|--------------|
| As at December 31, 2020 | the year | 2 years | 4 years | and later | Total |
| Principal | | | | | |
| Debt ^{(1),(2)} | \$ 274,526 | \$ 1,236,540 | \$ 1,325,709 | \$ 1,327,292 | \$ 4,164,067 |
| Convertible debentures | - | 172,400 | - | - | 172,400 |
| Due to Affiliate | - | _ | - | 300,000 | 300,000 |
| Interest | | | | | |
| Debt ⁽¹⁾ | 126,939 | 194,297 | 117,017 | 27,523 | 465,776 |
| Convertible debentures | 9,913 | 4,957 | _ | - | 14,870 |
| Due to Affiliate ⁽³⁾ | 17,250 | 34,500 | 34,500 | 161,896 | 248,146 |
| Total | \$ 428,628 | \$ 1,642,694 | \$ 1,477,226 | \$ 1,816,711 | \$ 5,365,259 |

(1) Certain mortgages' principal and interest repayments were translated to U.S. dollars at the year-end exchange rate.

(2) The contractual maturities reflect the maturity dates after all extensions have been exercised. The Company intends to exercise the extension options available on all loans.

(3) Reflects the contractual maturity date of September 3, 2032.

The details of the net liabilities are shown below:

| (in thousands of U.S. dollars) | December 31, 2020 | December 31, 2019 |
|---|-------------------|-------------------|
| Cash | \$ 55,158 | \$ 8,908 |
| Amounts receivable | 25,593 | 8,952 |
| Prepaid expenses and deposits | 13,659 | 796 |
| Current assets | 94,410 | 18,656 |
| Amounts payable and accrued liabilities | 98,290 | 26,190 |
| Resident security deposits | 45,157 | - |
| Dividends payable | 10,641 | 10,474 |
| Current portion of long-term debt | 274,190 | 284 |
| Current liabilities | 428,278 | 36,948 |
| Net current liabilities | \$ (333,868) | \$ (18,292) |

During the year ended December 31, 2020, the change in the Company's liquidity resulted in a working capital deficit of \$333,868 (December 31, 2019 – deficit of \$18,292). The working capital deficit is driven primarily by debt coming due in 2021, including \$116,000 relating to the SFR JV-1 subscription facility and \$109,890 relating to the U.S. multi-family credit facility. The SFR JV-1 subscription facility will be partially repaid with limited partners' capital contributions as per the joint venture agreement, and the U.S. multi-family credit facility is expected to be repaid prior to maturity in connection with the syndication of the Company's majority interest in its U.S. multi-family portfolio. The Company has determined that its current financial obligations and working capital deficit are adequately funded from the available borrowing capacity and from operating cash flows. In addition, the Company has set aside cash in separate bank accounts, presented as non-current restricted cash on the consolidated balance sheets, to settle its obligations for resident security deposits.

As of December 31, 2020, the outstanding amount under the corporate credit facility was \$26,000 (December 31, 2019 – \$297,000) and \$474,000 of the corporate credit facility remained available to the Company. During the year ended December 31, 2020, the Company received distributions of \$78,378 (2019 – \$200,631) from its investments.

34. CAPITAL MANAGEMENT

The Company's objectives when managing capital are: (i) to safeguard its ability to meet financial obligations and growth objectives, including future acquisitions; (ii) to provide an appropriate return to its shareholders; and (iii) to maintain an optimal capital structure that allows multiple financing options, should a financing need arise. The Company's capital consists of debt (including credit facilities, term loans, mortgages, securitizations, convertible debentures and Due to Affiliate), cash and shareholders' equity. In order to maintain or adjust the capital structure, the Company manages equity as capital and may adjust the amount of dividends paid to shareholders, return capital to shareholders, issue new shares or subsidiary entity interests, repurchase and cancel shares or sell assets.

The Company discussed the potential effect of the current COVID-19 pandemic in relation to the Company's liquidity risk in Note 33. Management believes that understanding the alternative funding options that are available during times of volatility and how to access those are also a prudent part of capital management of the Company.

As of December 31, 2020, the Company was in compliance with all financial covenants in its debt facilities (Note 17).

35. WORKING CAPITAL CHANGES AND OTHER NON-CASH ITEMS

The details of the adjustments for other non-cash items presented in operating activities of the cash flow statement are shown below:

| (in thousands of U.S. dollars) | | |
|---|-----------|--------------|
| For the years ended December 31 | 2020 | 2019 |
| Amortization of debt and debentures discount | | |
| and financing costs (Note 21) | \$ 10,594 | \$ 4,299 |
| Interest on lease obligation (Note 21) | 328 | 65 |
| Long-term incentive plan (Note 30) | 862 | 4,628 |
| Annual incentive plan (Note 30) | 17,787 | 13,855 |
| Unrealized foreign exchange gain | (7,231) | (1,965) |
| Accrued investment income from single-family rental | - | (162,193) |
| Accrued investment income from multi-family rental | - | (34,980) |
| Other non-cash items | \$ 22,340 | \$ (176,291) |

The following table presents the changes in non-cash working capital items for the years ended December 31, 2020 and December 31, 2019.

(in thousands of U.S. dollars)

| For the years ended December 31 | 2020 | 2019 |
|---|-------------|--------------|
| Amounts receivable | \$ (16,641) | \$ 8,982 |
| Prepaid expenses and deposits | (12,863) | 23 |
| Resident security deposits | 45,157 | - |
| Amounts payable and accrued liabilities | 72,100 | 19,626 |
| Non-cash working capital items acquired on Deemed Acquisition (Note 5) | (88,218) | - |
| Non-cash working capital items acquired with Canadian development properties (Note 8) | (4,878) | - |
| Changes in non-cash working capital items | \$ (5,343) | \$ 28,631 |

36. FINANCING ACTIVITIES

| | | | | Non-cash | changes | | | |
|--------------------------------|----------------------------------|-------------|---------------------------------|-----------------------|--------------------------|----------------------|-------------------------------|--|
| (in thousands of U.S. dollars) | As at December 31, 2019 Ca | Cash flows | Foreign exchange movement | Fair value changes | Additions ⁽¹⁾ | Other ⁽²⁾ | As at December 31, 2020 | |
| SFR JV-1 subscription | | | | | | | | |
| facility | \$ – | \$ (70,001) | \$ - | \$ – | \$ 185,161 | \$ 504 | \$ 115,664 | |
| SFR JV-1 warehouse | | | | | | | | |
| credit facility | - | (115,340) | - | _ | 209,998 | 1,292 | 95,950 | |
| Term Ioan 2 | - | - | - | - | 96,077 | - | 96,077 | |
| Warehouse credit facility | - | (19,770) | - | - | 29,864 | 16 | 10,110 | |
| Securitization debt 2016-1 | - | (357,478) | _ | _ | 357,478 | _ | - | |
| Securitization debt 2017-1 | - | (1,771) | _ | _ | 461,301 | _ | 459,530 | |
| Term Ioan | - | (255) | _ | _ | 375,000 | _ | 374,745 | |
| Securitization debt 2017-2 | - | (897) | _ | _ | 363,357 | 223 | 362,683 | |
| Securitization debt 2018-1 | _ | (1,403) | _ | _ | 313,093 | 223 | 311,913 | |
| SFR JV-1 securitization | | | | | | | | |
| debt 2019-1 | - | (10) | _ | _ | 325,511 | 1,266 | 326,767 | |
| SFR JV-1 securitization | | | | | | | | |
| debt 2020-1 | - | 543,001 | _ | _ | _ | 802 | 543,803 | |
| Securitization debt 2020-2 | _ | 432,662 | _ | _ | _ | 155 | 432,817 | |
| U.S. multi-family | | | | | | | | |
| credit facility | _ | (6,000) | _ | _ | 115,890 | _ | 109,890 | |
| Mortgage tranche A | _ | _ | _ | _ | 160,090 | _ | 160,090 | |
| Mortgage tranche B | _ | _ | _ | _ | 400,225 | _ | 400,225 | |
| Mortgage tranche C | _ | _ | _ | _ | 240,135 | _ | 240,135 | |
| Vendor take-back (VTB) | | | | | | | | |
| loan 2020 ⁽³⁾ | - | (10,880) | 566 | _ | 10,314 | _ | - | |
| Land loan | _ | _ | 940 | _ | 21,051 | _ | 21,991 | |
| Vendor take-back (VTB) | | | | | | | | |
| loan 2021 | - | _ | 1,680 | _ | 23,884 | _ | 25,564 | |
| Mortgage | _ | (379) | 817 | _ | 12,019 | 6 | 12,463 | |
| Corporate credit facility | 297,000 | (271,000) | _ | _ | _ | _ | 26,000 | |
| Corporate office mortgages | 11,153 | (275) | 211 | _ | _ | _ | 11,089 | |
| 2022 convertible | , | · · · · | | | | | , | |
| debentures | 161,311 | _ | _ | _ | _ | 4,645 | 165,956 | |
| Due to Affiliate | _ | 287,798 | _ | _ | (37,613) | 1,462 | 251,647 | |
| Derivative financial | | | | | , , , | , | | |
| instruments ⁽⁴⁾ | 657 | _ | _ | 6,422 | 37,574 | 841 | 45,494 | |
| Limited partners' interests | | | | | | | | |
| in rental business | _ | 19,950 | _ | 50,581 | 285,774 | _ | 356,305 | |
| Lease obligations | 1,089 | (2,415) | _ | _ | 7,401 | 328 | 6,403 | |
| Other liabilities | 13,375 | (14,922) | _ | 1,039 | 508 | _ | _ | |
| Total liabilities from | | | | | | | | |
| financing activities | \$ 484,585 | \$ 410,615 | \$ 4,214 | \$ 58,042 | \$ 3,994,092 | \$ 11,763 | \$ 4,963,311 | |

(1) Includes debt of \$3,647,108, lease liability of \$5,435 and derivative financial instruments of \$28 recognized as part of the Deemed Acquisition (Note 5) and \$53,340 of assumed debt and vendor take-back loans in connection with Tricon's purchase of the remaining ownership interests of 50% and 75% in The James and The Shops of Summerhill, respectively (Note 8).

(2) Includes amortization of transaction costs and debt discount and interest on lease obligations.

(3) Tricon entered into VTB loan 2020 as part of its acquisition of the remaining interests in the Canadian development properties (Note 8), and this loan was fully repaid during the year.

(4) As at December 31, 2020, the embedded derivative on the 2022 convertible debentures was an asset of \$841, and was reclassified from liability to asset on the consolidated balance sheet.

NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

for the year ended December 31, 2020

(in thousands of U.S. dollars, except per share amounts and percentage amounts)

| (in thousands of U.S. dollars) | As at December 31, 2018 | Cash flows | Foreign exchange movement | Fair value changes | Additions | Other ⁽¹⁾ | As at December 31, 2019 |
|--------------------------------|-------------------------------|----------------|---------------------------------|-----------------------|-----------|----------------------|-------------------------------|
| Corporate credit facility | \$ 209,250 | \$ 87,750 | \$ - | \$ - | \$ – | \$ - | \$ 297,000 |
| Corporate office mortgages | 7,150 | 3,567 | 436 | _ | _ | _ | 11,153 |
| 2022 convertible | | | | | | | |
| debentures | 157,112 | - | _ | - | _ | 4,199 | 161,311 |
| Derivative financial | | | | | | | |
| instruments | 3,936 | - | _ | (3,279) | - | _ | 657 |
| Lease obligations | 1,204 | (180) | - | - | - | 65 | 1,089 |
| Other liabilities | - | - | _ | 318 | 13,057 | _ | 13,375 |
| Total liabilities from | | | | | | | |
| financing activities | \$ 378,652 | \$ 91,137 | \$ 436 | \$ (2,961) | \$ 13,057 | \$ 4,264 | \$ 484,585 |

 Includes amortization of debentures discount and amortization of debentures issuance costs, offset by the conversion of \$100 principal amount of the 2022 convertible debentures into common shares, along with interest on lease obligation.

37. INDEMNIFICATION

Pursuant to Indemnification Agreements with certain General Partners of Limited Partnerships managed by the Company and certain shareholders of the Company (who are also officers and directors of the Company), the Company has agreed to indemnify the General Partners and those shareholders and, where applicable, any of their directors, officers, agents and employees (collectively, the Indemnified Parties) for any past, present or future amounts paid or payable by any of the Indemnified Parties to the Limited Partnership in the form of a capital contribution or clawback guarantee relating to performance fees for any claim or obligation, as set out in the Limited Partnership Agreements. There are no amounts payable in respect of this indemnification as of December 31, 2020 (December 31, 2019 – nil).

38. SUBSEQUENT EVENTS

U.S. multi-family rental portfolio syndication

On February 25, 2021, the Company announced that it had reached an agreement in principle to enter into a joint venture arrangement with two institutional investors. Under the joint venture, the investors will acquire a combined 80% ownership interest in Tricon's existing portfolio of 23 U.S. multi-family apartments and Tricon will retain a 20% ownership interest. The transaction reflects a total portfolio value of \$1,331,000 including in-place debt, and is expected to generate gross proceeds of approximately \$425,000 to Tricon, which will be used to repay outstanding debt and for general corporate purposes. The transaction is expected to close in March of 2021, subject to finalizing definitive documentation and customary closing conditions including obtaining the necessary lender consents.

Quarterly dividend

On March 2, 2021, the Board of Directors of the Company declared a dividend of seven cents per common share in Canadian dollars payable on or after April 15, 2021 to shareholders of record on March 31, 2021.

Corporate Information

Board of Directors

David Berman Executive Chairman and Co-Founder

Gary Berman President and Chief Executive Officer, Director

Frank Cohen Independent Director

Camille Douglas Independent Director

Ira Gluskin Independent Director

Michael Knowlton Chair of Audit Committee

Siân M. Matthews Chair of the Compensation, Nominating and Corporate Governance Committee

Geoff Matus Director and Co-Founder

Peter D. Sacks Independent Lead Director

Tracy Sherren Independent Director

Annual General Meeting Details

Annual and Special Shareholders Meeting

Date June 23, 2021 Time 10:00am ET Place Virtual Conference Details www.triconresidential.com/investors

Shareholder Information

Exchange and Symbol TSX: TCN

Corporate Head Office 7 St. Thomas Street, Suite 801 Toronto, Ontario M5S 2B7

Auditors PricewaterhouseCoopers LLP Toronto, Ontario

Legal Counsel Goodmans LLP Toronto, Ontario

Transfer Agent TSX Trust Company 100 Adelaide Street West Toronto, Ontario M5H 4H1 1 866 393 4891 www.tsxtrust.com

Investor Relations Contacts

Wissam Francis Executive Vice President and Chief Financial Officer 416 323 2484

Wojtek Nowak Managing Director, Capital Markets 416 925 2409



causes were supported by our employees in 2020.

being at the centre of everything we do.

We're committed to bettering the lives of the people who live in our communities. Our commitment to social responsibility extends to all of our stakeholders, including employees, customers, communities and investors. However, we believe that social responsibility starts at home.

Our ongoing efforts to provide new and meaningful philanthropic opportunities to our employees have brought all of us closer together. These include our Founders' Day event, where employees devote the day to tangibly investing in our communities; the Tricon Giving Back website, where employees can make payroll-deducted donations to a charity of their choice; and our Pay It Forward initiative, where – during the holidays – employees pass along \$100 or more to an organization or individual in need. In all, we support a broad range of meaningful causes that bring communities together and help others.





has been donated by Tricon employees through our Pay It Forward program.

Imagine a world where housing unlocks life's potential

7 St. Thomas Street, Suite 801 Toronto, Ontario M5S 2B7 T 416 925 7228 F 416 925 7964 www.triconresidential.com